STATE OF COLORADO OFFICE OF THE STATE ARCHITECT STATE BUILDINGS PROGRAM



REQUEST FOR PROPOSALS
FOR AN
INTEGRATED PROJECT DELIVERY METHOD
UTILIZING
CONSTRUCTION MANAGEMENT/GENERAL CONTRACTING (CM/GC)
SERVICES

For The

University of Colorado Denver | Anschutz Medical Campus

For The

Research 1 North – Vivarium Air Valve Replacement – Phase 1 of 2 / PN 23-180525

REQUEST FOR PROPOSALS FOR AN INTEGRATED PROJECT DELIVERY METHOD UTILIZING CONSTRUCTION MANAGEMENT/GENERAL CONTRACTING (CM/GC) SERVICES

TABLE OF CONTENTS

| | INFORMATION |
|--|-------------|
| | |
| | |
| | |

- A. Introduction
- **B.** Description of Project
- C. Minimum Qualifications
- D. Scope of Services

II. PREQUALIFICATION SUBMITTALS (STEP I)

- A. Schedule
- **B.** Mandatory Pre-submittal Conference
- C. Clarifications
- D. General Information
- E. Prequalification Submittals

III. ORAL INTERVIEWS/COST PROPOSAL (STEP II)

- A. Short List
- **B.** In-Person Interview
- C. Cost Proposals
- D. Method of Selection and Award

IV. EVALUATION CRITERIA

A. PREQUALIFICATION SUBMITTAL CRITERIA

- 1. Qualifications of the Firm(s)
- 2. Qualifications of the Management Team Members
- 3. Project Management Approach
- 4. Prior Project Experience/Success
 - a. Timeliness
 - b. Budget Considerations
 - c. Quality
 - d. Services Disruption
 - e. Project Acceptability
 - f. Compliance

5. Miscellaneous Considerations

- a. Claims/Litigation History
- b. Apprenticeship Training Program
- c. Other

B. IN-PERSON INTERVIEWS/COST PROPOSALS CRITERIA

- 1. Qualifications of the Firm(s)
- 2. Qualifications of the Management Team Members

- 3. Project Management Approach
- 4. Prior Project Experience/Success
- 5. Miscellaneous Considerations
- V. CM/GC CONTRACT INFORMATION
- VI. ACKNOWLEDGMENT AND ATTESTATION
- VII. COST PROPOSAL FORM

APPENDICES:

Appendix A: Prequalification Submittal/Evaluation Form

(To be completed by Selection Committee)

Appendix A1: In-Person Interview /Evaluation Form

(To be completed by Selection Committee)

Appendix A2: Submittal and Interview Ranking Matrix

(To be completed by Selection Committee)

Appendix B: Construction Manager/General Contractor (CM/GC) Agreement

(Sample) with CM/GC Designated Services and Method of Payment

Matrix and CM/GC Certification (Form SC- 6.4)

Appendix C: Not Used

Appendix D: Direct Labor Burden Calculation (SBP-6.18)

Appendix E: Not Used

Appendix F: Not Used

Appendix G: Program Plan, etc.

REQUEST FOR PROPOSALS FOR AN INTEGRATED PROJECT DELIVERY METHOD UTILIZING CONSTRUCTION MANAGEMENT/GENERAL CONTRACTING (CM/GC) SERVICES UNIVERSITY OF COLORADO DENVER | ANSCHUTZ MEDICAL CAMPUS

Settlement Notice

For all projects with a total dollar value above \$150,000 Notice of Final Settlement is required by C.R.S. §38-26-107(1). Final Settlement, if required, will be advertised in the same location as the original solicitation.

I. GENERAL INFORMATION

A. INTRODUCTION

PROJECT NAME: Research 1 North – Vivarium Air Valve Replacement – Ph 1 of 2

PROJECT NUMBER: 23-180525

The University of Colorado Anschutz Medical Campus is utilizing a Construction Manager/General Contractor (CM/GC) approach to project delivery. A Guaranteed Maximum Price (GMP) and an updated project duration schedule will be established by the Architect/Engineer and the Construction Manager/General Contractor in conjunction with the University of Colorado Anschutz Medical Campus. The CM/GC will evaluate, among other things, availability of materials and labor, project schedule, project costs as they relate to the established budget, constructability, and will work closely with the Architect/Engineer and the University of Colorado Anschutz Medical Campus throughout the planning, design and construction phases of the project.

The process to be used in the selection of the CM/GC is comprised of two steps. STEP I is the Submittal of Prequalification as described in Section II (D). STEP II is the In-Person Interview/Cost Proposal as described in detail in Section III. A Selection Committee of individuals who will be involved in the project and/or understand the required services associated with Construction Management/General Contracting will evaluate responses to this RFP for both STEPS. Upon completion of the evaluation of the Submittals of Prequalification, a limited number of firms will be invited to the in-person interviews. Sealed fee proposals will be required only from those firms who are interviewed and are to be submitted as indicated in this RFP. Both qualifications and cost will be considered in the final ranking of firms with qualifications given 70% of the value of the weighted criteria and fees for the Cost/ Proposal given 30%.

Selection and award of this project will be based on a combination of qualifications and costs that represents the best overall value to the State.

B. PROJECT DESCRIPTION

PROJECT NAME: Research 1 North – Vivarium Air Valve Replacement – Ph 1 of 2

PROJECT NUMBER: 23-180525

Estimated Construction Cost: \$1,463,271 (Fixed Limit of Construction Cost, FLCC) (Ph 1)

Total Project Budget: \$1,847,537 (combined hard and soft costs) (Ph 1)

The University of Colorado Anschutz Medical Campus seeks a highly qualified Construction Manager/General Contractor (CM/GC) to provide pre-construction services and construction services to replace the terminal box air valve system that serves the vivarium in Research 1 North (R1N). Phase 1 of this project is currently funded through the state-controlled maintenance program (FY 2023-2024). This is a federally funded project and required documentation will be included in the CM/GC agreement. Controlled-maintenance funding for Phase 2 of the project is expected to be received in FY 2024-2025. It's anticipated that the selected CM/GC firm for Phase 1 will continue on with Phase 2.

CU Anschutz medical research relies on modern vivarium facilities to ensure compliance with NIH grants policy and regulations. Laboratory animal facilities present complex building maintenance and repair challenges. Vivarium HVAC systems are critical to program operation and AAALAC accreditation. Protocol violations affecting HVAC operation can result in suspensions, fines, funding problems, program disruption, and facility loss of use. Inadequate HVAC systems risk loss of accreditation and faculty recruitment concerns.

The R1N vivarium ventilation system currently has operational problems requiring replacement of the terminal box air valve system. The air valves precisely control temperature and ventilation and resides in the interstitial space above the vivarium. There are 145 supply air valves with reheat coils, 114 general exhaust air valves, 39 cage rack exhaust valves, 16 Biosafety Cabinet exhaust valves and 5 other miscellaneous exhaust valves. The existing air valves are pressure independent venturi style with internal plunger and spring. These air valves do not have airflow measurement, therefore separate cross-ring airflow stations were provided to report airflow to the building automaton system (BAS).

The potential to disrupt operations or vivarium program delivery associated with this building system is intermittent. The two-phase solution includes new equipment to replace the ventilation system serving the vivarium space. Each of the two phases of this project will complete approximately half of the total required scope of work. Design and construction will need to be planned and coordinated between each phase to maintain continuous and uninterrupted vivarium operation throughout.

The interstitial space above the vivarium is comprised of catwalks to service the ventilation system components and other equipment. Most of the air valves are located adjacent to the catwalk. Therefore, most air valves are readily accessible for replacement. However, there are approximately 35 air valves that are too far away from the edge of the catwalk and will require scaffolding or other platforms to be installed to access air valves for replacement. Depending on how the contractor arranges scaffolding, there will likely need to be 20-30 areas of scaffolding or platforms constructed in order to replace these hard-to-reach air valves. The reheat coils do not have true isolation valves since "valve packs" were used where balancing valves are also intended for isolation. Therefore, drain-down of the heating water system and additional "quadrant" isolation valves are recommended to be installed for future isolation without having to affect other areas of the vivarium. Control valves for the coils, while still operational are beyond their useful life. Reheat coils will be replaced in the same location to minimize the cost of accessories.

University Point of Contact:

Name: Robert Holzwarth Phone: (720) 854-4664

Email: robert.holzwarth@cuanschutz.edu

C. MINIMUM QUALIFICATIONS

Notice is hereby given to all interested parties that all firms will be required to meet ALL of the minimum qualifications to be considered for these projects. To be considered as qualified, interested firms shall have, as a minimum:

- 1. Provided Construction Management/General Contracting services within the last three (3) years for at least three (3) projects each in excess of \$ 4,000,000 (hard costs), utilizing the expertise present in their Colorado Office; and
- 2. Demonstrated specific Construction Management/General Contracting experience in projects of similar scope and complexity; and
- 3. Demonstrated bonding capability up to \$4,000,000 for an individual project coincidentally with current and anticipated workloads; provide letter from surety that affirms this capacity.

Firms meeting the minimum requirements may obtain documents on the website accompanying this advertisement.

University of Colorado Denver | Anschutz Medical Campus Facilities Projects – Request for Qualifications website: https://www.cuanschutz.edu/offices/facilities-management/construction-projects/RFP

Colorado CORE/ColoradoVSS:

https://codpa-vss.cloud.cgifederal.com/webapp/PRDVSS2X1/AltSelfService

D. SCOPE OF SERVICES

The scope of services will include assistance to the State during the process of assessment, design, construction, and warranty period. Specific tasks to be performed by the Construction Manager/General Contractor (CM/GC) include those generally performed by the CM/GC construction community where the Construction Manager is also the Contractor. A sample copy of the State's CM/GC contract is contained within the RFP. A Guaranteed Maximum Price (GMP) will be required at the completion of the Design Development phase.

II. PREQUALIFICATION SUBMITTALS (STEP I)

A. SCHEDULE

1. The schedule of events for the RFP process and an outline of the schedule for the balance of the project is as follows:

| Advertisement | 9/8/2023 |
|---|---------------------------------------|
| Mandatory RFP Pre-submittal Conference & Tour | 9/22/2023 1:30 PM |
| Date Email Questions Due | 10/4/2023 3:00 PM |
| Date Email Answers Issued | 10/13/2023 3:00 PM |
| RFP Submittals Due (Prequalification: Step I) | 10/27/2023 3:00 PM |
| Interview Short List Announced | 11/10/2023 |
| Sealed Proposal Due (Evaluation and Award: Step II) | 11/17/2023 |
| In-Person CM/GC Interviews (projected) | 11/17/2023 |
| Selection Announced (projected) | 11/27/2023 |
| Negotiation of CM/GC Agreement (projected) | 12/8/2023 |
| CM/GC Agreement Approval (projected) | 12/22/2023 |
| Anticipated Design Start | 11/1/2023 |
| Anticipated CM/GC Start | 1/2/2024 |
| Anticipated Construction Start/Finish | 7/1/2024 to 7/1/2025 |
| | · · · · · · · · · · · · · · · · · · · |

Specific requirements for submittals and scoring criteria are detailed in II. (F).
 PREQUALIFICATION SUBMITTALS. In order to facilitate review, One (1) complete
 PDF copy of the submittal must be provided. PDF submittal size is limited to 25 MB
 by CU Anschutz submission website

SUBMITTAL (STEP I) DUE DATE/TIME: 10/27/2023 at 3:00 PM

Submittals shall be submitted online at the following website: https://ucdenverdata.formstack.com/forms/rfp_rfg_submission

2. The above schedule is tentative. Responding firms shall be notified of revisions in a timely manner by email. Respondents may elect to verify times and dates by email, but no earlier than 36 hours before the schedule date and time.

B. MANDATORY PRE-SUBMITTAL CONFERENCE

1. To ensure sufficient information is available to firms preparing submittals, a mandatory pre-submittal conference and tour has been scheduled. The intent of this conference is to tour the site and to have University of Colorado Anschutz Medical Campus staff available to discuss the project. Firms preparing submittals must attend and sign-in in order to have their submittals accepted. The pre-submittal conference and tour will be held as follows:

Pre-Submittal Conference & Tour: Friday 9/22/23 at 1:30 PM

Location:

University of Colorado Anschutz Medical Campus Research 1 North – Hensel Phelps East Classroom P18-1000 12800 East 19th Avenue

C. CLARIFICATIONS

- 1. Owner initiated changes to this RFP will be issued under numerically sequenced email addenda. Addenda generally consist of the following items:
 - a. Clarifications
 - b. Scope Changes
 - c. Time and/or Date Changes

Respondents must acknowledge all issued addenda in their submittal and proposal.

 Respondent initiated email requests for clarification will be received any time on or before <u>October 4, 2023 at 3:00 PM</u>. All State responses will be issued by email addenda on or before <u>October 13, 2023 at 3:00 PM</u>.

E. GENERAL INFORMATION

- 1. All respondents accept the conditions of this RFP, including, but not limited to, the following:
 - a. All submittals shall become the property of the State of Colorado and will not be returned.
 - b. Late submittals shall not be evaluated. Facsimile submittals shall not be accepted.
 - c. Any restriction as to the use of submitted materials must be clearly indicated as proprietary. The requested limitation or prohibition of use or release shall be identified in writing on a cover sheet. Blanket claims of proprietary submittals will not be honored. Fee proposals will be considered proprietary.
 - d. The State reserves the right to reject any or all proposals on the basis of being unresponsive to this RFP or for failure to disclose requested information.
 - e. The State shall not be liable for any costs incurred by respondents in the preparation of submittals and proposals nor in costs related to any element of the selection and contract negotiation process.
 - f. The respondent has reviewed Appendix B and by responding has agreed that the terms and conditions of the sample Construction Management/General Contracting Agreement are expressly workable without reservation.

F. PREQUALIFICATION SUBMITTALS (STEP I)

- 1. Respondent must comply with the following items, a through f. The State retains the right to waive any minor irregularity or requirement should it be judged to be in the best interest of the State. (Note that the primary focus of the Prequalification evaluation will be on the firm(s)' capabilities).
 - a. Submit **one** (1) complete PDF copy of all materials.

- b. Submittals shall be formatted and tabbed in the exact form and numeric sequence of the Evaluation Form (1 through 5) in Appendix A. A two-sided single page cover letter addressed to the University of Colorado Anschutz Medical Campus contact person outlining the firm(s) qualifications is required at the front of the submittal. (Not counting the cover letter and required Acknowledgement and Attestation form, the entire submittal is to be no more than 40 doubled sided 8 ½" x 11" sized pages in portrait format using at least 10 font). Please note the PDF submittal size is limited to 25 MB by the CU Anschutz submission website
- Submittals shall be evaluated in accordance with criteria as indicated in SECTION IV.
 A. PREQUALIFICATION SUBMITTAL CRITERIA and ranked on the corresponding Evaluation Form in Appendix A.
- d. Response to all items shall be complete.
- e. All references shall be current and relevant.
- f. Complete and execute the appropriate Acknowledgment and Attestation Form as provided in Section VI and submit at the back of the Prequalification Submittal.

III. IN-PERSON INTERVIEWS/COST PROPOSALS (STEP II)

A. SHORT LIST

From the submittals received, a short list of qualified respondents shall be identified using the scoring indicated on the enclosed Evaluation Form, Appendix A.

Firms failing to meet the minimum required qualifications will not receive further consideration.

B. IN-PERSON INTERVIEW

1. Mandatory in-person interviews shall be conducted for the short-listed firm(s) only. Interview times and location will be arranged by the University of Colorado Anschutz Medical Campus and all short-listed firms will be notified in advance. At the option of the State, a visit to the short listed firm(s) managing home office and/or representative field office may be required. (Note that the primary focus of the In-Person Interview evaluation, in addition to the Cost Proposal, will be the proposed Project Management Team members' capabilities).

C. COST PROPOSALS

- 1. Only those firms short listed for interview are required to submit their sealed proposals. (Only one copy is required on the scheduled submission date.) Cost Proposals will remain sealed until after the qualitative scoring and will then be opened. The Cost Proposal will then be considered (equivalent to 30 percent of the weighted criteria) in conjunction with the qualitative score from the response and interview (equivalent to 70 percent of the weighted criteria).
- Cost Proposals shall be submitted on the form provided in Section VII, without
 modification. A Cost Proposal shall be accompanied with sufficient detail to clearly identify
 the fee for service and include a detailed schedule of estimated (not-to-exceed)
 reimbursable and non-reimbursable costs. Percentage of the cost of work is not an

acceptable value. The Cost Proposal should be prepared independently in accordance with the following:

- a. Any specific services requested in the RFP and its appendices that are not included should be clearly identified. Exclusion of any required service may result in the proposal being found non-responsive.
- b. Provide a CM/GC staff schedule with staff by name, position and person-hours (assume 8-hour days) per month estimated on the project.
- c. Provide a detailed estimate of reimbursable costs including breakdown of direct salaries and payroll fringes (DPE) for on-site CM/GC personnel associated with the services. Not-to-exceed reimbursable expenses shall be provided at direct cost.
- d. Provide a detailed estimate of non-reimbursable expenses (included in fee).
- e. The State reserves the right to reject any Cost Proposal not prepared in the above manner. Proposals that exceed the available funds may be rejected outright but the State reserves the right to negotiate a reasonable fee for service within the available funds. The CM/GC contract will be a bonded lump sum contract including not-to-exceed reimbursables with a Guaranteed Maximum Price to encompass all construction work; some not-to-exceed allowances may be included as directed by the State.
- 3. This Fee Proposal is a binding offer to perform the services associated with the Scope of Services described in this RFP and the Designated Services and Method of Payment Matrix in Appendix B. The State reserves the right to negotiate a cost adjustment based on scope clarification subsequent to selection and prior to contract execution.

D. METHOD OF SELECTION AND AWARD

The Selection Committee shall complete a combined evaluation of qualifications and fee in accordance with the criteria as indicated in SECTION IV, B. IN-PERSON INTERVIEWS/COST PROPOSALS/EVALUATION CRITERIA. Numerical ranking and selection of the most qualified firm (including fee) will then occur on the corresponding evaluation forms in Appendix A1.

The final fee amount and scope of services may be negotiated at the State's discretion. Award and contract will be contingent on availability of key proposed Project Management Team staff.

IV. EVALUATION CRITERIA

A. PREQUALIFICATION SUBMITTAL CRITERIA

(Note that the primary focus of the Prequalification evaluation will be the Firm(s) capabilities).

1. QUALIFICATIONS OF THE FIRM(s)

□ Provide a description of the composition and management structure of your firm. Identify the firm's roles and responsibilities and relevant experience with projects of

| | | similar scope and complexity and similar fast track project delivery methods. Describe how the firm's experience will relate to the success of this project. Provide a description and separate graphic organizational chart complete with working titles identifying the lines of authority, responsibility and coordination. Provide a detailed description of the process of how your firm selects qualified subcontractors and manages them effectively on complex multi-phased projects. Provide a detailed description of how your firm will maximize the Colorado construction work force on this project. Provide your firms' safety record over the last ten years and describe your firms' efforts to retain and support employees. |
|----|------|---|
| 2. | QL | JALIFICATIONS OF THE MANAGEMENT TEAM MEMBERS |
| | | Describe the qualifications and relevant experience of the superintendent including demonstrated experience working on projects of similar scope and complexity and time commitment for this project. |
| | | Describe the qualifications and relevant experience of the project manager including demonstrated experience working on projects of similar scope and complexity and time commitment for this project. |
| | | Identify all current office locations of the assigned staff and any other resident expertise intended to be provided under this RFP. |
| 3. | Ы | ROJECT MANAGEMENT APPROACH |
| | _ | Provide a strategic project approach summary: Include discussion of your team's approach in providing successful Construction Management/General Contracting services based on prior experience in cost, schedule and quality effectiveness. Include specific examples (1-2 page excerpts) of actual products (estimates, progress reports, schedules, constructability reviews, value engineering studies, forms, general conditions budgets, organizational structures, etc.). |
| | | Provide a description of construction work that your team has capability to competitively bid and self-perform, including qualifications to do such. It is the perception of the University of Colorado Anschutz Medical Campus that subcontracting CM/GC construction work is in the State's best interest in terms of price competition. The University of Colorado Anschutz Medical Campus may, at its discretion, limit the types and amount of work that the CM/GC bids and self-performs. |
| 4. | Pl | RIOR PROJECT EXPERIENCE/SUCCESS |
| Se | lect | your three (3) most relevant projects and provide, at a minimum, the following: |
| | | The project/contract name Description of services provided Overall construction cost of project, as applicable, including initial contract value and |
| | | change orders including reasons for change orders Organizational structure of service delivery under the contract (include the owner's organization as it interfaced with the respondent's contract) |
| | | Key assigned in-house staff (name and title) Subcontracts (service) used in the performance of the contract Schedule history |
| | | Reference(s) for Owner and Architect as described in Section II. F. e Continuing services, if any |

a. Timeliness

In general, Construction Management/General Contracting work is seen as successful if it is on time, on budget, and of high quality of workmanship. Timeliness is generally based on completion by the originally scheduled date and is indicated by a Certificate of Occupancy. Please demonstrate for each of the above projects how timely delivery occurred.

b. Budget Considerations

Similar to timeliness, being on budget historically means the work was completed within the originally identified available budget. For purposes of this RFP, the State is interested not only in being within budget but also in the respondent's ability to address and implement the following issues as well:

- 1. Conceptual estimating
- 2. Value analysis
- 3. Alternate solutions
- 4. Scope reduction that maintains project function
- 5. Cost/benefit analysis

Demonstrate for the above projects examples of how you accomplished the above cost control services.

c. Quality

Construction quality has the obvious traditional connotations (workmanlike, in compliance with the specifications, normal standard of care, etc.). Demonstrate for the above project examples how a high quality of workmanship was achieved.

d. Services Disruption

Demonstrate how your services on the above project examples dealt with issues of disruption at existing facilities, etc. if applicable.

e. Project Acceptability

Please discuss how your Construction Management/General Contracting services helped achieve owner satisfaction with regard to project quality and acceptability on your project examples.

f. Compliance

Provide information on how compliance with industry standards of care, building codes, etc. was achieved.

5. MISCELLANEOUS CONSIDERATIONS

a. Claims/Litigation History of Firm

Provide information on any past, current or anticipated claims (i.e., knowledge of pending claims) on respondent contracts; explain the litigation, the issue, and its outcome or anticipated outcome.

b. Apprenticeship Training Program (Optional for Step I)

Where an Apprentice Training Program certified by the Office of Apprenticeship located in the Employment and Training Administration in the United States Department of Labor exists in the State, or a comparable program for the training of apprentices is available in the State:

- 1. Each submitter shall demonstrate access to the certified program or a comparable alternative (Note that it is the responsibility of the submitter to demonstrate the comparability of a non-certified program) and,
- 2. Each submitter's subcontractor at any tier with a contract value of two hundred fifty thousand dollars or more shall demonstrate access to the certified program or a comparable alternative.

c. Other

This category is included for other items provided by the submitter. Inclusions may include standard firm promotional literature, testimonials, awards, corporate memberships in professional organizations or sponsorships, additional project/contract histories, etc., intended to demonstrate why your firm is uniquely qualified for this project.

B. <u>IN-PERSON INTERVIEWS/COST PROPOSALS EVALUATION CRITERIA</u> (Note that the primary focus of the In-Person Interview evaluation in addition to the Cost Proposal will be the proposed project management team members' capabilities).

1. QUALIFICATIONS OF THE FIRM

| Explain the composition and structure of your project management team and how the |
|---|
| firm will support their efforts in the field throughout this project. |
| Are the lines of authority, responsibility and coordination clearly identified? |

2. QUALIFICATIONS OF THE MANAGEMENT TEAM MEMBERS

| Explain the prior experience with projects of similar scope and complexity and similar |
|--|
| fast track project delivery methods of the superintendent, project manager and all |
| other project management team members. Explain their roles and responsibilities and |
| authority and why they are the right team members for this project. |

| Explain | anticipated | project | management | team staff | current and | projected | workload |
|---------|-------------|---------|------------|------------|-------------|-----------|----------|
| | | | 9 | | | | |

| Identify all current office locations and the resident expertise intended to be provided |
|--|
| under this RFP. Identify the location of the staff for the performance of this contract, |
| their expertise, and generic equipment that will be located in Colorado and act in |
| support of the anticipated contract. |

3. PROJECT MANAGEMENT APPROACH

□ Explain the strategic project approach for this project in summary: Include discussion of your team's approach in providing successful CM/GC services based on the needs of this specific project utilizing the team's prior past experience including cost, schedule, and quality control.

| | competitively bid and self-perform including qualifications to do such work. Provide a detailed description of how your project management team will select qualified sub-contractors and manage them effectively on this project. |
|-----|--|
| PR | IOR PROJECT EXPERIENCE/SUCCESS |
| | Explain the most relevant projects the superintendent and the team members have completed together and/or separately and what their role was. The University of Colorado Anschutz Medical Campus at its discretion, may contact references and/or conduct independent performance analysis on projects on which the team member has worked). |
| | Provide descriptions of other related experience of superintendent and other project management team members. |
| MIS | SCELLANEOUS CONSIDERATIONS |
| | Craft Labor Capabilities Describe the availability of resources that will be utilized to successfully complete the project. |
| | Apprenticeship Training Program (Mandatory for Step II) Describe access to federal or state-approved apprenticeship programs, as available. |
| | Other |

This category is included for other items provided by the presenter. Inclusions may include testimonials, awards, corporate memberships in professional organizations or sponsorships, additional project/contract histories, etc., intended to demonstrate why

□ Explain the construction work the project management team has the capability to

V. CM/GC CONTRACT INFORMATION

4.

5.

A. Carefully review the CM/GC Contract sample (Appendix B) before initiating your response submittal. Any exceptions to the contract must be communicated formally in accordance with the written questions schedule in II.A.

this management team is uniquely qualified for this project.

- B. Appendix E and F of this RFP includes mandatory State apprenticeship and prevailing wage requirements based on the construction value of the project. (Not Used. This project is entirely federally funded)
- C. The State reserves the right to make non-material changes to the appended model agreement, including additions and /or modifications that may be necessary to more completely describe the services defined or implied herein.
- **D.** Any approved reimbursable expenses made under the terms of the final agreement shall be a direct pass-on cost with no adjustment to the fee described therein.
- **E.** Any and all products, systems, methods, and procedures developed, as a result of this agreement shall remain the exclusive property of the State.

VI. ACKNOWLEDGEMENT AND ATTESTATION FORM

A. Several versions of the Acknowledgment and Attestation Form follow this section. Proper completion of the appropriate form is a mandatory requirement for a respondent to be considered responsive to this RFP Pregualification Submittal.

B. Qualifications made by a respondent in executing this form may render a submittal non-responsive as determined by the State.

VII. COST PROPOSAL FORM

- A. Immediately following the Acknowledgement and Attestation Form is a Cost Proposal Form to be utilized to summarize the fee proposal for the services. Only those firms short-listed will be required to submit fee proposals as directed by the University of Colorado Anschutz Medical Campus.
- **B.** This RFP document, it's appendices, and any written addenda issued prior to the submittal of proposals, and written clarifications prior to the interview shall serve as the only basis for proposals.
- C. The respondent, by submitting this proposal, does hereby accept that minor changes by the State to the exhibited contract and its exhibits, which do not adversely affect the respondent, shall not be cause for withdrawal or modification of the amounts submitted herein. Exceptions to the RFP documents and/or modification of the proposal may render the proposal non-responsive.
- D. Upon due consideration and review of this document along with its appendices, written addenda, and written clarifications prior to the interview, the respondent does hereby submit the following proposal for Construction Management/General Contracting fees, consistent with the schedules provided in the Scope of Services. Respondents are hereby advised that it is the State's desire to accelerate design and construction schedules where reasonably possible, without adverse cost impact.
- **E.** Respondent should complete the Cost Proposal Form by filling in all blanks on the form that follows.
- **F.** Respondents should include a separate detailed not-to-exceed reimbursable estimate

End of RFP

ACKNOWLEDGEMENT AND ATTESTATION FORM (Partnership Format)

| Date: | | |
|-------|-------------|--|
| | Page 1 of 1 | |

By responding to this RFP, the respondent(s) certify that he/she has reviewed the Construction Management/General Contracting sample contract, and its exhibits contained herein, and is familiar with their terms and conditions and finds them expressly workable without change or modification.

We certify and declare that the foregoing is true and correct.

| Subscribed on | at | | | |
|----------------------|-------------------|-------|------|--|
| | Date | | City | |
| | , State of | | | |
| County | | State | | |
| 1) | | | _ | |
| | Partner Signature | | | |
| | Typed Name: | | _ | |
| 2) | | | | |
| | Partner Signature | | | |
| | Typed Name: | | _ | |
| Notary: | | | | |
| | | | Date | |
| Commission Expires:_ | | | | |

Note: Add additional signatures if there are more than two partners.

ACKNOWLEDGEMENT AND ATTESTATION FORM

(Joint Venture Format)

| Date: | | |
|-------|-------------|--|
| | Page 1 of 1 | |

By responding to this RFP, the respondent(s) certify that he/she has reviewed the Construction Manager/General Contractor sample contract, and its exhibits contained herein, and is familiar with their terms and conditions and finds them expressly workable without change or modification.

We certify and declare that the foregoing is true and correct.

| Subscribed on | | at | , |
|-----------------------|--------|-------------------|------|
| | Date | | City |
| | , Stat | te ofState | |
| County | | State | |
| 1) Venture Partner | | | |
| Venture Partner | | Binding Signature | Date |
| Type of Dusiness | | Typed Name: | |
| Type of Business | | Tille: | |
| | ī | Witness | Date |
| | - | Typed Name: | |
| 2) | | | |
| Venture Partner | I | Binding Signature | Date |
| Type of Business | | Typed Name: | |
| i ype oi busilless | | Title: | |
| | ī | Witness | Date |
| | | Typed Name: | |

Note:

- 1. Add additional venture partners as necessary.
- 2. Witnesses of venture partners shall be corporate secretary for corporations, partners for partnerships, and notaries for sole proprietorships.
- 3. Attach venture agreement
- 4. Type of business shall identify the venture partner as a corporation, venture, partnership, sole proprietorship, or other legal entity.

ACKNOWLEDGEMENT AND ATTESTATION FORM (CORPORATE FORMAT)

| Date: | | |
|-------|-------------|--|
| | Page 1 of 1 | |

By responding to this RFP, the respondent(s) certify that he/she has reviewed the Construction Management/General Contracting sample contract, and its exhibits contained herein, and is familiar with their terms and conditions and finds them expressly workable without change or modification.

We certify and declare that the foregoing is true and correct.

Note: Use full corporate name and attach corporate seal here.

| Subscribed on | | | | _ at | | , |
|-------------------------|-------|------------|-------|------|------|---|
| | Date | | | | City | |
| | | , State of | | | | |
| County | | _ | State | | | |
| | | | | | | |
| Corporate Officer Ciana | -t | | | Data | | |
| Corporate Officer Signa | ature | | | Date | | |
| | | | | | | |
| | | | | | | |
| Secretary | | | | Date | | |
| | | | | | | |
| | | | | | | |

(SEAL)

IPD CM/GC RFP Rev 7/2022

ACKNOWLEDGEMENT AND ATTESTATION FORM (Sole Proprietorship Format)

| Date: | | |
|-------|-------------|--|
| | Page 1 of 1 | |

By responding to this RFP, the respondent(s) certify that he/she has reviewed the Construction Management/General Contracting sample contract, and its exhibits contained herein, and is familiar with their terms and conditions and finds them expressly workable without change or modification.

We certify and declare that the foregoing is true and correct.

| Subscribed on | | | at | | , |
|---------------------|------|-------------|-------|------|---|
| | Date | | | City | , |
| | | _, State of | | | |
| County | | | State | | |
| | | | | | |
| Respondent | | | Date | | |
| Typed Name: | | | | | |
| | | | | | |
| Notary: | | | | | |
| | | | Date | | |
| Commission Expires: | | | | | |

COST PROPOSAL FORM CONSTRUCTION MANAGER/GENERAL CONTRACTING (CM/GC) SERVICES

| Date: | <u> </u> | | |
|--------------------|---|--|--|
| Proje | ct Title: | | |
| 1. | CM/GC Pre-Construction Fee | \$ | _ |
| 2. | CM/GC Construction Fee | \$ | _ |
| 3. | General Conditions - On-Site CM/GC Staff | \$ | _ |
| 4. | Other Reimbursable General Conditions (NTE) | \$ | _ |
| | Total CM/GC Fee | \$ | |
| Fees | are to be calculated per Exhibit A (SC-6.5), CM/GC December 2015 | esignated Services and Me | thod of Payment. |
| assoc | e provide a detailed breakdown to adequately describ ciated anticipated reimbursable costs so as to demons ces provided. | | |
| reimb | bursable general condition expenses are generally co oursed at direct personnel expense, plus those on-site cCM/GC staff and construction subcontractors. | | |
| Ackno | owledge receipt of Addendum Numbers: | | |
| If o se a | pates Services outside the United States or Colorado* Ye the respondent anticipates services under the contract or a r Colorado, the respondent shall provide in a written statemetrices that will be performed at a location outside the United dvantageous to go outside the United States or Colorado to eccives federal moneys) | any subcontracts will be perfor ent which must include, but ne ed States or Colorado and the | ed not be limited to the type of reason why it is necessary or |
| F "(R sı | omply with 80% Colorado Labor Yes No or State Public Works Project per C.R.S 8-17-10, Colorado Colorado Labor" means any person who is a resident of the espondents indicating that their bid proposal will not complubmit written justification along with the bid submission. A g Istification for the waiver on its web site. (Does not apply to a | e state of Colorado at the tim ly with the 80% Colorado Lab lovernmental body that allows | e of the Public Works project. or requirement are required to a waiver shall post notice and |
| A in re | r is a Service-Disabled Veteran Owned Small Business* Service-Disabled Veteran Owned Small Business (SDV acorporated or organized in Colorado or maintains a place agistered and verified by the Center for Veteran Enterprise vertification along with the proposal submission. | OSB) per C.R.S. 24-103-90 of business or has an office | e in Colorado and is officially |
| *Does | not apply to projects for Institutions of Higher Education tha | nt have opted out of the State F | Procurement Code. |
| Applic | ant or Corporate Officer Signature | | |
| | | | |

IPD CM/GC RFP Rev 7/2022

Appendix A

STATE BUILDINGS PROGRAM PREQUALIFICATION SUBMITTAL/EVALUATION FORM CONSTRUCTION MANAGEMENT/GENERAL CONTRACTING (CM/GC) SERVICES

| Name of Firm: | | | |
|---|---------------------|--------------|------------------|
| Name of Project: | | | |
| Evaluator No: | Date: | | |
| RFP REFERENCE | | | |
| MINIMUM REQUIREMENTS | | Υ_ | N |
| If the minimum requirements (including letter from surety) have no | ot been me | et specify | the reason(s): |
| | | , opeoy | |
| | | | |
| | | | |
| Acknowledgement and Attestation included: | | Υ_ | N |
| SCORE | | | |
| SCORE | Weiaht ² | x Ratin | g^3 = Score |
| | | 71 1 15.11. | 9 |
| 1. QUALIFICATIONS OF THE FIRM(s) ¹ | | | |
| □ Qualifications of the firm | 5 | v | _ |
| □ Qualifications of the firm □ Organizational structure/lines of authority | 4 | — <u>x</u> — | = = = = |
| □ Subcontractor selection and management | 5 | _ x | |
| □ Colorado workforce | 4 | x | = |
| □ Safety/employee support | 4 | x | = |
| 2. QUALIFICATIONS OF THE MANAGEMENT TEAM MEMBER | RS ¹ | | |
| 2. GOVERNOON THE WAY TO COME THE WINDER | <u></u> | | |
| ☐ Qualifications and relevant experience of superintendent | 5 | x | = |
| Qualifications and relevant experience of project manager | 5 | x | = = = |
| □ Location/Access | 3 | x | = |
| 3. PROJECT MANAGEMENT APPROACH ¹ | | | |
| <u></u> | | | |
| □ Approach to successful CM/GC Services | | | |
| a. Cost effectiveness | 5 | _ X | = |
| b. Schedule effectiveness | 5 | _ X | _ = |
| c. Quality effectiveness | 5 | | |
| □ Competitively Bid/Self Performed Work | 4 | Х | _ |

4. PRIOR PROJECT EXPERIENCE/SUCCESS¹

| | Project #1 | | 5 | х | _ = | |
|-------------|---------------------------------|--------------------------------|------------|---------------------------------------|-----|---|
| | a. Timeliness | d. Disruption | | | | |
| | b. Budget Considerations | e. Acceptability | | | | |
| | c. Quality | f. Compliance | | | | |
| | Project #2 | | 5 | X | = | |
| | a. Timeliness | d. Disruption | | | | |
| | b. Budget Considerations | e. Acceptability | | | | |
| | c. Quality | f. Compliance | | | | |
| | | | | | | |
| | Project #3 | . 5 | 5 | Х | _ = | |
| | a. Timeliness | d. Disruption | | | | |
| | b. Budget Considerations | e. Acceptability f. Compliance | | | | |
| | c. Quality | i. Compliance | | | | |
| | Related experience of the firm | | 4 | X | = | |
| | | | | · · · · · · · · · · · · · · · · · · · | | |
| 5. <u>M</u> | <u>ISCELLANEOUS¹</u> | | | | | |
| | | | | | | |
| | Claims/litigation history | | 4 | Х | _ = | |
| | Apprenticeship Training Program | | 3 | X | _ = | |
| | Other | | 3 | Х | _ = | |
| | | T | OTAL SCOPE | | | 4 |

NOTES:

- 1. **Criteria**: Agencies/Institutions are encouraged to include additional criteria that reflect unique characteristics of the project under each category to help determine the submitter's overall qualifications.
- **2. Weights**: Agency/Institutions to assign weights, using whole numbers, to all criteria on evaluation forms for inclusion into RFQ document and prior to evaluations.
- **3. Ratings**: Evaluator to assess the strength of each firms qualifications and assign a numerical rating of 1 to 5 with 5 being the highest rating, 0 is missing information. (Use whole numbers)
- **4. Total Score**: Includes the sum of all criteria. Note: a passing score (as a percentage of the total points available) is optional and should be assigned by the agency/institution prior to evaluation.

Appendix A1

STATE BUILDINGS PROGRAM IN-PERSON INTERVIEWS/COST PROPOSALS EVALUATION FORM CONSTRUCTION MANAGEMENT/GENERAL CONTRACTING (CM/GC) SERVICES

| Name of Firm: Name of Project: Evaluator No: Date: | | | | |
|--|---------------------|-------------------|--------|---|
| SCORE | | | | _ |
| Score | Weight ² | x Ratir | ng³ = | |
| 1. QUALIFICATIONS OF THE TEAM ¹ | 5 | _ x | = | |
| 2. QUALIFICATIONS OF THE MANAGEMENT TEAM MEMBERS | <u> 5</u> | _ x | = | |
| 3. PROJECT MANAGEMENT APPROACH ¹ | 4 | _ x | = | |
| 4. PRIOR PROJECT EXPERIENCE/SUCCESS ¹ | 4 | _ x | = | |
| 5. MISCELLANEOUS¹ □ Craft Labor Capabilities □ Apprenticeship Training Program □ Other | 3 3 3 | _ x _ x _ x | = = | |
| TOTA | L SCORE: | | | 4 |

NOTES:

- 1. **Criteria**: Agencies/Institutions are encouraged to include additional criteria that reflect unique characteristics of the project under each category to help determine the submitter's overall qualifications.
- **2. Weights**: Agency/Institutions to assign weights, using whole numbers, to all criteria on evaluation forms for inclusion into RFQ document and prior to evaluations.
- **3. Ratings**: Evaluator to assess the strength of each firms qualifications and assign a numerical rating of 1 to 5 with 5 being the highest rating, 0 is missing information. (Use whole numbers)
- **4. Total Score**: Includes the sum of all criteria. Note: a passing score (as a percentage of the total points available) is optional and should be assigned by the agency/institution prior to evaluation.

Appendix A2

STATE BUILDINGS PROGRAM SUBMITTAL AND IN-PERSON INTERVIEW RANKING MATRIX

QUALIFICATIONS 70%/FEE 30%

| FIRM | QUALIF | CATION | IS¹ | | | | AVERAGE QUALS ² | QUALS SCORE ³ | FEE SCORE ⁴ | QUALS & FEE | RANK ⁶ |
|------|------------|------------|------------|------------|------------|------------|-------------------------------|-----------------------------|---------------------------|--------------------|-------------------|
| | EVAL #1 | EVAL #2 | EVAL #3 | EVAL #4 | EVAL #5 | EVAL #6 | | | | SCORE ⁵ | |
| | | | | | | | | | | | |
| | | | | | | | | | | | |
| | | | | | | | | | | | |
| | | | | | | | | | | | |
| | | | | | | | | | | | |
| | | | | | | | | | | | |
| | | | | | | | | | | | |
| | | | | | | | | | | | |
| | | | | | | | | | | | |

NOTES:

- Insert total score from each evaluator's PREQUALIFICATION SUBMITTAL or IN-PERSON INTERVIEW/ COST PROPOSALS/EVALUATION FORMS. (Note that the use of the Matrix for the PREQUALIFICATION SUBMITTAL EVALUATION does not consider cost proposals only qualifications). DO NOT combine the scores of the two evaluation forms.
- 2. Add all evaluators' total scores and divide by the number of evaluators to determine the average score for each firm's qualifications.
- 3. The highest score for qualifications on the evaluation form is to receive 70 points and the other team scores are to be determined as a percentage of the 70 points. To score each average qualification score, use the example formula.

Assume the highest score is 700.

SCORING OF QUALIFICATIONS

FIRM B: $\frac{700}{2} \times 70$ points = 70 points

700

FIRM C: $\underline{600} \times 70 \text{ points} = 60 \text{ points}$

700

FIRM A: $500 \times 70 \text{ points} = 50 \text{ points}$

700

4. Determine score for each firm's sealed cost proposal with the lowest fee being equivalent to a score of 30 points. To score each fee, use the example formula.

Assume the lowest fee was \$100,000.

SCORING OF FEES

FIRM A: $$100,000 \times 30 \text{ points} = 30 \text{ points}$

\$100,000

FIRM B: $$100,000 \times 30 \text{ points} = 24 \text{ points}$

\$125,000

FIRM C: $$100,000 \times 30 \text{ points} = 20 \text{ points}$

\$150,000

- 5. Add the average qualification score to the fee score to determine cumulative qualifications and fee score.
- 6. Numerically rank all firms with the highest scoring firm being the most qualified.

Appendix B

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (FORM SC-6.5) (Sample)

STATE OF COLORADO OFFICE OF THE STATE ARCHITECT STATE BUILDINGS PROGRAM

Sample SC-6.5 agreement format for information only.



CONSTRUCTION MANAGER/GENERAL/CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

| STATE AGENCY: | University of Colorado Anschutz Medical Campus |
|-----------------|--|
| DEPARTMENT ID: | GFE |
| CONTRACT ID #s: | |
| PROJECT #: | |
| PROJECT NAME: | |
| VENDOR NAME: | |
| | |

STATE OF COLORADO OFFICE OF THE STATE ARCHITECT STATE BUILDINGS PROGRAM

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT

(STATE FORM SC-6.5)

| TABLE | E OF CONTENTS | Page |
|-------|--|------|
| Table | of Contents for the entire Agreement is located in THE GENERAL CONDITIONS OF THE | |
| CONS | TRUCTION MANAGER.GENERAL CONTRACTOR (CMGC) AGREEMENT (SC-6.51) | |
| | | |
| SIGNA | ATURE PAGE | 1 |
| RECIT | ALS: | 2 |
| 1 A | RTICLE 1 PERFORMANCE OF THE WORK | 3 |
| 1.1 | THE WORK | 3 |
| 1.2 | CONTRACT DOCUMENTS | 4 |
| 2 A | RTICLE 2 EXHIBITS TO THE AGREEMENT | 5 |
| 2.1 | EXHIBITS | 5 |
| 3 A | RTICLE 3 CONSTRUCTION MANAGER'S SERVICES | 6 |
| 3.1 | AVAILABLE FUNDS | 6 |
| 3.2 | BUDGETING AND FIXED LIMIT OF CONSTRUCTION COST | 6 |
| 3.3 | CONSULTATION AND VALUE ENGINEERING | 6 |
| 3.4 | CONSTRUCTION COSTS | 7 |
| 3.5 | OTHER PRE-CONSTRUCTION SERVICES | 10 |
| 3.6 | CONTROL OF THE WORK | 13 |
| 3.7 | SCHEDULE AND COORDINATION | 14 |
| 3.8 | AMENDMENTS AND CHANGE ORDERS | 14 |
| 3.9 | PRINCIPAL REPRESENTATIVE CONSULTANTS | 15 |
| 3.10 |) START UP | 15 |
| 4 A | RTICLE 4 TIME OF COMMENCEMENT AND COMPLETION | 15 |
| 4.1 | COMMENCEMENT | 15 |
| 4.2 | COMPLETION | 16 |
| 5 A | RTICLE 5 COMPENSATION | 16 |
| 5.1 | CONSTRUCTION MANAGER'S FEE | 16 |
| 5.2 | ADJUSTMENTS IN FEE | 18 |
| 5.3 | PAYMENT OF FEE | 18 |
| 5.4 | GUARANTEED MAXIMUM PRICE | 18 |
| 5.5 | CONTRACT SUM | 20 |
| 6 A | RTICLE 6 CONDITION PRECEDENT | 21 |
| 6.1 | FUTURE APPROPRIATIONS | 21 |
| 6.2 | OBLIGATIONS | 21 |
| 7 A | RTICLE 7 OPTIONAL PROVISIONS AND ELECTIONS | 21 |
| 7.1 | MODIFICATION OF ARTICLE 2. EXECUTION, CORRELATION, INTENT OF DOCUMENTS, | |
| | COMMUNICATION AND COOPERATION | 21 |
| 7.2 | MODIFICATION OF ARTICLE 45. GUARANTEE INSPECTIONS AFTER COMPLETION | 21 |
| 7.3 | MODIFICATION 1 OF ARTICLE 27. LABOR AND WAGES | 21 |
| 7.4 | MODIFICATION 2 OF ARTICLE 27. LABOR AND WAGES | 21 |

| | 7.5 | | FICATION OF | | | | | | | | |
|--------------------------------------|---|---|--|---|--|--|----------------|-----------|---------|---------------|----------------|
| | | | | | | | | | | | 21 |
| | 7.6 | | | | | | | • | | | ES22 |
| _ | | TICLE 8 | | | | | | | | | 22 |
| ΑT | | | | | | | | | | | GER/GENERAL |
| | | | | | | | | | | | 24 |
| 1 | AR | | | | | | | | | | 1 |
| 2 | 1.1 | | | | | | | | | | 1 |
| 2 | 1.2 | DEFIN | | | | | | | | | 2 |
| 2 | AR | TICLE 2 | EXECUT | ION, COR | RELATION, | , INTENT | OF D | OCUMENT | s, con | IMUNIC | ATION AND |
| | CO | | | | | | | | | | 7 |
| 2 | 2.1 | EXECL | TION | | | | | | | | 7 |
| 2 | 2.2 | | _ | | | | | | | | 7 |
| 2 | 2.3 | | | | | | | | | | 8 |
| 2 | 2.4 | PARTI | IERING, CON | /MUNICA | TIONS AND | COOPER | RATION | l | | | 8 |
| 3 | | | | _ | | | | | | | 9 |
| 4 | AR | TICLE 4 | OWNERSHI | P OF DRA | WINGS | | | | | | 9 |
| 5 | AR | TICLE 5 | ARCHITECT | /ENGINEE | R'S STATUS | S | | | | | 9 |
| 6 | AR | TICLE 6 | ARCHIT | ECT/ENG | INEER DEC | CISIONS | AND . | IUDGMEN1 | s, acc | ESS TO | WORK AND |
| | INS | PECTION | l | | ••••• | • | ••••• | | | | 10 |
| 6 | 5.1 | DECIS | ONS | | | | | | | | 10 |
| 6 | 5.2 | JUDGI | ИENTS | | | | | | | | 10 |
| 6 | 5.3 | ACCES | S TO WORK. | | | | | | | | 10 |
| 6 | 5.4 | INSPE | CTION | | | | | | | | 10 |
| | | | | | | | | | | | ±0 |
| 7 | AR | TICLE 7 | | | | | | | | | 12 |
| | | | CONSTRUCT | TION MAN | NAGER'S SU | JPERINTE | NDEN | CE AND SU | PERVISI | ON | |
| 8 | AR | TICLE 8 | CONSTRUCT | TION MAN AND EMI | NAGER'S SU PLOYEES | JPERINTE | NDEN | CE AND SU | PERVISI | ON | 12 |
| 8 9 | AR | TICLE 8 | CONSTRUCT MATERIALS SURVEYS, P | ΓΙΟΝ MAN AND EMI ERMITS, L | NAGER'S SUPLOYEES | JPERINTE | NDEN REGULA | CE AND SU | PERVISI | ON | 12 12 |
| 8 9 | ART | TICLE 8 TICLE 9 SURVI | CONSTRUCT MATERIALS SURVEYS, P | FION MAN AND EMI ERMITS, L | NAGER'S SU PLOYEES AWS, TAX | JPERINTE | NDEN REGULA | CE AND SU | PERVISI | ON | 12 12 13 |
| 8 9 | ART ART 9.1 | TICLE 8 TICLE 9 SURVI PERM | CONSTRUCT MATERIALS SURVEYS, P EYS TS AND LICE | FION MAN AND EMI ERMITS, L | NAGER'S SU PLOYEES AWS, TAX | JPERINTE | EGULA | CE AND SU | PERVISI | ON | 12 12 13 |
| 8 9 | ART ART 9.1 9.2 | TICLE 8 TICLE 9 SURVI PERM TAXES | CONSTRUCT MATERIALS SURVEYS, P TYS TS AND LICE | FION MAN AND EMI ERMITS, L | NAGER'S SUPLOYEES | JPERINTE | REGULA | ATIONS | PERVISI | ON | 121313 |
| 8 9 | ART 9.1 9.2 9.3 9.4 | TICLE 8 TICLE 9 SURVI PERM TAXES LAWS | CONSTRUCT MATERIALS SURVEYS, P TYS TS AND LICE AND REGUL | FION MAN AND EMI ERMITS, L ENSES | NAGER'S SUPLOYEES | ES AND R | EGULA | ATIONS | PERVISI | ON | 12131313 |
| 8 9 9 | ART 9.1 9.2 9.3 9.4 | TICLE 8 TICLE 9 SURVI PERM TAXES LAWS | CONSTRUCT MATERIALS SURVEYS, P EYS TS AND LICE AND REGUL PROTECTIO | FION MAN AND EMI ERMITS, L ENSES ATIONS ON OF WO | NAGER'S SUPLOYEES AWS, TAXI | ES AND R | REGULA | ATIONS | PERVISI | ON | 1213131313 |
| 8 9 9 | ART ART 9.1 9.2 9.3 9.4 ART | TICLE 8 TICLE 9 SURVI PERM TAXES LAWS TICLE 10 | CONSTRUCT MATERIALS SURVEYS, P EYS TS AND LICE AND REGUL PROTECTIO | ENSES ATIONS ON OF WO | NAGER'S SUPLOYEES AWS, TAXI | ES AND R | EGULA | ATIONS | PERVISI | ON | 1213131314 |
| 8 9 10 | ART ART 9.1 9.2 9.3 9.4 ART 10.1 | TICLE 8 TICLE 9 SURVI PERM TAXES LAWS TICLE 10 GENEI SAFET | CONSTRUCT MATERIALS SURVEYS, P EYS TS AND LICE AND REGUL PROTECTION Y PRECAUTION | ENSES ATIONS ON OF WO | NAGER'S SUPLOYEES AWS, TAXI | ES AND R | EGULA | ATIONS | PERVISI | ON | 121313131414 |
| 8 9 9 10 | ART ART 9.1 9.2 9.3 9.4 ART 10.1 10.2 | TICLE 8 TICLE 9 SURVE PERM TAXES LAWS TICLE 10 GENEE SAFET EMER | CONSTRUCT MATERIALS SURVEYS, P EYS ITS AND LICE AND REGUL PROTECTION PROTECTION Y PRECAUTION GENCIES | ENSES ATIONS ON OF WO | NAGER'S SUPLOYEES | ES AND R | EGULA | ATIONS | PERVISI | ON | |
| 8 9 10 11 | ART ART 9.1 9.2 9.3 9.4 ART 10.1 10.2 10.3 ART | TICLE 8 TICLE 9 SURVI PERM TAXES LAWS TICLE 10 GENEI SAFET EMER TICLE 11 | CONSTRUCT MATERIALS SURVEYS, P EYS TS AND LICE AND REGUL PROTECTION RAL PROVISION Y PRECAUTION GENCIES DRAWING | AND EMI ERMITS, L ENSES ATIONS ON OF WO | PLOYEESAWS, TAX | ES AND R | ENDEN | ATIONS | PERVISI | ON | |
| 8 9 10 11 11 12 | ART ART 9.1 9.2 9.3 9.4 ART 10.1 10.2 10.3 ART | TICLE 8 TICLE 9 SURVI PERM TAXES LAWS TICLE 10 GENEI SAFET EMER TICLE 11 TICLE 12 | MATERIALS SURVEYS, P EYS ITS AND LICE AND REGUL PROTECTION RAL PROVISION Y PRECAUTION GENCIES DRAWING REQUESTS | ENSES ATIONS ATIONS ON OF WOONS ONS S AND SPI | PLOYEES AWS, TAXI | ROPERTY ONS ON T | ENDEN | ATIONS | PERVISI | ON | |
| 8 9 10 11 11 12 | ART ART 9.1 9.2 9.3 9.4 ART 10.1 10.2 10.3 ART | TICLE 8 TICLE 9 SURVE PERM TAXES LAWS TICLE 10 GENEE SAFET EMER TICLE 11 TICLE 12 DETAI | CONSTRUCT MATERIALS SURVEYS, P EYS TS AND LICE AND REGUL PROTECTION AND PROVISION PRECAUTION GENCIES DRAWING REQUESTS L DRAWINGS | AND EMI ERMITS, L ENSES ATIONS ON OF WO ONS ONS S AND SPI FOR INFO | PLOYEES AWS, TAXI ORK AND P ECIFICATION TRUCTION | ROPERTY AND SCI | EGULA | ATIONS | PERVISI | ON | |
| 8 9 10 11 12 | ART 9.1 9.2 9.3 9.4 ART 10.2 10.3 ART 12.1 | TICLE 8 TICLE 9 SURVI PERM TAXES LAWS TICLE 10 GENEI SAFET EMER TICLE 11 TICLE 12 DETAI SCHEE | CONSTRUCT MATERIALS SURVEYS, P EYS TS AND LICE AND REGUL PROTECTION RAL PROVISION Y PRECAUTION GENCIES DRAWING REQUESTS L DRAWINGS OULES | AND EMI ERMITS, L ENSES ATIONS ON OF WO ONS ONS S AND SPI FOR INFO | PLOYEES AWS, TAXI ORK AND P ECIFICATION TRUCTION | ROPERTY ONS ON T | EGULA | ATIONS | PERVISI | ON | |
| 8 9 10 11 12 13 | ART 9.1 9.2 9.3 9.4 ART 10.2 10.3 ART 12.1 | TICLE 8 TICLE 9 SURVI PERM TAXES LAWS TICLE 10 GENEI SAFET EMER TICLE 11 TICLE 12 DETAI SCHEE TICLE 13 | MATERIALS SURVEYS, P TYS AND REGUL PROTECTION RAL PROVISION Y PRECAUTION GENCIES DRAWING REQUESTS L DRAWINGS DULES | AND EMI ERMITS, L ENSES ATIONS ON OF WO ONS S AND SPI FOR INFO S AND INS WINGS, P | PLOYEES AWS, TAXI ORK AND P ECIFICATION TRUCTIONS PRODUCT D | ROPERTY ONS ON TO | ENDEN EEGULA | ORK | PERVISI | ON | |
| 8 9 10 11 12 13 | ART ART 9.1 9.2 9.3 9.4 ART 10.1 10.2 10.3 ART 12.1 12.2 ART | TICLE 8 TICLE 9 SURVI PERM TAXES LAWS TICLE 10 GENEI SAFET EMER TICLE 11 TICLE 12 DETAI SCHEE TICLE 13 SUBM | CONSTRUCT MATERIALS SURVEYS, P EYS ITS AND LICE AND REGUL PROTECTION RAL PROVISION Y PRECAUTION GENCIES DRAWING REQUESTS L DRAWINGS OULES SHOP DRA | ENSES S AND SPICE AND INS S AND SPICE AND INS WINGS, PESS | PLOYEES AWS, TAXI DRK AND P ECIFICATION TRUCTIONS PRODUCT D | ROPERTY ONS ON TO | HE WO | ORK | PERVISI | ON | |
| 8 9 10 11 12 13 | ART ART 9.1 9.2 9.3 9.4 ART 10.2 10.3 ART 12.1 12.2 ART | TICLE 8 TICLE 9 SURVI PERM TAXES LAWS TICLE 10 GENEI SAFET EMER TICLE 11 TICLE 12 DETAI SCHEE TICLE 13 SUBM FABRI | CONSTRUCT MATERIALS SURVEYS, P EYS TS AND LICE AND REGUL PROTECTION RAL PROVISION Y PRECAUTION GENCIES DRAWING REQUESTS L DRAWINGS OULES SHOP DRA ITTAL PROCE | AND EMI ERMITS, L ENSES ATIONS ON OF WO ONS S AND SPI FOR INFO S AND INS WINGS, P ESS | PLOYEES AWS, TAXI DRK AND P ECIFICATION TRUCTION RODUCT D | ROPERTY ONS ON TO AND SCI | ENDEN EGULA | ORK | PERVISI | ON | |
| 8 9 10 11 12 13 | ART ART 9.1 9.2 9.3 9.4 ART 10.1 10.2 10.3 ART 12.1 12.2 ART 13.1 | TICLE 8 TICLE 9 SURVI PERM TAXES LAWS TICLE 10 GENEI SAFET EMER TICLE 11 TICLE 12 DETAI SCHEE TICLE 13 SUBM FABRI DEVIA | MATERIALS SURVEYS, P EYS AND REGUL PROTECTION RAL PROVISION PRECAUTION GENCIES DRAWING REQUESTS L DRAWINGS DULES SHOP DRA ITTAL PROCE | AND EMI ERMITS, L ENSES ATIONS ON OF WO ONS S AND SPI FOR INFO S AND INS WINGS, P ESS O ORDERING | PLOYEES AWS, TAXI DRK AND P ECIFICATION TRUCTIONS PRODUCT D | ROPERTY ONS ON TO AND SCORE OATA AND CIFICATION CIFICAT | HE WO | ORK | PERVISI | ON | |
| 8 9 10 11 12 13 | ART ART 9.1 9.2 9.3 9.4 ART 10.1 10.2 10.3 ART 12.1 12.2 ART 13.1 13.2 13.3 13.4 | TICLE 8 TICLE 9 SURVI PERM TAXES LAWS TICLE 10 GENEI SAFET EMER TICLE 11 TICLE 12 DETAI SCHEE TICLE 13 SUBM FABRI DEVIA CONS | CONSTRUCT MATERIALS SURVEYS, P EYS ITS AND LICE MATERIALS SURVEYS, P EYS ITS AND LICE MATERIALS AND REGULE PROTECTION GENCIES DRAWING REQUESTS L DRAWINGS OULES SHOP DRA ITTAL PROCE CATION AND TRUCTION M | AND EMI ERMITS, L ENSES ATIONS ON OF WO ONS S AND SPI FOR INFO S AND INS WINGS, P ESS ORDERING I DRAWIN IANAGER | PLOYEES AWS, TAXI DRK AND P ECIFICATION TRUCTIONS PRODUCT D IGS OR SPE REPRESENT | ROPERTY AND SCIENCE CIFICATIONS CIFICATIONS | HE WO | ORK | PERVISI | ON | |
| 8 9 10 11 12 13 | ART ART 9.1 9.2 9.3 9.4 ART 10.1 10.2 10.3 ART 12.1 12.2 ART 13.1 13.2 13.3 13.4 | TICLE 8 TICLE 9 SURVI PERM TAXES LAWS TICLE 10 GENEI SAFET EMER TICLE 11 TICLE 12 DETAI SCHEE TICLE 13 SUBM FABRI DEVIA CONST | CONSTRUCT MATERIALS SURVEYS, P EYS TS AND LICE AND REGUL PROTECTION RAL PROVISION Y PRECAUTION GENCIES DRAWING REQUESTS L DRAWINGS OULES SHOP DRA ITTAL PROCE CATION AND TRUCTION M SAMPLES | AND EMI ERMITS, L ENSES ATIONS ATIONS ON OF WO ONS ONS S AND SPI FOR INFO S AND INS O ORDERING I DRAWIN IANAGER AND TEST | PLOYEES AWS, TAXI ORK AND P ECIFICATION TRUCTION TRUCTION IG | ROPERTY ONS ON TO AND SCIENCE CATIONS | HE WOHEDUL | ORK | PERVISI | ON | |
| 8 9 10 11 12 13 14 | ART ART 9.1 9.2 9.3 9.4 ART 10.1 10.2 10.3 ART 12.1 12.2 ART 13.1 13.2 13.3 13.4 | TICLE 8 TICLE 9 SURVI PERM TAXES LAWS TICLE 10 GENEI SAFET EMER TICLE 11 TICLE 12 DETAI SCHEE TICLE 13 SUBM FABRI DEVIA CONS' TICLE 14 SAMP | CONSTRUCT MATERIALS SURVEYS, P EYS | AND EMI ERMITS, L ENSES ATIONS ON OF WO ONS ONS S AND SPI FOR INFO S AND INS WINGS, P ESS O ORDERIN I DRAWIN IANAGER AND TEST | PLOYEES AWS, TAXI ORK AND P ECIFICATION TRUCTION! PRODUCT D IG | ROPERTY ONS ON TO AND SCIENCE SAND SCIENCE S | HE WOHEDUL | ORK | PERVISI | ON | |

| 1 | 4.4 | TESTIN | G - OTHER | 20 |
|-----------|------|---------------|--|----|
| 15 | ARTI | CLE 15 | SUBCONTRACTS | 20 |
| 1 | 5.1 | CONTR | ACT PERFORMANCE OUTSIDE OF THE UNITED STATES OR COLORADO | 20 |
| 1 | 5.2 | SUBCO | NTRACTOR PREQUALIFICATION | 21 |
| 1 | 5.3 | SUBCO | NTRACTOR PROPOSALS | 21 |
| 1 | 5.4 | SUBCO | NTRACT FORMS | 21 |
| 1 | 5.5 | SUBCO | NTRACTOR SUBSTITUTION | 21 |
| 1 | 5.6 | CONST | RUCTION MANAGER RESPONSIBLE FOR SUBCONTRACTORS | 21 |
| 16 | ARTI | CLE 16 | RELATIONS OF CONSTRUCTION MANAGER AND SUBCONTRACTOR | 22 |
| 17 | ARTI | CLE 17 | MUTUAL RESPONSIBILITY OF CONTRACTORS | 22 |
| 18 | ARTI | CLE 18 | SEPARATE CONTRACTS | 22 |
| | | | USE OF PREMISES | |
| 20 | ARTI | CLE 20 | CUTTING, FITTING OR PATCHING | 23 |
| 21 | ARTI | CLE 21 | UTILITIES | 23 |
| 2 | 1.1 | TEMPO | RARY UTILITIES | 23 |
| 2 | 1.2 | PROTE | CTION OF EXISTING UTILITIES | 23 |
| 2 | 1.3 | CROSSI | NG OF UTILITIES | 24 |
| 22 | ARTI | CLE 22 | UNSUITABLE CONDITIONS | 24 |
| 23 | ARTI | CLE 23 | TEMPORARY FACILITIES | 24 |
| 2 | 3.1 | OFFICE | FACILITIES | 24 |
| 2 | 3.2 | TEMPO | RARY HEAT | 24 |
| 2 | 3.3 | WEATH | ER PROTECTION | 24 |
| 2 | 3.4 | DUST P | ARTITIONS | 24 |
| 2 | 3.5 | | MARKS | |
| 2 | 3.6 | SIGN | | 25 |
| 2 | 3.7 | SANITA | RY PROVISION | 25 |
| 24 | ARTI | CLE 24 | CLEANING UP | 25 |
| 25 | ARTI | CLE 25 | INSURANCE | 25 |
| 2 | 5.1 | | AL | |
| 2 | 5.2 | | ERCIAL GENERAL LIABILITY INSURANCE (CGL) | |
| 2 | 5.3 | AUTON | 1OBILE LIABILITY INSURANCE | 26 |
| 2 | 5.4 | | RS' COMPENSATION INSURANCE | |
| 2 | 5.5 | - | LLA LIABILITY INSURANCE: | |
| 2 | 5.6 | BUILDE | R'S RISK INSURANCE | 27 |
| 2 | 5.7 | | TON LIABILITY INSURANCE | |
| | 5.8 | | ONAL MISCELLANEOUS INSURANCE PROVISIONS | |
| | | | CONSTRUCTION MANAGER'S PERFORMANCE AND PAYMENT INSURANCE | |
| | | | LABOR AND WAGES | |
| 2 | 7.1 | | | |
| _ | 7.2 | | ADO WAGES | |
| | | | ROYALTIES AND PATENTS | |
| | | | ASSIGNMENT | |
| | | | CORRECTION OF WORK BEFORE ACCEPTANCE | |
| | | | APPLICATIONS FOR PAYMENTS | |
| _ | 1.1 | | RUCTION MANAGER'S SUBMITTALS | |
| _ | 1.2 | | ECT/ENGINEER CERTIFICATION | |
| | 1.3 | | AGE WITHHELD | |
| 3 | 1.4 | RELEAS | F OF RETAINAGE | 33 |

| 32 | ARTI | CLE 32 | CERTIFICATES FOR PAYMENTS | 34 |
|-----|--------------|---------------|--|--------|
| 33 | 3 ARTICLE 33 | | PAYMENTS WITHHELD | 34 |
| 34 | ARTI | CLE 34 | DEDUCTIONS FOR UNCORRECTED WORK | 35 |
| 35 | ARTI | CLE 35 | CHANGES IN THE WORK | 35 |
| | 35.1 | | LUE OF CHANGED WORK | |
| | 35.2 | | ED BREAKDOWN | |
| | 35.3 | | DOUS MATERIALS | |
| | 35.4 | | SENCY FIELD CHANGE ORDERED WORK | |
| | 35.5 | | PRIATION LIMITATIONS - C.R.S. § 24-91-103.6, as amended | |
| | | | CLAIMS | |
| | | | DIFFERING SITE CONDITIONS | |
| | 37.1 | | IN WRITING | |
| | 37.1 37.2 | | TIONS | |
| | | | DELAYS AND EXTENSIONS OF TIME | |
| | | | NON-BINDING DISPUTE RESOLUTION – FACILITATED NEGOTIATIONS | |
| | | | | |
| | | | RIGHT OF OCCUPANCY | |
| | | | COMPLETION, FINAL INSPECTION, ACCEPTANCE AND SETTLEMENT | |
| | 41.1 | | OF COMPLETION | |
| | 11.2 | | NSPECTION | |
| | 41.3 | | OF SUBSTANTIAL COMPLETION | |
| | 11.4 | | OF ACCEPTANCE | |
| | 41.5 | | MENT | |
| 42 | ARTI | CLE 42 | GENERAL WARRANTY AND CORRECTION OF WORK AFTER ACCEPTANCE | 50 |
| 43 | ARTI | CLE 43 | LIENS | 51 |
| 44 | ARTI | | ONE-YEAR GUARANTEE AND SPECIAL GUARANTEES AND WARRANTIES | |
| 4 | 14.1 | ONE-YE | EAR GUARANTEE OF THE WORK | 51 |
| 4 | 14.2 | SPECIA | L GUARANTEES AND WARRANTIES | 52 |
| 45 | ARTI | CLE 45 | GUARANTEE INSPECTIONS AFTER COMPLETION | 52 |
| 46 | ARTI | CLE 46 | TIME OF COMPLETION AND LIQUIDATED DAMAGES | 53 |
| 47 | ARTI | CLE 47 | DAMAGES | 54 |
| 48 | ARTI | CLE 48 | STATE'S RIGHT TO DO THE WORK; TEMPORARY SUSPENSION OF WORK; | DELAY |
| | DAM | IAGES | | 55 |
| 4 | 18.1 | STATE' | S RIGHT TO DO THE WORK | 55 |
| _ | 18.2 | | PRARY SUSPENSION OF WORK | |
| | 18.3 | | DAMAGES | |
| | | | STATE'S RIGHTS TO TERMINATE CONTRACT | |
| | 49.1 | | AL | |
| | 19.2 | | TIONS AND PROCEDURES | |
| | 19.3 | | ONAL CONDITIONS | |
| | | | TERMINATION FOR CONVENIENCE OF STATE | |
| | 50.1 | | OF TERMINATION | |
| - | 50.2 | | DURES | |
| | | | CONSTRUCTION MANAGER'S RIGHT TO STOP WORK AND/OR TERMINATE CON | |
| 21 | | CLE 21 | CONSTRUCTION WIANAGER S RIGHT TO STOP WORK AND/OR TERMINATE CO | VIRACI |
| E 2 | | | COLORADO SPECIAL PROVISIONS | 60 |
| | | | | |
| | 52.1 | | OLLER'S APPROVAL, C.R.S. § 24-30-202(1) | |
| | | | AVAILABILITY, C.R.S. § 24-30-202(5.5) | |
| | 2.3 | GUVER | NMENTAL IMMUNITY | b() |

| 52.4 | INDEPENDENT CONTRACTOR | 60 |
|---------|--|----|
| 52.5 | COMPLIANCE WITH LAW | 61 |
| 52.6 | CHOICE OF LAW, JURISDICTION, AND VENUE | 61 |
| 52.7 | PROHIBITED TERMS | |
| 52.8 | SOFTWARE PIRACY PROHIBITION. SOFTWARE PIRACY PROHIBITION | 61 |
| 52.9 | EMPLOYEE FINANCIAL INTEREST/CONFLICT OF INTEREST | 61 |
| 52.10 | VENDOR OFFSET AND ERRONEOUS PAYMENTS | 61 |
| 53 ARTI | CLE 53 MISCELLANEOUS PROVISIONS | 62 |
| 53.1 | PROFESSIONAL ASSOCIATION PERMITTED | 62 |
| 53.2 | DISSOLUTION OF PROFESSIONAL ASSOCIATION | 62 |
| 53.3 | WAGE RATES, in accordance with C.R.S. § 24-30-1404 (1) | 62 |
| 53.4 | PUBLIC ART LAW | 62 |
| 53.5 | ASSIGNMENT | 62 |
| 53.6 | SUBCONTRACTS | 62 |
| 53.7 | BINDING EFFECT | 63 |
| 53.8 | AUTHORITY | |
| 53.9 | CAPTIONS AND REFERENCES | |
| 53.10 | COUNTERPARTS | 63 |
| 53.11 | ENTIRE UNDERSTANDING | 63 |
| 53.12 | DIGITAL SIGNATURES | 63 |
| 53.13 | MODIFICATION | |
| 53.14 | STATUTES, REGULATIONS, FISCAL RULES AND OTHER AUTHORITY | 63 |
| 53.15 | EXTERNAL TERMS AND CONDITIONS | 63 |
| 53.16 | SEVERABILITY | |
| 53.17 | SURVIVIAL AND CERTAIN CONTRACT TERMS | |
| 53.18 | TAXES | |
| 53.19 | THIRD PARTY BENEFICIARIES | 64 |
| 53.20 | WAIVER | |
| 53.21 | CORA DISCLOSURE | |
| 53.22 | STANDARD AND MANNER OF PERFORMANCE | |
| 53.23 | LICENSES, PERMITS, AND OTHER AUTHORIZATIONS | |
| 53.24 | INDEMNIFICATION | |
| | ACCESSIBILITY | |
| 54 ARTI | CLE 54 CONFIDENTIAL INFORMATION-STATE RECORDS | |
| 54.1 | CONFIDENTIALITY | |
| 54.2 | OTHER ENTITY ACCESS AND NONDISCLOSURE AGREEMENTS | |
| 54.3 | USE, SECURITY, AND RETENTION | |
| 54.4 | INCIDENT NOTICE AND REMEDIATION | |
| 54.5 | DATA PROTECTION AND HANDLING | |
| 54.6 | SAFEGUARDING PERSONAL IDENTIFIABLE INFORMATION (PII) | |
| | A DESIGNATED SERVICES AND METHOD OF PAYMENT | |
| | B CONSTRUCTION MANAGER'S CERTIFICATION | |
| | C REQUEST FOR PROPOSAL | |
| | D CONSTRUCTION MANAGER'S FEE PROPOSAL | |
| | D.1 CONSTRUCTION MANAGER'S FEE PROPOSAL | |
| | D.2 CONSTRUCTION MANAGER'S CERTIFICATE OF INSURANCE | |
| | D.3 CONSTRUCTION MANAGER'S DIRECT LABOR BURDEN CALCULATION | |
| FXHIBIT | E BUILDING CODE COMPLIANCE POLICY | |

| EXHIBIT F LIST OF PRE-QUALIFIED SUBCONTRACTORS (BASED ON SCHEMATIC DESIGN DOCUMENTS) . $oldsymbol{I}$ |
|---|
| EXHIBIT G SCHEMATIC DESIGN ESTIMATE SUMMARY AND UPDATED SUMMARIES |
| EXHIBIT H AMENDMENT (INCORPORATING GMP) EXHIBITS BASED ON DESIGN DEVELOPMEN |
| DOCUMENTS |
| EXHIBIT H.1 GMP Documents, Drawings and Specifications including Addenda and Modifications I |
| EXHIBIT H.2 Schedule of Bid Package Descriptions and Issuance Dates (as applicable) |
| EXHIBIT H.3 Schedule of Values (prepared at the time of the GMP Amendment) |
| EXHIBIT H.4 Allowance Schedule (prepared at the time of the GMP Amendment) |
| EXHIBIT H.5 Detailed Critical Path Method Construction Schedule (Prepared at the time of the GMI |
| Amendment) I |
| EXHIBIT I SUBSEQUENT AMENDMENTS (INCORPORATING BID PACKAGES) EXHIBITS |
| EXHIBIT I.1 Drawings, Specifications and Addenda(When approved by the Principal Representative). |
| EXHIBIT I.2 Schedule of Values (Consistent with GMP Schedule of Values) |
| EXHIBIT I.3 Labor Overhead (SBP-6.18) for each Subcontractor to be applied to all change orders and |
| amendments) |
| EXHIBIT I.4 Allowance Schedule (consistent with GMP Allowance Schedule) |
| EXHIBIT I.5 Performance Bond (Form SC-6.22) |
| EXHIBIT I.6 Labor and Material Payment Bond (Form SC-6.221) |
| EXHIBIT I.7 Insurance Certificates |
| EXHIBIT I.8 Detailed Critical Path Method Construction Schedules (when approved by the Principal |
| Representative) |
| EXHIBIT I.9 Applicable Prevailing Wage Determinations (If applicable) |
| EXHIBIT I.10 Apprenticeship Utilization Certifications (SBP-2.1) (If applicable) |
| EXHIBIT J Notice to Proceed to Commence Construction Phase (Form SC-7.26) |
| EXHIBIT M SUPPLEMENTARY GENERAL CONDITIONS: FEDERAL PROVISIONS |
| EXHIBIT N NOTICE OF (PARTIAL) SUBSTANTIAL COMPLETION |
| EXHIBIT O NOTICE OF APPROVAL OF OCCUPANCY/USE |
| EXHIBIT P NOTICE OF (PARTIAL) FINAL ACCEPANCE |
| EXHIBIT Q NOTICE OF CONTRACTOR'S SETTLEMENT |
| EXHIBIT S University of Colorado Denver Anschutz Medical Campus Supplementary Genera |
| Conditions |
| EXHIBIT T SALES AND USE TAX FORM |
| EXHIBIT Y Service-Disabled Veteran-Owned Small Business and Minority/Women Business Enterprise |
| Participation Report |
| EXHIBIT 7. MISCELLANEOUS DOCUMENTS |

SIGNATURE PAGE

THE PARTIES HERETO HAVE EXECUTED THIS AGREEMENT

Each person signing this Agreement represents and warrants that the signer is duly authorized to execute this Agreement and to bind the Party authorizing such signature.

| Project Number/Name: _ / | | | | | |
|---|---|--|--|--|--|
| Contract ID No.: N/A | | | | | |
| | | | | | |
| CONTRACTOR* | STATE OF COLORADO Board of Regents of the University of Colorado, a body corporate, for and on behalf of the University of Colorado Denver | | | | |
| By: , Owner Date: | By: , Date: | | | | |
| DEPARTMENT OF PERSONNEL & ADMINISTRATION | LEGAL REVIEW | | | | |
| STATE BUILDINGS PROGRAM State Architect (or authorized delegate) | Steve Zweck-Bronner, Chris Puckett or Jenny Willits; Special Assistant Attorney General | | | | |
| By: , Senior Director Facilities Projects and State Buildings | By: | | | | |
| Delegate | 37 | | | | |
| Delegate | Date: | | | | |
| Date: | | | | | |
| Dutc | | | | | |
| In accordance with §24-30-202, C.R.S., this Contract is not val authorized delegate) or the Financial Of | | | | | |
| Associate Vice Chancellor for Financial Services and Controller | | | | | |
| By:, Associate Vice Chancellor for Financial Services and Controller or Delegate | | | | | |
| Effective Date: | | | | | |

STATE OF COLORADO OFFICE OF THE STATE ARCHITECT STATE BUILDINGS PROGRAM

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT

| Department ID: | GFE | Contract ID #: | Project #: | |
|----------------|-----|----------------|------------|--|
| _ opa | U | 00 | | |

PARTIES. THIS AGREEMENT is entered into by and between the STATE OF COLORADO, acting by and through the <u>Board of Regents of the University of Colorado</u>, a body corporate, for and on behalf of the <u>University of Colorado Denver</u> hereinafter referred to as the State or Principal Representative, and having its offices at hereinafter referred to as the Construction Manager or the contractor.

EFFECTIVE DATE AND NOTICE OF NONLIABILITY. This Agreement shall not be effective or enforceable until it is approved and signed by the State Controller or its designee (hereinafter called the "Effective Date"), but shall be effective and enforceable thereafter in accordance with its provisions. The State shall not be liable to pay or reimburse Construction Manager for any performance hereunder or be bound by any provision hereof prior to the Effective Date.

RECITALS:

WHEREAS, the Principal Representative intends to engage the services of the Construction Manager to construct the following: procure /,, hereinafter called the Project; and

WHEREAS, authority exists in the Law and Funds have been budgeted, appropriated, and otherwise made available, and a sufficient unencumbered balance thereof remains available for payment in In Fund Number N/A, Account Number N/A; and

WHEREAS, the State of Colorado has appropriated and the Principal Representative has been authorized to expend the total sum of Dollars (\$) for this Project including all professional services, construction management/general contractor services, construction/improvements, Project contingencies, reimbursables, furnishings, movable equipment, and miscellaneous expenses; and

WHEREAS, funds are available for only a portion of the services defined herein, as more fully described in the funding Condition Precedent clause in Article 6 hereof; and

WHEREAS, the Principal Representative has established the Fixed Limit of Construction Cost in the amount Dollars (\$); and

WHEREAS, the Construction Manager shall establish a Guaranteed Maximum Price (GMP) that is within the Fixed Limit of Construction Cost as established by the Principal Representative at the completion of the Design Development Phase; and

WHEREAS, in accordance with Article 5 Compensation the Construction Manager's Total Fee and General Conditions for the Project is **Dollars (\$)**; and

WHEREAS, the Architect/Engineer for the project is; and

WHEREAS, the Construction Manager acknowledges the statutory authority and responsibility of the Principal Representative within the State of Colorado; and

WHEREAS, the Construction Manager was selected after a determination that its proposal was the most advantageous to the Principal Representative pursuant to a request for proposal issued and awarded on ; and

WHEREAS, the Construction Manager and the Principal Representative have finalized the terms of this Agreement pursuant to the Colorado Procurement Code or the applicable procurement code for institutions of higher education.

WITNESSETH, that the Principal Representative and the Construction Manager agree as follows:

1 ARTICLE 1 PERFORMANCE OF THE WORK

1.1 THE WORK

- 1.1.1 The Construction Manager will construct the Project within the Fixed Limit of Construction Cost specified, and the Construction Manager will furnish all the services, labor and materials to perform all the Work, including design, for the complete and prompt execution of the Project in accordance with the Contract Documents as identified in Section 1.2.
- 1.1.2 In the performance of the Work under this Agreement, the Construction Manager acknowledges that time is critical for Project delivery and that portions of the Work could have their design completed as separate Bid Packages and under construction before other portions of the Work are fully designed. It is further recognized that this accelerated approach to construction utilizing the services of an Architect/Engineer and Construction Manager is defined as "Fast Track Construction" and requires maximum cooperation between all parties. It is also recognized that the services to be rendered by the Construction Manager and the inter-relationships and coordinative aspects thereof are in the developmental state and not fully defined. The Construction Manager has reviewed the Architect/Engineer's Agreement and accepts the terms thereof as expressing a workable concept. In furtherance thereof, in the event there appears to be a duplication, overlap or conflict of the responsibilities of or duties between the Architect/Engineer and Construction Manager or an absence of designation, the question shall be submitted to the Principal Representative for determination. The Construction Manager shall abide by the decision of the Principal Representative provided it does not require the performance of work beyond what was reasonably contemplated and accepted by the Construction Manager as its responsibility. If the Construction Manager claims any increase in the Work arises by virtue of such a decision.
- 1.1.3 The Construction Manager acknowledges that the Principal Representative shall provide () **Bid Packages** to accomplish the Work. In the event the Construction Manager for any reason within the Construction Manager's control, requests more than () **Bid Packages** to be furnished by the Principal Representative, the Principal Representative shall make arrangement with the Architect/Engineer for the additional Bid Packages desired and shall directly compensate the Architect/Engineer for all fees and cost associated therewith. The Construction Manager shall reimburse the Principal Representative for all of the Architect/Engineer's fees and costs associated therewith.
- 1.1.4 The Construction Manager agrees to use best efforts, to cooperate fully with the Principal Representative in the construction aspects of the Work, and to keep within the Principal Representative's monetary, schedule and quality limitations, as stipulated within this Agreement.

- 1.1.5 The organization of the Specifications into division, section, and article, and the arrangement of Drawings shall not control the Construction Manager in dividing the Work among any level of Subcontractors or in establishing the extent of the Work to be performed by any trade.
- 1.1.6 The Construction Manager understands the relationship of trust and confidence established between it and the Principal Representative and accepts those responsibilities as described in this Agreement. The Construction Manager covenants with the Principal Representative to furnish its best skill and judgment and to cooperate with the Architect/Engineer in furthering the interests of the Principal Representative. The Construction Manager agrees to furnish efficient business administration and superintendence and to use its best efforts to complete the Work in an expeditious and economical manner consistent with the interest of the Principal Representative.
- 1.1.7 The Construction Manager, the Principal Representative, and the Architect/Engineer, called the Construction Team, shall work during design through to construction completion. The Construction Manager shall provide leadership to the Construction Team on all matters relating to construction.
- 1.1.8 The Architect/Engineer is a representative of the Principal Representative as provided in the Contract Documents and its Agreement is with the Principal Representative. In case of termination of employment or death of the Architect/Engineer, the Principal Representative shall appoint a capable and reputable Architect/Engineer against whom the Construction Manager makes no reasonable objection, whose status under the Agreement shall be the same as that of the former Architect/Engineer.
- 1.1.9 The Architect/Engineer shall not be responsible for or have control or charge of construction means, methods, techniques, sequences or procedures, or for safety precautions and programs in connection with the Work and except for the Architect/Engineer's specifically enumerated Contract Administration duties such as observation of the Work, shall not be responsible for the Construction Manager's failure to carry out the Work in accordance with the Contract Documents. The Architect/Engineer shall not be responsible for or have control or charge over the acts or omissions of the Subcontractors of any tier or any of their agents or employees, or any other persons performing any of the Work.
- 1.1.10 The Contract Documents shall not be deemed to create any contractual relationship between the Architect/Engineer and the Construction Manager or any separate contractors, subcontractors of any tier or suppliers on the Project; nor shall anything contained in the Contract Documents be deemed to give any third party any claim or right of action against the Principal Representative, the Architect/Engineer or Construction Manager which does not otherwise exist without regard to the Contract Documents.
- 1.1.11 The initial Work of the Construction Manager shall consist of its services in connection with the Preconstruction Phase. The Preconstruction Phase of the Work shall be parallel and coincidental with the Schematic Design, Design Development, and Construction Document Phases of the Architect/Engineer's Services. As the Bid Packages are prepared and prices are established for the work to be performed within each respective Bid Package, the parties contemplate that the work to be performed by the Construction Manager shall be adjusted by Amendment or Change Order to this Agreement to place the work contained within the various Bid Packages within the work to be performed by the Construction Manager with corresponding adjustments made to the Contract Sum, Guaranteed Maximum Price and Contract Time.

1.2 CONTRACT DOCUMENTS

1.2.1 The Contract Documents are described in Article 1.1 of the General Conditions of the Construction Manager/General Contractor Agreement (SC-6.51) and are essential parts of this Agreement and are fully incorporated herein.

2 ARTICLE 2 EXHIBITS TO THE AGREEMENT

2.1 EXHIBITS

The following Exhibits are, or will be, attached to this Agreement and are or shall become when approved and accepted, part of the Contract Documents.

- 1. The Drawings released for Construction (Exhibit I.1);
- 2. The Specifications released for Construction (Exhibit I.1);
- 3. Exhibit A, Construction Manager's Designated Services and Method of Payment;
- 4. **Exhibit B,** Construction Manager Certification;
- 5. Exhibit C, Request for proposal (minus blank template forms) dated Aug 29, 2023;
- 6. Exhibit D1, Construction Manager's Fee Proposal dated;
- 7. **Exhibit D2,** Construction Manager's Certificates of Insurance;
- 8. **Exhibit D3,** Construction Manager's Direct Labor Burden Calculation;
- 9. Exhibit E, State Sales and Use Tax Forms;

Amendment Exhibits based on Schematic Design Documents

- 10. **Exhibit F,** List of Pre-Qualified Subcontractors (Add by amendment when approved by the Principal Representative and prior to bidding);
- 11. **Exhibit G,** Schematic Design Estimate Summary and Updated Summaries (Add by amendment when approved by the Principal Representative);

Amendment (incorporating GMP) Exhibits based on Design Development Documents

- 12. **Exhibit H.1,** GMP Documents, Drawings and Specifications including Addenda and Modifications (When approved by the Principal Representative);
- 13. Exhibit H.2, Schedule of Bid Package Descriptions and Issuance Dates (as applicable);
- 14. Exhibit H.3, Schedule of Values (prepared at the time of the GMP Amendment);
- 15. Exhibit H.4, Allowance Schedule (prepared at the time of the GMP Amendment);
- 16. **Exhibit H.5,** Detailed Critical Path Method Construction Schedule (Prepared at the time of the GMP Amendment);

Subsequent Amendments (incorporating Bid Packages) Exhibits

- 17. **Exhibit I.1,** Drawings, Specifications and Addenda(When approved by the Principal Representative);
- 18. Exhibit 1.2, Schedule of Values (Consistent with GMP Schedule of Values);
- 19. **Exhibit I.3,** Labor Overhead (SBP-6.18) for each Subcontractor to be applied to all change orders and amendments);
- 20. Exhibit 1.4, Allowance Schedule (consistent with GMP Allowance Schedule);
- 21. Exhibit I.5, Performance Bond (Form SC-6.22);
- 22. Exhibit I.6, Labor and Material Payment Bond (Form SC-6.221);
- 23. **Exhibit 1.7**, Insurance Certificates;
- 24. **Exhibit 1.8**, Detailed Critical Path Method Construction Schedules (when approved by the Principal Representative);
- 25. **Exhibit I.9**, Applicable Prevailing Wage Determinations (If applicable)
- 26. Exhibit I.10, Apprenticeship Utilization Certifications (SBP-2.1) (If applicable)

Miscellaneous Exhibits

- 27. Exhibit J, Notice to Proceed to Commence Construction Phase (Form SC-7.26) (when issued);
- 28. **Exhibit K**, Building Code Compliance Policy: Coordination of Approved Building Codes, Plan Reviews, and Building Inspections.
- 29. Exhibit M, Supplementary General Conditions: Federal Provisions (if applicable);
 - a. SLFRF Federal Funds: Contractor Terms and Conditions Certification (if applicable); and

- b. SLFRF Federal Funds: Contractor Terms and Conditions (if applicable).
- 30. Exhibit N, Notice of (Partial) Substantial Completion
- 31. Exhibit O, Notice of Approval of Occupancy/Use
- 32. Exhibit P, Notice of (Partial) Final Acceptance
- 33. Exhibit Q, Notice of Contractor's Settlement
- 34. **Exhibit S**, University of Colorado Denver | Anschutz Medical Campus Supplementary General Conditions.
- 35. **Exhibit Z,** Service-Disabled Veteran-Owned Small Business and Minority/Women Business Enterprise Participation Report

3 ARTICLE 3 CONSTRUCTION MANAGER'S SERVICES

The Construction Manager shall perform the following services under this Agreement in each of the phases described below:

PRE-CONSTRUCTION SERVICES

3.1 AVAILABLE FUNDS

3.1.1 The Construction Manager acknowledges that the Principal Representative is limited in the sum available to design and construct the Project. Should funding of a lesser amount be made available for the Project, it is the obligation of the Principal Representative to revise the Project Scope consistent with the ultimate appropriation.

3.2 BUDGETING AND FIXED LIMIT OF CONSTRUCTION COST

- 3.2.1 The Construction Manager shall assist the Architect/Engineer in evaluating the Principal Representative's preliminary budget. Based on consultation with the Architect/Engineer and the Construction Manager, the Principal Representative shall furnish a Project Budget to the Construction Manager which shall set forth a dollar amount available for the total Construction Cost of the Project and include contingencies for bidding and construction.
 - a) The Fixed Limit of Construction Cost has been established by the Principal Representative, converting the applicable portion of the Project Budget into the Fixed Limit of Construction Cost, as set forth in the Recitals above.
 - b) The Fixed Limit of Construction Cost may be revised only by approved Amendments and Change Orders issued after execution of the Contract Documents.

3.3 CONSULTATION AND VALUE ENGINEERING

- 3.3.1 The Construction Manager shall provide consultation throughout the Preconstruction and Construction Phases including but not limited to the furnishing of Value Engineering Services to identify cost effective changes in the State's specifications that will result in reducing the Contract Price without impairing essential functions or characteristics. The objective of Value Engineering is to achieve optimum value for each construction dollar spent and keep the time of completion and cost of the Work within the time and fiscal constraints set forth throughout the Contract Documents. In cooperation with the Principal Representative, the Construction Manager shall:
 - a) Formulate and evaluate alternative designs, systems, materials, etc.;
 - b) Provide cost estimates of the alternatives to be evaluated. Cost estimates shall include industry standard operating and maintenance costs when appropriate to evaluate life-cycle costs of the alternatives. Cost estimates shall take into consideration all cost impacts related to alternatives including but not limited to construction costs. The Construction Manager shall, at a minimum, review the cost estimate at the completion of the Schematic Design, and

- Design Development Phases and include an analysis and commentary as to any discrepancies observed in the report referenced in paragraph 3.3.1.d below;
- c) Evaluate the alternatives on the basis of costs, time schedules, availability of labor and materials, construction feasibility, etc.;
- d) With the assistance of the Architect/Engineer to prepare written reports at the end of the Schematic Design and Design Development Phases summarizing the Value Engineering activities accomplished and any recommendations developed within each phase;
- e) If Estimates of Construction and/or bids received for the Work contained in any Bid Package cause the anticipated cost of the Work to exceed the then current Estimate of Construction Cost, the Fixed Limit of Construction Cost, the Guaranteed Maximum Price or Schedule of Values, the Construction Manager shall, at no additional cost to the Principal Representative unless caused by an increase in the Construction Manager 's Work requested by the Principal Representative, provide additional Value Engineering services in conjunction with any and all appropriate items in the Estimate of Construction, the Fixed Limit of Construction Cost, the Guaranteed Maximum Price, and/or the Schedule of Values for the Work; and
- f) Lead a formal Value Engineering workshop as requested by the Principal Representative, at the end of the Schematic Design, Design Development and Construction Documents Phases review and estimating tasks, bringing multidiscipline cost/construction experts to evaluate alternative designs, systems and materials.
- 3.3.2 The Principal Representative shall participate in the formulation and evaluation of alternatives in the Value Engineering activity, and shall approve Value Engineering alternatives accepted in each design phase.

3.4 CONSTRUCTION COSTS

- 3.4.1 It is the desire of the Principal Representative to incorporate as many alternate bid items into the Project as reasonable, in order to maximize the scope for the Fixed Limit of Construction Cost. All parties recognize that the availability of costs to perform the Work depend, in part, upon favorable market conditions. With thorough and careful planning, cost estimating and cooperation, funds may become available for the alternates through the procurement process at less than the Construction Manager's estimated cost therefore. Together with savings through the unexpended portion of the bidding contingency, the Principal Representative may authorize alternates and/or additional scope, all within the Fixed Limit of Construction Cost.
- 3.4.2 To accomplish the inclusion of alternates and/or increases, the project contingency shall be as follows and included in the GMP:
 - a) The **bidding contingency** for all Bid Packages together with the construction of the Work shall be equal to **two-point five percent (2.5%)** of the total Guaranteed Maximum Price, all within the Fixed Limit of Construction Cost.
 - The bidding contingency shall be allocated between the presently () Bid Packages, equally proportionate to the value associated with each Bid Package. The Construction Manager shall notify, in writing, the Principal Representative of the allocation of the bidding contingency for each Bid Package.
 - b) The **construction contingency** for the Work shall be equal to **three percent (3%)** of the total Guaranteed Maximum Price, all within the Fixed Limit of Construction Cost.
- 3.4.3 At the conclusion and award of the Bid Packages, all differences between the Construction Manager's estimated cost of the Work contained within the Bid Packages, exclusive of contingency, versus the actual cost thereof as determined by bidding and award (buyouts) shall be promptly calculated and totaled. If the total of all of the buyouts exceed the Construction Manager estimated cost therefore, the bidding contingency identified in paragraph 3.4.2.a shall

- be applied by the Construction Manager, after prior written notice to the Principal Representative, to cover any overrun per Bid Package.
- 3.4.4 After all of the Bid Packages have been bought out, and subcontracts and purchase agreements have been executed, any and all savings achieved through the buyouts of the Bid Packages together with all unexpended sums remaining in the bidding contingencies shall forthwith accrue to the Principal Representative to be applied by the Principal Representative, in its sole and absolute discretion, to the inclusion of desired alternates into the Work or to otherwise increase the scope of Work to be performed by the Construction Manager, and/or to reduce the Guaranteed Maximum Price.
- 3.4.5 The construction contingency shall only be used to cover costs for labor, materials, equipment and similar costs for items or Work to be furnished during the construction phase of the Project. It is not the intent of this Agreement to use the construction contingency for costs incurred during the Pre-Construction phase or bidding phase or for costs to correct any errors, omissions, mistakes or rejected Work caused by Subcontractors. The construction contingency may be used to cover the Construction Manager's costs (i) arising from estimating cost overruns in the costs of Exhibit H.4 Allowance Schedule; (ii) unexpected additional trade coordination costs incurred for Work directly performed by the Construction Manager that could not have been reasonably contemplated; (iii) items required and reasonably inferable from the Contract Documents, or items included within the Contract Documents but missed within the Subcontractor buy-out which the Construction Manager can show were not specifically called out within the estimate or bid documents of the Construction Manager or any Subcontractor; (iv) losses or damages to property related to the Work not covered by insurance provided by the Construction Manager and including any insurance deductible(s); (v) arising from expediting or acceleration of the Project schedule where such cost is not a basis for an increase in the GMP under Article 7.6 of the agreement,; (vi) Bid Package buyout overrun costs for additional Bid Packages that were not part of anticipated Construction Phases defined in 1.1.3, but only if bidding procedures contemplated hereunder were followed and the bidding contingency has been exhausted; or (vii) other costs incurred not reasonably to have been expected that are approved by the Principal Representative in the Principal Representative's sole discretion, so long as those costs are not recovered under any insurance policy provided pursuant to this Agreement and so long as the total costs under this Agreement do not exceed the Guaranteed Maximum Price. No expenditure from the construction contingency for any matters or Work activities shall be made without the prior written approval of the Principal Representative, which approval, with the exception of item (vii) above, shall not be unreasonably withheld. Expenditures from the construction contingency shall be made only by Change Order.

Construction contingency shall include all costs associated with a stated scope including, if applicable Direct Work, Insurance, Bonds, Fee, and General Conditions (if appropriate).

- 3.4.6 When preparing any estimates of Construction and in development of the Schedule of Values, such documents shall include, without duplication:
 - a) All labor, materials, equipment, tools, construction equipment and machinery, water, heat, utilities, transportation, and other facilities and services necessary for the proper execution and completion of the Work, whether temporary or permanent, and whether or not incorporated or to be incorporated in the Work;
 - b) Any allowance designated by the Principal Representative;
 - c) Any Principal Representative furnished equipment which has been designed, specified, selected or specifically provided for by the Architect/Engineer;
 - d) The Construction Manager's fee and the cost of work provided by the Construction Manager;
 - e) All bonds, insurance premiums and applicable taxes;

- f) Contingencies for bidding, price escalation, and construction;
- g) Plumbing and electrical building permits from appropriate entities and any other building permits as directed by the Principal Representative; and
- h) Administrative expenses directly related to the Work. (Refer to Section 12.2, Schedules of the General Conditions of the Contract (SC-5.51)
- 3.4.7 Estimates of Construction Cost shall not include the compensation of the Architect/Engineer and, the Architect/Engineer mechanical, electrical, plumbing, structural, civil, and any other consultants and subconsultants required in the Request For Proposal or any other sums due the Architect/Engineer and it's consultants, and shall not include the costs of land, right of way, financing or other costs, which are the responsibility of the Principal Representative.
- 3.4.8 The Construction Manager, in preparing its Estimates of Construction cost and providing the Guaranteed Maximum Price, shall consult with the Architect/Engineer to determine what materials, equipment, component systems and types of construction are to be included in the Contract Documents, to make reasonable adjustments in the scope of the Work, and to include in the Contract Documents alternate items, as approved by the Principal Representative in writing, for bid so as to permit the adjustment of the Estimate of Construction Cost.
- 3.4.9 The Construction Manager shall prepare an Estimate of Construction as soon as major Project requirements have been identified and update it periodically. For the Schematic Design Phases, the Construction Manager shall prepare a quantity take-off cost estimate based on building systems, assemblies, components, etc., and update periodically. During the Design Development Phase, the Construction Manager shall prepare a final cost estimate in preparation for a Guaranteed Maximum Price and update periodically. All Estimates of Construction shall include separate defined allowances for bidding and Construction price escalation. During the Construction Documents Phase, the Construction Manager shall continually monitor the cost estimates and develop a cost estimate to help assure that the cost of the Work remains within the applicable portion of the Project Budget, Fixed Limit of Construction Cost and Guaranteed Maximum Price.
- 3.4.10 Estimates shall be prepared and shall be based on quantitative takeoffs whenever possible and shall be supported in sufficient depth and organization to be used in preparing budgets based on Construction Specifications Institute (CSI) Division, funding sources, sub-trades, combinations of sub-trades, building systems, Bid Packages or combinations thereof. The specific cost coding structure, estimating guidelines, assumptions, and contents of the cost estimates shall be mutually resolved between the Construction Manager and the Architect/Engineer prior to development of the first cost estimate to assure that estimates developed by all parties can be compared and reconciled. Lump sum estimates are not acceptable.
- 3.4.11 During the preparations of cost estimates, the Construction Manager shall notify the Principal Representative if it appears that the Estimate of Construction will exceed the applicable portion of the Project Budget or Fixed Limit of Construction Cost as may be applicable, satisfactorily demonstrate the accuracy of its estimate in such detail as shall be reasonably required by the Principal Representative, and make reasonable recommendations for corrective action consistent with the Project Budget or Fixed Limit of Construction Cost, as may be applicable. The Construction Manager shall submit Estimates of Construction Cost to the Principal Representative for review and acceptance at each design milestone and other times as required by the Principal Representative to analyze various building systems and components. Concurrently, the Construction Manager shall provide copies to the Architect/Engineer for review and verification.
- 3.4.12 The Principal Representative shall reasonably cooperate with the Construction Manager to keep the Work within the applicable portions of the Project Budget or Fixed Limit of Construction Cost,

as may be applicable, including but not limited to the giving of appropriate and reasonable consideration to all reasonable recommendations of the Construction Manager, approving redesign, only for Principal Representative directed scope changes, deductive alternatives or reductions in Work, requesting additional Value Engineering, making modifications to the Contract Documents or exercising such other rights or remedies as may be available elsewhere under this Agreement including termination for convenience. However, the Principal Representative shall be under no duty to reduce the Work to accommodate for any projected costs over or beyond the Guaranteed Maximum Price that is the responsibility of the Construction Manager or allow access to the construction contingency to cover costs to correct errors, omissions, mistakes, rejected Work or warranty Work.

3.4.13 Architect/Engineer/Construction Manager Cooperation: The Architect/Engineer, by the terms of its agreement with the Principal Representative, is obligated to provide reasonable cooperation to the Construction Manager in the development of Estimates of Construction Cost and the Guaranteed Maximum Price. Conversely, the Construction Manager, by the terms of this Agreement is obligated to provide reasonable cooperation to the Architect/Engineer in the development of Statements of Probable Construction Cost and the Guaranteed Maximum Price. Additionally, both Architect/Engineer and Construction Manager are obligated to reconcile their respective cost estimates at the completion of each design phase of the Work including the Guaranteed Maximum Price in a timely manner so as not to negatively impact the Project Schedule.

3.5 OTHER PRE-CONSTRUCTION SERVICES

- 3.5.1 The Construction Manager shall perform those items designated as Required Services as set forth in the Designated Services and Method of Payment schedule designated as **Exhibit A**. In addition, and not in limitation, the Construction Manager shall also perform the other Pre-Construction Services designated in this Article 3 together with such other services as are normally and customarily provided by a Construction Manager.
- 3.5.2 Meeting Attendance: The Construction Manager shall attend all regular meetings with the Principal Representative and such additional meetings as the Principal Representative may request. All regular meetings shall be scheduled by the Architect/Engineer with the Construction Manager and approval of the Principal Representative. All additional meetings shall be requested by the Principal Representative.
- 3.5.3 Copies for Review: The Principal Representative through the Architect/Engineer and consistent with the Principal Representative's contract with the Architect/Engineer, shall furnish the Construction Manager a sufficient quantity of documents required for the Preconstruction Services.
- 3.5.4 The Construction Manager shall make recommendations to the Principal Representative and the Architect/Engineer regarding the Division of Work in the Drawings and Specifications to facilitate the bidding and awarding of subcontracts, allowing for phased construction and funding, if applicable, taking into consideration such factors as time of performance, availability of labor, overlapping trade jurisdictions, provisions for temporary facilities, etc.
- 3.5.5 The Construction Manager shall review Drawings and Specifications with the Architect/Engineer to (1) eliminate areas of conflict, overlapping trade jurisdictions, and overlapping in the Work to be performed by the various Subcontractors, (2) endeavor to confirm that all Work has been included, and (3) allow for phased construction as applicable.
- 3.5.6 The Construction Manager shall participate in Project Design Review Sessions at the close of the Schematic Design Phase, the Design Development Phase, and as Construction Documents are finalized for each Bid Package. The Project design review sessions shall be attended by the

Architect/Engineer and representatives of the Principal Representative. The purposes of the Project design review sessions are to (1) assure consistency with the design intent; (2) confirm complete, coordinated, constructible and cost-effective designs for all disciplines (e.g. architectural, structural, mechanical); (3) assure that the design documents are code compliant per **Exhibit K** Approved Building Codes Plan Reviews and Building Inspections; (4) endeavor to confirm that all Work has been included and described in sufficient detail to assure complete pricing of Work; and (5) allow for phased construction. The Architect/Engineer shall collect all design review comments from the various participants, provide reports to the Principal Representative, and confirm that with the issuance of each progress set of design documents all comments have either been incorporated or resolved to the satisfaction of the Principal Representative.

- 3.5.7 The Construction Manager recognizes that the Principal Representative is a Governmental Body with certain procedural requirements to be satisfied. The Construction Manager has and shall make reasonable allowance in its performance of the Work for such additional time as may be required for approvals and decisions by the Principal Representative, in addition to the times specifically provided in paragraph 3.5.8.
- 3.5.8 In the Review Process of the final Design Development Documents and Construction Documents for each Bid Package, the Construction Manager expressly agrees to the following review times by the Principal Representative:
 - a) A period of fourteen (14) days for the review of the Design Development Documents; and
 - b) A period of fourteen (14) days prior to completion of the Construction Documents together with an additional seven (7) days after receipt of all bid documents for each Bid Package, commencing with the date of receipt by the Principal Representative of all documents and any other items which are required to be furnished to the Principal Representative by the terms of the Principal Representative's contract with the Architect/Engineer.
- 3.5.9 As part of the Schematic Design review and estimating tasks, the Construction Manager shall develop a preliminary detailed Critical Path Method (CPM) Project Schedule as described in Article 12 of the General Conditions of the Contract (SC-6.51), that is coordinated with the milestone dates specified in **Exhibit H.2**, the Date of Completion specified in paragraph 4.2.1, the scope of Work described within the Contract Documents, and the Work described within the Schematic Design Documents. The Construction Manager shall utilize the Project Management Software as described in paragraph 3.6.4 to develop and manage the schedule.
- 3.5.10 Principal Representative Purchasing: The Construction Manager shall investigate and recommend materials and equipment that could be purchased by the Principal Representative; consider long lead time procurement and mass purchasing power in making such recommendations; recommend a schedule for such purchases after coordination with the schedule for preparation of Contract Documents; and expedite and coordinate delivery of these purchases to facilitate their delivery by the required dates.
- 3.5.11 The Construction Manager shall prepare necessary bidding information, bidding forms, and prequalification criteria for bidders; develop subcontractor interest in the Project; establish bidding schedules; advertise for bids; and conduct pre-bid conferences to familiarize bidders with the bidding documents and management techniques and with any special systems, materials, or methods. As soon as the Construction Manager becomes aware prior to any bid date that less than three (3) pre-qualified subcontractors plan to bid any portion of any Bid Package or that anticipated bids from previously approved or pre-qualified subcontractors listed on **Exhibit F**, are likely to exceed the then current Schedule of Values or Estimate of Construction Cost, the Construction Manager shall promptly so notify the Principal Representative and Principal

- Representative shall be entitled to treat the situation as an unforeseeable circumstance pursuant to Article 15.5 of the General Conditions of the Contract (SC-6.51)
- 3.5.12 The Construction Manager shall receive and open bids when advertised, prepare a bid analysis, conduct pre-award conferences, and notify the Principal Representative concerning which bids shall be accepted. The Principal Representative shall be notified in advance of the time and place of all bid openings and may elect to attend such openings with their representatives. Should the Construction Manager submit a proposal for subcontract Work (Work not included in the Construction Manager's Construction Phase Fee and/or General Conditions) herein referred to as "Self-Perform Work", the proposal conditions shall be the same as for all subcontractor proposals. These Construction Manager proposals for subcontract Work shall be submitted to the Principal Representative twenty-four (24) hours prior to receipt of other subcontractor proposals and all opened with the other proposals. The Construction Manager's team performing such work may include its employees, material providers, and sub-subcontractors. If the Construction Manager is chosen to perform the work, then such work shall be performed for the Construction Manager's bid amount on the basis of a stipulated lump sum. A proposal to accept other than a low bid shall be justified in writing by the Construction Manager and subject to prior written approval by the Principal Representative.
- 3.5.13 The Construction Manager shall provide the requirements and assignment of responsibilities for safety precautions and programs as required for the execution of the Work, temporary project facilities and for equipment, materials and services for common use of subcontractors and verify that all are included in the Contract Documents.
- 3.5.14 The Construction Manager shall provide not later than the first of each month, unless requested otherwise by the Principal Representative, a monthly report utilizing the Project Management Software described in paragraph 3.6.4 documenting the current status of the project's schedule, costs, requests for information, submittals, manpower, safety, and other pertinent information. The report shall include a narrative discussion of the progress achieved, activities anticipated for the next month, and issues that are affecting the rate of progress. Progress photographs should be attached/included. This monthly report shall be provided in Construction Phases of the project. The schedule status shall include the following minimum items:
 - a) Cost report showing activity dollar value, dollar value of Work in place to-date and dollar value for current period.
 - b) Cost report showing activity dollar value, dollar value of Work in place to-date, and dollar value for current period summarizing to schedule of values.
 - c) Resource report showing man-day allocations by specific trade on each activity.
 - d) Variance report comparing current dates to target dates.
 - e) Cash flow report showing monthly projections of expenditures.

A narrative schedule report shall document:

- a) Description of the actual Work accomplished during the reporting period.
- b) Description of any problem areas.
- c) Description of current and anticipated delays with recommended corrective actions to mitigate such delays.
- d) A list of proposed modifications, additions, deletions, and changes in logic to the approved schedule.

CONSTRUCTION PHASE SERVICES

3.6 CONTROL OF THE WORK

- 3.6.1 The Construction Manager shall supervise and direct the Work of its Subcontractors and shall coordinate the Work with the activities and responsibilities of the Principal Representative to complete the Project in accordance with the Principal Representative's objectives of cost, time and quality and subject to the terms and conditions of the General Conditions of the Contract (SC-6.51).
- 3.6.2 The Construction Manager shall establish on-site organization and lines of authority in order to carry out the overall plans of the Construction Team.
- 3.6.3 The Construction Manager shall schedule and conduct weekly progress meetings at which the Principal Representative, Construction Manager, Architect/Engineer, and Architect/Engineer's Consultants, can discuss jointly such matters as procedures, progress, schedule, costs, quality control and problems. The Construction Manager shall record and distribute minutes of all construction meetings within 48 hours of the meeting.
- 3.6.4 A contract-control/project-management software (hereafter "Project Management Software") approved by the Principal Representative, shall be used as a primary tool for project control, communication and documentation control by all the project participants, to include the Principal Representative, the Construction Manager and the Architect/Engineer. The Construction Manager shall utilize the Project Management Software to implement a cost forecasting, monitoring, control and reporting system for the Project. The Project Management Software shall be maintained throughout the project, both during the pre-construction and construction phases. Cost analyses shall be based upon data analyses as developed/described within Section 3.3 and shall include analyses of all trades and Project components making a significant contribution for total Project costs. The Project Management Software shall provide for development of a Project cost model, monitoring the design process and periodic reviews of the cost estimates/forecasts to identify variances from the cost model. Additionally, the Project Management Software shall identify variances between actual and budgeted costs and the Fixed Limit of Construction Cost and the Contract Sum.
 - The Construction Manager shall use the Project Management Software for the major contract administration processes to include, but not limited to:
- 3.6.4.1 Submittals: Construction Manager shall create a Submittal log and Submittal schedule. Submittals shall be directly submitted to the Architect/Engineer and Principal Representative and directly returned from the Architect/Engineer.
- 3.6.4.2 Requests for Information: Construction Manager shall submit requests for information using the Project Management Software. Architect/Engineer shall answer requests for information via the Project Management Software. Requests for Information responses that have cost impact will have corresponding Change Order Bulletin (Form SC-6.311) issued by the Architect/Engineer.
- 3.6.4.3 Change Management: Entire change management process including Notices, and Change Orders shall be managed using the Project Management Software and utilizing Contract Amendment (Form SC-6.0A,B or C), Change Order (Form SC-6.31B), Change Order Bulletin (Form SC-6.311), Change Order Proposal (Form SC-6.312) and Change Order Log.
- 3.6.4.4 Pay Applications: Construction Manager shall be responsible for creating and distributing pay application in the Project Management Software using an earned-value calculation through the CPM Schedule & utilizing Application and Certificate for Contractor's Payment (SBP-7.2).
- 3.6.4.5 Meeting Minutes: Construction Manager shall be responsible for creating and distributing construction-meeting minutes in the Project Management Software.

- 3.6.4.6 Reports: Construction Manager shall be responsible to prepare and distribute reports in the Project Management Software.
- 3.6.4.7 Insurance Certificate: Construction Manager shall responsible for storing all the insurance related information of Subcontractors in the Project Management Software.
- 3.6.4.8 Punchlist: Construction Manager shall be responsible to update the Substantial Completion Punchlist status using the Project Management Software.
- 3.6.4.9 Construction Schedule: Critical Path Method as described in Article 12 of the General Conditions of the Contract (SC-6.51).
- 3.6.4.10 All project correspondence with Principal Representative shall be in the Project Management Software.
- 3.6.4.11 The Construction Manager shall propose and implement an approved procedure for coordinating and tracking all required Code Compliance Building Inspections as indicated on the Building Inspection Record (BIR) as provided by the State Buildings Program approved Code Review Agent at the appropriate Construction Phase(s) as described in the attached **Exhibit K**.

3.7 SCHEDULE AND COORDINATION

3.7.1 The Construction Manager shall begin the construction Work upon receiving the Notice to Proceed to Commence Construction Phase (SBP-7.26), in accordance with Article 4.1. The Construction Manager shall schedule and coordinate the Work of all of its Subcontractors on the Project including their use of the site. The Construction Manager shall keep the Subcontractors informed of the Project construction schedule to enable the Subcontractors to plan and perform the Work properly. The Construction Manager shall carry the Work forward expeditiously with adequate forces and shall achieve Completion of the Work prior to the Contract Completion Date specified in Section 4.2, as adjusted by Change Orders and Amendments.

3.7.2 Schedule Management

- 3.7.2.1 Schedule Modifications: If, as a result of the monthly schedule update the Project Schedule no longer represents the actual / logical progression of the Work or the Construction Manager's plan for prosecution and progress of the Work, the Principal Representative shall require the Construction Manager to submit a revision to the Project Schedule. Such revisions to the Schedule shall not alter any of the Project Milestone dates.
- 3.7.2.2 Schedule Impacts, Schedule Delays, Time Extensions: During the course of the Project, it may be appropriate to revise the Schedule to incorporate impacts or delay issues into the Project Schedule. If the Construction Manager determines it has encountered schedule impacts that may warrant a time extension, the Construction Manager shall present an Impacted Schedule in accordance with Article 38 of the General Conditions of the Contract (SC6.51), to the Principal Representative supporting its claim. Note that time extension change order requests, due to weather or other construction delays, do not necessarily or automatically translate into budgetary change order increases.
- 3.7.2.3 Recovery Schedule: In the event progress falls behind schedule dates, the Construction Manager shall prepare a recovery schedule indicating its revised plan to assure the timely completion of the Work. The recovery schedule shall be subject to the Principal Representative's approval.

3.8 AMENDMENTS AND CHANGE ORDERS

3.8.1 The Construction Manager shall assist in developing and implementing a system for the preparation, processing and tracking of Amendments and Change Orders using the Project Management Software as described in paragraph 3.6.4 and recommend necessary or desirable changes to the Principal Representative. Fully executed and approved Change Orders shall

constitute obligations of the Principal Representative to pay as part of the Contract Sum the amounts identified by such modifications so long as such amounts do not exceed the Guaranteed Maximum Price. Change Orders shall only be used to effect changes in the Work which apply to bidding and construction contingency amounts set forth in paragraphs 3.4. 1through 3.4.5. Any Changes in the Work that result in an increase in said contingencies shall be added to this Agreement by an amendment pursuant to Article 35 of the General Conditions of the Contract (SC-6.51).

3.9 PRINCIPAL REPRESENTATIVE CONSULTANTS

3.9.1 If required, the Construction Manager shall assist the Principal Representative in selecting and retaining the professional services including but not limited to a surveyor, geotechnical, testing and inspection and other special consultants, and coordinate these services, without assuming any responsibility or liability of or for these consultants.

3.10 START UP

- 3.10.1 The Construction Manager, with the Principal Representative's maintenance staff and/or consultant, shall direct the checkout of utilities, operations, systems and equipment for readiness and assist in their initial start-up and testing/commissioning as required in the Scope Narrative with the Subcontractors of all tiers.
- 3.10.2 Prior to the Date of Completion of the Work or earlier date for phased occupation of the Work as requested by the Principal Representative, the Construction Manager shall schedule and conduct with the Principal Representative and the Architect/Engineer a complete review, commissioning, demonstration, start-up and operational testing of all equipment and mechanical and electrical systems installed by the Construction Manager or its Subcontractors on the Project, and shall also review the operation and maintenance of such systems with the Principal Representative's maintenance personnel.
- 3.10.3 Subsequent to this review, the Construction Manager, with reasonable promptness and at no cost to the Principal Representative, shall make all adjustments or corrections required by the Principal Representative or the Architect/Engineer and shall balance all systems in order to make all equipment and systems perform as required by the Contract Documents and to reflect the actual use and occupancy of the Project. If necessary or requested by the Architect/Engineer or Principal Representative, the Construction Manager shall require the Subcontractor, supplier of material supplier to make adjustments, corrections or balancing required by this process, at no additional cost to the Principal Representative.

4 ARTICLE 4 TIME OF COMMENCEMENT AND COMPLETION

4.1 COMMENCEMENT

- 4.1.1 The Contract Time shall commence on the Effective Date of this Agreement but no Work shall be performed prior to the delivery of all bonds, and insurance certificates as required to be furnished by the Construction Manager and described on the Notice of Award.
- 4.1.2 The Construction Phase shall commence on the date the first Bid Package is added to this Agreement by Amendment as approved by the Principal Representative in accordance with Section 1.1.3.
- **4.1.3** The commencement of the Construction Phase is expressly conditioned upon and shall not commence until:

SC-6.5

- The Guaranteed Maximum Price and Schedule of Values shall have been timely submitted (or such timeliness shall have been waived in writing by the Principal Representative and the State Architect) and shall have been approved and accepted by the Principal Representative;
- b) The date for Completion of the Work has been approved and accepted by the Principal Representative;
- c) All required Performance and Labor and Material Payment Bonds and insurance certificates have been approved and accepted by the Principal Representative; and
- d) Exhibit J, Notice To Proceed to Commence Construction Phase (SBP-7.26) has been issued by the Principal Representative and made a part of the Contract Documents.

 If any of the preceding material conditions to be performed by the Construction Manager have not been fully satisfied by reason of any act or omission on the part of the Construction Manager through no fault of the Principal Representative, the Principal Representative shall give the Construction Manager written notice of any and all such deficiencies and allow ten (10) days from the date of such notice to correct and cure such deficiency or deficiencies, and in the event the deficiency or deficiencies are not fully corrected and cured within the ten (10) day period, the Principal Representative may declare the Construction Manager to be in default of this Agreement.

4.2 COMPLETION

4.2.1 The Construction Manager agrees to Substantially Complete the Project in the number of days established in Exhibit I.8, Detailed Critical Path Method Construction Schedules and as described in Article 12 of the General Conditions of the Contract (SC-6.51). The Date of Completion may be revised by mutual Agreement. The Construction Manager shall perform the Work with due diligence to completion.

5 ARTICLE 5 COMPENSATION

5.1 CONSTRUCTION MANAGER'S FEE

5.1.1 Subject to the provisions of this Agreement and of the General Conditions of the Construction Contract (SC-6.51), and in consideration of the performance of this Agreement, the Principal Representative shall pay the Construction Manager in current funds as compensation for its services as listed below:

| Construction Manager's Fee | |
|---|----|
| 1 Pre-Construction Phase Fee | \$ |
| 2 Construction Phase Fee | \$ |
| 3 Total Fee (1+2) | \$ |
| General Conditions | |
| 4 Direct Personal Expenses of On-Site CM/GC Staff (Not to Exceed) | \$ |
| 5 Other Reimbursable General Conditions (Not to Exceed per Pgh 9.1.3) | \$ |
| 6 Total GCs(including Direct Personnel Expenses of Staff) (4+5) | \$ |
| Total Fee and General Conditions (3+6) | \$ |

- 5.1.2 The Construction Manager's Fee shall include all job indirect costs, and General Conditions costs as defined in **Exhibit A**, Construction Manager Designated Services and Method of Payment, home office overhead, and profit, included but not limited to the following:
 - a) Salaries or other compensation of the Construction Manager's employees at the principal office and branch offices;
 - b) General operating expenses of the Construction Manager's principal and branch offices other than the field office;
 - c) Any part of the Construction Manager's capital expenses, including interest on the Construction Manager's capital employed for the Project;
 - d) Overhead or general expenses of any kind;
 - e) Salaries of the Construction Manager's employees engaged on the road in expediting the production or transportation of materials and equipment;
 - f) Cost of all employee benefits and taxes for such items as unemployment compensation and social security, insofar as such cost is based on wages, salaries or other remuneration paid to employees of the Construction Manager and included in the fee under paragraphs 5.1.2.a through 5.1.2.f;
 - g) All transportation, traveling, moving, and hotel expenses of the Construction Manager or its officers or employees incurred in discharge of duties connected with the Work;
 - h) Costs, including transportation and maintenance, of all materials, supplies, equipment, temporary facilities, and hand tools not owned by the workmen, which are employed or consumed in the performance of the Work;
 - i) Cost of the premium for all insurance which the Construction Manager is required to procure by this Agreement or is deemed necessary by the Construction Manager;
 - j) Minor expenses such as facsimile messages, telegrams, long distance telephone call telephone service at the site, express mail, and similar petty cash items in connection with the Work;
 - k) All other items set forth in Exhibit A, Construction Manager Designated Services and Method of Payment that are specifically designated as Pre-Construction Services Fee, Construction Services Fee or General Conditions. All Items listed in the columns designated Direct Cost of Work shall be included in the separate Bid Packages.
 - I) Except as expressly provided to the contrary elsewhere in this Agreement, approved costs in excess of the Guaranteed Maximum Price.
- 5.1.3 General conditions items, as set forth in paragraph 5.1.2, shall generally include the cost of Construction Phase on-site construction management staff and those temporary facilities, services and equipment to support the Work of construction Subcontractors. General conditions items are more fully identified in **Exhibit A**, Designated Services and Methods of Payment, and shall be reimbursed at cost, without mark-up, based upon pre-approved not-to-exceed budgets. General conditions (exclusive of the Construction Manager's staff) provided directly by the Construction Manager must be at market competitive rates. Each monthly request for progress payment shall be justified with reasonable support for expenses to include:
 - a) Invoice or receipt for any vendors or suppliers for material, rented equipment, etc.
 - b) Labor/timesheet reports (by task number) for direct labor, provide bare labor rate and itemized breakdown of labor burden prior to initial billing.
 - c) Owned equipment shall be compensated per pre-negotiated rates established in accordance with the Colorado Procurement or applicable procurement code for institutions of higher education. In no case shall cumulative/total cost of owned equipment exceed the value of the

- equipment minus salvage value. The Principal Representative shall approve all rental rates and salvage values in writing prior to initial billing.
- d) Labor, material and equipment cost may be audited by the Principal Representative.
- e) Principal Representative agrees Construction Manager's wage, equipment and insurance rates included in the Guaranteed Maximum Price proposal are fixed rates not subject to further review or audit by either party.

5.2 ADJUSTMENTS IN FEE

- 5.2.1 Adjustments in the Construction Phase Fee shall be made as follows: If, after the total Guaranteed Maximum Price is accepted, in writing, by the Principal Representative, the Principal Representative directs additions to or other changes made in the Work, the Construction Manager's fee shall be adjusted as follows:
 - a) If the changes in the aggregate increase the total Guaranteed Maximum Price, the Construction Manager's fee for any and all other changes in the Work shall be calculated at the rate of FOUR percent (4%) (plus appropriate General Condition costs) of the estimated cost of such work and shall be agreed upon between the Construction Manager and the Principal Representative as a fixed fee for the effect of the change (or changes), prior to starting the changed Work. The adjustments stated above shall only be deemed valid after the Principal Representative accepts the adjustments in writing and, are the only adjustments to the fee that shall be granted for changes authorized to the GMP. Adjustments to these fees beyond these values shall not be granted. However, General Condition costs directly attributable to time extensions may be charged in accordance with Article 38 of the General Conditions of the Contract (SC-6.51).
- 5.2.2 The Construction Manager shall also be paid an additional fee at the rate as set forth in paragraph 5.2.1 if the Construction Manager is placed in charge of the reconstruction of any insured loss.
- 5.2.3 If there is a material reduction in the scope of Work greater than fifteen percent (15%) of the Fixed Limit of Construction Cost, the Construction Manager's Fees shall be reduced proportionally after the fifteen percent (15%).

5.3 PAYMENT OF FEE

- 5.3.1 Preconstruction Services Phase Fee: For the performance of the Preconstruction Services ending with the execution of the first amendment establishing and accepting the Guaranteed Maximum Price of the Work, the fee therefore as set forth in paragraphs 5.1.1 shall be paid monthly and as described in Article 31 of the General Conditions of the Contract (SC-6.51) with the total payment not to exceed the fee for such services as set forth in paragraph 5.1.1.
- 5.3.2 Construction Services Phase Fee: For the Performance of the Construction Services after the execution of the first amendment establishing and accepting the Guaranteed Maximum Price of the Work and with the addition of the second Amendment incorporating the first Bid Package and subsequent Amendments incorporating subsequent Bid Packages, the fee therefore as set forth in paragraph 5.1.1 shall be paid monthly as described in Article 31 of the General Conditions of the Contract (SC-6.51) with the total payment not to exceed the fee for such services as set forth in paragraph 5.1.1. and the portion of the fee to be paid shall be equivalent to the ratio of the dollar value of each Bid Package to the Guaranteed Maximum Price including the premiums for the Performance and Labor and Materials Payment Bonds with coverage up to the value of the Contract Sum.

5.4 GUARANTEED MAXIMUM PRICE

5.4.1 At the conclusion of the Design Development Phase, the Construction Manager shall deliver to the Principal Representative, a Guaranteed Maximum Price proposal which shall agree to perform

all of the Work even though all of the Construction Documents have not all been finalized and released for construction, and guarantee the maximum price to the Principal Representative for the entire cost of the Work, as adjusted by deductive alternates required to maintain the Guaranteed Maximum Price below the Fixed Limit of Construction Cost which have been previously approved by the Principal Representative pursuant to paragraph 3.2.1.b.

- 5.4.2 The Guaranteed Maximum Price shall include all of the Construction Manager's obligations to be performed pursuant to the terms of the Contract Documents and may include, but not be limited to, the total of the following:
 - a) The total of all prices already received for all items bid before the establishment of the Guaranteed Maximum Price;
 - b) The Construction Manager's estimate of the cost of all other Work to be performed but not yet bid, excluding the approved deductive alternates unless said Work can be incorporated into the Contract Documents by application of the contingency per the provisions of paragraphs 3.4.1 through 3.4.5, with the consent of the Construction Manager which consent shall not be unreasonably withheld;
 - c) The installation cost of items to be procured by the Principal Representative and assigned to the Construction Manager for installation, as defined in the Contract Documents;
 - d) The estimated maximum cost of all Work to be performed by the Construction Manager;
 - e) Construction Manager's Fee as provided under this Agreement;
 - f) The cost of all Performance and Labor and Material Payment Bonds furnished by the Construction Manager pursuant to Article 26 of the General Conditions of the Contract (SC 6.51);
 - g) The premiums for insurance to protect the Project pursuant to the Article 25 of the General Conditions of the Contract (SC 6.51); and
 - h) Authorized adjustments as set forth elsewhere in this Agreement, to include but may not be limited to: taxes; fees for licenses, and royalties; special conditions, commissioning, start-up services, and warranty support; and contingencies.
- 5.4.3 The Guaranteed Maximum Price proposal as set forth in Section 5.4 shall:
 - a) Set forth a stated not to exceed dollar amount;
 - b) Set forth the Schedule of Values therefore which shall be consistent with previously approved Schedules of Values, and a adjusted as required pursuant to Design Development cost estimating;
 - c) Contain no conditions or exceptions; except as set forth in the Guaranteed Maximum Price proposal;
 - d) Not exceed the Fixed Limit of Construction Cost;
 - e) Contain no allowances except for those set forth in **Exhibit H.4**, Allowance Schedule, of which all allowances are to be a not-to-exceed dollar amount; whenever costs are more than or less than allowances, the Guaranteed Maximum Price shall be adjusted accordingly by Amendment. The amount of the Amendment shall reflect the difference between actual costs and the allowances; and
 - f) Be substantiated with complete supporting documentation acceptable to the Principal Representative, to clearly define the anticipated Work to be performed by the Construction Manager and facilitate a determination thereafter when final drawings and specifications are released for construction, as to whether or not there has been an increase in the Work required of the Construction Manager in the documents released for construction from the Design Development documents on which the Guaranteed Maximum Price was based. If at

any time thereafter, any Claim is asserted by the Construction Manager for an increase to the Contract Sum or Guaranteed Maximum Price and/or extension of the Contract Time because of an alleged increase in the Work to be performed by the Construction Manager as contained in the drawings or specifications released for construction, the Construction Manager shall be required to satisfactorily demonstrate the increase in the Work; otherwise the Construction Manager shall be entitled to no increase in the Contract Sum, Guaranteed Maximum Price or extension of the Contract Time.

- 5.4.4 If, through no fault on the part of the Construction Manager, and after receiving reasonable cooperation by the Principal Representative, the Construction Manager submits a Guaranteed Maximum Price proposal contrary to the provisions of paragraphs 5.4.2 and 5.4.3, the proposal may be rejected by the Principal Representative; the Principal Representative shall be under no obligation to award subsequent Bid Packages; the Principal Representative may declare the Construction Manager to be in default; and payment may be withheld from the Construction Manager, excepting the Construction Manager's reasonable costs incurred, up and until a Guaranteed Maximum Price is furnished in accordance with the foregoing.
- 5.4.5 If, in developing a Guaranteed Maximum Price, the Construction Manager believes any documentation or information, consistent with the Design Development level of documentation, is not sufficiently complete to clearly define the anticipated Work, the Construction Manager shall be responsible for making all necessary inquires and requests to establish the same.
- S.4.6 When the Guaranteed Maximum Price is agreed upon and accepted by the Principal Representative, it shall be made a part of the Contract Documents by Amendment, shall supersede updated summaries, and all documents relating to Schedules of Values and Estimates of Construction; and shall be subject to modification for Changes in the Work as provided in the Article 35 of the General Conditions of the Contract (SC-6.51). If the Construction Manager, in good faith, furnishes the Principal Representative with a Guaranteed Maximum Price proposal which meets the criteria of paragraphs 5.4.1, 5.4.2, and 5.4.3 and the parties fail to mutually agree to that number as set forth above, the parties expressly agree that default termination of the Construction Manager shall not be a remedy therefore under this Agreement, and, the Principal Representative shall be entitled to proceed with the Project and Work as set forth elsewhere in this Agreement.
- 5.4.7 When the Construction Manager provides a Guaranteed Maximum Price, the trade contracts for the Work shall either be with the Construction Manager or shall contain the necessary provisions to allow the Construction Manager to control the performance of the Work. The Principal Representative shall also authorize the Construction Manager to take all steps necessary in the name of the Principal Representative to assure that any separate contractors, having separate contracts with the Principal Representative for the Project, perform their contracts in accordance with their terms.

5.5 CONTRACT SUM

- 5.5.1 Subject to the provisions of Article 3 and Article 5 of the Agreement and Article 50, of the General Conditions of the Contract (SC-6.51) the Contract Sum shall be the total amount payable by the Principal Representative to the Construction Manager for the performance of all Work under the Contract Documents and equal to the total of:
 - a) The Construction Costs as set forth in paragraphs 3.4.6 and 3.4.7.
 - b) Authorized adjustments as set forth elsewhere in this Agreement;

ARTICLE 6 CONDITION PRECEDENT

6.1 FUTURE APPROPRIATIONS

Financial obligations of the Principal Representative payable after the current fiscal year are contingent upon funds for the purpose being appropriated, budgeted, and otherwise made available.

6.2 OBLIGATIONS

At the time of the execution of this Agreement, there are sufficient funds budgeted and appropriated to compensate the Construction Manager only for performance of the Work through and including. Therefore, it shall be a Condition Precedent to the Construction Manager's performance of the remaining Work specified in and the State's liability to pay for such performance, sufficient funding must be made available to the Principal Representative for the Project prior to and, as a further Condition Precedent, a written Amendment to this Agreement is entered into in accordance with the State of Colorado Fiscal Rules, stating that additional funds are lawfully available for the Project. If either Condition Precedent is not satisfied by , the Construction Manager's obligation to perform Work for and the State obligation to pay for such Work is discharged without liability to each other. If funding is eventually made available after, the Construction Manager has no right to perform the Work under of this Agreement and the State has no right to require the Construction Manager to perform said Work.

ARTICLE 7 OPTIONAL PROVISIONS AND ELECTIONS

The provisions of this Article 7 alter or enlarge upon the following Articles of the General Conditions of the Contract (SC-6.51).

7.1 MODIFICATION OF ARTICLE 2. EXECUTION, CORRELATION, INTENT OF DOCUMENTS, COMMUNICATION AND COOPERATION If the box below is marked, certification of apprenticeship utilization is required for all mechanical, sheet metal, fire suppression, sprinkler fitting, electrical and plumbing work on the project. Principal Representative initial 7.2 MODIFICATION OF ARTICLE 45. GUARANTEE INSPECTIONS AFTER COMPLETION If the box below is marked the six-month guarantee inspection is not required. Principal Representative initial 7.3 MODIFICATION 1 OF ARTICLE 27. LABOR AND WAGES If the box is marked the Federal Davis-Bacon Act shall be applicable to the Project. The minimum wage rates to be paid on the Project shall be furnished by the Principal Representative and included in the Contract Documents. ☐ Principal Representative initial

7.4 MODIFICATION 2 OF ARTICLE 27. LABOR AND WAGES

If the box is marked, the State prevailing wage statute shall be applicable to the Project. The minimum wage rates to be paid on the Project shall be furnished by the Principal Representative and included in the Contract Documents.

Principal Representative initial

7.5 MODIFICATION OF ARTICLE 39. NON-BINDING DISPUTE RESOLUTION - FACILITATED **NEGOTIATIONS**

If the box is marked, and initialed by the State as noted, the requirement to participate in facilitated negotiations shall be deleted from this Contract. Article 39, Non-Binding Dispute Resolution – Facilitated Negotiations, shall be deleted in its entirety and all references to the right

| to the same wherever they appear in the contract shall be similarly deleted. The box may be | эe |
|---|----|
| marked only for projects with an estimated value of less than \$500,000. | |
| ☐ Principal Representative initial | |

7.6 MODIFICATION OF ARTICLE 46. TIME OF COMPLETION AND LIQUIDATED DAMAGES

If an amount is indicated immediately below, liquidated damages shall be applicable to this Project as, and to, the extent shown below. Where an amount is indicated below, liquidated damages shall be assessed in accordance with and pursuant to the terms of Article 46, Time of Completion and Liquidated Damages, in the amounts and as here indicated. The election of liquidated damages shall limit and control the party's right to damages only to the extent noted.

- 7.6.1 For the inability to use the Project, for each day after the number of calendar days specified in the Construction Manager's proposal for the Project and the Agreement for achievement of Substantial Completion, until the day that the Project has achieved Substantial Completion and the Notice of Substantial Completion is issued, the Construction Manager agrees that an amount equal to Dollars (\$); shall be assessed against Construction Manager from amounts due and payable to the Construction Manager under the Agreement, or the Construction Manager and the Construction Manager's Surety shall pay to the Principal Representative such sum for any deficiency, if amounts on account thereof are deducted from remaining amounts due, but amounts remaining are insufficient to cover the entire assessment.
- 7.6.2 For damages related to or arising from additional administrative, technical, supervisory and professional expenses related to and arising from the extended closeout period, for each day in excess of the number of calendar days specified in the Construction Manager's proposal for the Project and the Agreement to finally complete the Project as defined by the issuance of the Notice of Final Acceptance) after the issuance of the final Notice of Substantial Completion, the Construction Manager agrees that an amount equal to **Dollars (\$)**; shall be assessed against Construction Manager from amounts due and payable to the Construction Manager under the Agreement, or the Construction Manager and the Construction Manager's Surety shall pay to the Principal Representative such sum for any deficiency, if amounts on account thereof are deducted from remaining amounts due but amounts remaining are insufficient to cover the entire assessment.

8 ARTICLE 8 NOTICE IDENTIFICATION

All Notices pertaining to the Agreement or the General Conditions of the Contract (SC-6.51) or otherwise required to be given shall be transmitted in writing, to the individuals at the addresses listed below, and shall be deemed duly given when received by the parties at their addresses below or any subsequent persons or addresses provided to the other party in writing.

Notice to Principal Representative:

With copies to State Buildings Program (or Delegate): , Senior Director Facilities Projects and State Buildings Delegate 1945 Wheeling Street, MailStop F418, Aurora, CO 80045 todd.akey@cuanschutz.edu

Notice to Contractor:

, Owner

With copies to:

Robert Holzwarth, robert.holzwarth@cuanschutz.edu

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

ATTACHMENT 1 THE GENERAL CONDITIONS OF THE CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC 6.51)



THE GENERAL CONDITIONS OF THE CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT

(STATE FORM SC 6.51)

THE GENERAL CONDITIONS OF THE CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.51)

| 1 ARTICLE 1 DEFINITIONS | |
|---|----------------------|
| | 1 |
| 1.1 CONTRACT DOCUMENTS | |
| 1.2 DEFINITIONS OF WORDS AND TERMS USED | 2 |
| 2 ARTICLE 2 EXECUTION, CORRELATION, INTENT OF DOCUMENTS, COMMUNICA | TION AND |
| COOPERATION | 7 |
| 2.1 EXECUTION | 7 |
| 2.2 CORRELATION | |
| 2.3 INTENT OF DOCUMENTS | |
| 2.4 PARTNERING, COMMUNICATIONS AND COOPERATION | |
| 3 ARTICLE 3 COPIES FURNISHED | |
| 4 ARTICLE 4 OWNERSHIP OF DRAWINGS | |
| 5 ARTICLE 5 ARCHITECT/ENGINEER'S STATUS | |
| 6 ARTICLE 6 ARCHITECT/ENGINEER DECISIONS AND JUDGMENTS, ACCESS TO | |
| INSPECTION | |
| 6.1 DECISIONS | |
| 6.3 ACCESS TO WORK | |
| 6.4 INSPECTION | |
| 7 ARTICLE 7 CONSTRUCTION MANAGER'S SUPERINTENDENCE AND SUPERVISION | - |
| ARTICLE / CONSTRUCTION WARRACK S SOI EMITTED ENCE AND SOI ENVISION | |
| 8 ARTICLE 8 MATERIALS AND EMPLOYEES | 12 |
| 8 ARTICLE 8 MATERIALS AND EMPLOYEES 9 ARTICLE 9 SURVEYS, PERMITS, LAWS, TAXES AND REGULATIONS | |
| 8 ARTICLE 8 MATERIALS AND EMPLOYEES 9 ARTICLE 9 SURVEYS, PERMITS, LAWS, TAXES AND REGULATIONS 9.1 SURVEYS | 13 |
| 9 ARTICLE 9 SURVEYS, PERMITS, LAWS, TAXES AND REGULATIONS | 13 |
| 9 ARTICLE 9 SURVEYS, PERMITS, LAWS, TAXES AND REGULATIONS | 13 13 |
| 9 ARTICLE 9 SURVEYS, PERMITS, LAWS, TAXES AND REGULATIONS | 13 13 13 |
| 9 ARTICLE 9 SURVEYS, PERMITS, LAWS, TAXES AND REGULATIONS | 13 13 13 13 |
| 9 ARTICLE 9 SURVEYS, PERMITS, LAWS, TAXES AND REGULATIONS | 131313131414 |
| 9 ARTICLE 9 SURVEYS, PERMITS, LAWS, TAXES AND REGULATIONS | |
| 9 ARTICLE 9 SURVEYS, PERMITS, LAWS, TAXES AND REGULATIONS | |
| 9 ARTICLE 9 SURVEYS, PERMITS, LAWS, TAXES AND REGULATIONS | |
| 9 ARTICLE 9 SURVEYS, PERMITS, LAWS, TAXES AND REGULATIONS | |
| 9 ARTICLE 9 SURVEYS, PERMITS, LAWS, TAXES AND REGULATIONS. 9.1 SURVEYS | |
| 9 ARTICLE 9 SURVEYS, PERMITS, LAWS, TAXES AND REGULATIONS. 9.1 SURVEYS | |
| 9 ARTICLE 9 SURVEYS, PERMITS, LAWS, TAXES AND REGULATIONS. 9.1 SURVEYS | |
| 9 ARTICLE 9 SURVEYS, PERMITS, LAWS, TAXES AND REGULATIONS. 9.1 SURVEYS | |
| 9 ARTICLE 9 SURVEYS, PERMITS, LAWS, TAXES AND REGULATIONS. 9.1 SURVEYS | |

| 14 | ARTIC | LE 14 | SAMPLES AND TESTING | 19 |
|-----------|-------|--------------------|---|----|
| 14 | 1.1 | SAMPL | ES | 19 |
| 14 | | | G - GENERAL | |
| 14 | 1.3 | TESTIN | G - CONCRETE AND SOILS | 20 |
| 14 | 1.4 | TESTIN | G - OTHER | 20 |
| 15 | ARTIC | LE 15 | SUBCONTRACTS | 20 |
| 15 | 5.1 | CONTR | ACT PERFORMANCE OUTSIDE OF THE UNITED STATES OR COLORADO | 20 |
| 15 | 5.2 | SUBCO | NTRACTOR PREQUALIFICATION | 21 |
| 15 | 5.3 | SUBCO | NTRACTOR PROPOSALS | 21 |
| 15 | 5.4 | SUBCO | NTRACT FORMS | 21 |
| 15 | 5.5 | SUBCO | NTRACTOR SUBSTITUTION | 21 |
| 15 | 5.6 | CONST | RUCTION MANAGER RESPONSIBLE FOR SUBCONTRACTORS | 21 |
| 16 | ARTIC | LE 16 | RELATIONS OF CONSTRUCTION MANAGER AND SUBCONTRACTOR | 22 |
| 17 | ARTIC | LE 17 | MUTUAL RESPONSIBILITY OF CONTRACTORS | 22 |
| 18 | ARTIC | LE 18 | SEPARATE CONTRACTS | 22 |
| 19 | ARTIC | LE 19 | USE OF PREMISES | 23 |
| 20 | ARTIC | LE 20 | CUTTING, FITTING OR PATCHING | 23 |
| 21 | ARTIC | LE 21 | UTILITIES | 23 |
| 21 | L.1 | TEMPO | RARY UTILITIES | 23 |
| 21 | L.2 | PROTE | CTION OF EXISTING UTILITIES | 23 |
| 21 | L.3 | CROSS | NG OF UTILITIES | 24 |
| 22 | ARTIC | LE 22 | UNSUITABLE CONDITIONS | 24 |
| 23 | ARTIC | LE 23 | TEMPORARY FACILITIES | 24 |
| 23 | 3.1 | OFFICE | FACILITIES | 24 |
| 23 | 3.2 | TEMPO | PRARY HEAT | 24 |
| 23 | 3.3 | WEATH | IER PROTECTION | 24 |
| 23 | 3.4 | DUST F | ARTITIONS | 24 |
| 23 | 3.5 | BENCH | MARKS | 25 |
| 23 | 3.6 | SIGN | | 25 |
| 23 | 3.7 | SANITA | RY PROVISION | 25 |
| 24 | ARTIC | LE 24 | CLEANING UP | 25 |
| 25 | ARTIC | LE 25 | INSURANCE | 25 |
| 25 | 5.1 | GENER | AL | 25 |
| 25 | 5.2 | COMM | ERCIAL GENERAL LIABILITY INSURANCE (CGL) | 25 |
| 25 | 5.3 | AUTON | MOBILE LIABILITY INSURANCE | 26 |
| 25 | 5.4 | WORK | ERS' COMPENSATION INSURANCE | 26 |
| 25 | 5.5 | UMBRI | ELLA LIABILITY INSURANCE: | 27 |
| 25 | 5.6 | BUILDE | R'S RISK INSURANCE | 27 |
| 25 | 5.7 | POLLU ¹ | FION LIABILITY INSURANCE | 28 |
| 25 | 5.8 | ADDITI | ONAL MISCELLANEOUS INSURANCE PROVISIONS | 28 |
| 26 | ARTIC | LE 26 | CONSTRUCTION MANAGER'S PERFORMANCE AND PAYMENT INSURANCE. | 29 |
| 27 | ARTIC | LE 27 | LABOR AND WAGES | 29 |
| 27 | 7.1 | LABOR | | 29 |
| 27 | 7.2 | COLOR | ADO WAGES | 29 |
| 28 | ARTIC | LE 28 | ROYALTIES AND PATENTS | 30 |
| | | | ASSIGNMENT | |
| | | | CORRECTION OF WORK BEFORE ACCEPTANCE | |
| | | | APPLICATIONS FOR PAYMENTS | 32 |

| | | 50 | | |
|----|------|---------------|--|---------|
| 51 | ARTI | CLE 51 | CONSTRUCTION MANAGER'S RIGHT TO STOP WORK AND/OR TERMINATE CO | ONTRACT |
| _ | 0.2 | | DURES | |
| 5 | 0.1 | NOTICE | OF TERMINATION | 58 |
| 50 | ARTI | CLE 50 | TERMINATION FOR CONVENIENCE OF STATE | |
| | 9.3 | | ONAL CONDITIONS | |
| | 9.2 | | TIONS AND PROCEDURES | |
| | 9.1 | | AL | |
| | | | STATE'S RIGHTS TO TERMINATE CONTRACT | |
| | 8.3 | | DAMAGES | |
| | | | RARY SUSPENSION OF WORK | |
| - | 8.1 | _ | S RIGHT TO DO THE WORK | |
| _ | | | AMAGES | |
| 48 | | CLE 48 | • | |
| | | | DAMAGES | |
| | | | TIME OF COMPLETION AND LIQUIDATED DAMAGES | |
| | | | GUARANTEE INSPECTIONS AFTER COMPLETION | |
| | 4.2 | | L GUARANTEES AND WARRANTIES | |
| | | | EAR GUARANTEE OF THE WORK | |
| | | | ONE-YEAR GUARANTEE AND SPECIAL GUARANTEES AND WARRANTIES | |
| | | | LIENS | |
| | | | GENERAL WARRANTY AND CORRECTION OF WORK AFTER ACCEPTANCE | |
| | 1.5 | | MENT | |
| | 1.4 | | OF ACCEPTANCE | |
| | 1.3 | | OF ACCEPTANCE | |
| - | 1.2 | | NSPECTION | |
| | 1.1 | | | |
| | | | OF COMPLETION | |
| | | | RIGHT OF OCCUPANCY COMPLETION, FINAL INSPECTION, ACCEPTANCE AND SETTLEMENT | |
| | | | | |
| | | | NON-BINDING DISPUTE RESOLUTION – FACILITATED NEGOTIATIONS | |
| _ | | | DELAYS AND EXTENSIONS OF TIME | |
| _ | | | TIONS | |
| | 37.1 | | IN WRITING | |
| | | | DIFFERING SITE CONDITIONS | |
| | | | CLAIMS | |
| _ | - | | PRIATION LIMITATIONS - C.R.S. § 24-91-103.6, as amended | |
| _ | 5.4 | | ENCY FIELD CHANGE ORDERED WORK | |
| 3 | 5.3 | | DOUS MATERIALS | |
| 3 | 5.2 | | ED BREAKDOWN | |
| 3 | 5.1 | THE VA | LUE OF CHANGED WORK | 36 |
| 35 | ARTI | CLE 35 | CHANGES IN THE WORK | |
| 34 | ARTI | CLE 34 | DEDUCTIONS FOR UNCORRECTED WORK | |
| 33 | ARTI | CLE 33 | PAYMENTS WITHHELD | |
| 32 | ARTI | CLE 32 | CERTIFICATES FOR PAYMENTS | |
| 3 | 31.4 | RELEAS | E OF RETAINAGE | 33 |
| | 31.3 | | AGE WITHHELD | |
| 3 | 1.2 | ARCHIT | ECT/ENGINEER CERTIFICATION | 33 |
| 3 | 31.1 | CONST | RUCTION MANAGER'S SUBMITTALS | 32 |

| 52 ART | ICLE 52 COLORADO SPECIAL PROVISIONS | 60 |
|--------|--|----|
| 52.1 | CONTROLLER'S APPROVAL, C.R.S. § 24-30-202(1) | 60 |
| 52.2 | FUND AVAILABILITY, C.R.S. § 24-30-202(5.5) | 60 |
| 52.3 | GOVERNMENTAL IMMUNITY | 60 |
| 52.4 | INDEPENDENT CONTRACTOR | |
| 52.5 | COMPLIANCE WITH LAW | |
| 52.6 | CHOICE OF LAW, JURISDICTION, AND VENUE | 61 |
| 52.7 | PROHIBITED TERMS | |
| 52.8 | SOFTWARE PIRACY PROHIBITION. SOFTWARE PIRACY PROHIBITION | |
| 52.9 | EMPLOYEE FINANCIAL INTEREST/CONFLICT OF INTEREST | |
| | VENDOR OFFSET AND ERRONEOUS PAYMENTS | |
| 53 ART | ICLE 53 MISCELLANEOUS PROVISIONS | 62 |
| 53.1 | PROFESSIONAL ASSOCIATION PERMITTED | |
| 53.2 | DISSOLUTION OF PROFESSIONAL ASSOCIATION | |
| 53.3 | WAGE RATES, in accordance with C.R.S. § 24-30-1404 (1) | 62 |
| 53.4 | PUBLIC ART LAW | |
| 53.5 | ASSIGNMENT | 62 |
| 53.6 | SUBCONTRACTS | |
| 53.7 | BINDING EFFECT | 63 |
| 53.8 | AUTHORITY | |
| 53.9 | CAPTIONS AND REFERENCES | 63 |
| 53.10 | | |
| 53.11 | ENTIRE UNDERSTANDING | 63 |
| 53.12 | DIGITAL SIGNATURES | 63 |
| 53.13 | MODIFICATION | |
| 53.14 | , | |
| | EXTERNAL TERMS AND CONDITIONS | |
| | SEVERABILITY | |
| | SURVIVIAL AND CERTAIN CONTRACT TERMS | |
| | TAXES | |
| | THIRD PARTY BENEFICIARIES | |
| 53.20 | WAIVER | |
| 53.21 | | |
| 55.22 | STANDARD AND MANNER OF PERFORMANCE | |
| 53.23 | LICENSES, PERMITS, AND OTHER AUTHORIZATIONS | |
| 53.24 | INDEMNIFICATION | |
| 53.25 | ACCESSIBILITY | |
| 54 ART | ICLE 54 CONFIDENTIAL INFORMATION-STATE RECORDS | |
| 54.1 | CONFIDENTIALITY | |
| 54.2 | OTHER ENTITY ACCESS AND NONDISCLOSURE AGREEMENTS | |
| 54.3 | USE, SECURITY, AND RETENTION | |
| 54.4 | INCIDENT NOTICE AND REMEDIATION | |
| 54.5 | DATA PROTECTION AND HANDLING | |
| 54.6 | SAFEGUARDING PERSONAL IDENTIFIABLE INFORMATION (PII) | 67 |

THE GENERAL CONDITIONS OF THE CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT

(STATE FORM SC-6.51)

1 ARTICLE 1 DEFINITIONS

1.1 CONTRACT DOCUMENTS

The Contract Documents consist of the following some of which are procedural documents used in the administration and performance of the Agreement:

- a) Construction Manager/General Contractor (CM/GC) Agreement; (SC-6.5.);
- b) Construction Manager's Fee Proposal, All Appendices, Addenda and Clarifications
- c) The Request for Proposals, All Appendices, Addenda and Clarifications;
- Performance Bond (SC-6.22) and Labor and Material Payment Bond (SC-6.221);
- e) These General Conditions of the Construction Manager/General Contractor (CM/GC) Agreement (SC- 6.5.1) and if applicable, Supplementary General Conditions;
- f) Drawings, including all addenda issued prior to the Notice to Proceed to Commence Construction Phase (SBP-8.261);
- g) Change Orders (SC-6.31) and Amendments (SC-6.0B), if any, when properly executed;
- h) Notice of Award (SBP-6.15);
- i) Builder's Risk insurance certificates of insurance (ACORD 25-S);
- j) Liability and Workers' compensation certificates of insurance;
- k) Notice to Proceed to Commence Construction Phase (SBP-6.261);
- I) Notice of Approval of Occupancy/Use (SBP-01);
- m) Notice of Partial Substantial Completion (SBP-071);
- n) Notice of Substantial Completion (SBP-07);
- o) Notice of Partial Final Acceptance (SC-6.27);
- p) Notice of Final Acceptance (SBP-6.271);
- q) Notice of Partial Contractor's Settlement (SC-7.3);
- r) Notice of Contractor's Settlement (SBP-7.31);
- s) Application and Certificate for Contractor's Payment (SBP-7.2);
- t) Other procedural and reporting documents or forms referred to in the General Conditions, the Supplementary General Conditions, and the Specifications or required by the State Buildings Program or the Principal Representative, including but not necessarily limited to Pre-Acceptance Check List (SBP-05) and the Building Inspection Record (SBP-BIR). A list of the current standard State Buildings Program forms applicable to this Contract may be obtained from the Principal Representative on request.

1.2 DEFINITIONS OF WORDS AND TERMS USED

Agreement

The term "Agreement" shall mean the written agreement entered into by the State of Colorado acting by and through the Principal Representative and the Contractor for the performance of the Work and payment therefore, on State Form SC-6.21. The term Agreement when used without reference to State Form SC-6.21 may also refer to the entirety of the parties' agreement to perform the Work described in the Contract Documents or reasonably inferable there from. The term "Contract" shall be interchangeable with this latter meaning of the term Agreement

Amendment

The term "Amendment" shall be defined as provided in Article 3.8 of the Agreement, Amendments and Change Orders.

Architect/Engineer

The term "Architect/Engineer" shall mean either the architect of record or the engineer of record under contract to the State of Colorado for the Project identified in the Contract Documents.

Bid Package

Bid Package describes all documents that relate to a specific scope of work, including the drawings, specifications, documents, estimates, bid forms and bid bonds relevant to a discrete portion of or a complete construction Project.

Change Order

The term "Change Order" means a written order directing the Construction Manager to make changes in the Work, in accordance with Article 35, The Value of Changed Work and Article 3.8 of the Agreement (SC-6.5), Amendments and Change Orders.

Colorado Labor

The term "Colorado labor", as provided in C.R.S. § 8-17-101(2) (a), as amended, means any person who is a resident of the state of Colorado, at the time of the public Works project, without discrimination as to race, color, creed, sex, sexual orientation, marital status, national origin, ancestry, age, or religion except when sex or age is a bona fide occupational qualification. A resident of the state of Colorado is a person who can provide a valid Colorado driver's license, a valid Colorado state-issued photo identification, or documentation that he or she has resided in Colorado for the last thirty days.

Construction Manager/General Contractor

The term "Construction Manager / General Contractor" shall mean the person, company, firm, corporation or other legal entity entering into a contract with the State of Colorado acting by and through the Principal Representative. The Construction Manager/General Contractor may also be referred to as the "Construction Manager" or "Contractor" in this agreement or in related exhibits, attachments, contract modification or procedural documents.

Construction Manager/General Contractor's Scope Narrative

Shall be defined as the bilateral agreement concerning final scope, which is developed cumulatively and simultaneously with each of the Bid Packages and is agreed upon during review of the final scope as it pertains to each Bid Package.

Consultant

The term "Consultant" shall mean a person, firm or corporation supplying design/consulting services for the Project. Design professionals and consultants are directly contracted to the Architect/Engineer or the Principal Representative.

CORA

The term "CORA" refers to the Colorado Open Records Act, §§24-72-200.1, et seq., C.R.S.

Days

The term "days" whether singular or plural shall mean calendar days unless expressly stated otherwise. Where the term "business days" is used it shall mean business days of the State of Colorado.

Direct Cost of Work

The term "Direct Cost of Work" shall mean those costs directly linked to the physical construction of the project including material, labor and equipment and subcontractor costs and as further defined in Designated Services and Method of Payment, Exhibit A.

Drawings

The term "Drawings" shall mean all drawings approved by appropriate State officials which have been prepared by the Architect/Engineer showing the Work to be done, except that where a list of drawings is specifically enumerated in the Supplementary General Conditions or Division 1 of the Specifications, the term shall mean the drawings so enumerated, including all addenda drawings.

Emergency Field Change Order

The term "Emergency Field Change Order" shall mean a written change order for extra Work or a change in the Work necessitated by an emergency as defined in Article 35.4 executed on State form SC 6.31 and identified as an Emergency Field Change Order. The use of such orders is limited to emergencies and to the amounts shown in Article 35.4.

Fast Track Construction

The term "Fast Track Construction" is a methodology where portions of the Work could have their design completed as separate Construction Phase(s) and may be under construction before other portions of the Work are fully designed.

Final Acceptance

The terms "final acceptance" or "finally complete" mean the stage in the progress of the Work, after substantial completion, when all remaining items of Work have been completed, all requirements of the Contract Documents are satisfied and the Notice of Acceptance can be issued. Discrete physical portions of the Project may be separately and partially deemed finally complete at the discretion of the Principal Representative when that portion of the Project reaches such stage of completion and a partial Notice of Acceptance can be issued.

Fixed Limit of Construction Cost

The term "Fixed Limit of Construction Cost" shall set forth a dollar amount available for the total Construction Cost of all elements of the Work as specified by the Principal Representative.

Guaranteed Maximum Price

The term "Guaranteed Maximum Price" shall mean the maximum amount for which the Work shall be accomplished and it shall be computed by the Construction Manager in accordance with the provisions of paragraph 5.4 of the Agreement and as approved by the Principal Representative.

Incident

The term 'incident' means any accidental or deliberate event that results in or constitutes an imminent threat of the unauthorized access, loss, disclosure, modification, disruption, or destruction of any communications or information resources of the State, which are included as part of the Work, as described in §§24-37.5-401, et seq., C.R.S. Incidents include, without limitation, (i) successful attempts to gain unauthorized access to a State system or State Records regardless of where such information is located; (ii) unwanted disruption or denial of service; (iii) the unauthorized use of a State system for the processing or storage of data; or (iv) changes to State system hardware, firmware, or software characteristics without the State's knowledge, instruction, or consent.

Notice

The term "Notice" shall mean any communication in writing from either contracting party to the other by such means of delivery that receipt cannot properly be denied. Notice shall be provided to the person identified to receive it in Article 8 of the Agreement. Notice Identification, or to such other person as either party identifies in writing to receive Notice. Notice by facsimile transmission where proper transmission is evidence shall be adequate where facsimile numbers are included in Article 7 of the Agreement, or to such other person as either party identifies in writing to receive Notice or in the absence of the identified party, a principal of the Construction Manager. Notice by facsimile transmission where proper transmission is evidenced shall be adequate where facsimile numbers are included in Article 7 of the Agreement. Notwithstanding an email delivery or return receipt, email Notice shall not be adequate. Acknowledgment of receipt of a voice message shall not be deemed to waive the requirement that Notice, where required, shall be in writing.

Occupancy

The term "Occupancy" means occupancy taken by the State as Owner after the Date of Substantial Completion at a time when a building or other discrete physical portion of the Project is used for the purpose intended. The Date of Occupancy shall be the date of such first use, but shall not be prior to the date of execution of the Notice of Approval of Occupancy/Use. Prior to the date of execution of a Notice of Approval of Occupancy/Use, the state shall have no right to occupy and the project may not be considered safe for occupancy for the intended use.

Owner

The term "Owner" shall mean the Principal Representative.

PII

The term "PII" shall be defined as personally identifiable information including, without limitation, any information maintained by the State about an individual that can be used to distinguish or trace an individual's identity, such as name, social security number, date and place of birth, mother's maiden name, or biometric records; and any other information that is linked or linkable to an individual, such as medical, educational, financial, and employment information. PII includes, but is not limited to, all information defined as personally identifiable information in §§24-72-501 and 24-73-101, C.R.S. "PII" shall also mean "personal identifying information" as set forth at § 24-74-102, et. seq., C.R.S.

Preconstruction

The term "Preconstruction" shall mean the Work done by the Construction Manager in the management and definition of the project prior to the awarding of construction contracts for any bid package.

Principal Representative

The term "Principal Representative" shall be defined, as provided in C.R.S. § 24-30-1301(14), as the governing board of a state department, institution, or agency; or if there is no governing board, then the executive head of a state department, institution, or agency, as designated by the governor or the general assembly and as specifically identified in the Contract Documents, or shall have such other meaning as the term may otherwise be given in C.R.S. § 24-30-1301(14), as amended. The Principal Representative may delegate authority. The Construction Manager shall have the right to inquire regarding the delegated authority of any of the Principal Representative's representatives on the project and shall be provided with a response in writing when requested.

Product Data

The term "Product Data" shall mean all submittals in the form of printed manufacturer's literature, manufacturer's specifications, and catalog cuts.

4

Project

The "Project" is the total construction of which the Work performed under the Contract Documents is a part, and may include construction by the Principal Representative or by separate contractors.

Reasonably Inferable

The phrase "reasonably inferable" means that if an item or system is either shown or specified, all material and equipment normally furnished with such items or systems and needed to make a complete installation shall be provided whether mentioned or not, omitting only such parts as are specifically excepted, and shall include only components which the Construction Manager could reasonably anticipate based on his or her skill and knowledge using an objective, industry standard, not a subjective standard. This term takes into consideration the normal understanding that not every detail is to be given on the Drawings and Specifications If there is a difference of opinion, the Principal Representative shall make the determination as to the standards of what reasonably inferable.

Samples

The term "Samples" shall mean examples of materials or Work provided to establish the standard by which the Work will be judged.

SBP

The term "SBP" means "State Buildings", which is used in connection with labeling applicable State form documents (e.g., "SBP-01" is the form number for Notice of Approval of Occupancy/Use).

SC

The term "SC" means "State Contract" which is used in connection with labeling applicable State form documents (e.g. "SC 6.23" is the State form number for these General Conditions of the Construction Manager/General Contractor's (CM/GC) Agreement).

Schedule of Values

The term "Schedule of Values" is defined as the itemized listing of description of the Work by Division and Section of the Specifications. The format shall be similar to Form SC-7.2. Included shall be the material costs, and the labor and other costs plus the sum of both.

Shop Drawings

The term "Shop Drawings" shall mean any and all detailed drawings prepared and submitted by Construction Manager, Subcontractor at any tier, vendors or manufacturers providing the products and equipment specified on the Drawings or called for in the Specifications.

Specifications

The term "Specifications" shall mean the requirements of the CSI divisions of the project manual prepared by the Architect/Engineer describing the Work to be accomplished.

State Buildings Program

The term "State Buildings Program" shall refer to the Office of the State Architect within the Department of Personnel & Administration of Colorado State government responsible for project administration, review, approval and coordination of plans, construction procurement policy, contractual procedures, and code compliance and inspection of all buildings, public Works and improvements erected for state purposes; except public roads and highways and projects under the supervision of the division of wildlife and the division of parks and outdoor recreation as provided in C.R.S. § 24-30-1301, et seq. The term State Buildings Program shall also mean that individual within a State Department agency or institution, including institutions of higher education, who has signed an agreement accepting delegation to perform all or part of the responsibilities and functions of State Buildings Program.

State Confidential Information

The term "State Confidential Information" shall mean any and all State Records not subject to disclosure under CORA. State Confidential Information shall include, but is not limited to, PII, and State personnel records not subject to disclosure under CORA. State Confidential Information shall not include information or data concerning individuals that is not deemed confidential but nevertheless belongs to the State, which has been communicated, furnished, or disclosed by the State to Contractor which (i) is subject to disclosure pursuant to CORA; (ii) is already known to Contractor without restrictions at the time of its disclosure to Contractor; (iii) is or subsequently becomes publicly available without breach of any obligation owed by Contractor to the State; (iv) is disclosed to Contractor, without confidentiality obligations, by a third party who has the right to disclose such information; or (v) was independently developed without reliance on any State Confidential Information.

State Fiscal Rules

"State Fiscal Rules" means the fiscal rules promulgated by the Colorado State Controller pursuant to §24-30-202(13)(a), C.R.S.

State Records

The term "State Records" shall mean any and all State data, information, and records, regardless of physical form, including, but not limited to, information subject to disclosure under CORA.

Subconsultant

The term "Subconsultant" shall mean a person, firm or corporation supplying design/consulting services for the Project. Design and other professionals directly contracted to the Architect/Engineers are considered subconsultants.

Subcontractor

The term "Subcontractor" shall mean a person, firm or corporation supplying labor, materials, equipment and/or Services for Work at the site of the Project for, and under separate contract or agreement with the Construction Manager.

Submittals

The term "submittals" means drawings, lists, tables, documents and samples prepared by the Construction Manager to facilitate the progress of the Work as required by these General Conditions or the Drawings and Specifications. They consist of Shop Drawings, Product Data, Samples, and various administrative support documents including but not limited to lists of subcontractors, construction progress schedules, schedules of values, applications for payment, inspection and test results, requests for information, various document logs, and as-built drawings. Submittals are required by the Contract Documents, but except to the extent expressly specified otherwise are not themselves a part of the Contract Documents.

Substantial Completion

The terms "substantial completion" or "substantially complete" mean the stage in the progress of the Work when the construction is sufficiently complete, in accordance with the Contract Documents as modified by any Change Orders, so that the Work, or at the discretion of the Principal Representative, any designated portion thereof, is available for its intended use by the Principal Representative and a Notice of Substantial Completion can be issued. Portions of the Project may, at the discretion of the Principal Representative, be designated as substantially complete.

Supplier

The term "Supplier" shall mean any manufacturer, fabricator, distributor, material man or vendor.

Surety

The term "Surety" shall mean the company providing the labor and material payment and performance bonds for the Construction Manager as obligor.

Value Engineering

"Value Engineering" or "VE" is defined as an analysis and comparison of cost versus value of building materials, equipment, and systems. VE considers the initial cost of construction, coupled with the estimated cost of maintenance, energy use, life expectancy and replacement cost. VE related to this Project shall include the analysis and comparison of building elements in an effort to reduce overall Project costs, while maintaining or enhancing the quality of the design intent, whenever possible.

Work (

The term "Work" shall mean all or part of the labor, materials, equipment, and other services required by the Contract Documents or otherwise required to be provided by the Construction Manager to meet the Construction Manager's obligations under the Contract.

Work Product

The phrase "Work Product" means the tangible and intangible results of the Work, whether finished or unfinished, including drafts. Work Product includes, but is not limited to, documents, text, software (including source code), research, reports, proposals, specifications, plans, notes, studies, data, images, photographs, negatives, pictures, drawings, designs, models, surveys, maps, materials, ideas, concepts, know-how, and any other results of the Work. "Work Product" does not include any material that was developed prior to the Effective Date that is used, without modification, in the performance of the Work.

2 ARTICLE 2 EXECUTION, CORRELATION, INTENT OF DOCUMENTS, COMMUNICATION AND COOPERATION

2.1 EXECUTION

The Construction Manager, within ten (10) days from the date of Notice of Award, will be required to:

- a) Execute the Agreement, State Form SC-6.5.
- b) Furnish fully executed Performance and Labor and Material Payment Bonds on State Forms SC-6.22 and SC-6.221; and
- c) Furnish certificates of insurance evidencing all required insurance on standard Acord forms designed for such purpose.
- d) Furnish certified copies of any insurance policies requested by the Principal Representative.
- e) If Article 7.1 of the Construction Manager/General Contractor Agreement (SC-6.5) applies, furnish documentation that identifies the subcontractors that will be used for all mechanical, sheet metal, fire suppression, sprinkler fitting, electrical, and plumbing work required on the project and certify that that all firms identified participate in apprenticeship programs registered with the United States Department of Labor's Employment and Training Administration or state apprenticeship councils recognized by the United States Department of Labor and have a proven record of graduating a minimum of fifteen percent of its apprentices for at least three of the past five years.

2.2 CORRELATION

By execution of the Agreement the Construction Manager represents that the Construction Manager has visited the site, has become familiar with local conditions and local requirements under which the Work is to be performed, including the building code programs of the State Buildings Program as implemented by the Principal Representative, and has correlated personal observations with the requirements of the Contract Documents.

2.3 INTENT OF DOCUMENTS

The Contract Documents are complementary, and what is called for by any one document shall be as binding as if called for by all. The intention of the documents is to include all labor, materials, equipment and transportation necessary for the proper execution of the Work. Words describing materials or Work which have a well-known technical or trade meaning shall be held to refer to such recognized standards.

In any event, if any error exists, or appears to exist, in the requirements of the Drawings or Specifications, or if any disagreement exists as to such requirements, the Construction Manager shall submit a request for information to the Architect/Engineer before proceeding with the Work in question. In the event of the Construction Manager's failure to give prior written Notice of any such errors or disagreements of which the Construction Manager or the Subcontractors at any tier are aware, the Construction Manager shall, at no additional cost to the Principal Representative, make good any damage to, or defect in, Work which is caused by such omission.

Where a conflict occurs between or within standards, Specifications or Drawings, which is not resolved by reference to the precedence between the Contract Documents, the more stringent or higher quality requirements shall apply so long as such more stringent or higher quality requirements are reasonably inferable. The Principal Representative, or the Architect/Engineer with consent of the Principal Representative, shall decide which requirements will provide the best installation.

With the exception noted in the following paragraph, the precedence of the Contract Documents is in the following sequence:

- a) The Supplementary General Conditions, if any;
- b) The Minimum Requirements of the Request for Proposals;
- c) The Construction Manager's Fee Proposals;
- d) The Agreement (SC-6.5);
- e) The General Conditions (SC-6.51);
- f) Drawings and Specifications, all as modified by any addenda; and
- g) Any additional Exhibit to this agreement.

Change Orders and Amendments, if any, to the Contract Documents take precedence over the original Contract Documents.

Notwithstanding the foregoing order of precedence, the Colorado Special Provisions of Article 52 of the General Conditions, Colorado Special Provisions, shall take precedence, rule and control over all other provisions of the Contract Documents.

Unless the context otherwise requires, form numbers in this document are for convenience only. In the event of any conflict between the form required by name or context and the form required by number, the form required by name or context shall control. The Construction Manager may obtain State forms from the Principal Representative upon request.

2.4 PARTNERING, COMMUNICATIONS AND COOPERATION

In recognition of the fact that conflicts, disagreements and disputes often arise during the performance of construction contracts, the Construction Manager and the Principal Representative aspire to encourage a relationship of open communication and cooperation

between the employees and personnel of both, in which the objectives of the Contract may be better achieved and issues resolved in a more fully informed atmosphere.

The Construction Manager and the Principal Representative each agree to assign an individual who shall be fully authorized to negotiate and implement a voluntary partnering plan for the purpose of facilitating open communications between them. Within thirty days (30) of the Notice to Proceed, the assigned individuals shall meet to discuss development of an informal agreement to accomplish these goals.

The assigned individuals shall endeavor to reach an informal agreement, but shall have no such obligation. Any plans these parties voluntarily agree to implement shall result in no change to the contract amount, and no costs associated with such plan or its development shall be recoverable under any contract clause. In addition, no plan developed to facilitate open communication and cooperation shall alter, amend or waive any of the rights or duties of either party under the Contract unless and except by written Amendment to the Contract, nor shall anything in this clause or any subsequently developed partnering plan be deemed to create fiduciary duties between the parties unless expressly agreed in a written Amendment to the Contract. It is also recognized that projects with relatively low contract values may not justify the expense or special efforts required. In the case of small projects with an initial Contract value under \$500,000, the requirements of the preceding paragraph shall not apply.

3 ARTICLE 3 COPIES FURNISHED

The Construction Manager will be furnished, free of charge, the number of copies of Drawings and Specifications as specified in the Contract Documents, or if no number is specified, all copies reasonably necessary for the execution of the Work.

4 ARTICLE 4 OWNERSHIP OF DRAWINGS

Drawings or Specifications, or copies of either, furnished by the Architect/Engineer, are not to be used on any other Work. At the completion of the Work, at the written request of the Architect/Engineer, the Construction Manager shall endeavor to return all Drawings and Specifications.

The Construction Manager may retain the Construction Manager's Contract Document set, copies of Drawings and Specifications used to contract with others for any portion of the Work and a marked up set of as-built drawings.

5 ARTICLE 5 ARCHITECT/ENGINEER'S STATUS

The Architect/Engineer is the representative of the Principal Representative for purposes of administration of the Contract, as provided in the Contract Documents and the Agreement. In case of termination of employment or the death of the Architect/Engineer, the Principal Representative will appoint a capable Architect/Engineer against whom the Construction Manager makes no reasonable objection, whose status under the Contract shall be the same as that of the former Architect/Engineer.

INSPECTION

6.1 DECISIONS

The Architect/Engineer shall, within a reasonable time, make decisions on all matters relating to the execution and progress of the Work or the interpretation of the Contract Documents, and in the exercise of due diligence shall be reasonably available to the Construction Manager to timely interpret and make decisions with respect to questions relating to the design or concerning the Contract Documents. The Principal Representative may consent with such decision by the Design Build Entity's Architect/Engineer or amend/revise such decision at the discretion of the Principal Representative.

6.2 JUDGMENTS

The Architect/Engineer is, in the first instance, the judge of the performance required by the Contract Documents as it relates to compliance with the Drawings and Specifications and quality of Workmanship and materials.

The Architect/Engineer shall make judgments regarding whether directed Work is extra or outside the scope of Work required by the Contract Documents at the time such direction is first given. If, in the Construction Manager's judgment, any performance directed by the Architect/Engineer is not required by the Contract Documents or if the Architect/Engineer does not make the judgment required, it shall be a condition precedent to the filing of any claim for additional cost related to such directed Work that the Construction Manager, before performing such Work, shall first obtain in writing, the Architect/Engineer's written decision that such directed Work is included in the performance required by the Contract Documents. If the Architect/Engineer's direction to perform the Work does not state that the Work is within the performance required by the Contract Documents, the Construction Manager shall, in writing, request the Architect/Engineer to advise in writing whether the directed Work will be considered extra Work or Work included in the performance required by the Contract Documents.

The Architect/Engineer shall respond to any such written request for such a decision within three (3) business days and if no response is provided, or if the Architect/Engineer's written decision is to the effect that the Work is included in the performance required by the Contract Documents, the Construction Manager may file with the Principal Representative and the Architect/Engineer a Notice of claim in accordance with Article 36, Claims. Whether or not a Notice of claim is filed, the Construction Manager shall proceed with the ordered Work. Disagreement with the decision of the Architect/Engineer shall not be grounds for the Construction Manager to refuse to perform the Work directed or to suspend or terminate performance.

6.3 ACCESS TO WORK

The Architect/Engineer, the Principal Representative and representatives of State Buildings Program shall at all times have access to the Work. The Construction Manager shall provide proper facilities for such access and for their observations or inspection of the Work.

6.4 INSPECTION

The Architect/Engineer has agreed to make, or that structural, mechanical, electrical engineers or other consultants will make, periodic visits to the site to generally observe the progress and quality of the Work to determine in general if the Work is proceeding in accordance with the Contract Documents. Observation may extend to all or any part of the Work and to the preparation, fabrication or manufacture of materials.

Without in any way meaning to be exclusive or to limit the responsibilities of the Architect/Engineer or the Construction Manager, the Architect/Engineer has agreed to observe, among other aspects of the Work, the following for compliance with the Contract Documents:

- a) Compaction testing reports based upon the findings and recommendations of the Principal Representative's testing consultant;
- b) Bearing surfaces of excavations before concrete is placed based upon the findings and recommendations of the Principal Representative's soils engineering consultant;
- c) Reinforcing steel after installation and before concrete is poured;
- d) Structural concrete;
- e) Laboratory reports on all concrete testing based upon the findings and recommendations of the Principal Representative's testing consultant;
- f) Structural steel during and after erection and prior to its being covered or enclosed;
- g) Steel welding; Principal Representative will furnish steel welding inspection consultant/agency if required or necessary for the project;
- h) Mechanical and plumbing Work following its installation and prior to its being covered or enclosed;
- i) Electrical Work following its installation and prior to its being covered or enclosed; and
- j) Any special or quality control testing required in the Contract Documents provided by the Principal Representative's testing consultant.

If the Specifications, the Architect/Engineer's instructions, laws, ordinances of any public authority require any Work to be specifically tested or approved, the Construction Manager shall give the Principal Representative, Architect/Engineer and appropriate testing agency (if necessary) timely notice of its readiness for observation by the Architect/Engineer or inspection by another authority, and if the inspection is by another authority, of the date fixed for such inspection, required certificates of inspection being secured by the Construction Manager. The Construction Manager shall give all required Notices to the Principal Representative or his or her designee for inspections required for the building inspection program. It shall be the responsibility of the Construction Manager to determine the Notice required by the State pursuant to Building Inspection Record for the Project, according to State form SBP-B.I.R., or the equivalent form required by the Principal Representative as approved by the State Buildings Program. If any portion of the Work should be covered contrary to the reasonable request of the Architect/Engineer, or to requirements specifically expressed in the Contract Documents, it must, if required in writing by the Architect/Engineer, be uncovered for its observation and shall be replaced at the Construction Manager's expense.

If any other portion of the Work has been covered which the Architect/Engineer has not specifically requested to observe prior to it's being covered, it may request to see such work and it shall be uncovered by the Construction Manager. If such work is found in accordance with the Contract Documents, the cost of uncovering and replacement shall, by appropriate Amendment or Change Order, be charged to the Principal Representative. If such work is found not in accordance with the Contract Documents, the Construction Manager shall pay such costs unless it is found that this condition was caused by the Principal Representative or a separate Construction Manager as provided in Article 18, in which event, the Principal Representative shall be responsible for the payment of such costs.

7 ARTICLE 7 CONSTRUCTION MANAGER'S SUPERINTENDENCE AND SUPERVISION

The Construction Manager shall employ, and keep present (as applicable) on the Project during its progress, a competent project manager as satisfactory to the Principal Representative. The project manager shall not be changed except with the consent of the Principal Representative, unless the project manager proves to be unsatisfactory to the Construction Manager and ceases to be in his or her employ. The project manager shall represent the Construction Manager for the Project, and in the absence of the Construction Manager, all directions given to the project manager shall be as binding as if given to the Construction Manager. Directions received by the project manager shall be documented by the project manager and communicated in writing with the Construction Manager.

The Construction Manager shall employ, and keep present on the Project during its progress, a competent superintendent and any necessary assistants, all satisfactory to the Architect/Engineer and the Principal Representative. The superintendent shall not be changed except with the consent of the Architect/Engineer and the Principal Representative, unless the superintendent proves to be unsatisfactory to the Project Manager/Construction Manager and ceases to be in his or her employ. The superintendent shall represent the Project Manager/Construction Manager in his or her absence and all directions given to the superintendent shall be as binding as if given to the Project Manager/Construction Manager. Directions received by the superintendent shall be documented by the superintendent and confirmed in writing with the Project Manager/Construction Manager.

The Construction Manager shall give efficient supervision to the Work, using his or her best skill and attention. He or she shall carefully study and compare all Drawings, Specifications and other written instructions and shall without delay report any error, inconsistency or omission which he or she may discover in writing to the Architect/Engineer. The Construction Manager shall not be liable to the Principal Representative for damage to the extent it results from errors or deficiencies in the Contract Documents or other instructions by the Architect/Engineer, unless the Construction Manager knew or had reason to know, that damage would result by proceeding and the Construction Manager fails to so advise the Architect/Engineer.

The superintendent shall see that the Work is carried out in accordance with the Contract Documents and in a uniform, thorough and first-class manner in every respect. The Construction Manager's superintendent shall establish all lines, levels, and marks necessary to facilitate the operations of all concerned in the Construction Manager's Work. The Construction Manager shall lay out all Work in a manner satisfactory to the Architect/Engineer, making permanent records of all lines and levels required for excavation, grading, foundations, and for all other parts of the Work.

8 ARTICLE 8 MATERIALS AND EMPLOYEES

Unless otherwise stipulated, the Construction Manager shall provide and pay for all materials, labor, water, tools, equipment, light, power, transportation and other facilities necessary for the execution and completion of the Work.

Unless otherwise specified, all materials shall be new and both workmanship and materials shall be first class and of uniform quality. The Construction Manager shall, if required, furnish satisfactory evidence as to the kind and quality of materials.

SC-6.51 Rev. 08/2022 The Construction Manager is fully responsible for all acts and omissions of the Construction Manager's employees and shall at all times enforce strict discipline and good order among employees on the site. The Construction Manager shall not employ on the Work any person reasonably deemed unfit by the Principal Representative or anyone not skilled in the Work assigned to him.

9 ARTICLE 9 SURVEYS, PERMITS, LAWS, TAXES AND REGULATIONS

9.1 SURVEYS

The Principal Representative shall furnish all surveys, property lines and benchmarks deemed necessary by the Architect/Engineer, unless otherwise specified.

9.2 PERMITS AND LICENSES

Permits and licenses necessary for the prosecution of the Work shall be secured and paid for by the Construction Manager. Unless otherwise specified in the Specifications, no local municipal or county building permit shall be required. However, State Buildings Program requires each Principal Representative to administer a building code inspection program, the implementation of which may vary at each agency or institution of the State. The Construction Manager s' employees shall become personally familiar with these local conditions and requirements and shall fully comply with such requirements. State electrical and plumbing permits are required, unless the requirement to obtain such permits is altered by State Building's Programs. The Construction Manager shall obtain and pay for such permits.

Easements for permanent structures or permanent changes in existing facilities shall be secured and paid for by the Principal Representative, unless otherwise specified.

9.3 TAXES

9.3.1 Refund of Sales and Use Taxes

The Construction Manager shall pay all local taxes required to be paid, including but not necessarily limited to all sales and use taxes. If requested by the Principal Representative prior to issuance of the Notice to Proceed or directed in the Supplementary General Conditions or the Specifications, the Construction Manager shall maintain records of such payments in respect to the Work, which shall be separate and distinct from all other records maintained by the Construction Manager, and the Construction Manager shall furnish such data as may be necessary to enable the State of Colorado, acting by and through the Principal Representative, to obtain any refunds of such taxes which may be available under the laws, ordinances, rules or regulations applicable to such taxes. When so requested or directed, the Construction Manager shall require Subcontractors at all tiers to pay all local sales and use taxes required to be paid and to maintain records and furnish the Construction Manager with such data as may be necessary to obtain refunds of the taxes paid by such Subcontractors. No State sales and use taxes are to be paid on material to be used in this Project. On application by the purchaser or seller, the Department of Revenue shall issue to a Construction Manager or to a Subcontractor at any tier, a certificate or certificates of exemption per C.R.S. § 39-26-703(2)(b), and C.R.S. § 39-26-708.

9.3.2 Federal Taxes

The Construction Manager shall exclude the amount of any applicable federal excise or manufacturers' taxes from the proposal. The Principal Representative will furnish the Construction Manager, on request exemption certificates.

9.4 LAWS AND REGULATIONS

The Construction Manager shall give all notices and comply with all laws, ordinances, rules and regulations bearing on the conduct of the Work as drawn or specified. If the Construction Manager observes that the Drawings or Specifications require Work which is at variance therewith, the Construction Manager shall without delay notify the Architect/Engineer in writing and any necessary changes shall be adjusted as provided in Article 35, Changes in the Work.

The Construction Manager shall bear all costs arising from the performance of Work required by the Drawings or Specifications that the Construction Manager knows to be contrary to such laws, ordinances, rules or regulations, if such Work is performed without giving Notice to the Architect/Engineer.

10 ARTICLE 10 PROTECTION OF WORK AND PROPERTY

10.1 GENERAL PROVISIONS

The Construction Manager shall continuously maintain adequate protection of all Work and materials, protect the property from injury or loss arising in connection with this Contract and adequately protect adjacent property as provided by law and the Contract Documents. The Construction Manager shall make good any damage, injury or loss, except to the extent:

- a) Directly due to errors in the Contract Documents;
- b) Caused by agents or employees of the Principal Representative; and,
- Due to causes beyond the Construction Manager's control and not to fault or negligence; provided such damage, injury or loss would not be covered by the insurance required to be carried by the Construction Manager;

10.2 SAFETY PRECAUTIONS

The Construction Manager shall take all necessary precautions for the safety of employees on the Project, and shall comply with all applicable provisions of federal, State and municipal safety laws and building codes to prevent accidents or injury to persons on, about or adjacent to the premises where the Work is being performed. He or she shall erect and properly maintain at all times, as required by the conditions and progress of the Work, all necessary safeguards for the protection of Workers and the public and shall post danger signs warning against the hazards created by such features of construction as protruding nails, hoists, well holes, elevator hatchways, scaffolding, window openings, stairways and falling materials; and he or she shall designate a responsible member of his or her organization on the Project, whose duty shall be the prevention of accidents. The name and position of any person so designated shall be reported to the Architect/Engineer by the Construction Manager.

The Construction Manager shall provide all necessary bracing, shoring and tying of all structures, decks and framing to prevent any structural failure of any material which could result in damage to property or the injury or death of persons; take all precautions to insure that no part of any structure of any description is loaded beyond its carrying capacity with anything that will endanger its safety at any time during the execution of this Contract; and provide for the adequacy and safety of all scaffolding and hoisting equipment. The Construction Manager shall not permit open fires within the building enclosure. The Construction Manager shall construct and maintain all necessary temporary drainage and do all pumping necessary to keep excavations and floors, pits and trenches free of water. The Construction Manager shall be solely responsible

for all construction means, methods, techniques, sequences and procedures, and for coordinating all portions of the Work, except as otherwise noted.

The Construction Manager shall take due precautions when obstructing sidewalks, streets or other public ways in any manner, and shall provide, erect and maintain barricades, temporary walkways, roadways, trench covers, colored lights or danger signals and any other devices necessary or required to assure the safe passage of pedestrians and automobiles.

10.3 EMERGENCIES

In an emergency affecting the safety of life or of the Work or of adjoining property, the Construction Manager without special instruction or authorization from the Architect/Engineer or Principal Representative, is hereby permitted to act, at his or her discretion, to prevent such threatened loss or injury; and he or she shall so act, without appeal, if so authorized or instructed. Provided the Construction Manager has no responsibilities for the emergency, if the Construction Manager incurs additional cost not otherwise recoverable from insurance or others on account of any such emergency Work, the Contract sum shall be equitably adjusted in accordance with Article 35, Changes in the Work.

11 ARTICLE 11 DRAWINGS AND SPECIFICATIONS ON THE WORK

The Construction Manager shall keep on the job site one copy of the Contract Documents in good order, including current copies of all Drawings and Specifications for the Work, and any approved Shop Drawings, Product Data or Samples, and as-built drawings. As-built drawings shall be updated weekly by the Construction Manager and Subcontractors to reflect actual constructed conditions including dimensioned locations of underground Work and the Construction Manager 's failure to maintain such updates may be grounds to withhold portions of payments otherwise due in accordance with Article 33, Payments Withheld. All such documents shall be available to the Architect/Engineer and representatives of the State. In addition, the Construction Manager shall keep on the job site one copy of all approved addenda, Change Orders and requests for information issued for the Work.

The Construction Manager shall develop procedures to insure the currency and accuracy of asbuilt drawings and shall maintain on a current basis a log of requests for information and responses thereto, a Shop Drawing and Product Data submittal log, and a Sample submittal log to record the status of all necessary and required submittals.

12 ARTICLE 12 REQUESTS FOR INFORMATION AND SCHEDULES

12.1 DETAIL DRAWINGS AND INSTRUCTIONS

The Architect/Engineer shall furnish additional instructions with reasonable promptness, by means of drawings or otherwise, necessary for the proper execution of the Work. All such drawings and instructions shall be consistent with the Contract Documents and reasonably inferable there from. The Architect/Engineer shall determine what additional instructions or drawings are necessary for the proper execution of the Work.

The Work shall be executed in conformity with such instructions and the Construction Manager shall do no Work without proper drawings, specifications or instructions. If the Construction Manager believes additional instructions, specifications or drawings are needed for the performance of any portion of the Work, the Construction Manager shall give Notice of such need

in writing through a request for information furnished to the Architect/Engineer sufficiently in advance of the need for such additional instructions, specifications or drawings to avoid delay and to allow the Architect/Engineer a reasonable time to respond. The Construction Manager shall maintain a log of the requests for information and the responses provided.

The Construction Manager, the Architect/Engineer, and the Principal Representative shall jointly prepare a schedule, subject to change from time to time in accordance with the progress of the Work, fixing the dates at which the various detail drawings will be required, and the Architect/Engineer shall furnish them in accordance with that schedule. Under like conditions, a schedule shall be prepared, fixing the dates for the submission of shop drawings, for the beginning of manufacture and installation of materials and for the completion of the various parts of the Work.

The Principal Representative may consent with such instructions by the Architect/Engineer or amend/revise such instructions at the discretion of the Principal Representative.

12.2 SCHEDULES

12.2.1 Submittal Schedules

Prior to the Notice to Proceed to Commence Construction for the first construction phase, a schedule shall be prepared which may be preliminary to the extent required, fixing the dates for the submission and initial review of required Shop Drawings, Product Data and Samples for the beginning of manufacture and installation of materials, and for the completion of the various parts of the Work. It shall be prepared so as to cause no delay in the Work or in the Work of any other contractor engaged by the Principal Representative. The schedule shall be subject to change from time to time in accordance with the progress of the Work, and it shall be subject to the review and approval by the Architect/Engineer. It shall fix the dates at which the various Shop Drawings Product Data and Samples will be required from the Architect/Engineer. The Architect/Engineer, after review and agreement as to the time provided for initial review, shall review and comment on the Shop Drawings, Product Data and Samples in accordance with that schedule. The schedule shall be finalized, prepared and submitted with respect to each of the elements of the Work in time to avoid delay, considering reasonable periods for review, manufacture or installation.

At the time the schedule is prepared, the Construction Manager, the Architect/Engineer and Principal Representative shall jointly identify the Shop Drawing, Product Data and Samples, if any, which the Principal Representative shall receive simultaneously with the Architect/Engineer for the purposes of owner coordination with existing facility standards and systems. The Construction Manager shall furnish a copy for the Principal Representative when so requested. Transmittal of Shop Drawings and Product Data copies to the Principal Representative shall be solely for the convenience of the Principal Representative and shall neither create nor imply responsibility or duty of review by the Principal Representative.

The Construction Manager may also, or at the direction of the Principal Representative at any time shall, prepare and maintain a schedule, which may also be preliminary and subject to change to the extent required, fixing the dates for the initial responses to requests for information or for detail drawings which will be required from the Architect/Engineer to allow the beginning of manufacture, installation of materials and for the completion of the various parts of the Work. The schedule shall be subject to review and approval by the Architect/Engineer. The Architect/Engineer shall, after review and agreement, furnish responses and detail drawings in accordance with that schedule. Any such schedule shall be prepared and approved in time to avoid delay, considering reasonable periods for review, manufacture or installation, but so long

as the request for information schedule is being maintained, it shall not be deemed to transfer responsibility to the Construction Manager for errors or omissions in the Contract Documents where circumstances make timely review and performance impossible.

The Architect/Engineer shall not unreasonably withhold approval of the Construction Manager's schedules and shall inform the Construction Manager and the Principal Representative of the basis of any refusal to agree to the Construction Manager's schedules. The Principal Representative shall attempt to resolve any disagreements.

12.2.2 Schedule of Values

Prior to the Notice to Proceed to Commence Construction for the first construction phase, the Construction Manager shall submit to the Architect/Engineer and Principal Representative, for approval, and to the State Buildings Program when specifically requested, a complete itemized schedule of the values of the various parts of the Work, as estimated by the Construction Manager, aggregating the total price. The schedule of values shall be in such detail as the Architect/Engineer or the Principal Representative shall require, prepared on forms acceptable to the Principal Representative. It shall, at a minimum, identify on a separate line each division of the Specifications including the general conditions costs to be charged to the Project. The Construction Manager shall revise and resubmit the schedule of values for approval when, in the opinion of the Architect/Engineer or the Principal Representative, such resubmittal is required due to changes or modifications to the Contract Documents or the Contract sum.

The total cost of each line item so separately identified shall, when requested by the Architect/Engineer or the Principal Representative, be broken down into reasonable estimates of the value as indicated in Article 3.4.6 of the Construction Manager/General Contractor (CMGC) Agreement (SC-6.5).

The cost of subcontracts shall be incorporated in the Construction Manager's schedule of values, and when requested by the Architect/Engineer or the Principal Representative, shall be separately shown as line items.

The Architect/Engineer shall review the proposed schedules and approve it after consultation with the Principal Representative, or advise the Construction Manager of any required revisions within ten (10) days of its receipt. In the event no action is taken on the submittal within ten days, the Construction Manager may utilize the schedule of values as its submittal for payment until it is approved or until revisions are requested.

When the Architect/Engineer deems it appropriate to facilitate certification of the amounts due to the Construction Manager, further breakdown of subcontracts, including breakdown by labor and materials, may be directed.

This schedule of values, when approved, will be used in preparing Construction Manager's applications for payment on State Form SC-7.2, Application for Payment.

12.2.3 Construction Schedules

Prior to the Notice to Proceed to Commence Construction for the first construction phase, the Construction Manager shall submit to the Architect/Engineer and the Principal Representative, and to the State Buildings Program when specifically requested, on a form acceptable to them, an overall timetable of the construction schedule for the Project. Unless the Supplementary General Conditions or the Specifications allow scheduling with bar charts or other less sophisticated scheduling tools, the Construction Manager's schedule shall be a critical-path method (CPM)

construction schedule. The CPM schedule shall start with the date of the Notice to Proceed to Commence Construction for the first construction phase and include submittals activities, the various construction activities, change order Work (when applicable), close-out, testing, demonstration of equipment operation when called for in the Specifications, and acceptance. The CPM schedule shall at a minimum correlate to the schedule of values line items and shall be cost loaded if requested by the Architect/Engineer or Principal Representative. The completion time shall be the time specified in the Agreement and all Project scheduling shall allocate float utilizing the full period available for construction as specified in the Agreement on State Form SC 6.13, without indication of early completion, unless such earlier completion is approved in writing by the Principal Representative and State Building Programs.

The time shown between the starting and completion dates of the various elements within the construction schedule shall represent one hundred per cent (100%) completion of each element.

All other elements of the CPM schedule shall be as required by the Specifications. In addition, the Construction Manager shall submit monthly updates or more frequently, if required by the Principal Representative, updates of the construction schedule. These updates shall reflect the Construction Manager's "Work in place" progress.

When requested by the Architect/Engineer, the Principal Representative or the State Buildings Program, the Construction Manager shall revise the construction schedule to reflect changes in the schedule of values.

When the testing of materials is required by the Specifications, the Construction Manager shall also prepare and submit to the Architect/Engineer and the Principal Representative a schedule for testing in accordance with Article 14, Samples and Testing.

13 ARTICLE 13 SHOP DRAWINGS, PRODUCT DATA AND SAMPLES

13.1 SUBMITTAL PROCESS

The Construction Manager shall check and field verify all dimensions. The Construction Manager shall check, approve and submit to the Architect/Engineer in accordance with the schedule described in Article 12, Requests for Information and Schedules, all Shop Drawings, Product Data and Samples required by the specifications or required by the Construction Manager for the Work of the various trades. All Drawings and Product Data shall contain identifying nomenclature and each submittal shall be accompanied by a letter of transmittal identifying in detail all enclosures. The number of copies of Shop Drawings and Product Data to be submitted shall be as specified in the Specifications and if no number is specified then three copies shall be submitted.

The Architect/Engineer shall review and comment on the Shop Drawings and Product Data within the time provided in the agreed upon schedule for conformance with information given and the design concept expressed in, or reasonably inferred from, the Contract Documents. The nature of all corrections to be made to the Shop Drawings and Product Data, if any, shall be clearly noted, and the submittals shall be returned to the Construction Manager for such corrections. If a change in the scope of the Work is intended by revisions requested to any Shop Drawings and Product Data, the Construction Manager shall be requested to prepare a change proposal in accordance with Article 35, Changes in the Work. On resubmitted Shop Drawings, Product Data or Samples, the Construction Manager shall direct specific attention in writing on the transmittal cover to revisions other than those corrections requested by the Architect/Engineer on any previously

checked submittal. The Architect/Engineer shall promptly review and comment on, and return, the resubmitted items.

The Construction Manager shall thereafter furnish such other copies in the form approved by the Architect/Engineer as may be needed for the prosecution of the Work.

13.2 FABRICATION AND ORDERING

Fabrication shall be started by the Construction Manager only after receiving approved Shop Drawings from the Architect/Engineer. Materials shall be ordered in accordance with approved Product Data. Work which is improperly fabricated, whether through incorrect Shop Drawings, faulty workmanship or materials, will not be acceptable.

13.3 DEVIATIONS FROM DRAWINGS OR SPECIFICATIONS

The review and comments of the Architect/Engineer of Shop Drawings, Product Data or Samples shall not relieve the Construction Manager from responsibility for deviations from the Drawings or Specifications, unless he or she has in writing called the attention of the Architect/Engineer to such deviations at the time of submission, nor shall it relieve the Construction Manager from responsibility for errors of any sort in Shop Drawings or Product Data. Review and comments on Shop Drawings or Product Data containing identified deviations from the Contract Documents shall not be the basis for a Change Order or a claim based on a change in the scope of the Work unless Notice is given to the Architect/Engineer and Principal Representative of all additional costs, time and other impacts of the identified deviation by bring it to their attention in writing at the time the submittals are made, and any subsequent change in the Contract sum or the Contract time shall be limited to cost, time and impacts so identified.

13.4 CONSTRUCTION MANAGER REPRESENTATIONS

By preparing, approving, and/or submitting Shop Drawings, Product Data and Samples, the Construction Manager represents that the Construction Manager has determined and verified all materials, field measurements, and field construction criteria related thereto, and has checked and coordinated the information contained within each submittal with the requirements of the Work, the Project and the Contract Documents and prior reviews and approvals.

14 ARTICLE 14 SAMPLES AND TESTING

14.1 SAMPLES

The Construction Manager shall furnish for approval, with such promptness as to cause no delay in his or her Work or in that of any other Construction Manager, all Samples as directed by the Architect/Engineer. The Architect/Engineer shall check and approve such Samples, with reasonable promptness, but only for conformance with the design intent of the Contract Documents and the Project, and for compliance with any submission requirements given in the Contract Documents.

14.2 TESTING - GENERAL

The Construction Manager shall provide such equipment and facilities as the Architect/Engineer may require for conducting field tests and for collecting and forwarding samples to be tested. Samples themselves shall not be incorporated into the Work after approval without the permission of the Architect/Engineer.

All materials or equipment proposed to be used may be tested at any time during their preparation or use. The Construction Manager shall furnish the required samples without charge

and shall give sufficient Notice of the placing of orders to permit the testing thereof. Products may be sampled either prior to shipment or after being received at the site of the Work.

Tests shall be made by an accredited testing laboratory. Except as otherwise provided in the Specifications, sampling and testing of all materials, and the laboratory methods and testing equipment, shall be in accordance with the latest standards and tentative methods of the American Society of Testing Materials (ASTM). The cost of testing which is in addition to the requirements of the Specifications shall be paid by the Construction Manager if so directed by the Architect/Engineer, and the Contract sum shall be adjusted accordingly by Change Order; provided however, that whenever testing shows portions of the Work to be deficient, all costs of testing including that required to verify the adequacy of repair or replacement Work shall be the responsibility of the Construction Manager.

14.3 TESTING - CONCRETE AND SOILS

Unless otherwise specified or provided elsewhere in the Contract Documents, the Principal Representative will contract for and pay for the testing of concrete and for soils compaction testing through an independent laboratory or laboratories selected and approved by the Principal Representative. The Construction Manager shall assume the responsibility of arranging, scheduling and coordinating the concrete sample collection efforts and soils compaction efforts in an efficient and cost effective manner. Testing shall be performed in accordance with the requirements of the Specifications, and if no requirements are specified, the Construction Manager shall request instructions and testing shall be as directed by the Architect/Engineer or the soils engineer, as applicable, and in accordance with standard industry practices.

The Principal Representative and the Architect/Engineer shall be given reasonable advance notice of each concrete pour and reserve the right to either increase or decrease the number of cylinders or the frequency of tests.

Soil compaction testing shall be at random locations selected by the soils engineer. In general, soils compaction testing shall be as directed by the soils engineer and shall include all substrate prior to backfill or construction.

14.4 TESTING - OTHER

Additional testing required by the Specifications will be accomplished and paid for by the Principal Representative in a manner similar to that for concrete and soils unless noted otherwise in the Specifications. In any case, the Construction Manager will be responsible for arranging, scheduling and coordinating additional tests. Where the additional testing will be contracted and paid for by the Principal Representative the Construction Manager shall give the Principal Representative not less than one-month advance written Notice of the date the first such test will be required.

15 ARTICLE 15 SUBCONTRACTS

15.1 CONTRACT PERFORMANCE OUTSIDE OF THE UNITED STATES OR COLORADO

After the contract is awarded, Construction Manager is required to provide written notice to the Principal Representative no later than twenty (20) days after deciding to perform services under this contract outside the United States or Colorado or to subcontract services under this contract to a subcontractor that will perform such services outside the United States or Colorado. The written notification must include, but need not be limited to, a statement of the type of services that will be performed at a location outside the United States or Colorado and the reason why it

is necessary or advantageous to go outside the United States or Colorado to perform the services. All notices received by the State pursuant to outsourced services shall be posted on the Colorado Department of Personnel & Administration's website. *If Construction Manager knowingly fails to notify the Principal Representative of any outsourced services as specified herein, the Principal Representative, at its discretion, may terminate this contract as provided in the Colorado Procurement Code or the applicable procurement code for institutions of higher education (Does not apply to any project that receives federal moneys).*

15.2 SUBCONTRACTOR PREQUALIFICATION

Prior to the Notice to Proceed to Commence Construction for the first and subsequent construction phases, the Construction Manager shall submit to the Principal Representative and State Buildings Program a complete list of all known Subcontractors, planned for the Work. The list shall be supplemented as other Subcontractors are determined by the Construction Manager and any such supplemental list shall be submitted to the Principal Representative and State Buildings Program not less than ten (10) days before the Subcontractors commence Work.

15.3 SUBCONTRACTOR PROPOSALS

If Construction Manager utilizes any Subcontractor on this Project, Construction Manager shall request and receive proposals from the Subcontractors and subcontracts will be awarded after the proposals are tabulated in a pre-approved format which compares to the Fixed Limit of Construction Cost budgeted by line item, as indicated in the finalized Construction Manager's Fee Proposal, and, reviewed by the Construction Manager, and Principal Representative.

Should the construction Manager submit a proposal for subcontract Work, the proposal conditions used shall be the same as for all subcontractor proposals. These Construction Manager proposals for subcontract Work shall be submitted to the Principal Representative twenty-four (24) hours prior to receipt of other subcontractor proposals and be opened with the other proposals.

15.4 SUBCONTRACT FORMS

All subcontracts will be between Construction Manager and the Subcontractors. The form of subcontracts shall be furnished to the Principal Representative for review and consent as to form, which consent shall not be unreasonably withheld.

15.5 SUBCONTRACTOR SUBSTITUTION

The substitution of any Subcontractor listed in the Construction Manager's bid shall be justified in writing not less than ten (10) days after the date of the Notice to Proceed to commence construction for the first and subsequent construction phases, and shall be subject to the approval of the Principal Representative. For reasons such as the Subcontractor's refusal to perform as agreed, subsequent unavailability or later discovered bid errors, or other similar reasons, but not including the availability of a lower Subcontract price, such substitution may be approved. The Construction Manager shall bear any additional cost incurred by such substitutions.

15.6 CONSTRUCTION MANAGER RESPONSIBLE FOR SUBCONTRACTORS

The Construction Manager shall not employ any Subcontractor that the Architect/Engineer, within ten (10) days after the date of receipt of the Construction Manager's list of Subcontractors or any supplemental list, objects to in writing as being unacceptable to either the Architect/Engineer, the Principal Representative or State Buildings Program. If a Subcontractor is deemed unacceptable, the Construction Manager shall propose a substitute Subcontractor and the Contract sum shall be adjusted by any demonstrated difference between the Subcontractor's bids, except where the Subcontractor has been debarred by the State or fails to meet qualifications of the Contract Documents to perform the Work proposed.

The Construction Manager shall be fully responsible to the Principal Representative for the acts and omissions of Subcontractors and of persons either directly or indirectly employed by them. All instructions or orders in respect to Work to be done by Subcontractors shall be given to the Construction Manager.

16 ARTICLE 16 RELATIONS OF CONSTRUCTION MANAGER AND SUBCONTRACTOR

The Construction Manager agrees to bind each Subcontractor to the terms of these General Conditions and to the requirements of the Drawings and Specifications, and any Addenda thereto, and also all the other Contract Documents, so far as applicable to the Work of such Subcontractor. The Construction Manager further agrees to bind each Subcontractor to those terms of the General Conditions which expressly require that Subcontractors also be bound, including without limitation, requirements that Subcontractors waive all rights of subrogation, provide adequate general commercial liability and property insurance, automobile insurance and workers' compensation insurance as provided in Article 25, Insurance.

Nothing contained in the Contract Documents shall be deemed to create any contractual relationship whatsoever between any Subcontractor and the State of Colorado acting by and through its Principal Representative.

17 ARTICLE 17 MUTUAL RESPONSIBILITY OF CONTRACTORS

Should the Construction Manager cause damage to any separate contractor on the Work, the Construction Manager agrees, upon due Notice, to settle with such contractor by agreement, if he or she will so settle. If such separate contractor sues the Principal Representative on account of any damage alleged to have been so sustained, the Principal Representative shall notify the Construction Manager, who shall defend such proceedings if requested to do so by Principal Representative. If any judgment against the Principal Representative arises there from, the Construction Manager shall pay or satisfy it and pay all costs and reasonable attorney fees incurred by the Principal Representative, in accordance with Article 53.24, Indemnification, provided the Construction Manager was given due Notice of an opportunity to settle.

18 ARTICLE 18 SEPARATE CONTRACTS

The Principal Representative reserves the right to enter into other contracts in connection with the Project or the Contract. The Construction Manager shall afford other contractors reasonable opportunity for the introduction and storage of their materials and the execution of their Work, and shall properly connect and coordinate his or her Work with theirs. If any part of the Construction Manager's Work depends, for proper execution or results, upon the Work of any other contractor, the Construction Manager shall inspect and promptly report to the Architect/Engineer any defects in such Work that render it unsuitable for such proper execution and results. Failure of the Construction Manager to so inspect and report shall constitute an acceptance of the other contractor's Work as fit and proper for the reception of Work, except as to defects which may develop in the other Construction Manager's Work after the execution of the Construction Manager's Work.

To insure the proper execution of subsequent Work, the Construction Manager shall measure Work already in place and shall at once report to the Architect/Engineer any discrepancy between the executed Work and the Drawings.

SC-6.51

Rev. 08/2022 22

19 ARTICLE 19 USE OF PREMISES

The Construction Manager shall confine apparatus, the storage of materials and the operations of workmen to limits indicated by law, ordinances, permits and any limits lines shown on the Drawings. The Construction Manager shall not unreasonably encumber the premises with materials.

The Construction Manager shall enforce all of the Architect/Engineer's instructions and prohibitions regarding, without limitation, such matters as signs, advertisements, fires, smoking and security.

20 ARTICLE 20 CUTTING, FITTING OR PATCHING

The Construction Manager shall do all cutting, fitting or patching of Work that may be required to make its several parts come together properly and fit it to receive or be received by Work of other contractors shown upon, or reasonably inferred from, the Drawings and Specifications for the complete structure, and shall provide for such finishes to patched or fitted Work as the Architect/Engineer may direct. The Construction Manager shall not endanger any Work by cutting, excavating or otherwise altering the Work and shall not cut or alter the Work of any other contractor save with the consent of the Architect/Engineer.

21 ARTICLE 21 UTILITIES

21.1 TEMPORARY UTILITIES

Unless otherwise specifically stated in the Specifications or on the Drawings, the Principal Representative shall be responsible for the locations of all utilities as shown on the Drawings or indicated elsewhere in the Specifications, subject to the Construction Manager's compliance with all statutory or regulatory requirements to call for utility locates. When actual conditions deviate from those shown the Construction Manager shall comply with the requirements of Article 37, Differing Site Conditions. The Construction Manager shall provide and pay for the installation of all temporary utilities required to supply all the power, light and water needed by him and other contractors for their Work and shall install and maintain all such utilities in such manner as to protect the public and Workmen and conform with any applicable laws and regulations. Upon completion of the Work, he or she shall remove all such temporary utilities from the site. The Construction Manager shall pay for all consumption of power, light and water used by him or her and the other contractors, without regard to whether such items are metered by temporary or The Superintendent shall have full authority over all trades and permanent meters. Subcontractors at any tier to prevent waste. The cut-off date on permanent meters shall be either the agreed date of the date of the Notice of Substantial Completion or the Notice of Approval of Occupancy/Use of the Project.

21.2 PROTECTION OF EXISTING UTILITIES

Where existing utilities, such as water mains, sanitary sewers, storm sewers and electrical conduits, are shown on the Drawings, the Construction Manager shall be responsible for the protection thereof, without regard to whether any such utilities are to be relocated or removed as a part of the Work. If any utilities are to be moved, the moving must be conducted in such manner as not to cause undue interruption or delay in the operation of the same.

21.3 CROSSING OF UTILITIES

When new construction crosses highways, railroads, streets, or utilities under the jurisdiction of State, city or other public agency, public utility or private entity, the Construction Manager shall secure proper written permission before executing such new construction. The Construction Manager will be required to furnish a proper release before final acceptance of the Work.

22 ARTICLE 22 UNSUITABLE CONDITIONS

The Construction Manager shall not Work at any time, or permit any Work to be done, under any conditions contrary to those recommended by manufacturers or industry standards which are otherwise proper, unsuited for proper execution, safety and performance. Any cost caused by ill-timed Work shall be borne by the Construction Manager unless the timing of such Work shall have been directed by the Architect/Engineer or the Principal Representative, after the award of the Contract, and the Construction Manager provided Notice of any additional cost.

23 ARTICLE 23 TEMPORARY FACILITIES

23.1 OFFICE FACILITIES

The Construction Manager shall provide and maintain without additional expense for the duration of the Project temporary office facilities, as required and as specified, for its own use and the use of the Architect/Engineer, representatives of the Principal Representative and State Buildings Program.

23.2 TEMPORARY HEAT

The Construction Manager shall furnish and pay for all the labor, facilities, equipment, fuel and power necessary to supply temporary heating, ventilating and air conditioning, except to the extent otherwise specified, and shall be responsible for the installation, operation, maintenance and removal of such facilities and equipment. Unless otherwise specified, the permanent HVAC system shall not be used for temporary heat in whole or in part. If the Construction Manager desires to put the permanent system into use, in whole or in part, the Construction Manager shall set it into operation and furnish the necessary fuel and manpower to safely operate, protect and maintain that HVAC system. Any operation of all or any part of the permanent HVAC system including operation for testing purposes shall not constitute acceptance of the system, nor shall it relieve the Construction Manager of his or her one-year guarantee of the system from the date of the Notice of Substantial Completion of the entire Project, and if necessary due to prior operation, the Construction Manager shall provide manufacturers' extended warranties from the date of the Construction Manager's use prior to the date of the Notice of Substantial Completion.

23.3 WEATHER PROTECTION

The Construction Manager shall, at all times, provide protection against weather, so as to maintain all Work, materials, apparatus and fixtures free from injury or damages. The Construction Manager shall provide weather-tight storage on substantial floors at least six (6) inches off the ground for all materials requiring protection from the weather.

23.4 DUST PARTITIONS

If the Work involves Work in an occupied existing building, the Construction Manager shall erect and maintain during the progress of the Work, suitable dust-proof temporary partitions, or more permanent partitions as specified, to protect such building and the occupants thereof.

23.5 BENCH MARKS

The Construction Manager shall maintain any site bench marks provided by the Principal Representative and shall establish any additional benchmarks specified by the Architect/Engineer as necessary for the Construction Manager to layout the Work and ascertain all grades and levels as needed.

23.6 SIGN

The Construction Manager shall erect and permit one 4' x 8' sign only at the site to identify the Project as specified or directed by the Architect/Engineer which shall be maintained in good condition during the life of the Project.

23.7 SANITARY PROVISION

The Construction Manager shall provide and maintain suitable, clean, temporary sanitary toilet facilities for any and all workmen engaged on the Work, for the entire construction period, in strict compliance with the requirement of all applicable codes, regulations, laws and ordinances, and no other facilities, new or existing, may be used by any person on the Project. When the Project is complete the Construction Manager shall promptly remove them from the site, disinfect, and clean or treat the areas as required. If any new construction surfaces in the Project other than the toilet facilities provided for herein are soiled at any time, the entire areas so soiled shall be completely removed from the Project and rebuilt. In no event may present toilet facilities of any existing building at the site of the Work be used by employees of the Construction Manager or any subcontractors.

24 ARTICLE 24 CLEANING UP

The Construction Manager shall keep the building and premises free from all surplus material, waste material, dirt and rubbish caused by employees or Work, and at the completion of the Work shall remove all such surplus material, waste material, dirt, and rubbish, as well as all tools, equipment and scaffolding, and shall wash and clean all window glass and plumbing fixtures, perform cleanup and cleaning required by the Specifications and leave all of the Work clean unless more exact requirements are specified.

25 ARTICLE 25 INSURANCE

25.1 GENERAL

The Construction Manager shall procure and maintain all insurance requirements and limits as set forth below, at his or her own expense, for the length of time set forth in Contract requirements. The Construction Manager shall continue to provide evidence of such coverage to State of Colorado on an annual basis during the aforementioned period including all of the terms of the insurance and indemnification requirements of this agreement. All below insurance policies shall include a provision preventing cancellation without thirty (30) days' prior notice by certified mail. A completed Certificate of Insurance shall be filed with the Principal Representative and State Buildings Program within ten (10) days after the date of the Notice of Award, said Certificate to specifically state the inclusion of the coverages and provisions set forth herein and shall state whether the coverage is "claims made" or "per occurrence".

25.2 COMMERCIAL GENERAL LIABILITY INSURANCE (CGL)

This insurance must protect the Construction Manager from all claims for bodily injury, including death and all claims for destruction of or damage to property (other than the Work itself), arising out of or in connection with any operations under this Contract, whether such operations be by

the Construction Manager or by any Subcontractor under him or anyone directly or indirectly employed by the Construction Manager or by a Subcontractor. All such insurance shall be written with limits and coverages as specified below and shall be written on an occurrence form.

| General Aggregate | \$2,000,000 |
|---|-------------|
| Products – Completed Operations Aggregate | \$2,000,000 |
| Each Occurrence | \$1,000,000 |
| Personal Injury | \$1,000,000 |

The following coverages shall be included in the CGL:

- a) Per project general aggregate (CG 25 03 or similar)
- b) Additional Insured status in favor of the State of Colorado and any other parties as outlined in The Contract and must include both ONGOING Operations AND COMPLETED Operations per CG2010 10/01 and CG 2037 10/01 or equivalent as permitted by law.
- c) The policy shall be endorsed to be **primary and non-contributory** with any insurance maintained by Additional Insureds.
- d) A waiver of Subrogation in favor of all Additional Insured parties.
- e) Personal Injury Liability
- f) Contractual Liability coverage to support indemnification obligation per Article 53.24
- g) Explosion, collapse and underground (xcu)

The following exclusionary endorsements are prohibited in the CGL policy:

- a) Damage to Work performed by Subcontract/Vendor (CG 22-94 or similar)
- b) Contractual Liability Coverage Exclusion modifying or deleting the definition of an "insured contract" from the unaltered SO CG 0001 1001 policy from (CG 24 26 or similar)
- c) If applicable to the Work to be performed: Residential or multi-family
- d) If applicable to the Work to be performed: Exterior insulation finish systems
- e) If applicable to the Work to be performed: Subsidence or Earth Movement

The Construction Manager shall maintain general liability coverage including Products and Completed Operations insurance, and the Additional Insured with primary and non-contributory coverage as specified in this Contract for three (3) years after completion of the project.

25.3 AUTOMOBILE LIABILITY INSURANCE

Automobile liability insurance and business auto liability covering liability arising out of any auto (including owned, hired and non-owned autos).

Combined Bodily Injury and Property Damage Liability (Combined Single Limit): \$1,000,000 each accident

Coverages: Specific waiver of subrogation

25.4 WORKERS' COMPENSATION INSURANCE

The Construction Manager shall procure and maintain Workers' Compensation Insurance at his or her own expense during the life of this Contract, including occupational disease provisions for all employees per statutory requirements. Policy shall contain a waiver of subrogation in favor of the State of Colorado.

The Construction Manager shall also require each Subcontractor to furnish Workers' Compensation Insurance, including occupational disease provisions for all of the latter's employees, and to the extent not furnished, the Construction Manager accepts full liability and responsibility for Subcontractor's employees.

In cases where any class of employees engaged in hazardous Work under this Contract at the site of the Project is not protected under the Workers' Compensation statute, the Construction Manager shall provide, and shall cause each Subcontractor to provide, adequate and suitable insurance for the protection of employees not otherwise protected.

25.5 UMBRELLA LIABILITY INSURANCE:

For construction projects exceeding \$10,000,000, provide the following coverage: The Construction Manager shall maintain umbrella/excess liability insurance on an occurrence basis in excess of the underlying insurance described in Section B-D above. Coverage shall follow the terms of the underlying insurance, included the additional insured and waiver of subrogation provisions. The amounts of insurance required in Sections above may be satisfied by the Construction Manager purchasing coverage for the limits specified or by any combination of underlying and umbrella limits, so long as the total amount of insurance is not less than the limits specified in each section previously mentioned.

Each occurrence \$5,000,000 Aggregate \$5,000,000

25.6 BUILDER'S RISK INSURANCE

Unless otherwise expressly stated in the Supplementary General Conditions (e.g. where the State elects to provide for projects with a completed value of less than \$1,000,000), the Construction Manager shall purchase and maintain, in a company or companies lawfully authorized to do business in the jurisdiction in which the Project is located, property insurance written on a builder's risk "all-risk" or equivalent policy form in the amount of the initial Contract Sum, plus value of subsequent Contract Modifications and cost of materials supplied or installed by others, comprising total value for the entire Project at the site on a replacement cost basis without optional deductibles. Such property insurance shall be maintained, unless otherwise provided in the Contract Documents or otherwise agreed in writing by all persons and entities who are beneficiaries of such insurance, until final payment has been made or until no person or entity other than the Owner has an insurable interest in the property, or the Date of Notice specified on the Notice of Acceptance, State Form SBP-6.27 or whichever is later.

This insurance shall include interests of the Owner, the Construction Manager, Subcontractors and Sub-subcontractors in the Project as named insureds.

All associated deductibles shall be the responsibility of the Construction Manager. Such policy may have a deductible clause but not to exceed ten thousand dollars (\$10,000.00).

Property insurance shall be on an "all risk" or equivalent policy form and shall include, without limitation, insurance against the perils of fire (with extended coverage) and physical loss or damage including, without duplication of coverage, theft, vandalism, malicious mischief, collapse, earthquake, flood, windstorm, false Work, testing and startup, temporary buildings and debris removal including demolition occasioned by enforcement of any applicable legal requirements, and shall cover reasonable compensation for Architect's and Construction Manager's services and expenses required as a result of such insured loss.

SC-6.51

Rev. 08/2022 27

Construction Manager shall maintain Builders Risk coverage including partial use by Owner.

The Construction Manager shall waive all rights of subrogation as regards the State of Colorado and the Principal Representative, its officials, its officers, its agents and its employees, all while acting within the scope and course of their employment for damages caused by fire or other causes of loss to the extent covered by property insurance obtained pursuant to this Section or other property insurance applicable to the Work. The Construction Manager shall require all Subcontractors at any tier to similarly waive all such rights of subrogation and shall expressly include such a waiver in all subcontracts.

Upon request, the amount of such insurance shall be increased to include the cost of any additional Work to be done on the Project, or materials or equipment to be incorporated in the Project, under other independent contracts let or to be let. In such event, the Construction Manager shall be reimbursed for this cost as his or her share of the insurance in the same ratio as the ratio of the insurance represented by such independent contracts let or to be let to the total insurance carried.

The Principal Representative, with approval of the State Controller, shall have the power to adjust and settle any loss. Unless it is agreed otherwise, all monies received shall be applied first on rebuilding or repairing the destroyed or injured Work.

25.7 POLLUTION LIABILITY INSURANCE

If Construction Manager is providing directly or indirectly Work with pollution/environmental hazards, the Construction Manager must provide or cause those conducting the Work to provide Pollution Liability Insurance coverage. Pollution Liability policy must include contractual liability coverage. State of Colorado must be included as additional insureds on the policy. The policy limits shall be in the amount of \$1,000,000 with maximum deductible of \$25,000 to be paid by the Subcontractor/Vendor.

25.8 ADDITIONAL MISCELLANEOUS INSURANCE PROVISIONS

Certificates of Insurance and/or insurance policies required under this Contract shall be subject to the following stipulations and additional requirements:

- 25.8.1 Any and all deductibles or self-insured retentions contained in any Insurance policy shall be assumed by and at the sole risk of the Construction Manager;
- 25.8.2 If any of the said policies shall fail at any time to meet the requirements of the Contract Documents as to form or substance, or if a company issuing any such policy shall be or at any time cease to be approved by the Division of Insurance of the State of Colorado, or be or cease to be in compliance with any stricter requirements of the Contract Documents, the Construction Manager shall promptly obtain a new policy, submit the same to the Principal Representative and State Building Programs for approval if requested, and submit a Certificate of Insurance as hereinbefore provided. Upon failure of the Construction Manager to furnish, deliver and maintain such insurance as provided herein, this Contract, in the sole discretion of the State of Colorado, may be immediately declared suspended, discontinued, or terminated. Failure of the Construction Manager in obtaining and/or maintaining any required insurance shall not relieve the Construction Manager from any liability under the Contract, nor shall the insurance

- requirements be construed to conflict with the obligations of the Construction Manager concerning indemnification;
- 25.8.3 All requisite insurance shall be obtained from financially responsible insurance companies, authorized to do business in the State of Colorado and acceptable to the Principal Representative;
- 25.8.4 Receipt, review or acceptance by the Principal Representative of any insurance policies or certificates of insurance required by this Contract shall not be construed as a waiver or relieve the Construction Manager from its obligation to meet the insurance requirements contained in these General Conditions.

26 ARTICLE 26 CONSTRUCTION MANAGER'S PERFORMANCE AND PAYMENT INSURANCE

The Construction Manager shall furnish a Performance Bond and a Labor and Material Payment Bond on State Forms SC-6.22, Performance Bond, and SC-6.221, Labor and Material Payment Bond, or such other forms as State Buildings Program may approve for the Project, executed by a corporate Surety authorized to do business in the State of Colorado and in the full amount of the Contract sum. The expense of these bonds shall be borne by the Contractor and the bonds shall be filed with State Buildings Program.

If, at any time, a Surety on such a bond is found to be, or ceases to be in strict compliance with any qualification requirements of the Contract Documents or the bid documents, or loses its right to do business in the State of Colorado, another Surety will be required, which the Construction Manager shall furnish to State Buildings Program within ten (10) days after receipt of Notice from the State or after the Construction Manager otherwise becomes aware of such conditions.

27 ARTICLE 27 LABOR AND WAGES

27.1 LABOR

In accordance with laws of Colorado, C.R.S. § 8-17-101(1), as amended, Colorado labor shall be employed to perform at least eighty percent of the Work. If the Federal Davis-Bacon Act shall be applicable to the Project, as indicated in Article 7.3 of the Agreement, Modification of Article 27, the minimum wage rates to be paid on the Project will be specified in the Contract Documents.

27.2 COLORADO WAGES

In accordance with laws of Colorado, C.R.S. § 24-92 Part 2, if prevailing wage rates are applicable as indicated in Article 7.4 of the Agreement, Modification of Article 27, to this project:

- 27.2.1 The contractor shall in conspicuous places on the project post an owner provided poster with the current prevailing rate of payments as provided in the project solicitation.
 - A contractor who fails to comply shall be deemed guilty of a class 3 misdemeanor and shall pay the State one hundred dollars (\$100) for each calendar day of noncompliance as determined by the State.
- 27.2.2 The contractor and any subcontractors shall pay all the employees employed directly on the site of the work, unconditionally and not less often than once a week, and without subsequent deduction or rebate on any account, the full amounts accrued at time of payment computed at wage rates not less than those stated in the competitive solicitation, regardless of any contractual

- relationships that may be alleged to exist between the contractor or subcontractor and the employees.
- 27.2.3 The contractor and any subcontractors shall prepare and submit electronic payroll reports to the State in a format approved by OSA on a monthly basis that disclose all relevant payroll information, including the name and address of any entities to which fringe benefits are paid.
- 27.2.4 The contractor and any subcontractors shall maintain on the site where public projects are being constructed a daily log of employees employed each day on the public project. The log shall include, at a minimum, for each employee his or her name, primary job title, and employer, and shall be kept on a form prescribed by the director. The log shall be available for inspection on the site at all times by the State.
- 27.2.5 If the contractor or any subcontractor fails to pay wages as are required by the contract, the State shall not approve a warrant or demand for payment to the contractor until the contractor furnishes the State evidence satisfactory to such agency of government that such wages have been paid; except that the State shall approve and pay any portion of a warrant or demand for payment to the contractor to the extent the State has been furnished satisfactory evidence that the contractor or one or more subcontractors has paid such wages required by the contract, The contractor or subcontractor may use the following procedure in order to satisfy the requirements of this section:
- 27.2.6 The contractor or subcontractor may submit to the State, for each employee to whom such wages are due, a check payable to that employee or to the State so it is negotiable by either party. Each such check shall be in an amount representing the difference between the accrued wages required to be paid to that employee by the contract and the wages actually paid by the contractor or subcontractor.
- 27.2.7 If any check submitted cannot be delivered to the employee within a reasonable period, then it shall be negotiated by the State and the proceeds deposited in the unclaimed property trust fund created in Section 38-13-116.6. Nothing in this subsection (1) shall be construed to lessen the responsibility of the contractor or subcontractor to attempt to locate and pay any employee to whom wages are due.

28 ARTICLE 28 ROYALTIES AND PATENTS

The Construction Manager shall be responsible for assuring that all rights to use of products and systems have been properly arranged and shall take such action as may be necessary to avoid delay, at no additional charge to the Principal Representative, where such right is challenged during the course of the Work. The Construction Manager shall pay all royalties and license fees required to be paid and shall defend all suits or claims for infringement of any patent rights and shall save the State of Colorado harmless from loss on account thereof, in accordance with Article 53.24, Indemnification; provided, however, the Construction Manager shall not be responsible for such loss or defense for any copyright violations contained in the Contract Documents prepared by the Architect/Engineer or the Principal Representative of which the Construction Manager is unaware, or for any patent violations based on specified processes that the Construction Manager is unaware are patented or that the Construction Manager should not have had reason to believe were patented.

29 ARTICLE 29 ASSIGNMENT

Except as otherwise provided hereafter the Construction Manager shall not assign the whole or any part of this Contract without the written consent of the Principal Representative. This provision shall not be construed to prohibit assignments of the right to payment to the extent permitted by C.R.S. § 4-9-406, et. seq., as amended, provided that written Notice of assignment adequate to identify the rights assigned is received by the Principal Representative and the controller for the agency, department, or institution executing this Contract (as distinguished from the State Controller). Such assignment of the right to payment shall not be deemed valid until receipt by the Principal Representative and such controller and the Construction Manager assumes the risk that such written Notice of assignment is received by the Principal Representative and the controller for the agency, department, or institution involved. In case the Construction Manager assigns all or part of any moneys due or to become due under this Contract, the instrument of assignment shall contain a clause substantially to the effect that it is agreed that the right of the assignee in and to any moneys due or to become due to the Construction Manager shall be subject to all claims of all persons, firms, and corporations for services rendered or materials supplied for the performance of the Work called for in this Contract, whether said service or materials were supplied prior to or after the assignment. Nothing in this Article shall be deemed a waiver of any other defenses available to the State against the Construction Manager or the assignee.

30 ARTICLE 30 CORRECTION OF WORK BEFORE ACCEPTANCE

The Construction Manager shall promptly remove from the premises all Work or materials condemned or declared irreparably defective as failing to conform to the Contract Documents on receipt of written Notice from the Architect/Engineer or the Principal Representative, whether incorporated in the Work or not. If such materials shall have been incorporated in the Work, or if any unsatisfactory Work is discovered, the Construction Manager shall promptly replace and reexecute his or her Work in accordance with the requirements of the Contract Documents without expense to the Principal Representative, and shall also bear the expense of making good all Work of other contractors destroyed or damaged by the removal or replacement of such defective material or Work.

Should any defective Work or material be discovered during the process of construction, or should reasonable doubt arise as to whether certain material or Work is in accordance with the Contract Documents, the value of such defective or questionable material or Work shall not be included in any application for payment, or if previously included, shall be deducted by the Architect/Engineer from the next application submitted by the Construction Manager r.

If the Construction Manager does not perform repair, correction and replacement of defective Work, in lieu of proceeding by issuance of a Notice of intent to remove condemned Work as outlined above, the Principal Representative may, not less than seven (7) days after giving the original written Notice of the need to repair, correct, or replace defective Work, deduct all costs and expenses of replacement or correction as instructed by the Architect/Engineer from the Construction Manager's next application for payment in addition to the value of the defective Work or material. The Principal Representative may also make an equitable deduction from the Contract sum by unilateral Change Order, in accordance with Article 33, Payments Withheld and Article 35, Changes in the Work.

If the Construction Manager does not remove such condemned or irreparably defective Work or material within a reasonable time, the Principal Representative may, after giving a second seven

(7) day advance Notice to the Construction Manager and the Surety, remove them and may store the material at the Construction Manager's expense. The Principal Representative may accomplish the removal and replacement with its own forces or with another contractor. If the Construction Manager does not pay the expense of such removal and pay all storage charges within ten (10) days thereafter, the Principal Representative may, upon ten (10) days' written Notice, sell such material at auction or at private sale and account for the net proceeds thereof, after deducting all costs and expenses which should have been borne by the Construction Manager. If the Construction Manager shall commence and diligently pursue such removal and replacement before the expiration of the seven-day period, or if the Construction Manager shall show good cause in conjunction with submittal of a revised CPM schedule showing when the Work will be performed and why such removal of condemned Work should be scheduled for a later date, the Principal Representative shall not proceed to remove or replace the condemned Work.

If the Construction Manager disagrees with the Notice to remove Work or materials condemned or declared irreparably defective, the Construction Manager may request facilitated negotiation of the issue and the Principal Representative's right to proceed with removal and to deduct costs and expenses of repair shall be suspended and tolled until such time as the parties meet and negotiate the issue

During construction, whenever the Architect/Engineer has advised the Construction Manager in writing, in the Specifications, by reference to Article 6, Architect/Engineer Decisions And Judgments, of these General Conditions or elsewhere in the Contract Documents of a need to observe materials in place prior to their being permanently covered up, it shall be the Construction Manager's responsibility to notify the Architect/Engineer at least forty-eight (48) hours in advance of such covering operation. If the Construction Manager fails to provide such notification, Construction Manager shall, at his or her expense, uncover such portions of the Work as required by the Architect/Engineer for observation, and reinstall such covering after observation. When a covering operation is continued from day to day, notification of the commencement of a single continuing covering operation shall suffice for the activity specified so long as it proceeds regularly and without interruption from day to day, in which event the Construction Manager shall coordinate with the Architect/Engineer regarding the continuing covering operation.

31 ARTICLE 31 APPLICATIONS FOR PAYMENTS

31.1 CONSTRUCTION MANAGER'S SUBMITTALS

On or before the first day of each month and no more than five days prior thereto, the Construction Manager may submit applications for payment for the Work performed during such month covering the portion of the Work completed as of the date indicated, and payments on account of this Contract shall be due per C.R.S. § 24-30-202(24) (correct notice of amount due), within forty-five (45) days of receipt by the Principal Representative of application for payments that have been certified by the Architect/Engineer. The Construction Manager shall submit the application for payment to the Architect/Engineer on State forms SBP-7.2, Application and Certificate for Contractor's Payment, or such other format as the State Buildings Program shall approve, in an itemized format in accordance with the schedule of values or a cost loaded CPM schedule when required, supported to the extent reasonably required by the Architect/Engineer or the Principal Representative by receipts or other vouchers, showing payments for materials and labor, prior payments and payments to be made to Subcontractors and such other evidence

of the Construction Manager's right to payments as the Architect/Engineer or Principal Representative may direct.

If payments are made on account of materials not incorporated in the Work but delivered and suitably stored at the site, or at some other location agreed upon in writing, such payments shall be conditioned upon submission by the Construction Manager of bills of sale or such other procedure as will establish the Principal Representative's title to such material or otherwise adequately protect the Principal Representative's interests, and shall provide proof of insurance whenever requested by the Principal Representative or the Architect/Engineer, and shall be subject to the right to inspect the materials at the request of either the Architect/Engineer or the Principal Representative.

All applications for payment, except the final application, and the payments there under, shall be subject to correction in the next application rendered following the discovery of any error.

31.2 ARCHITECT/ENGINEER CERTIFICATION

In accordance with the Architect/Engineer's agreement with the Principal Representative, the Architect/Engineer after appropriate observation of the progress of the Work shall certify to the Principal Representative the amount that the Construction Manager is entitled to, and forward the application to the Principal Representative. If the Architect/Engineer certifies an amount different from the amount requested or otherwise alters the Construction Manager's application for payment, a copy shall be forwarded to the Construction Manager.

If the Architect/Engineer is unable to certify all or portions of the amount requested due to the absence or lack of required supporting evidence, the Architect/Engineer shall advise the Construction Manager of the deficiency. If the deficiency is not corrected at the end of ten (10) days, the Architect/Engineer may either certify the remaining amounts properly supported to which the Construction Manager is entitled, or return the application for payment to the Construction Manager for revision with a written explanation as to why it could not be certified.

31.3 RETAINAGE WITHHELD

Unless otherwise provided in the Supplementary General Conditions, an amount equivalent to five percent (5%) of the amount shown to be due the Construction Manager on each application for payment shall be withheld until the Work required by the Contract has been performed. The withheld percentage of the contract price of any such Work, improvement, or construction shall be administered according to C.R. S. § 24-91-103, as amended, and C.R.S. § 38-26-107, as amended, and Article 31.4, shall be retained until the Work or discrete portions of the Work, have been completed satisfactorily, finally or partially accepted, and advertised for final settlement as further provided in Article 41.

31.4 RELEASE OF RETAINAGE

The Construction Manager may, for satisfactory and substantial reasons shown to the Principal Representative's satisfaction, make a written request to the Principal Representative and the Architect/Engineer for release of part or all of the withheld percentage applicable to the Work of a Subcontractor which has completed the subcontracted Work in a manner finally acceptable to the Architect/Engineer, the Construction Manager, and the Principal Representative. Any such request shall be supported by a written approval from the Surety furnishing the Construction Manager's bonds and any surety that has provided a bond for the Subcontractor. The release of any such withheld percentage shall be further supported by such other evidence as the Architect/Engineer or the Principal Representative may require, including but not limited to, evidence of prior payments made to the Subcontractor, copies of the Subcontractor's contract

with the Construction Manager, any applicable warranties, as-built information, maintenance manuals and other customary close-out documentation. Neither the Principal Representative nor the Architect Engineer shall be obligated to review such documentation nor shall they be deemed to assume any obligations to third parties by any review undertaken.

The Construction Manager's obligation under these General Conditions to guarantee Work for one year from the date of the Notice of Substantial Completion or the date of any Notice of Partial Substantial Completion of the applicable portion or phase of the Project, shall be unaffected by such partial release; unless a Notice of Partial Substantial Completion is issued for the Work subject to the release of retainage.

Any rights of the Principal Representative which might be terminated by or from the date of any final acceptance of the Work, whether at common law or by the terms of this Contract, shall not be affected by such partial release of retainage prior to any final acceptance of the entire Project.

The Construction Manager remains fully responsible for the Subcontractor's Work and assumes any risk that might arise by virtue of the partial release to the Subcontractor of the withheld percentage, including the risk that the Subcontractor may not have fully paid for all materials, labor and equipment furnished to the Project.

If the Principal Representative considers the Construction Manager's request for such release satisfactory and supported by substantial reasons, the Architect/Engineer shall make a "final inspection" of the applicable portion of the Project to determine whether the Subcontractor's Work has been completed in accordance with the Contract Documents. A final punch list shall be made for the Subcontractor's Work and the procedures of Article 41, Completion, Final Inspection, Acceptance and Settlement, shall be followed for that portion of the Work, except that advertisement of the intent to make final payment to the Subcontractor shall be required only if the Principal Representative has reason to believe that a supplier or Subcontractor to the Subcontractor for which the request is made, may not have been fully paid for all labor and materials furnished to the Project.

32 ARTICLE 32 CERTIFICATES FOR PAYMENTS

State Form SBP-7.2, Certificate For Contractor's Payment, and its continuation detail sheets, when submitted, shall constitute the Construction Manager 's Application for Payment, and shall be a representation by the Construction Manager to the Principal Representative that the Work has progressed to the point indicated, the quality of the Work is in accordance with the Contract Documents, and materials for which payment is requested have been incorporated into the Project except as noted in the application. If requested by the Principal Representative the Certificate of Contractor's Application for Payment shall be sworn under oath and notarized.

33 ARTICLE 33 PAYMENTS WITHHELD

The Architect/Engineer, the Principal Representative or State Buildings Program may withhold, or on account of subsequently discovered evidence nullify, the whole or any part of any application on account of, but not limited to any of the following:

- a) Defective Work not remedied;
- b) Claims filed or reasonable evidence indicating probable filing of claims;

SC-6.51

Rev. 08/2022 34

- c) Failure of the Construction Manager to make payments to Subcontractors for material or labor;
- d) A reasonable doubt that the Contract can be completed for the balance of the contract price then unpaid;
- e) Damage or injury to another contractor or any other person, persons or property except to the extent of coverage by a policy of insurance;
- f) Failure to obtain necessary permits or licenses or to comply with applicable laws, ordinances, codes, rules or regulations or the directions of the Architect/Engineer;
- g) Failure to submit a monthly construction schedule;
- h) Failure of the Construction Manager to keep Work progressing in accordance with the time schedule;
- i) Failure to keep a superintendent on the Work;
- j) Failure to maintain as built drawings of the Work in progress;
- k) Unauthorized deviations by the Construction Manager from the Contract Documents; or
- I) On account of liquidated damages.

In addition, the Architect Engineer, Principal Representative or State Buildings Program may withhold or nullify the whole or any part of any application for any reason noted elsewhere in these General Conditions of the Construction Manager/General Contractor (CM/GC) Agreement. Nullification shall mean reduction of amounts shown as previously paid on the application. The amount withheld or nullified may be in such amount as the Architect/Engineer or the Principal Representative estimates to be required to allow the State to accomplish the Work, cure the failure and cover any damages or injuries, including an allowance for attorneys' fees and costs where appropriate. When the grounds for such withholding or nullifying are removed, payment shall be made for the amounts thus withheld or nullified on such grounds.

34 ARTICLE 34 DEDUCTIONS FOR UNCORRECTED WORK

If the Architect/Engineer and the Principal Representative deem it inexpedient to correct Work damaged or not performed in accordance with the Contract Documents, the Principal Representative may, after consultation with the Architect/Engineer and ten (10) days' Notice to the Construction Manager of intent to do so, make reasonable reductions from the amounts otherwise due the Construction Manager on the next application for payment. Notice shall specify the amount or terms of any contemplated reduction. The Construction Manager may during this period correct or perform the Work. If the Construction Manager does not correct or perform the Work, an equitable deduction from the Contract sum shall be made by Change Order, in accordance with Article 35, Changes in the Work, unilaterally if necessary. If either party elects' facilitation of this issue after Notice is given, the ten-day (10) notice period shall be extended and tolled until facilitation has occurred.

35 ARTICLE 35 CHANGES IN THE WORK

The Principal Representative may designate, without invalidating the Agreement, and with the approval of State Buildings Program and the State Controller, may order extra Work or make changes with or without the consent of the Construction Manager as hereafter provided, by altering, adding to or deducting from the Work, the Contract sum being adjusted accordingly. All such changes in the Work shall be within the general scope of and be executed under the conditions of the Contract, except that any claim for extension of time made necessary due to the

change or any claim of other delay or other impacts caused by or resulting from the change in the Work shall be presented by the Construction Manager and adjusted by Change Order to the extent known at the time such change is ordered and before proceeding with the extra or changed Work. Any claims for extension of time or of delay or other impacts, and any costs associated with extension of time, delay or other impacts, which are not presented before proceeding with the change in the Work, and which are not adjusted by Change Order to the extent known, shall be waived.

The Architect/Engineer shall have authority to make minor changes in the Work, not involving extra cost, and not inconsistent with the intent of the Contract Documents, but otherwise, except in an emergency endangering life or property, no extra Work or change in the Contract Documents shall be made unless by 1) a written Change Order, approved by the Principal Representative, State Buildings Program, and the State Controller prior to proceeding with the changed Work; or 2) by an Emergency Field Change Order approved by the Principal Representative and State Buildings Program as hereafter provided in Article 35.4, Emergency Field Ordered Changed Work; or 3) by an allocation in writing of any allowance already provided in the encumbered contract amount, the Contract sum being later adjusted to decrease the Contract sum by any unallocated or unexpended amounts remaining in such allowance. No change to the Contract sum shall be valid unless so ordered.

35.1 THE VALUE OF CHANGED WORK

- 35.1.1 The value of any extra Work or changes in the Work shall be determined by agreement in one or more of the following ways:
 - a) By estimate and acceptance of a lump-sum amount;
 - b) By unit prices specified in the Agreement, or subsequently agreed upon, that are extended by specific quantities;
 - c) By actual cost plus a fixed fee in a lump sum amount for profit, overhead and all indirect and off-site home office costs, the latter amount agreed upon in writing prior to starting the extra or changed Work.
- 35.1.2 Where the Construction Manager and the Principal Representative cannot agree on the value of extra Work, the Principal Representative may order the Construction Manager to perform the changes in the Work and a Change Order may be unilaterally issued based on an estimate of the change in the Work prepared by the Architect/Engineer. The value of the change in the Work shall be the Principal Representative's determination of the amount of equitable adjustment attributable to the extra Work or change. The Principal Representative's determination shall be subject to appeal by the Construction Manager pursuant to the claims process in Article 36, Claims.
- 35.1.3 Except as otherwise provided in Article 35.2, Detailed Breakdown, below, the Cost Principles of the Colorado Procurement Code or the applicable procurement code for institutions of higher education, shall govern all Contract changes.

35.2 DETAILED BREAKDOWN

In all cases where the value of the extra or changed Work is not known based on unit prices in the Construction Manager's bid or the Agreement, a detailed change proposal shall be submitted by the Construction Manager on a Change Order Proposal (SC-6.312), or in such other format as the State Buildings Program approves, with which the Principal Representative may require an itemized list of materials, equipment and labor, indicating quantities, time and cost for completion of the changed Work.

Such detailed change proposals shall be stated in lump sum amounts and shall be supported by a separate breakdown, which shall include estimates of all or part of the following when requested by the Architect/Engineer or the Principal Representative:

- a) Materials, indicating quantities and unit prices including taxes and delivery costs if any (separated where appropriate into general, mechanical and electrical and/or other Subcontractors' Work; and the Principal Representative may require in its discretion any significant subcontract costs to be similarly and separately broken down).
- b) Labor costs, indicating hourly rates and time and labor burden to include Social Security and other payroll taxes such as unemployment, benefits and other customary burdens.
- c) Costs of project management time and superintendence time of personnel stationed at the site, and other field supervision time, but only where a time extension, other than a weather delay, is approved as part of the Change Order, and only where such project management time and superintendence time is directly attributable to and required by the change; provided however that additional cost of on-site superintendence shall be allowable whenever in the opinion of the Architect/Engineer the impact of multiple change requests to be concurrently performed will result in inadequate levels of supervision to assure a proper result unless additional superintendence is provided.
- d) Construction equipment (including small tools). Expenses for equipment and fuel shall be based on customary commercially reasonable rental rates and schedules. Equipment and hand tool costs shall not include the cost of items customarily owned by workers.
- e) Workers' compensation costs, if not included in labor burden.
- f) The cost of commercial general liability and property damage insurance premiums but only to the extent charged the Construction Manager as a result of the changed Work.
- g) Overhead and profit, as hereafter specified.
- h) Builder's risk insurance premium costs.
- i) Bond premium costs.
- j) Testing costs not otherwise excluded by these General Conditions.
- k) Subcontract costs.

Adjustments to the construction services fee, as defined in Exhibit A of the Agreement shall be per Article 5.2 of the Agreement.

Unless modified in the Supplementary General Conditions, overhead and profit shall not exceed the percentages set forth in the table below.

| | | | OVERHEAD | PROFIT | COMMISSION | | | |
|--|---|-----------------------------------|------------|--------|------------|----|------|--|
| for the p | construction M portion of Work ir own forces: | anager or to Subco k performed | ontractors | 10% | 5% | 0% | | |
| To the Construction Manager or to Subcontractors | | | 5% | 0% | 5% | | | |
| for | Work | performed | by | others | at | а | tier | |
| immediately below either of them: | | | | | | | | |

Overhead shall include: a) insurance premium for policies not purchased for the Project and itemized above, b) home office costs for office management, administrative and supervisory personnel and assistants, c) estimating and change order preparation costs, d) incidental job burdens, e) legal costs, f) data processing costs, g) interest costs on capital, h) general office

expenses except those attributable to increased rental expenses for temporary facilities, and all other indirect costs, but shall not include the Social Security tax and other direct labor burdens. The term "Work" as used in the proceeding table shall include labor, materials and equipment and the "Commission" shall include all costs and profit for carrying the subcontracted Work at the tiers below except direct costs as listed in items a. through k. above if any.

On proposals for Work involving both additions and credits in the amount of the Contract sum, the overhead and profit will be allowed on the net increase only. On proposals resulting in a net deduct to the amount of the Contract sum, profit on the deducted amount shall be returned to the Principal Representative at fifty percent (50%) of the rate specified. The inadequacy of the profit specified shall not be a basis for refusal to submit a proposal.

Except in the case of Change Orders or Emergency Field Change Orders agreed to on the basis of a lump sum amount or unit prices as described in paragraphs 35.1.1, The Value of Changed Work, the Construction Manager shall keep and present a correct and fully auditable account of the several items of cost, together with vouchers, receipts, time cards and other proof of costs incurred, summarized on a Change Order form (SC-6.31) using such format for supporting documentation as the Principal Representative and State Buildings Program approve. This requirement applies equally to Work done by Subcontractors. Only auditable costs shall be reimbursable on Change Orders where the value is determined on the basis of actual cost plus a fixed fee pursuant to paragraph 35A3 above, or where unilaterally determined by the Principal Representative on the basis of an equitable adjustment in accordance with the Procurement Rules, as described above in Article 35, The Value Of Changed Work.

Except for proposals for Work involving both additions and credits, changed Work shall be adjusted and considered separately for Work either added or omitted. The amount of adjustment for Work omitted shall be estimated at the time it is directed to be omitted, and when reasonable to do so, the agreed adjustment shall be reflected on the schedule of values used for the next Construction Manager's application for payment.

The Principal Representative reserves the right to contract with any person or firm other than the Construction Manager for any or all extra Work; however, unless specifically required in the Contract Documents, the Construction Manager shall have no responsibility without additional compensation to supervise or coordinate the Work of persons or firms separately contracted by the Principal Representative.

35.3 HAZARDOUS MATERIALS

The Principal Representative represents that it has undertaken an examination of the site of the Work and has determined that there are no hazardous substances, as defined below, which the Construction Manager could reasonably encounter in its performance of the Work. In the event the Principal Representative so discovers hazardous substances, the Principal Representative shall render harmless such hazards before the Construction Manager commences the Work.

In the event the Construction Manager encounters any materials reasonably believed to be hazardous substances which have not been rendered harmless, the Construction Manager shall immediately stop Work in the area affected and report the condition to the Principal Representative, in writing. For purposes of this Agreement, "hazardous substances" shall include asbestos, lead, polychlorinated biphenyl (PCB) and any or all of those substances defined as "hazardous substance", "hazardous waste", or "dangerous or extremely hazardous wastes" as those terms are used in the Comprehensive Environmental Response, Compensation and Liability Act (CERCLA) and the Resource Conservation and Recovery Act (RCRA), and shall also include

materials regulated by the Toxic Substances Control Act (TSCA), the Clean Air Act, the Air Quality Act, the Clean Water Act, and the Occupational Safety and Health Act. The Work in the affected area shall not therefore be resumed except by written agreement of the Principal Representative and the Construction Manager, if in fact materials that are hazardous substances have not been rendered harmless. The Work in the affected area shall be resumed only in the absence of the hazardous substances or when it has been rendered harmless or by written agreement of the Principal Representative and the Construction Manager.

The Construction Manager shall not be required to perform Work without consent in any areas where it reasonably believes hazardous substances that have not been rendered harmless are present.

35.4 EMERGENCY FIELD CHANGE ORDERED WORK

The Principal Representative, without invalidating the Agreement, and with the approval of State Buildings Program and without the approval of the State Controller, may order extra Work or make changes in the case of an emergency that is a threat to life or property or where the likelihood of delays in processing a normal Change Order will result in substantial delays and or significant cost increases for the Project. Emergency Field Orders are not to be used solely to expedite normal Change Order processing absent a clear showing of a high potential for significant and substantial cost or delay. Such changes in the Work may be directed through issuance of an Emergency Field Change Order signed by the Construction Manager, the Principal Representative (or by a designee specifically appointed to do so in writing), and approved by the Office of the State Architect or their delegate. The change shall be directed using an Emergency Field Change Order form (SC-6.31E).

If the amount of the adjustment of the Contract price and time for completion can be determined at the time of issuance of the Emergency Field Change Order, those adjustments shall be reflected on the face of the Emergency Field Change Order. Otherwise, the Emergency Field Change Order shall reflect a not to exceed (NTE) amount for any schedule adjustment (increasing or decreasing the time for completion) and an NTE amount for any adjustment to Contract sum, which NTE amount shall represent the maximum amount of adjustment to which the Construction Manager will be entitled, including direct and indirect costs of changed Work, as well as any direct or indirect costs attributable to delays, inefficiencies or other impacts arising out of the change. Emergency Field Change Orders directed in accordance with this provision need not bear the approval signatures of the State Controller.

On Emergency Field Change Orders where the price and schedule have not been finally determined, the Construction Manager shall submit final costs for adjustment as soon as practicable. No later than seven (7) days after issuance, except as otherwise permitted, and every seven days thereafter, the Construction Manager shall report all costs to the Principal Representative and the Architect/Engineer. The final adjustment of the Emergency Field Change Order amount and the adjustment to the Project time for completion shall be prepared on a normal Change Order from (SC-6.31) in accordance with the procedures described in Article 35, The Value of Changed Work, and B, Detailed Breakdown, above. Unless otherwise provided in writing signed by the Director of State Buildings Program to the Principal Representative and the Construction Manager, describing the extent and limits of any greater authority, individual Emergency Field Change Orders shall not be issued for more than \$25,000, nor shall the cumulative value of Emergency Field Change Orders exceed an amount of \$100,000.

35.5 APPROPRIATION LIMITATIONS - C.R.S. § 24-91-103.6, as amended

The amount of money appropriated, as shown on the Construction Manager/General Contractor (CM/GC) Agreement (SC-6.5), is equal to or in excess of the Contract amount. No Change Order, Emergency Field Change Order, or other type of order or directive shall be issued by the Principal Representative, or any agent acting on his or her behalf, which directs additional compensable Work to be performed, which Work causes the aggregate amount payable under the Contract to exceed the amount appropriated for the original Contract, as shown on the Agreement (SC-6.5), unless one of the following occurs: (1) the Construction Manager is provided written assurance from the Principal Representative that sufficient additional lawful appropriations exist to cover the cost of the additional Work; or (2) the Work is covered by a Construction Manager remedy provision under the Contract, such as a claim for extra cost. By way of example only, no assurance is required for any order, directive or instruction by the Architect/Engineer or the Principal Representative to perform Work which is determined to be within the performance required by the Contract Documents; the Construction Manager's remedy shall be as described elsewhere in these General Conditions.

Written assurance shall be in the form of an Amendment to the Contract reciting the source and amount of such appropriation available for the Project. No remedy granting provision of this Contract shall obligate the Principal Representative to seek appropriations to cover costs in excess of the amounts recited as available to pay for the Work to be performed.

36 ARTICLE 36 CLAIMS

It is the intent of these General Conditions to provide procedures for speedy and timely resolution of disagreements and disputes at the lowest level possible. In the spirit of on the job resolution of job site issues, the parties are encouraged to use the partnering processes of Article 2.4, Partnering, Communications and Cooperation, before turning to the more formal claims processes described in this Article 36, Claims. The use of non-binding dispute resolution, whether through the formal processes described in Article 39, Non-Binding Dispute Resolution – Facilitated Negotiations, or through less formal alternative processes developed as part of a partnering plan, are also encouraged. Where such process cannot resolve the issues in dispute, the claims process that follows is intended to cause the issues to be presented, decided and where necessary, documented in close proximity to the events from which the issues arise. To that end, and in summary of the remedy granting process that follows commencing with the next paragraph of this Article 36, Claims, the Construction Manager shall 1) first, seek a decision by the Architect/Engineer, and 2) shall second, informally present the claim to Principal Representative as described hereafter, and 3) failing resolution in the field, give Notice of intent to exercise statutory rights of review of a formal contract controversy, and 4) seek resolution outside the Contract as provided by the Colorado Procurement Code or the applicable procurement code for institutions of higher education.

If the Construction Manager claims that any instructions, by detailed drawings, or otherwise, or any other act or omission of the Architect/Engineer or Principal Representative affecting the scope of the Construction Manager's Work, involve extra cost, extra time or changes in the scope of the Work under this Contract, the Construction Manager shall have the right to assert a claim for such costs or time, provided that before either proceeding to execute such Work (except in an emergency endangering life or property), or filing a Notice of claim, the Construction Manager shall have obtained or requested a written decision of the Architect/Engineer following the procedures as provided in Article 6, Architect/Engineer Decisions and Judgments, respectively; provided, however, that in the case of a directed Change in the Work pursuant to Article 35, no

written judgment or decision of the Architect/Engineer is required. If the Construction Manager is delayed by the lack of a response to a request for a decision by the Architect/Engineer, the Construction Manager shall give Notice in accordance with Article 38, Delays and Extensions of Time.

Unless it is the Architect/Engineer's judgment and determination that the Work is not included in the performance required by the Contract Documents, the Construction Manager shall proceed with the Work as originally directed. Where the Construction Manager's claim involves a dispute concerning the Value of Work unilaterally directed pursuant to Article 35.1.2 the Construction Manager shall also proceed with the Work as originally directed while his or her claim is being considered.

The Construction Manager shall give the Principal Representative and the Architect/Engineer Notice of any claim promptly after the receipt of the Architect/Engineer's decision, but in no case later than three (3) business days after receipt of the Architect/Engineer's decision (or no later than ten (10) days from the date of the Construction Manager's request for a decision when the Architect/Engineer fails to decide as provided in Article 6). The Notice of claim shall state the grounds for the claim and the amount of the claim to the extent known in accordance with the procedures of Article 35, Changes in the Work. The period in which Notice must be given may be extended by the Principal Representative if requested in writing by the Construction Manager with good cause shown, but any such extension to be effective shall be in writing.

The Principal Representative shall respond in writing, with a copy to the Architect/Engineer, within a reasonable time, and except where a request for facilitation of negotiation has been made as hereafter provided, in no case later than fourteen (14) business days (or at such other time as the Construction Manager and Principal Representative agree) after receipt of the Construction Manager's Notice of claim regarding such instructions or alleged act or omission. If no response to the Construction Manager's claim is received within fourteen (14) business days of Construction Manager's Notice (or at such other time as the Construction Manager and Principal Representative agree) and the instructions have not been retracted, it shall be deemed that the Principal Representative has denied the claim.

The Principal Representative may grant or deny the claim in whole or in part, and a Change Order shall be issued if the claim is granted. To the extent any portion of claim is granted where costs are not clearly shown, the Principal Representative may direct that the value of that portion of the Work be determined by any method allowed in Article 35.1, The Value of Changed Work. Except in the case of a deemed denial, the Principal Representative shall provide a written explanation regarding any portion of the Construction Manager's claim that is denied.

If the Construction Manager disagrees with the Principal Representative's judgment and determination on the claim and seeks an equitable adjustment of the Contract sum or time for performance, he or she shall give Notice of intent to exercise his or her statutory right to seek a decision on the contract controversy within ten (10) days of receipt of the Principal Representative's decision denying the claim. A "contract controversy," as such term is used in the Colorado Procurement Code or the applicable procurement code for institutions of higher education, shall not arise until the initial claim process described above in this Article 36 has been properly exhausted by the Construction Manager. The Construction Manager's failure to proceed with Work directed by the Architect/Engineer or to exhaust the claim process provided above in this Article 36, shall constitute an abandonment of the claim by the Construction Manager and a waiver of the right to contest the decision in any forum.

SC-6.51

Rev. 08/2022 41

At the time of filing the Notice of intent to exercise his or her statutory right to seek a decision on the contract controversy, the Construction Manager may request that the Principal Representative defer a decision on the contract controversy until a later date or until the end of the Project. If the Principal Representative agrees, he or she shall so advise the Construction Manager in writing. If no such request is made, or if the Principal Representative does not agree to such a request, the Principal Representative shall render a written decision within twenty (20) business days and advise the Construction Manager of the reasons for any denial. Unless the claim has been decided by the Principal Representative (as opposed to delegates of the Principal Representative), the person who renders the decision on this statutory contract controversy shall not be the same person who decided the claim. To the extent any portion of the contract controversy is granted where costs are not clearly shown, the Principal Representative may direct that the value of that portion of the Work be determined by any method allowed in Article 35.1, The Value of Changed Work. In the event of a denial the Principal Representative shall give Notice to the Construction Manager of his or her right to administrative and judicial reviews as provided in the Colorado Procurement Code or the applicable procurement code for institutions of higher education. If no decision regarding the contract controversy is issued within twenty (20) business days of the Construction Manager's giving Notice (or such other date as the Construction Manager and Principal Representative have agreed), and the instructions have not been retracted or the alleged act or omission have not been corrected, it shall be deemed that the Principal Representative has ruled by denial on the contract controversy. Except in the case of a deemed denial, the Principal Representative shall provide an explanation regarding any portion of the contract controversy that involves denial of the Construction Manager's claim.

Either the Construction Manager or the Principal Representative may request facilitation of negotiations concerning the claim or the contract controversy, and if requested, the parties shall consult and negotiate before the Principal Representative decides the issue. Any request for facilitation by the Construction Manager shall be made at the time of the giving of Notice of the claim or Notice of the contract controversy. Facilitation shall extend the time for the Principal Representative to respond by commencing the applicable period at the completion of the facilitated negotiation, which shall be the last day of the parties' meeting, unless otherwise agreed in writing.

Disagreement with the decision of the Architect Engineer, or the decision of the Principal Representative to deny any claim or denying the contract controversy, shall not be grounds for the Construction Manager to refuse to perform the Work directed or to suspend or terminate performance. During the period that any claim or contract controversy decision is pending under this Article 36, Claims, the Construction Manager shall proceed diligently with the Work directed.

In all cases where the Construction Manager proceeds with the Work and seeks equitable adjustment by filing a claim and or statutory appeal, the Construction Manager shall keep a correct account of the extra cost, in accordance with Article 35.2, Detailed Breakdown supported by receipts. The Principal Representative shall be entitled to reject any claim or contract controversy whenever the foregoing procedures are not followed and such accounts and receipts are not presented.

The payments to the Construction Manager in respect of such extra costs shall be limited to reimbursement for the current additional expenditure by the Construction Manager made necessary by the change in the Work, plus a reasonable amount for overhead and profit,

determined in accordance with Article 35.2, Detailed Breakdown, determined solely with reference to the additional Work, if any, required by the change.

37 ARTICLE 37 DIFFERING SITE CONDITIONS

37.1 NOTICE IN WRITING

The Construction Manager shall promptly, and where possible before conditions are disturbed, give the Architect/Engineer and the Principal Representative Notice in writing of:

- 37.1.1 Subsurface or latent physical conditions at the site differing materially from those indicated in or reasonably assumed from the information provided in the Contract Documents; and, (NUMBERING)
- 37.1.2 Unknown physical conditions at the site, of an unusual nature, differing materially from those ordinarily encountered and generally recognized as inherent in Work of the character provided for in the Contract Documents.

The Architect/Engineer shall promptly investigate the conditions, and if it is found that such conditions do materially so differ and cause an increase or decrease in the Construction Manager's costs of performance of any part of the Work required by the Contract Documents, whether or not such Work is changed as a result of such conditions, an equitable adjustment shall be made and the Contract sum shall be modified in accordance with Article 35, Changes In The Work.

If the time required for completion of the Work affected by such materially differing conditions will extend the Work on the critical path as indicated on the CPM schedule, the time for completion shall also be equitably adjusted.

37.2 LIMITATIONS

No claim of the Construction Manager under this clause shall be allowed unless the Construction Manager has given the Notice required in Article 37.1, Notice in Writing, above. The time prescribed for presentation and adjustment in Articles 36, Claims and 38, Delays And Extensions Of Time, shall be reasonably extended by the State to the extent required by the nature of the differing conditions; provided, however, that even when so extended no claim by the Construction Manager for an equitable adjustment hereunder shall be allowed if not quantified and presented prior to the date the Construction Manager requests a final inspection pursuant to Article 41.1, Notice Of Completion.

38 ARTICLE 38 DELAYS AND EXTENSIONS OF TIME

If the Construction Manager is delayed at any time in the progress of the Work by any act or neglect of the State of Colorado or the Architect/Engineer, or of any employee or agent of either, or by any separately employed Construction Manager or by strikes, lockouts, fire, unusual delay in transportation, unavoidable casualties or any other causes beyond the Construction Manager's control, including weather delays as defined below, the time of Completion of the Work shall be extended for a period equal to such portion of the period of delays directly affecting the completion of the Work as the Construction Manager shall be able to show he or she could not have avoided by the exercise of due diligence.

The Construction Manager shall provide Notice in writing to the Architect/Engineer, the Principal Representative and State Buildings Program within three (3) business days from the beginning of

Rev. 08/2022 43

such delay and shall file a written claim for an extension of time within seven (7) business days after the period of such delay has ceased, otherwise, any claim for an extension of time is waived.

Provided that the Construction Manager has submitted reasonable schedules for approval when required by Article 12, Requests for Information and Schedules, if no schedule is agreed to fixing the dates on which the responses to requests for information or detail drawings will be needed, or Shop Drawings, Product Data or Samples are to be reviewed as required or allowed by Article 12.2, Schedules, no extension of time will be allowed for the Architect/ Engineer's failure to furnish such detail drawings as needed, or for the failure to initially review Shop Drawings, Product Data or Samples, except in respect of that part of any delay in furnishing detail drawings or instructions extending beyond a reasonable period after written demand for such detailed drawings or instructions is received by the Architect/Engineer. In any event, any claim for an extension of time for such cause will be recognized only to the extent of delay directly caused by failure to furnish detail drawings or instructions or to review Shop Drawings, Product Data or Samples pursuant to schedule, after such demand.

All claims for extension of time due to a delay claimed to arise or result from ordered changes in the scope of the Work, or due to instructions claimed to increase the scope of the Work, shall be presented to the Architect/Engineer, the Principal Representative and State Buildings Program as part of a claim for extra cost, if any, in accordance with Article 36, Claims, and in accordance with the Change Order procedures required by Article 35, Changes In The Work.

Except as otherwise provided in this paragraph, no extension of time shall be granted when the Construction Manager has failed to utilize a CPM schedule or otherwise identify the Project's critical path as specified in Article 12, Requests for Information and Schedules, or has elected not to do so when allowed by the Supplementary General Conditions or the Specifications to use less sophisticated scheduling tools, or has failed to maintain such a schedule. Delay directly affecting the completion of the Work shall result in an extension of time only to the extent that completion of the Work was affected by impacts to the critical path shown on Construction Manager's CPM schedule. Where the circumstances make it indisputable in the opinion of the Architect/Engineer that the delay affected the completion of the Work so directly that the additional notice of the schedule impact by reference to a CPM schedule was unnecessary, a reasonable extension of time may be granted.

Extension of the time for completion of the Work will be granted for delays due to weather conditions only when the Construction Manager demonstrates that such conditions were more severe and extended than those reflected by the ten-year average for the month, as evidenced by the Climatological Data, U. S. Department of Commerce, for the Project area.

Extensions of the time for completion of the Work due to weather will be granted on the basis of one and three tenths (1.3) calendar days for every day that the Construction Manager would have Worked but was unable to Work, with each separate extension figured to the nearest whole calendar day.

For weather delays and delays caused by events, acts or omissions not within the control of the Principal Representative or any person acting on the Principal Representative's behalf, the Construction Manager shall be entitled to an extension of time only and shall not be entitled to recovery of additional cost due to or resulting from such delays. This Article does not, however, preclude the recovery of damages for delay by either party under other provisions in the Contract Documents.

39 ARTICLE 39 NON-BINDING DISPUTE RESOLUTION – FACILITATED NEGOTIATIONS

The Construction Manager and Principal Representative agree to designate one or more mutually acceptable persons willing and able to facilitate negotiations and communications for the resolution of conflicts, disagreements or disputes between them at the specific request of either party with regard to any Project decision of either of them or any decision of the Architect/Engineer. The designation of such person(s) shall not carry any obligation to use their services except that each party agrees that if the other party requests the intervention of such person(s) with respect to any such conflict, dispute or disagreement, the non-requesting party shall participate in good faith attempts to negotiate a resolution of the issue in dispute. If the parties cannot agree on a mutually acceptable person to serve in this capacity one shall be so appointed; provided, however, that either party may request the State Architect or their delegate to appoint such a person, who, if appointed, shall be accepted for this purpose by both the Construction Manager and the Principal Representative.

The cost, if any, of the facilitative services of the person(s) so designated shall be shared if the parties so agree in any partnering plan; or in the absence of agreement the cost shall be borne by the party requesting the facilitation of negotiation.

Any dispute, claim, question or disagreement arising from or relating to the Contract or an alleged breach of the Contract may be subject to a request by either party for facilitated negotiation subject to the limitations hereafter listed, and the parties shall participate by consultation and negotiation with each other, as guided by the facilitator and with recognition of their mutual interests, in an attempt to reach an equitable solution satisfactory to both parties.

The obligation to participate in facilitated negotiations shall be as described above and elsewhere in these General Conditions, as by way of example in Article 36, Claims, or Article 34, Deductions for Uncorrected Work and to the extent not more particularly described or limited elsewhere, each party's obligations shall be as follows:

- a) A party shall not initiate communication with the facilitator regarding the issues in dispute;
 except that any request for facilitation shall be made in writing with copies sent, faxed or delivered to the other party;
- A party shall prepare a brief written description of its position if so requested by the facilitator (who may elect to first discuss the parties' positions with each party separately in the interest of time and expense);
- A party shall respond to any reasonable request for copies of documents requested by the facilitator, but such requests, if voluminous, may consist of an offer to allow the facilitator access to the parties' documents;
- d) A party shall review any meeting agenda proposed by a facilitator and endeavor to be informed on the subjects to be discussed;
- e) A party shall meet with the other party and the facilitator at a mutually acceptable place and time, or, if none can be agreed to, at the time and place designated by the facilitator for a period not to exceed four hours unless the parties agree to a longer period;
- f) A party shall endeavor to assure that any facilitation meeting shall be attended by any other persons in their employ that the facilitator requests be present, if reasonably available, including the Architect/Engineer;
- g) Each party shall participate in such facilitated face-to-face negotiations of the issues in dispute through persons fully authorized to resolve the issue in dispute;

- h) Each party shall be obligated to participate in negotiations requested by the other party and to perform the specific obligations described in paragraphs (a) through (j) this Article 39, Facilitated Negotiation, no more than three times during the course of the Project;
- i) Neither party shall be under any obligation to resolve any issue by facilitated negotiation, but each agrees to participate in good faith and the Principal Representative shall direct the Architect/Engineer to appropriately document any resolution or agreement reached and to execute any Amendment or Change Order to the Contract necessary to implement their agreement; and,
- j) Any discussions and documents prepared exclusively for use in the negotiations shall be deemed to be matters pertaining to settlement negotiations and shall not be subsequently available in further proceedings except to the extent of any documented agreement.

In accordance with State Fiscal Rules and Article 52.7, Prohibited Terms, nothing in this Article 39 shall be deemed to call for arbitration or otherwise obligate the State to participate in any form of binding alternative dispute resolution.

A partnering plan developed as described in Article 2.4, Partnering, Communications and Cooperation, may modify or expand the requirements of this Article but may not reduce the obligation to participate in facilitated negotiations when applicable. In the case of small projects estimated to be valued under \$500,000, the requirements of this Article may be deleted from this Contract by modification in Article 7.5, Construction Manager/General Contractor (CM/GC) Agreement (SC-6.5), Optional Provisions and Elections. When so modified, the references to the parties' right to elect facilitated negotiation elsewhere in these General Conditions shall be deleted.

40 ARTICLE 40 RIGHT OF OCCUPANCY

The Principal Representative shall have the right to take possession of and to use any completed or partially completed portions of the Work, even if the time for completing the entire Work or portions of the Work has not expired and even if the Work has not been finally accepted, and the Construction Manager shall fully cooperate with the Principal Representative to allow such possession and use. Such possession and use shall not constitute an acceptance of such portions of the Work.

Prior to any occupancy of the Project, an inspection shall be made by the Principal Representative, the Architect/Engineer, State Buildings Program and the Construction Manager. Such inspection shall be made for the purpose of ensuring that the building is secure, protected by operation safety systems as designed, operable exits, power, lighting and HVAC systems, and otherwise ready for the occupancy intended and the Notice of Substantial Completion has been issued for the occupancy intended. The inspection shall also document existing finish conditions to allow assessment of any damage by occupants. The Construction Manager shall assist the Principal Representative in completing and executing State Form SBP-01, Approval of Occupancy/Use, prior to the Principal Representative's possession and use. Any and all areas so occupied will be subject to a final inspection when the Construction Manager complies with Article 41, Completion, Final Inspection, Acceptance and Settlement.

41 ARTICLE 41 COMPLETION, FINAL INSPECTION, ACCEPTANCE AND SETTLEMENT

41.1 NOTICE OF COMPLETION

When the Work, or a discrete physical portion of the Work (as hereafter described) which the Principal Representative has agreed to accept separately, is substantially complete and ready for final inspection, the Construction Manager shall file a written Notice with the Architect/Engineer that the Work, or such discrete physical portion, in the opinion of the Construction Manager, is substantially complete under the terms of the Contract. The Construction Manager shall prepare and submit with such Notice a comprehensive list of items to be completed or corrected prior to final payment, which shall be subject to review and additions as the Architect/Engineer or the Principal Representative shall determine after inspection. If the Architect/Engineer or the Principal Representative believe that any of the items on the list of items submitted, or any other item of Work to be corrected or completed, or the cumulative number of items of Work to be corrected or completed, will prevent a determination that the Work is substantially complete, those items shall be completed by the Construction Manager and the Notice shall then be resubmitted.

41.2 FINAL INSPECTION

Within ten (10) days after the Construction Manager files written Notice that the Work is substantially complete, the Architect/Engineer, the Principal Representative, and the Construction Manager shall make a "final inspection" of the Project to determine whether the Work is substantially complete and has been completed in accordance with the Contract Documents. State Buildings Program shall be notified of the inspection not less than three (3) business days in advance of the inspection. The Construction Manager shall provide the Principal Representative and the Architect/Engineer an updated punch list in sufficient detail to fully outline the following:

- a) Work to be completed, if any; and
- b) Work not in compliance with the Drawings or Specifications, if any.

A final punch list shall be made by the Architect/Engineer in sufficient detail to fully outline to the Construction Manager:

- a) Work to be completed, if any;
- b) Work not in compliance with the Drawings or Specifications, if any; and
- c) Unsatisfactory Work for any reason, if any.

The required number of copies of the final punch list will be countersigned by the authorized representative of the Principal Representative and will then be transmitted by the Architect/Engineer to the Construction Manager, the Principal Representative, and State Buildings Program. The Architect/Engineer's final punch list shall control over the Construction Manager's preliminary punch list.

41.3 NOTICE OF SUBSTANTIAL COMPLETION

Notice of Substantial Completion shall establish the date of substantial completion of the Project. The Construction Manager acknowledges and agrees that because the departments, agencies and institutions of the State of Colorado are generally involved with the business of the public at large, greater care must be taken in establishing the date of substantial completion than might otherwise be the case to confirm that a project or building or discrete physical portion of the Work is fully usable and safe for public use, and that such care necessarily raises the standard by which the concept of substantial completion is applied for a public building.

SC-6.51

- The Notice of Substantial Completion shall not be issued until the following have been fully established:
- 41.3.1 All required building code inspections have been called for and the appropriate code officials have affixed their signatures to the Building Inspection Record indicating successful completion of all required code inspections;
- 41.3.2 All required corrections noted on the Building Inspection Record shall have been completed unless the Architect/Engineer, the Principal Representative and State Buildings Program, in their complete and absolute discretion, all concur that the condition requiring the remaining correction is not in any way life threatening, does not otherwise endanger persons or property, and does not result in any undue inconvenience or hardship to the Principal Representative or the public;
- 41.3.3 The building, structure or Project can be fully and comfortably used by the Principal Representative and the public without undue interference by the Construction Manager's employees and Workers during the completion of the final punch list taking into consideration the nature of the public uses intended and taking into consideration any stage or level of completion of HVAC system commissioning or other system testing required by the Specifications to be completed prior to issuance of the Notice of Substantial Completion;
- 41.3.4 The Project has been fully cleaned as required by these General Conditions, and as required by any stricter requirements of the Specifications, and the overall state of completion is appropriate for presentation to the public; and
- 41.3.5 The Construction Manager has provided a schedule for the completion of each and every item identified on the punch list which specifies the Subcontractor or trade responsible for the Work, and the dates the completion or correction of the item will be commenced and finished; such schedule will show completion of all remaining final punch list items within the period indicated in the Contract for final punch list completion prior to Final Acceptance, with the exception of only those items which are beyond the control of the Construction Manager despite due diligence. The schedule shall provide for a reasonable punch list inspection process. Unless liquidated damages have been specified in Article 7.6 of the Construction Manager's Design/Bid/Build Agreement SC-6.21), the cost to the Principal Representative, if any, for re-inspections due to failure to adhere to the Construction Manager's proposed punch-list completion schedule shall be the responsibility of the Construction Manager and may be deducted by the Principal Representative from final amounts due to the Construction Manager.

Substantial completion of the entire Project shall not be conclusively established by a decision by the Principal Representative to take possession and use of a portion, or all of the Project, where portions of the Project cannot meet all the criteria noted above. Notice of Substantial Completion for the entire Project shall, however, only be withheld for substantial reasons when the Principal Representative has taken possession and uses all of the Project in accordance with the terms of Article 40, Right of Occupancy. Failure to furnish the required completion schedule shall constitute a substantial reason for withholding the issuance of any Notice of Substantial Completion.

The Construction Manager shall have the right to request a final inspection of any discrete physical portion of the Project when in the opinion of the Principal Representative, The Architect/Engineer and State Buildings Program a final punch list can be reasonably prepared, without confusion as to which portions of the Project are referred to in any subsequent Notice of Partial Final Settlement which might be issued after such portion is finally accepted. Discrete physical portions of the Project may be, but shall not necessarily be limited to, such portions of the Project as separate buildings where a Project consists of multiple buildings. Similarly, an addition to an

existing building where the Project also calls for renovation or remodeling of the existing building may constitute a discrete physical portion of the Project. In such circumstances, when in the opinion of the Principal Representative, the Architect/Engineer and State Buildings Program, the requirements for issuance of a Notice of Substantial Completion can be satisfied with respect to the discrete portion of the Project, a partial Notice of Substantial Completion may be issued for such discrete physical portion of the Project.

41.4 NOTICE OF ACCEPTANCE

The Notice of Acceptance shall establish the completion date of the Project. It shall not be authorized until the Construction Manager shall have performed all of the Work to allow completion and approval of the Pre-Acceptance Checklist (SBP-05).

Where partial Notices of Substantial Completion have been issued, partial Notices of Final Acceptance may be similarly issued when appropriate for that portion of the Work. Partial Notice of Final Acceptance may also be issued to exclude the Work described in Change Orders executed during late stages of the Project where a later completion date for the Change Ordered Work is expressly provided for in the Contract as amended by the Change Order, provided the Work can be adequately described to allow partial advertisement of any Notice of Partial Final Settlement to be issued without confusion as to the Work included for which final payment will be made.

41.5 SETTLEMENT

Final payment and settlement shall be made on the date fixed and published for such payment except as hereafter provided. The Principal Representative shall not authorize final payment until all items on the Pre-Acceptance check list (SBP-05) have been completed, the Notice of Acceptance issued, and the Notice of Contractors Settlement published. If the Work shall be substantially completed, but Final Acceptance and completion thereof shall be prevented through delay in correction of minor defects, or unavailability of materials or other causes beyond the control of the Construction Manager, the Principal Representative in his or her discretion may release all amounts due to the Construction Manager except such amounts as may be in excess of three times the cost of completing the unfinished Work or the cost of correcting the defective Work, as estimated by the Architect/Engineer and approved by State Buildings Program. Before the Principal Representative may issue the Notice of Contractor's Settlement and advertise the Project for final payment, the Construction Manager shall have corrected all items on the punch list except those items for which delayed performance is expressly permitted, subject to withholding for the cost thereof, and shall have:

41.5.1 Delivered to the Principal Representative:

- a) All guarantees and warranties;
- b) All statements to support local sales tax refunds, if any;
- c) Required operating maintenance instructions as per the Principal Representative; and,
- d) One (1) set of hard copy as-built Contract Documents, and one (1) electronic copy showing all job changes.

- 41.5.2 Demonstrated to the operating personnel of the Principal Representative the proper operation and maintenance of all equipment.
- 41.5.3 Delivered to the State of Colorado Department of Personnel & Administration in accordance with the Colorado Procurement Code or the applicable procurement code for institutions of higher education:

A written disclosure of the five most costly goods incorporated into the project, including iron, steel, or related manufactured goods and the total cost and country of origin of those five goods and whether the project was subject to any existing domestic content preferences.

Upon completion of the foregoing the Project shall be advertised in accordance with the Notice of Contractor's Settlement by publications of Notice, the publication appearing at least ten (10) days prior to the time of final settlement. Publication and final settlement should not be postponed or delayed solely by virtue of unresolved claims against the Project or the from Subcontractors, suppliers or materialmen based on good faith disputes; the resolution of the question of payment in such cases being directed by statute.

Except as hereafter provided, on the date of final settlement thus advertised, provided the Construction Manager has submitted a written Notice to the Architect/Engineer that no claims have been filed, and further provided the Principal Representative shall have received no claims, final payments and settlement shall be made in full. If any unpaid claim for labor, materials, rental machinery, tools, supplies or equipment is filed before payment in full of all sums due the Construction Manager, the Principal Representative and the State Controller shall withhold from the Construction Manager on the date established for final settlement, sufficient funds to insure the payment of such claim, until the same shall have been paid or withdrawn, such payment or withdrawal to be evidenced by filing a receipt in full or an order for withdrawal signed by the claimant or his or her duly authorized agent or assignee. The amount so withheld may be in the amount of 125% of the claims or such other amount as the Principal Representative reasonably deems necessary to cover expected legal expenses. Such withheld amounts shall be in addition to any amount withheld based on the cost to compete unfinished Work or the cost to repair defective Work. However, as provided by statute, such funds shall not be withheld longer than ninety (90) days following the date fixed for final settlement with the Construction Manager, as set forth in the published Notice of Contractor's Settlement, unless an action at law shall be commenced within that time to enforce such unpaid claim and a Notice of such action at law shall have been filed with the Principal Representative and the State Controller. At the expiration of the ninety (90) day period, the Principal Representative shall authorize the State Controller to release to the Construction Manager all other money not the subject of such action at law or withheld based on the cost to compete unfinished Work or the cost to repair defective Work.

Notices of Partial Final Settlement may be similarly advertised, provided all conditions precedent have been satisfied as though that portion of the Work affected stood alone, a Notice of Partial Acceptance has been issued, and the consent of surety to the partial final settlement has been obtained in writing. Thereafter, partial final payments may be made to the Construction Manager subject to the same conditions regarding unpaid claims.

42 ARTICLE 42 GENERAL WARRANTY AND CORRECTION OF WORK AFTER ACCEPTANCE

The Construction Manager warrants that the materials used and the equipment furnished shall be new and of good quality unless specified to the contrary. The Construction Manager further warrants that the Work shall, in all respects, be free from material defects not permitted by the Specifications and shall be in accordance with the requirements of the Contract Documents. Neither the final certificate for payment nor any provision in the Contract Documents shall relieve the Construction Manager of responsibility for defects or faulty materials or Workmanship. The Construction Manager shall be responsible to the Principal Representative for such warranties for the longest period permitted by any applicable statute of limitations.

In addition to these general warranties, and without limitation of these general warranties, for a period of one year after the date of any Notice of Substantial Completion, or any Notice of Partial Substantial Completion if applicable, the Construction Manager shall remedy defects, and faulty Workmanship or materials, and Work not in accordance with the Contract Documents which was not accepted at the time of the Notice of Final Acceptance, all in accordance with the provisions of Article 44, One-Year Guarantee And Special Guarantees And Warranties.

43 ARTICLE 43 LIENS

Colorado statutes do not provide for any right of lien against public buildings. In lieu thereof, C.R.S. § 38-26-107, provides adequate relief for any claimant having furnished labor, materials, rental machinery, tools, equipment, or services toward construction of the particular public Work in that final payment may not be made to a Construction Manager until all such creditors have been put on Notice by publication in the public press of such pending payment and given opportunity for a period of up to ninety (90) days to stop payment to the Construction Manager in the amount of such claims.

44 ARTICLE 44 ONE-YEAR GUARANTEE AND SPECIAL GUARANTEES AND WARRANTIES

44.1 ONE-YEAR GUARANTEE OF THE WORK

The Construction Manager shall guarantee to remedy defects and repair or replace the Work for a period of one year from the date of the Notice of Substantial Completion or from the dates of any partial Notices of Substantial Completion issued for discrete physical portions of the Work. The Construction Manager shall remedy any defects due to faulty materials or Workmanship and shall pay for, repair and replace any damage to other Work resulting there from, which shall appear within a period of one year from the date of such Notice(s) of Substantial Completion. The Construction Manager shall also remedy any deviation from the requirements of the Contract Documents which shall later be discovered within a period of one year from the date of the Notice of Substantial Completion; provided, however, that the Construction Manager shall not be required to remedy deviations from the requirements of the Contract Documents where such deviations were obvious, apparent and accepted by the Architect/Engineer or the Principal Representative at the time of the Notice of Final Acceptance. The Principal Representative shall give Notice of observed defects or other Work requiring correction with reasonable promptness. Such Notice shall be in writing to the Architect/Engineer and the Construction Manager.

The one year guarantee of the Construction Manager's Work may run separately for discrete physical portions of the Work for which partial Notices of Substantial Completion have been issued, however, it shall run from the last Notice of Substantial Completion with respect to all or any systems common to the Work to which more than one Notice of Substantial Completion may apply.

This one-year guarantee shall not be construed to limit the Construction Manager's general warranty described in Article 42, General Warranty and Correction of Work After Acceptance, that

all materials and equipment are new and of good quality, unless specified to the contrary, and that the Work shall in all respects be free from material defects not permitted by the Specifications and in accordance with the requirements of the Contract Documents.

44.2 SPECIAL GUARANTEES AND WARRANTIES

In case of Work performed for which product, manufacturers or other special warranties are required by the Specifications, the Construction Manager shall secure the required warranties and deliver copies thereof to the Principal Representative through the Architect/Engineer upon completion of the Work.

These product, manufacturers or other special warranties, as such, do not in any way lessen the Construction Manager's responsibilities under the Contract. Whenever guarantees or warranties are required by the Specifications for a longer period than one year, such longer period shall govern.

45 ARTICLE 45 GUARANTEE INSPECTIONS AFTER COMPLETION

The Architect/Engineer, the Principal Representative and the Construction Manager together shall make at least two (2) complete inspections of the Work after the Work has been determined to be substantially complete and accepted. One such inspection, the "Six-Month Guarantee Inspection," shall be made approximately six (6) months after date of the Notice of Substantial Completion, unless in the case of smaller projects valued under \$500,000 this inspection is declined in Article 7.2, Construction Manager/General Contractor's (CM/GC) Agreement (SC-6.5), Modification of Article 45, in which case the inspection to occur at six months shall not be required. Another such inspection, the "Eleven-Month Guaranty Inspection" shall be made approximately eleven (11) months after the date of the Notice of Substantial Completion. The Construction Manager shall schedule and so notify all parties concerned, and the Principal Representative shall so notify State Buildings Program, of these inspections. If more than one Notice of Substantial Completion has been issued at the reasonable discretion of the Principal Representative separate eleven month inspections may be required where the one year guarantees do not run reasonably concurrent.

Written punch lists and reports of these inspections shall be made by the Architect/Engineer and forwarded to the Construction Manager, the Principal Representative, State Buildings Program, and all other participants within ten (10) days after the completion of the inspections. The punch list shall itemize all guarantee items, prior punch list items still to be corrected or completed and any other requirements of the Contract Documents to be completed which were not waived by final acceptance because they were not obvious or could not reasonably have been previously observed. The Construction Manager shall immediately initiate such remedial Work as may be necessary to correct any deficiencies or defective Work shown by this report, and shall promptly complete all such remedial Work in a manner satisfactory to the Architect/Engineer, the Principal Representative and State Buildings Program.

If the Construction Manager fails to promptly correct all deficiencies and defects shown by this report, the Principal Representative may do so, after giving the Construction Manager ten (10) days written Notice of intention to do so.

The State of Colorado, acting by and through the Principal Representative, shall be entitled to collect from the Construction Manager all costs and expenses incurred by it in correcting such deficiencies and defects, as well as all damages resulting from such deficiencies and defects.

46 ARTICLE 46 TIME OF COMPLETION AND LIQUIDATED DAMAGES

It is hereby understood and mutually agreed, by and between the parties hereto, that the date of beginning, rate of progress, and the time for completion of the Work to be done hereunder are ESSENTIAL CONDITIONS of this Agreement, and it is understood and agreed that the Work embraced in this Contract shall be commenced at the time specified in the Notice to Proceed (SC-7.26).

It is further agreed that time is of the essence of each and every portion of this Contract, and of any portion of the Work described on the Drawings or Specifications, wherein a definite and certain length of time is fixed for the performance of any act whatsoever. The parties further agree that where under the Contract additional time is allowed for the completion of the Work or any identified portion of the Work, the new time limit or limits fixed by such extension of the time for completion shall be of the essence of this Agreement.

The Construction Manager acknowledges that subject to any limitations in the Advertisement for Bids, issued for the Project, the Construction Manager's bid is consistent with and considers the number of days to substantially complete the Project and the number of days to finally complete the Project to which the parties may have stipulated in the Agreement, which stipulation was based on the Construction Manager's bid. The Construction Manager agrees that Work shall be prosecuted regularly, diligently and uninterruptedly at such rate of progress as will confirm the Project will be substantially complete, and fully and finally complete, as recognized by the issuance of all required Notices of Substantial Completion and Notices of Final Acceptance, within any times stipulated and specified in the Agreement, as the same may be amended by Change Order or other written modification, and that the Principal Representative will be damaged if the times of completion are delayed.

It is expressly understood and agreed, by and between the parties hereto, that the times for the Substantial Completion of the Work or for the final acceptance of the Work as may be stipulated in the Agreement, and as applied here and in Article 4.2, Construction Manager/General Contractor (CM/GC) Agreement (SC-6.5), Modifications of Article 46, are reasonable times for these stages of completion of the Work, taking into such consideration all factors, including the average climatic range and usual industrial conditions prevailing in the locality of the building operations.

If the Construction Manager shall neglect, fail or refuse to complete the Work within the times specified in the Agreement, such failure shall constitute a breach of the terms of the Contract and the State of Colorado, acting by and through the Principal Representative, shall be entitled to liquidated damages for such neglect, failure or refusal, as specified in Article 7.6, Construction Manager/General Contractor (CM/GC) Agreement (SC-6.5), Modification of Article 46.

The Construction Manager and the Construction Manager's Surety shall be jointly liable for and shall pay the Principal Representative, or the Principal Representative may withhold, the sums hereinafter stipulated as liquidated damages for each calendar day of delay until the entire Project is 1) substantially completed, and the Notice (or all Notices) of Substantial Completion are issued, 2) finally complete and accepted and the Notice (or all Notices) of Acceptance are issued, or 3)

both. Delay in substantial completion shall be measured from the Date of the Notice to Proceed and delay in final completion and acceptance shall be measured from the Date of the Notice of Substantial Completion.

In the first instance, specified in Article 7.6.1 of the Construction Manager/General Contractor (CM/GC) Agreement (SC-6.5), Modification of Article 46, liquidated damages, if any, shall be the amount specified therein, for each calendar day of delay beginning after the stipulated number of days for Substantial Completion from the date of the Notice to Proceed, until the date of the Notice of Substantial Completion. Unless otherwise specified in any Supplementary General Conditions, in the event of any partial Notice of Substantial Completion, liquidated damages shall accrue until all required Notices of Substantial Completion are issued.

In the second instance, specified in Article 7.6.2 of the Construction Manager/General Contractor (CM/GC) Agreement (SC-6.5), Modification of Article 46, liquidated damages, if any, shall be the amount specified in Article 7.6.2 of the Construction Manager/General Contractor (CM/GC) Agreement (SC-6.5), Modification of Article 46, for each calendar day in excess of the number of calendar days specified in the Construction Manager's bid for the Project and stipulated in the Agreement to finally complete the Project (as defined by the issuance of the Notice of Acceptance) after the final Notice of Substantial Completion has been issued.

In the third instance, when so specified in both Articles 7.6.1 and 7.6.2 of the Construction Manager/General Contractor (CM/GC) Agreement (SC-6.5), both types of liquidated damages shall be separately assessed where those delays have occurred.

The parties expressly agree that said amounts are a reasonable estimate of the presumed actual damages that would result from any of the breaches listed, and that any liquidated damages that are assessed have been agreed to in light of the difficulty of ascertaining the actual damages that would be caused by any of these breaches at the time this Contract was formed; the liquidated damages in the first instance representing an estimate of damages due to the inability to use the Project; the liquidated damages in the second instance representing an estimate of damages due to the additional administrative, technical, supervisory and professional expenses related to and arising from the extended closeout period including delivery of any or all guarantees and warranties, the submittals of sales and use tax payment forms, the calling for the final inspection and the completion of the final punch list.

The parties also agree and understand that the liquidated damages to be assessed in each instance are separate and distinct, although potentially cumulative, damages for the separate and distinct breaches of delayed substantial completion or final acceptance. Such liquidated damages shall not be avoided by virtue of the fact of concurrent delay caused by the Principal Representative, or anyone acting on behalf of the Principal Representative, but in such event the period of delay for which liquidated damages are assessed shall be equitably adjusted in accordance with Article 38, Delays and Extensions of Time.

47 ARTICLE 47 DAMAGES

If either party to this Contract shall suffer damage under this Contract in any manner because of any wrongful act or neglect of the other party or of anyone employed by either of them, Notice of Claim shall be made in writing to such other party within a ten working days after the first observance of such damage, except to the extent of damages liquidated for the Construction Manager's failure to achieve timely completion as set forth in Article 46, Time of Completion and

SC-6.51

Liquidated Damages, the Principal Representative shall be responsible for, and at his or her option may insure against, loss of use of any existing property not included in the Work, due to fire or otherwise, however caused. Notwithstanding the foregoing, or any other provision of this Contract, to the contrary, no term or condition of this contract shall be construed or interpreted as a waiver, express or implied, of any of the immunities, rights, benefits, protection, or other provisions of the Colorado Governmental Immunity Act, Section 24-10-101, et seq., CRS, as now or hereafter amended. The parties understand and agree that liability for claims for injuries to persons arising out of negligence of the State of Colorado, its departments, institutions, agencies, boards, officials and employees is controlled and limited by the provisions of Section 24-101-101, et seq., CRS, as now or hereafter amended and the risk management statutes, Section 24-30-1501, et seq., CRS, as now or hereafter amended.

Notice of intent to file a claim under this clause shall be made in writing to the party liable within a reasonable time of the first observance of such damage, except that in the case of claims by the Principal Representative involving warranties against faulty Work or materials Notice shall be required only to the extent stipulated elsewhere in these General Conditions. Claims made to the Principal Representative involving extra cost or extra time arising by virtue of instructions to the Construction Manager to which Article 36, Claims, applies shall be made in accordance with Article 36. Other claims arising under the Contract involving extra cost or extra time which are made to the Principal Representative under this clause shall also be made in accordance with the procedures of Article 36, whether or not arising by virtue of instructions to the Construction Manager; provided however that it shall not be necessary to first obtain or request a written judgment of the Architect/Engineer.

Provided written Notice of intent to file a claim is provided as required in the preceding paragraph, nothing in this Article shall limit or restrict the rights of either party to bring an action at law or to seek other relief to which either party may be entitled, including consequential damages, if any, and shall not be construed to limit the time during which any action might be brought. Nothing in these General Conditions shall be deemed to limit the period of time during which any action may be brought as a matter of contract, tort, warranty or otherwise, it being the intent of the parties to allow any and all actions at law or in equity for such periods as the law permits. All such rights shall, however be subject to the obligation to assert claims and to appeal denials pursuant to Article 36, Claims, where applicable.

48 ARTICLE 48 STATE'S RIGHT TO DO THE WORK; TEMPORARY SUSPENSION OF WORK; DELAY DAMAGES

48.1 STATE'S RIGHT TO DO THE WORK

If after receipt of Notice to do so, the Construction Manager should neglect to prosecute the Work properly or fail to perform any provision of the Contract, the Principal Representative, after a second seven (7) days' advance written Notice to the Construction Manager and the Surety may, without prejudice to any other remedy the Principal Representative may have, take control of all or a portion of the Work, as the Principal Representative deems necessary and make good such deficiencies deducting the cost thereof from the payment then or thereafter due the Construction Manager, as provided in Article 30, Correction Of Work Before Acceptance and Article 33, Payments Withheld, provided, however, that the Architect/Engineer shall approve the amount charged to the Construction Manager by approval of the Change Order.

48.2 TEMPORARY SUSPENSION OF WORK

The State, acting for itself or by and through the Architect/Engineer, shall have the authority to suspend the Work, either wholly or in part, for such period or periods as may be deemed necessary due to:

- a) Unsuitable weather;
- b) Faulty Workmanship;
- c) Improper superintendence or project management;
- d) Construction Manager's failure to carry out orders or to perform any provision of the Contract Documents;
- e) Loss of, or restrictions to, appropriations;
- f) Conditions, which may be considered unfavorable for the prosecution of the Work.

If it should become necessary to stop Work for an indefinite period, the Construction Manager shall store materials in such manner that they will not become an obstruction or become damaged in any way; and he or she shall take every precaution to prevent damage to or deterioration of the Work, provide suitable drainage and erect temporary structures where necessary.

Notice of suspension of Work shall be provided to the Construction Manager in writing stating the reasons therefore. The Construction Manager shall again proceed with the Work when so notified in writing.

The Construction Manager understands and agrees that the State of Colorado cannot predict with certainty future revenues and could ultimately lack the revenue to fund the appropriations applicable to this Contract. The Construction Manager further acknowledges and agrees that in such event that State may, upon Notice to the Construction Manager, suspend the Work in anticipation of a termination of the Contract for the convenience of the State, pursuant to Article 50, Termination for Convenience of State. If the Contract is not so terminated the Contract sum and the Contract time shall be equitably adjusted at the time the Principal Representative directs the Work to be recommenced and gives Notice that the revenue to fund the appropriation is available.

48.3 DELAY DAMAGES

The Principal Representative and the State of Colorado shall be liable to the Construction Manager for the payment of any claim for extra costs, extra compensation or damages occasioned by hindrances or delays encountered in the Work only when and to the limited extent that such hindrance or delay is caused by an act or omission within the control of the Principal Representative, the Architect/Engineer or other persons or entities acting on behalf of the Principal Representative. Further, the Principal Representative and the State of Colorado shall be liable to the Construction Manager for the payment of such a claim only if the Construction Manager has provided required Notice of the delay or impact, or has presented its claim for an extension of time or claim of other delay or other impact due to changes ordered in the Work before proceeding with the changed Work. Except as otherwise provided, claims for extension of time shall be Noticed and filed in accordance with Article 38, Delays and Extensions of Time, within three (3) business days of the beginning of the delay with any claim filed within seven (7) days after the delay has ceased, or such claim is waived. Claims for extension of time or for other delay or other impact resulting from changes ordered in the Work shall be presented and adjusted as provided in Article 35, Changes in the Work.

49 ARTICLE 49 STATE'S RIGHTS TO TERMINATE CONTRACT

49.1 GENERAL

If the Construction Manager should be adjudged bankrupt, or if he or she should make a general assignment for the benefit of his or her creditors, or if a receiver should be appointed to take over his affairs, or if he or she should fail to prosecute his or her Work with due diligence and carry the Work forward in accordance with the construction schedule and the time limits set forth in the Contract Documents, or if he or she should fail to subsequently perform one or more of the provisions of the Contract Documents to be performed by him, the Principal Representative may serve written Notice on the Construction Manager and the Surety on performance and payment bonds, stating his or her intention to exercise one of the remedies hereinafter set forth and the grounds upon which the Principal Representative bases his or her right to exercise such remedy.

In such event, unless the matter complained of is satisfactorily cleared within ten (10) days after delivery of such Notice, the Principal Representative may, without prejudice to any other right or remedy, exercise one of such remedies at once, having first obtained the concurrence of the Architect/Engineer in writing that sufficient cause exists to justify such action.

49.2 CONDITIONS AND PROCEDURES

- 49.2.1 The Principal Representative may terminate the services of the Construction Manager, which termination shall take effect immediately upon service of Notice thereof on the Construction Manager and his or her Surety, whereupon the Surety shall have the right to take over and perform the Contract. If the Surety does not provide Notice to the Principal Representative of its intent to commence performance of the Contract within ten (10) days after delivery of the Notice of termination, the Principal Representative may take over the Work, take possession of and use all materials, tools, equipment and appliances on the premises and prosecute the Work to completion by such means as he or she shall deem best. In the event of such termination of his or her service, the Construction Manager shall not be entitled to any further payment under the Contract until the Work is completed and accepted. If the Principal Representative takes over the Work and if the unpaid balance of the contract price exceeds the cost of completing the Work, including compensation for any damages or expenses incurred by the Principal Representative through the default of the Construction Manager, such excess shall be paid to the Construction Manager. If, however, the cost, expenses and damages as certified by the Architect/Engineer exceed such unpaid balance of the contract price, the Construction Manager and his or her Surety shall pay the difference to the Principal Representative.
- 49.2.2 The Principal Representative may require the Surety on the Construction Manager 's bond to take control of the Work and see to it that all the deficiencies of the Construction Manager are made good, with due diligence within ten (10) days of delivery of Notice to the Surety to do so. As between the Principal Representative and the Surety, the cost of making good such deficiencies shall all be borne by the Surety. If the Surety takes over the Work, either by election upon termination of the services of the Construction Manager pursuant to Section 49.2.1 of this Article 49, State's Right To Terminate Contract, or upon instructions from the Principal Representative to do so, the provisions of the Contract Documents shall govern the Work to be done by the Surety, the Surety being substituted for the Construction Manager as to such provisions, including provisions as to payment for the Work, the times of completion and provisions of this Article as to the right of the Principal Representative to do the Work or to take control of all or a portion of the Work.
- 49.2.3 The Principal Representative may take control of all or a portion of the Work and make good the deficiencies of the Construction Manager, or the Surety if the Surety has been substituted for the Construction Manager, with or without terminating the Contract, employing such additional help

as the Principal Representative deems advisable in accordance with the provisions of Article 48.1, State's Right To Do The Work; Temporary Suspension Of Work; Delay Damages. In such event, the Principal Representative shall be entitled to collect from the Construction Manager and his or her Surety, or to deduct from any payment then or thereafter due the Construction Manager, the costs incurred in having such deficiencies made good and any damages or expenses incurred through the default of Construction Manager, provided the Architect/Engineer approves the amount thus charged to the Construction Manager. If the Contract is not terminated, a Change Order to the Contract shall be executed, unilaterally if necessary, in accordance with the procedures of Article 35, Changes in the Work.

49.3 ADDITIONAL CONDITIONS

If any termination by the Principal Representative for cause is later determined to have been improper, the termination shall be automatically converted to and deemed to be a termination by the Principal Representative for convenience and the Construction Manager shall be limited in recovery to the compensation provided for in Article 50, Termination for Convenience of State. Termination by the Construction Manager shall not be subject to such conversion.

50 ARTICLE 50 TERMINATION FOR CONVENIENCE OF STATE

50.1 NOTICE OF TERMINATION

The performance of Work under this Contract may be terminated, in whole or from time to time in part, by the State whenever for any reason the Principal Representative shall determine that such termination is in the best interest of State. Termination of Work hereunder shall be effected by delivery to the Construction Manager of a Notice of such termination specifying the extent to which the performance of Work under the Contract is terminated and the date upon which such termination becomes effective.

50.2 PROCEDURES

After receipt of the Notice of termination, the Construction Manager shall, to the extent appropriate to the termination, cancel outstanding commitments hereunder covering the procurement of materials, supplies, equipment and miscellaneous items. In addition, the Construction Manager shall exercise all reasonable diligence to accomplish the cancellation or diversion of all applicable outstanding commitments covering personal performance of any Work terminated by the Notice. With respect to such canceled commitments, the Construction Manager agrees to:

- 50.2.1 Settle all outstanding liabilities and all claims arising out of such cancellation of commitments, with approval or ratification of the Principal Representative, to the extent he or she may require, which approval or ratification shall be final for all purposes of this clause; and,
- 50.2.2 Assign to the State, in the manner, at the time, and to the extent directed by the Principal Representative, all of the right, title, and interest of the Construction Manager under the orders and subcontracts so terminated, in which case the State shall have the right, in its discretion, to settle or pay any or all claims arising out of the termination of such orders and subcontracts.

The Construction Manager shall submit his or her termination claim to the Principal Representative promptly after receipt of a Notice of termination, but in no event later than three (3) months from the effective date thereof, unless one or more extensions in writing are granted by the Principal Representative upon written request of the Construction Manager within such three-month period or authorized extension thereof. Upon failure of the Construction Manager to submit his or her termination claim within the time allowed, the Principal Representative may

SC-6.51

determine, on the basis of information available to him, the amount, if any, due to the Construction Manager by reason of the termination and shall thereupon pay to the Construction Manager the amount so determined.

Costs claimed, agreed to, or determined pursuant to the preceding and following paragraph shall be in accordance with the provisions of the Colorado Procurement Code or the applicable procurement code for institutions of higher education.

Subject to the preceding provisions, the Construction Manager and the Principal Representative may agree upon the whole or any part of the amount or amounts to be paid to the Construction Manager by reason of the termination under this clause, which amount or amounts may include any reasonable cancellation charges thereby incurred by the Construction Manager and any reasonable loss upon outstanding commitments for personal services which he or she is unable to cancel; provided, however, that in connection with any outstanding commitments for personal services which the Construction Manager is unable to cancel, the Construction Manager shall have exercised reasonable diligence to divert such commitments to other activities and operations. Any such agreement shall be embodied in an Amendment to this Contract and the Construction Manager shall be paid the agreed amount.

The State may from time to time, under such terms and conditions as it may prescribe, make partial payments against costs incurred by the Construction Manager in connection with the termination portion of this Contract, whenever, in the opinion of the Principal Representative, the aggregate of such payments is within the amount to which the Construction Manager will be entitled hereunder.

The Construction Manager agrees to transfer title and deliver to the State, in the manner, at the time, and to the extent, if any, directed by the Principal Representative, such information and items which, if the Contract had been completed, would have been required to be furnished to the State, including:

- a) Completed or partially completed plans, Drawings and information; and,
- b) Materials or equipment produced or in process or acquired in connection with the performance of the Work terminated by the Notice.

Other than the above, any termination inventory resulting from the termination of the Contract may, with written approval of the Principal Representative, be sold or acquired by the Construction Manager under the conditions prescribed by and at a price or prices approved by the Principal Representative. The proceeds of any such disposition shall be applied in reduction of any payments to be made by the State to the Construction Manager under this Contract or shall otherwise be credited to the price or cost of Work covered by this Contract or paid in such other manners as the Principal Representative may direct. Pending final disposition of property arising from the termination, the Construction Manager agrees to take such action as may be necessary, or as the Principal Representative may direct, for the protection and preservation of the property related to this Contract which is in the possession of the Construction Manager and in which the State has or may acquire an interest.

Any disputes as to questions of fact, which may arise hereunder, shall be subject to the Remedies provisions of the Colorado Procurement Code or the applicable procurement code for institutions of higher education.

51 ARTICLE 51 CONSTRUCTION MANAGER'S RIGHT TO STOP WORK AND/OR TERMINATE CONTRACT

If the Work shall be stopped under an order of any court or other public authority for a period of three (3) months through no act or fault of the Construction Manager or of any one employed by him, then the Construction Manager may on seven (7) days' written Notice to the Principal Representative and the Architect/Engineer stop Work or terminate this Contract and recover from the Principal Representative payment for all Work executed, any losses sustained on any plant or material, and a reasonable profit only for the Work completed. If the Architect/Engineer shall fail to issue or otherwise act in writing upon any certificate for payment within ten (10) days after it is presented and received by the Architect/Engineer, as provided in Article 31, Applications For Payments, or if the Principal Representative shall fail to pay the Construction Manager any sum certified that is not disputed in whole or in part by the Principal Representative in writing to the Construction Manager and the Architect/Engineer within thirty (30) days after the Architect/Engineer's certification, then the Construction Manager may on ten (10) days' written Notice to the Principal Representative and the Architect/Engineer stop Work and/or give written Notice of intention to terminate this Contract.

If the Principal Representative shall thereafter fail to pay the Construction Manager any amount certified by the Architect/Engineer and not disputed in writing by the Principal Representative within ten (10) days after receipt of such Notice, then the Construction Manager may terminate this Contract and recover from the Principal Representative payment for all Work executed, any losses sustained upon any plant or materials, and a reasonable profit only for the Work completed. The Principal Representative's right to dispute an amount certified by the Architect/Engineer shall not relieve the Principal Representative of the obligation to pay amounts not in dispute as certified by the Architect/Engineer.

52 ARTICLE 52 COLORADO SPECIAL PROVISIONS

52.1 CONTROLLER'S APPROVAL, C.R.S. § 24-30-202(1)

This contract shall not be valid until it has been approved by the Colorado State Controller or designee.

52.2 FUND AVAILABILITY, C.R.S. § 24-30-202(5.5)

Financial obligations of the State payable after the current fiscal year are contingent upon funds for that purpose being appropriated, budgeted, and otherwise made available.

52.3 GOVERNMENTAL IMMUNITY

Liability for claims for injuries to persons or property arising from the negligence of the State, its departments, boards, commissions committees, bureaus, offices, employees and officials shall be controlled and limited by the provisions of the Colorado Governmental Immunity Act, §24-10-101, et seq., C.R.S.; the Federal Tort Claims Act, 28 U.S.C. Pt. VI, Ch. 171 and 28 U.S.C. 1346(b), and the State's risk management statutes, §§24-30-1501, et seq. C.R.S. No term or condition of this Contract shall be construed or interpreted as a waiver, express or implied, of any of the immunities, rights, benefits, protections, or other provisions, contained in these statutes.

52.4 INDEPENDENT CONTRACTOR

Contractor shall perform its duties hereunder as an independent Contractor and not as an employee. Neither Contractor nor any agent or employee of Contractor shall be deemed to be an agent or employee of the State. Contractor shall not have authorization, express or implied, to bind the State to any agreement, liability or understanding, except as expressly set forth herein.

Contractor and its employees and agents are not entitled to unemployment insurance or workers compensation benefits through the State and the State shall not pay for or otherwise provide such coverage for Contractor or any of its agents or employees. Contractor shall pay when due all applicable employment taxes and income taxes and local head taxes incurred pursuant to this Contract. Contractor shall (i) provide and keep in force workers' compensation and unemployment compensation insurance in the amounts required by law, (ii) provide proof thereof when requested by the State, and (iii) be solely responsible for its acts and those of its employees and agents.

52.5 COMPLIANCE WITH LAW

Contractor shall comply with all applicable federal and State laws, rules, and regulations in effect or hereafter established, including, without limitation, laws applicable to discrimination and unfair employment practices.

52.6 CHOICE OF LAW, JURISDICTION, AND VENUE

Colorado law, and rules and regulations issued pursuant thereto, shall be applied in the interpretation, execution, and enforcement of this Contract. Any provision included or incorporated herein by reference which conflicts with said laws, rules, and regulations shall be null and void. All suits or actions related to this Contract shall be filed and proceedings held in the State of Colorado and exclusive venue shall be in the City and County of Denver.

52.7 PROHIBITED TERMS

Any term included in this Contract that requires the State to indemnify or hold Contractor harmless; requires the State to agree to binding arbitration; limits Contractor's liability for damages resulting from death, bodily injury, or damage to tangible property; or that conflicts with this provision in any way shall be void ab initio. Nothing in this Contract shall be construed as a waiver of any provision of §24-106-109, C.R.S.

52.8 SOFTWARE PIRACY PROHIBITION. SOFTWARE PIRACY PROHIBITION

State or other public funds payable under this Contract shall not be used for the acquisition, operation, or maintenance of computer software in violation of federal copyright laws or applicable licensing restrictions. Contractor hereby certifies and warrants that, during the term of this Contract and any extensions, Contractor has and shall maintain in place appropriate systems and controls to prevent such improper use of public funds. If the State determines that Contractor is in violation of this provision, the State may exercise any remedy available at law or in equity or under this Contract, including, without limitation, immediate termination of this Contract and any remedy consistent with federal copyright laws or applicable licensing restrictions.

52.9 EMPLOYEE FINANCIAL INTEREST/CONFLICT OF INTEREST

C.R.S. § 24-18-201 and C.R.S. § 24-50-507

The signatories aver that to their knowledge, no employee of the State has any personal or beneficial interest whatsoever in the service or property described in this contract. Contractor has no interest and shall not acquire any interest, direct or indirect, that would conflict in any manner or degree with the performance of Contractor services and Contractor shall not employ any person having such known interests.

52.10 VENDOR OFFSET AND ERRONEOUS PAYMENTS

C.R.S. § 24-30-202(1) & C.R.S. § 24-30-202.4

Subject to §24-30-202.4(3.5), C.R.S., the State Controller may withhold payment under the State's vendor offset intercept system for debts owed to State agencies for: (i) unpaid child support debts or child support arrearages; (ii) unpaid balances of tax, accrued interest, or other charges specified in §§39-21-101, et seq., C.R.S.; (iii) unpaid loans due to the Student Loan Division of the

Department of Higher Education; (iv) amounts required to be paid to the Unemployment Compensation Fund; and (v) other unpaid debts owing to the State as a result of final agency determination or judicial action. The State may also recover, at the State's discretion, payments made to Contractor in error for any reason, including, but not limited to, overpayments or improper payments, and unexpended or excess funds received by Contractor by deduction from subsequent payments under this Contract, deduction from any payment due under any other contracts, grants or agreements between the State and Contractor, or by any other appropriate method for collecting debts owed to the State.

53 ARTICLE 53 MISCELLANEOUS PROVISIONS

53.1 PROFESSIONAL ASSOCIATION PERMITTED

The Contractor may, with the prior written consent of the Principal Representative, join with him in the performance of this Agreement any other duly licensed Architect or Architects or registered Engineers with whom he may, in good faith, and enter into an association.

53.2 DISSOLUTION OF PROFESSIONAL ASSOCIATION

In the event there is dissolution of the association, other than by death of a member, the State of Colorado, acting by and through the Principal Representative, shall designate which former member shall continue with the work and may make all payments thereafter falling due in connection with the work directly to the person or persons so designated and without being required to look to the application of such payments as among the former members.

53.3 WAGE RATES, in accordance with C.R.S. § 24-30-1404 (1)

As amended, the Contractor has executed a schedule, which is attached hereto and made a part hereof by reference as **Exhibit B**, Wage Rates Schedule, and by doing so is certifying that wage rates and other factual unit costs supporting the compensation paid by the State for these professional services are accurate, complete and current.

The original contract price and any additions thereto shall be adjusted to exclude any significant sums by which the Principal Representative determines the contract price had been increased due to inaccurate, incomplete, or non-current wage rates and other factual unit costs. All such contract adjustments shall be made within one year following the end of this contract.

53.4 PUBLIC ART LAW

In recognition of the Public Art Law, C.R.S. § 24-48.5-312, as amended, if the State determines that this project is eligible for the acquisition of artworks in accordance with this law, the Contractor agrees to participate in the art selection process as an art jury member and to cooperate with and to advise the State in working with the commissioned artist(s) for this Capital Construction Project.

53.5 ASSIGNMENT

Contractor's rights and obligations under this Contract are personal and may not be transferred or assigned without the prior, written consent of the State. Any attempt at assignment or transfer without such consent shall be void. Any assignment or transfer of Contractor's rights and obligations approved by the State shall be subject to the provisions of this Contract.

53.6 SUBCONTRACTS

Contractor shall not enter into any subcontract in connection with its obligations under this Contract without the prior, written approval of the State. Contractor shall submit to the State a copy of each such subcontract upon request by the State. All subcontracts entered into by

Contractor in connection with this Contract shall comply with all applicable federal and state laws and regulations, shall provide that they are governed by the laws of the State of Colorado, and shall be subject to all provisions of this Contract.

53.7 BINDING EFFECT

Except as otherwise provided in §53.5, all provisions of this Contract, including the benefits and burdens, shall extend to and be binding upon the Parties' respective successors and assigns.

53.8 AUTHORITY

Each Party represents and warrants to the other that the execution and delivery of this Contract and the performance of such Party's obligations have been duly authorized.

53.9 CAPTIONS AND REFERENCES

The captions and headings in this Contract are for convenience of reference only, and shall not be used to interpret, define, or limit its provisions. All references in this Contract to sections (whether spelled out or using the § symbol), subsections, exhibits or other attachments, are references to sections, subsections, exhibits or other attachments contained herein or incorporated as a part hereof, unless otherwise noted.

53.10 COUNTERPARTS

This Contract may be executed in multiple, identical, original counterparts, each of which shall be deemed to be an original, but all of which, taken together, shall constitute one and the same agreement.

53.11 ENTIRE UNDERSTANDING

This Contract represents the complete integration of all understandings between the Parties related to the Work, and all prior representations and understandings related to the Work, oral or written, are merged into this Contract. Prior or contemporaneous additions, deletions, or other changes to this Contract shall not have any force or effect whatsoever, unless embodied herein.

53.12 DIGITAL SIGNATURES

If any signatory signs this Contract using a digital signature in accordance with the Colorado State Controller Contract, Grant and Purchase Order Policies regarding the use of digital signatures issued under the State Fiscal Rules, then any agreement or consent to use digital signatures within the electronic system through which that signatory signed shall be incorporated into this Contract by reference.

53.13 MODIFICATION

Except as otherwise provided in this Contract, any modification to this Contract shall only be effective if agreed to in a formal amendment to this Contract, properly executed and approved in accordance with applicable Colorado State law and State Fiscal Rules. Modifications permitted under this Contract, other than contract amendments, shall conform to the policies issued by the Colorado State Controller.

53.14 STATUTES, REGULATIONS, FISCAL RULES AND OTHER AUTHORITY

Any reference in this Contract to a statute, regulation, State Fiscal Rule, fiscal policy or other authority shall be interpreted to refer to such authority then current, as may have been changed or amended since the Effective Date of this Contract.

53.15 EXTERNAL TERMS AND CONDITIONS

Notwithstanding anything to the contrary herein, the State shall not be subject to any provision included in any terms, conditions, or agreements appearing on Contractor's or a Subcontractor's

website or any provision incorporated into any click-through or online agreements related to the Work unless that provision is specifically referenced in this Contract.

53.16 SEVERABILITY

The invalidity or unenforceability of any provision of this Contract shall not affect the validity or enforceability of any other provision of this Contract, which shall remain in full force and effect, provided that the Parties can continue to perform their obligations under this Contract in accordance with the intent of this Contract.

53.17 SURVIVIAL AND CERTAIN CONTRACT TERMS

Any provision of this Contract that imposes an obligation on a Party after termination or expiration of this Contract shall survive the termination or expiration of this Contract and shall be enforceable by the other Party.

53.18 TAXES

The State is exempt from federal excise taxes under I.R.C. Chapter 32 (26 U.S.C., Subtitle D, Ch. 32) (Federal Excise Tax Exemption Certificate of Registry No. 84-730123K) and from State and local government sales and use taxes under §§39-26-704(1), et seq., C.R.S. (Colorado Sales Tax Exemption Identification Number 98-02565). The State shall not be liable for the payment of any excise, sales, or use taxes, regardless of whether any political subdivision of the state imposes such taxes on Contractor. Contractor shall be solely responsible for any exemptions from the collection of excise, sales or use taxes that Contractor may wish to have in place in connection with this Contract.

53.19 THIRD PARTY BENEFICIARIES

Except for the Parties' respective successors and assigns described in § 53.5, this Contract does not and is not intended to confer any rights or remedies upon any person or entity other than the Parties. Enforcement of this Contract and all rights and obligations hereunder are reserved solely to the Parties. Any services or benefits which third parties receive as a result of this Contract are incidental to this Contract, and do not create any rights for such third parties.

53.20 WAIVER

A Party's failure or delay in exercising any right, power, or privilege under this Contract, whether explicit or by lack of enforcement, shall not operate as a waiver, nor shall any single or partial exercise of any right, power, or privilege preclude any other or further exercise of such right, power, or privilege.

53.21 CORA DISCLOSURE

To the extent not prohibited by federal law, this Contract and the performance measures and standards required under §24-106-107, C.R.S., if any, are subject to public release through the CORA.

53.22 STANDARD AND MANNER OF PERFORMANCE

Contractor shall perform its obligations under this Contract in accordance with the highest standards of care, skill and diligence in Contractor's industry, trade, or profession.

53.23 LICENSES, PERMITS, AND OTHER AUTHORIZATIONS

Contractor shall secure, prior to the Effective Date, and maintain at all times during the term of this Contract, at its sole expense, all licenses, certifications, permits, and other authorizations required to perform its obligations under this Contract, and shall ensure that all employees, agents and Subcontractors secure and maintain at all times during the term of their employment, agency or subcontract, all license, certifications, permits and other authorizations required to perform their obligations in relation to this Contract.

53.24 INDEMNIFICATION

53.24.1 General Indemnification

Contractor shall indemnify, save, and hold harmless the State, its employees, agents and assignees (the "Indemnified Parties"), against any and all costs, expenses, claims, damages, liabilities, court awards and other amounts (including attorneys' fees and related costs) incurred by any of the Indemnified Parties in relation to any act or omission by Contractor, or its employees, agents, Subcontractors, or assignees in connection with this Contract.

53.24.2 Confidential Information Indemnification

Disclosure or use of State Confidential Information by Contractor in violation of Article 54 may be cause for legal action by third parties against Contractor, the State, or their respective agents. Contractor shall indemnify, save, and hold harmless the Indemnified Parties, against any and all claims, damages, liabilities, losses, costs, expenses (including attorneys' fees and costs) incurred by the State in relation to any act or omission by Contractor, or its employees, agents, assigns, or Subcontractors in violation of Article 54.

53.24.3 Intellectual Property Indemnification

Contractor shall indemnify, save, and hold harmless the Indemnified Parties, against any and all costs, expenses, claims, damages, liabilities, and other amounts (including attorneys' fees and costs) incurred by the Indemnified Parties in relation to any claim that any Deliverable, Good or Service, software, or Work Product provided by Contractor under this Contract (collectively, "IP Deliverables"), or the use thereof, infringes a patent, copyright, trademark, trade secret, or any other intellectual property right. Contractor's obligations hereunder shall not extend to the combination of any IP Deliverables provided by Contractor with any other product, system, or method, unless the other product, system, or method is (a) provided by Contractor or Contractor's subsidiaries or affiliates; (b) specified by Contractor to work with the IP Deliverables; (c) reasonably required in order to use the IP Deliverables in its intended manner and the infringement could not have been avoided by substituting another reasonably available product, system, or method capable of performing the same function; or (d) is reasonably expected to be used in combination with the IP Deliverables.

53.24.4 Accessibility Indemnification

Contractor shall indemnify, save, and hold harmless the state, its employees, agents and assignees (collectively, the "Indemnified Parties"), against any and all costs, expenses, claims, damages, liabilities, court awards and other amounts (including attorneys' fees and related costs) incurred by any of the Indemnified Parties in relation to Contractor's failure to comply with §§24-85-101, et seq., C.R.S., or the Accessibility Standards for Individuals with a Disability as established by the Office of Information Technology pursuant to Section §24-85-103 (2.5), C.R.S.

53.25 ACCESSIBILITY

- 53.25.1 Contractor shall comply with and the Work Product provided under this Contract shall be in compliance with all applicable provisions of §§24-85-101, et seq., C.R.S., and the Accessibility Standards for Individuals with a Disability, as established by the Governor's Office Of Information Technology (OIT), pursuant to Section §24-85-103 (2.5), C.R.S. Contractor shall also comply with all State of Colorado technology standards related to technology accessibility and with Level AA of the most current version of the Web Content Accessibility Guidelines (WCAG), incorporated in the State of Colorado technology standards.
- 53.25.2 The State may require Contractor's compliance to the State's Accessibility Standards to be determined by a third party selected by the State to attest to Contractor's Work Product and software is in compliance with §§24-85-101, et seq., C.R.S., and the Accessibility Standards for

Individuals with a Disability as established by the Office of Information Technology pursuant to Section §24-85-103 (2.5), C.R.S.

54 ARTICLE 54 CONFIDENTIAL INFORMATION-STATE RECORDS

54.1 CONFIDENTIALITY

Contractor shall keep confidential, and cause all Subcontractors to keep confidential, all State Records, unless those State Records are publicly available. Contractor shall not, without prior written approval of the State, use, publish, copy, disclose to any third party, or permit the use by any third party of any State Records, except as otherwise stated in this Contract, permitted by law or approved in writing by the State. Contractor shall provide for the security of all State Confidential Information in accordance with all policies promulgated by the Colorado Office of Information Security and all applicable laws, rules, policies, publications, and guidelines. If Contractor or any of its Subcontractors will or may receive the following types of data, Contractor or its Subcontractors shall provide for the security of such data according to the following: (i) the most recently promulgated IRS Publication 1075 for all Tax Information and in accordance with the Safeguarding Requirements for Federal Tax Information attached to this Contract as an Exhibit, if applicable, (ii) the most recently updated PCI Data Security Standard from the PCI Security Standards Council for all PCI, (iii) the most recently issued version of the U.S. Department of Justice, Federal Bureau of Investigation, Criminal Justice Information Services Security Policy for all CJI, and (iv) the federal Health Insurance Portability and Accountability Act for all PHI and the HIPAA Business Associate Agreement attached to this Contract, if applicable. Contractor shall immediately forward any request or demand for State Records to the State's Principal Representative.

54.2 OTHER ENTITY ACCESS AND NONDISCLOSURE AGREEMENTS

Contractor may provide State Records to its agents, employees, assigns and Subcontractors as necessary to perform the Work, but shall restrict access to State Confidential Information to those agents, employees, assigns and Subcontractors who require access to perform their obligations under this Contract. Contractor shall ensure all such agents, employees, assigns, and Subcontractors sign agreements containing nondisclosure provisions at least as protective as those in this Contract, and that the nondisclosure provisions are in force at all times the agent, employee, assign or Subcontractor has access to any State Confidential Information. Contractor shall provide copies of those signed nondisclosure provisions to the State upon execution of the nondisclosure provisions if requested by the State.

54.3 USE, SECURITY, AND RETENTION

Contractor shall use, hold, and maintain State Confidential Information in compliance with any and all applicable laws and regulations only in facilities located within the United States, and shall maintain a secure environment that ensures confidentiality of all State Confidential Information. Contractor shall provide the State with access, subject to Contractor's reasonable security requirements, for purposes of inspecting and monitoring access and use of State Confidential Information and evaluating security control effectiveness. Upon the expiration or termination of this Contract, Contractor shall return State Records provided to Contractor or destroy such State Records and certify to the State that it has done so, as directed by the State. If Contractor is prevented by law or regulation from returning or destroying State Confidential Information, Contractor warrants it will guarantee the confidentiality of, and cease to use, such State Confidential Information.

54.4 INCIDENT NOTICE AND REMEDIATION

If Contractor becomes aware of any Incident, Contractor shall notify the State immediately and cooperate with the State regarding recovery, remediation, and the necessity to involve law enforcement, as determined by the State. Unless Contractor can establish that Contractor and its Subcontractors are not the cause or source of the Incident, Contractor shall be responsible for the cost of notifying each person who may have been impacted by the Incident. After an Incident, Contractor shall take steps to reduce the risk of incurring a similar type of Incident in the future as directed by the State, which may include, but is not limited to, developing and implementing a remediation plan that is approved by the State at no additional cost to the State. The State may adjust or direct modifications to this plan in its sole discretion, and Contractor shall make all modifications as directed by the State. If Contractor cannot produce its analysis and plan within the allotted time, the State, in its discretion, may perform such analysis and produce a remediation plan, and Contractor shall reimburse the State for the actual costs thereof. The State may, in its sole discretion and at Contractor's sole expense, require Contractor to engage the services of an independent, qualified, State-approved third party to conduct a security audit. Contractor shall provide the State with the results of such audit and evidence of Contractor's planned remediation in response to any negative findings.

54.5 DATA PROTECTION AND HANDLING

Contractor shall ensure that all State Records and Work Product in the possession of Contractor or any Subcontractors are protected and handled in accordance with the requirements of this Contract, including the requirements of any Exhibits hereto, at all times.

54.6 SAFEGUARDING PERSONAL IDENTIFIABLE INFORMATION (PII)

If Contractor or any of its Subcontractors will or may receive Personal Identifiable Information (PII) under this Contract, Contractor shall provide for the security of such PII, in a manner and form acceptable to the State, including, without limitation, State non-disclosure requirements, use of appropriate technology, security practices, computer access security, data access security, data storage encryption, data transmission encryption, security inspections, and audits. Contractor shall be a "Third-Party Service Provider" as defined in §24-73-103(1)(i), C.R.S. and shall maintain security procedures and practices consistent with §§24-73-101 et seq., C.R.S. In addition, as set forth in § 24-74-102, et. seq., C.R.S., Contractor, including, but not limited to, Contractor's employees, agents and Subcontractors, agrees not to share any PII with any third parties for the purpose of investigating for, participating in, cooperating with, or assisting with Federal immigration enforcement. If Contractor is given direct access to any State databases containing PII, Contractor shall execute, on behalf of itself and its employees, a certification as provided by the Office of the State Controller on an annual basis Contractor's duty and obligation to certify shall continue as long as Contractor has direct access to any State databases containing PII. If Contractor uses any Subcontractors to perform services requiring direct access to State databases containing PII, the Contractor shall require such Subcontractors to execute and deliver the certification to the State on an annual basis, so long as the Subcontractor has access to State databases containing PII.

STATE OF COLORADO OFFICE OF THE STATE ARCHITECT STATE BUILDINGS PROGRAM

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT A DESIGNATED SERVICES AND METHOD OF PAYMENT

DESIGNATED SERVICES AND METHOD OF PAYMENT

(Attached)

Exhibit A, Designated Services and Method of Payment, requests the CM/GC to include the following items in their estimates for the general conditions. This additional information/clarification is intended to augment the descriptions and requirements included within the actual matrix. Wherever the matrix includes more specific information or this document conflicts with the requirements noted in the matrix, the requirements included in the matrix govern.

- 1. A/E Trailer (Temporary Facilities) Will not be required.
- 2. Construction Manager's Payment & Performance Bonds (Insurance and Bonds) Base upon preliminary budget, will be adjusted if necessary.
- 3. General Liability, Automobile, Product Liability, and Excess Liability Insurance (Insurance and Bonds) Base upon preliminary budget, will be adjusted as necessary.
- 4. Builder's Risk Insurance (Insurance and Bonds) Base upon preliminary budget, will be adjusted as necessary. Costs begin when the construction starts.
- 5. Construction (Site) Fencing (Temporary Facilities) Assume fencing the entire site for the duration of the construction period.
- 6. Handrails & Toe Boards (Temporary Facilities) Provide allowance based on what would be reasonable for a project of this type and schedule.
- 7. Opening Protection (Temporary Facilities) Provide allowance based on what would be reasonable for a project of this type and schedule.
- 8. Temporary Stairs (Temporary Facilities) Provide allowance based on what would be reasonable for a project of this type and schedule.
- 9. Temporary Power Service (On-Site Utilities and Services) Provide allowance based on what would be reasonable for a project of this type and schedule.
- 10. Temporary Heating (Temporary Heating) The Temporary Heating Phase of Exhibit A contains and allocates many elements of anticipated reimbursable general conditions and direct costs. Provide appropriate allowances for these elements of reimbursable general conditions costs based upon what would be reasonable for a project of this type and schedule.
- 11. Field Inspector and trailer (Quality Control) The Quality Control Phase of Exhibit A contains and allocates many elements of anticipated reimbursable general conditions and direct costs. If the CM/GC submitting the proposal feels they will require Field Inspectors as part of their staff for managing the project, the appropriate costs should be reflected in the DPE for staff and reimbursable general condition expenses for transportation, office, and equipment. Please note and include as appropriate the other elements of general conditions cost included in Exhibit A's Quality Control Phase, project photographs, operator on-site training, and prepare operation/maintenance manuals.

SC-6.5

Rev. 03/2023 Ex A

Designated Services and Method of Payment

| CONSTRUCTION MANAGEMENT SERVICES | REQUIRE | D OF CONS | REQUIRED OF ARCH | REQUIRED OF OWNER | | |
|--|------------------------------|----------------------|---------------------|---------------------------|---|---|
| PRECONSTRUCTION | PRE- CONST SVCS FEE | CONST SRVS FEE | GEN CONDS. | DIRECT COST OF WORK | | |
| ARCHITECTURAL SELECTION | | | | | | Х |
| CIVIL, STRUCTURAL, MECHANICAL, PLUMBING, ELECTRICAL AND OTHER SUB- CONSULTANTS AS APPLICABLE | | | | | 1 | 2 |
| SPECIAL CONSULTANT SELECTION | | | | | | Х |
| SURVEYOR SELECTION | | | | | | Х |
| SITE SELECTION RECOMMENDATIONS | | | | | 2 | 1 |
| REVIEW DESIGN CONCEPTS | Х | | | | | |
| DEVELOP BID PACKAGES/SUB- CONTRACTING STRATEGY | 1 | | | | 2 | |
| SITE USE RECOMMENDATIONS | 2 | | | | 1 | |
| MATERIAL SELECTION RECOMMENDATIONS | 2 | | | | 1 | |
| BUILDINGS SYSTEMS RECOMMENDATONS | 2 | | | | 1 | |
| BUILDING EQUIPMENT RECOMMENDATIONS (MOVEABLE) | 2 | | | | 2 | 1 |
| BUILDING EQUIPMENT RECOMMENDATIONS (FIXED) | 2 | 2 | | | 1 | |
| CONSTRUCTION FEASIBILITY RECOMMENDATIONS | 1 | | | | 2 | |
| PROJECT MASTER SCHEDULING | Х | | | | | |
| BID PACKAGE RECOMMENDATIONS | 1 | | | | 2 | |
| LIFE CYCLE COSTING ANALYSIS | 2 | | | | 1 | |
| INFORMAL AND FORMAL VALUE ENGINEERING | Х | | | | 2 | |
| ENERGY USE ANALYSIS AND RECOMMENDATIONS | 2 | | | | 1 | |
| PRELIMINARY TOTAL COST FEASIBILITY REVIEW | 1 | | | | 2 | |
| LABOR AVAILABILITY REVIEW (SUBCONTRACTORS) | Х | | | | | |
| MATERIAL EQUIPMENT AND CONTRACTOR AVAILABILITY | Х | | | | | |

Responsibility:

Designated Services and Method of Payment

| CONSTRUCTION MANAGEMENT SERVICES | REQUIRI | ED OF CONS | STRUCTION | MANAGER | REQUIRED OF ARCH | REQUIRED OF OWNER |
|-------------------------------------|------------------------------|----------------------|---------------|---------------------------|---------------------|----------------------|
| PROJECT BUDGETING AND COST CONTROL | PRE- CONST SVCS FEE | CONST SVCS FEE | GEN CONDS. | DIRECT COST OF WORK | | |
| TOTAL PROJECT COST BUDGET | | | | | | Х |
| CONSTRUCTION COST BUDGET | Х | | | | | |
| CONSTRUCTION COST BUDGET ESTIMATES | Х | | | | | |
| PRELIMINARY COST MODEL | Х | | | | | |
| SCHEMATIC DESIGN PHASE ESTIMATES | Х | | | | | |
| DESIGN DEVELOPMENT PHASE ESTIMATES | Х | | | | | |
| BID PACKAGE/SUBCONTRACT ESTIMATES | Х | | | | | |
| CASH FLOW PROJECTIONS | Х | | | | | |
| PHASE FUNDING MODELING | Х | | | | | |
| MATERIAL SURVEYS | Х | | | | | |
| TRADE CONTRACTOR ESTIMATES | Х | | | | | |
| CHANGE ORDER ESTIMATES | | | Х | | | |
| SET-UP COST ACCOUNTING | | | Х | | | |
| SET-UP REPORTING METHODS | | | Х | | | |
| SET-UP PAYMENT PROCEDURES | | | 2 | | | 1 |
| SET-UP CHANGE ORDER PROCEDURES | | | 1 | | 2 | 1 |
| CONTINUAL PROJECT COST MONITORING | | | 1 | | 2 | 1 |

| Responsibility: | | | |
|-----------------|-----------|-------------|---------------|
| | x = Total | 1 = Primary | 2 = Secondary |

Designated Services and Method of Payment

| CONSTRUCTION MANAGEMENT SERVICES | REQUIRE | D OF CONS | TRUCTION | MANAGER | REQUIRED OF ARCH | REQUIRED OF OWNER |
|---|------------------------------|----------------------|---------------|---------------------------|---------------------|----------------------|
| SUB-CONTRACTING SELECTION AND PURCHASING | PRE- CONST SVCS FEE | CONST SVCS FEE | GEN CONDS. | DIRECT COST OF WORK | | |
| SET PRE-QUALIFICATION CRITERIA | 1 | | | | | 2 |
| RECOMMEND SUBCONTRACTOR SELECTION METHODS | Х | | | | | |
| RECOMMEND SUBCONTRACTOR AWARD SELECTION METHODS | Х | | | | | |
| DEVELOP CONTRACTOR INTEREST | Х | | | | | |
| PREPARE BIDDING SCHEDULES | X | | | | | |
| CONDUCT PRE-BID CONFERENCE AND ISSUE PLANS | Х | | | | | |
| RECEIVE BIDS | 1 | | | | | 2 |
| ANALYZE BIDS | 1 | | | | | 2 |
| RECOMMEND AWARD | 1 | | | | 2 | 2 |
| VERIFY UNIT COSTS | X | | | | | |
| NEGOTIATE UNION RATES AND MANPOWER COSTS REQUIRED | Х | | | | | |
| CONDUCT PRE-AWARD CONFERENCES | | | Х | | | |
| PREPARE CONTRACTS | Х | | | | | |
| SUPPLIER AND SUBCONTRACTOR REVIEW | Х | | | | | |
| ORIGINATE RFI'S AFTER SCREENING | | | Х | | | |
| PREPARE CHANGE ORDERS | | | 1 | | 2 | |
| VERIFY CORRECTNESS OF QUANTITIES AND PRICES OF CHANGE ORDERS | | | 1 | | 2 | |
| COORDINATE OWNER-SUPPLIED FIXED EQUIPMENT | | | 2 | | 1 | 1 |

Responsibility:

Designated Services and Method of Payment

| CONSTRUCTION MANAGEMENT SERVICES | REQUIRED | OF CONST | FRUCTION I | MANAGER | REQUIRED OF ARCH | REQUIRED OF OWNER |
|---|---------------------------|----------------------|---------------|---------------------------|---------------------|-------------------|
| CONTRACT DOCUMENTS COORDINATION | PRE- CONST SVCS FEE | CONST SVCS FEE | GEN CONDS. | DIRECT COST OF WORK | | |
| FEASIBILITY REVIEW AND RECOMMENDATIONS | Х | | | | | |
| CONSTRUCTIBILITY REVIEW AND RECOMMENDATIONS | Х | | | | | |
| SUBCONTRACTOR WORK SCOPING | Х | | | | | |
| RESPONSIBILITY FOR: SAFETY PRECAUTIONS | | | х | | | |
| SAFETY PROGRAMS | | | Х | | | |
| TEMPORARY FACILITIES | | | Х | | | |
| COMMON USE EQUIPMENT | | | Х | | | |
| COMMON USE SERVICES | | | X | | | |
| REVIEW FOR: JURISDICTIONAL OVERLAP | Х | | | | | |
| INCLUSION OF ALL WORK | Х | | | | | |
| PHASE CONSTRUCTION COORD. | Х | | | | | |
| IDENTIFY LONG LEAD ITEMS | Х | | | | | |
| OBTAIN AGENCY APPROVALS | | | | | 2 | 1 |
| ASSIST IN OBTAINING PERMITS (AS NEEDED) | | | X | | | |

Responsibility:

Designated Services and Method of Payment

| CONSTRUCTION MANAGEMENT SERVICES | REQUIRE | OF CONST | ANAGER | REQUIRED OF ARCH | REQUIRED OF OWNER | |
|--|------------------------------|----------------------|---------------|---------------------------|----------------------|--|
| CONSTRUCTION PHASE STAFF | PRE- CONST SVCS FEE | CONST SVCS FEE | GEN CONDS. | DIRECT COST OF WORK | | |
| PROJECT MANAGER/ASSISTANT PROJECT MANAGER (AS REQUIRED) | | | Х | | | |
| PROJECT SUPERINTENDENT (AS REQUIRED) | | | Х | | | |
| ASSISTANT PROJECT SUPERINTENDENT | | | Х | | | |
| MECHANICAL COORDINATOR (AS REQUIRED) | | | Х | | | |
| ELECTRICAL COORDINATOR (AS REQUIRED) | | | Х | | | |
| OFFICE ENGINEER (AS REQUIRED) | | | Х | | | |
| ENGINEERING AND LAYOUT (AS REQUIRED) FIELD ENGINEER-LINE AND GRADE (AS REQUIRED) | | | | X X | | |
| DRAWING CHECKER (AS REQUIRED) | | | х | | | |
| RODMAN AND HELPERS (AS REQUIRED) | | | | Х | | |
| TIME KEEPER/CHECKER (AS REQUIRED) | | | X | | | |
| SCHEDULING ENGINEER (AS REQUIRED) | | | Х | | | |
| PROJECT COORDINATOR | | | Х | | | |
| COST ENGINEER (AS REQUIRED) | | | X | | | |
| CLERK-TYPIST (AS REQUIRED) | | | Х | | | |
| SAFETY ENGINEER (AS REQUIRED) | | | X | | | |

Responsibility:

Designated Services and Method of Payment

| CONSTRUCTION MANAGEMENT SERVICES | ANAGER | REQUIRED OF ARCH | | | | |
|--|------------------------------|----------------------|---------------|---------------------------|---------------------|----------------------|
| TRAVEL AND LODGING | PRE- CONST SVCS FEE | CONST SVCS FEE | GEN CONDS. | DIRECT COST OF WORK | | |
| STAFF TRAVEL COST | | Х | | | | |
| STAFF TRANSPORTATION | | Х | | | | |
| PROJECT STAFF MOVING EXPENSES | | Х | | | | |
| PROJECT STAFF SUBSISTENCE COSTS | | | Х | | | |
| CONSTRUCTION MANAGEMENT SERVICES | REQUIRE | D OF CONS | TRUCTION | MANAGER | REQUIRED OF ARCH | REQUIRED OF OWNER |
| PHASE: TEMPORARY FACILITIES | PRE- CONST SVCS FEE | CONST SVCS FEE | GEN CONDS. | DIRECT COST OF WORK | | |
| SAFETY EQUIPMENT AND FIRST AID SUPPLIES | | | Х | | | |
| HANDRAILS AND TOE BOARDS | | | Х | | | |
| OPENING PROTECTION | | | Х | | | |
| FIRE EXTINGUISHERS | | | Х | | | |
| WATCHMAN SERVICE | | | | Х | | |
| OFFICE OR TRAILER RENTAL | | | Х | | | |
| HYDRATION STATION CUPS | | | Х | | | |
| TEMPORARY STAIRS | | | Х | | | |
| PROJECT SIGNS | | | Х | | | |
| BULLETIN BOARDS | | | Х | | | |
| CONSTRUCTION FENCING | | | Х | | | |
| BARRICADES AND COVERED WALKWAYS (AS REQUIRED) | | | | Х | | |
| SAFETY NETS (AS REQUIRED) | | | | Х | | |
| A/E TEMPORARY OFFICE (AS REQUIRED) | | | Х | | | |
| TEMPORARY TOILETS | | | Х | | | |

Responsibility:

Designated Services and Method of Payment

| CONSTRUCTION MANAGEMENT SERVICES | REQUIRE | D OF CONS | TRUCTION I | MANAGER | REQUIRED OF ARCH | REQUIRED OF OWNER |
|--|------------------------------|----------------------|---------------|---------------------------|---------------------|----------------------|
| ON-SITE UTILITIES AND SERVICES | PRE- CONST SVCS FEE | CONST SVCS FEE | GEN CONDS. | DIRECT COST OF WORK | | |
| TEMPORARY TELEPHONE INSTALLATION AND EXPENSE (INCLUDING LOCAL A/E) | | | Х | | | |
| TEMPORARY POWER SERVICE | | | Х | | | |
| POWER SERVICE | | | Х | | | |
| TEMPORARY WATER AND HEATING SERVICE | | | Х | | | |
| HEATING ENERGY CHARGES | | | Х | | | |
| TEMPORARY WIRING | | | | х | | |
| LIGHT BULBS | | | | Х | | |
| DAILY CLEAN-UP | | | 1 | 2 | | |
| WEEKLY TRASH-REMOVAL | | | 1 | 2 | | |
| FINAL CLEAN-UP | | | 1 | 2 | | |
| DUMP PERMITS AND FEES | | | | Х | | |
| DEBRIS HAULING/REMOVAL | | | | Х | | |
| FLAGMAN/TRAFFIC CONTROL (AS REQUIRED) | | | | Х | | |
| FUELS FOR INITIAL TANK FILLING | | | | Х | | |
| TEMPORARY ROADS | | | | Х | | |
| ROADWAY MAINTENANCE | | | | Х | | |
| DUST CONTROLS | | | | Х | | |
| TEMPORARY EROSION CONTROL TEMP. WATER /SEWER EXPENSE & WATER | | | | Х | | |
| EXPENSES - SITE GRADING & COMPACTION TWO-WAY RADIO EQUIPMENT | | | | Х | | |
| (AS REQUIRED) | | | Х | | | |
| TRASH CHUTE AND HOPPERS (AS REQUIRED) | | | | Х | | |

Responsibility:

Designated Services and Method of Payment

| CONSTRUCTION MANAGEMENT SERVICES | REQUI | RED OF CONS | TRUCTION M | ANAGER | REQUIRED OF ARCH | REQUIRED OF OWNER |
|--|------------------------------|-------------------|---------------|---------------------------|---------------------|-------------------|
| ON-SITE EQUIPMENT | PRE- CONST SVCS FEE | CONST SVCS FEE | GEN CONDS. | DIRECT COST OF WORK | | |
| AUTOMOBILE AND FUEL (AS REQUIRED) | | | Х | | | |
| PICK-UP TRUCK AND FUEL (AS REQUIRED) | | | Х | | | |
| FLATBED TRUCK AND FUEL (AS REQUIRED) | | | Х | | | |
| WATER TRUCK (AS REQUIRED) | | | | Х | | |
| AIR COMPRESSOR AND FUEL (AS REQUIRED) | | | | Х | | |
| DEWATERING EQUIPMENT AND FUEL (AS REQUIRED) | | | | Х | | |
| TEMPORARY GENERATOR AND FUEL (AS REQUIRED) | | | Х | | | |
| DEBRIS REMOVAL/HAULING EQUIPMENT (AS REQUIRED) | | | | Х | | |
| SNOW REMOVAL (AS REQUIRED) | | | Х | | | |
| TIRES AND MAINTENANCE COST (AS REQUIRED) | | | Х | | | |
| FORKLIFT OPERATOR | | | | Х | | |
| MATERIAL HOIST OPERATOR | | | Х | | | |
| PERSONNEL OPERATOR | | | Х | | | |
| FIXED CRANE OPERATOR | | | | Х | | |
| TRAVEL CRANE OPERATOR | | | | Х | | |

Responsibility:

Designated Services and Method of Payment

| CONSTRUCTION MANAGEMENT SERVICES | REQUIRE | D OF CONS | TRUCTION | MANAGER | REQUIRED OF ARCH | REQUIRED OF OWNER |
|---|------------------------------|----------------------|---------------|---------------------------|---------------------|----------------------|
| TEMPORARY HEATING | PRE- CONST SVCS FEE | CONST SVCS FEE | GEN CONDS. | DIRECT COST OF WORK | | |
| REMOVE SNOW AND ICE (AS REQUIRED) | | | Х | | | |
| TEMPORARY ENCLOSURES (AS REQUIRED) | | | | Х | | |
| PIPING COST IN BUILDING (AS REQUIRED) | | | Х | | | |
| FUEL COST FOR HEATING (AS REQUIRED) | | | Х | | | |
| POWER COST FOR HEATING (AS REQUIRED) | | | Х | | | |
| FURNACE RENTAL (AS REQUIRED) | | | Х | | | |
| HEATER RENTAL (AS REQUIRED) | | | Х | | | |
| BOILER RENTAL (AS REQUIRED) | | | Х | | | |
| OPERATOR - TEMPORARY SYSTEMS (AS REQUIRED) | | | Х | | | |
| OPERATION FIRE WATCH (AS REQUIRED) | | | | Х | | |
| CLEANING COST (AS REQUIRED) | | | | Х | | |
| MAINTENANCE COST (AS REQUIRED) | | | | х | | |
| EXTENDED WARRANTY COST (AS REQUIRED) | | | | Х | | |
| FILTER CHANGE (AS REQUIRED) | | | | Х | | |
| TEMPORARY OFFICE HEATING (AS REQUIRED) | | | Х | | | |
| TEMP WEATHER PROTECTION & HEATING FOR SUBCONTRACTORS (AS REQ'D) | | | | Х | | |

Responsibility:

Designated Services and Method of Payment

| CONSTRUCTION MANAGEMENT SERVICES | REQUIRE | D OF CONS | TRUCTION I | MANAGER | REQUIRED OF ARCH | REQUIRED OF OWNER |
|---|------------------------------|----------------------|---------------|---------------------------|---------------------|----------------------|
| REPRODUCTION/PRINTING AND DATA PROCESSING | PRE- CONST SVCS FEE | CONST SVCS FEE | GEN CONDS. | DIRECT COST OF WORK | | |
| COST STUDY DOCUMENTS | | | | | X | |
| SYSTEMS STUDY DOCUMENTS | | | | | Х | |
| BID PACKAGE SETS (SEE PARAGRAPH 3.5.9) | Х | | | | | |
| BIDDING INSTRUCTIONS | Х | | | | | |
| CONSTRUCTION DOCUMENTS ORIGINAL POSTAGE AND EXPRESS COSTS (CM/GC ISSUES PLANS) | | | X | | Х | |
| AS-BUILT SUB-DOCUMENTS | | | | Х | | |
| AS-BUILT DOCUMENTS | | | Х | | | |
| ACCOUNTING FORMS | | Х | | | | |
| FIELD REPORTING FORMS | | | х | | | |
| SUBCONTRACT AGREEMENT FORMS | Х | | | | | |
| SCHEDULE REPORT FORMS | | | Х | | | |
| ESTIMATING FORMS | Х | | | | | |
| COST REPORTING FORMS | Х | | | | | |
| VALUE ANALYSIS STUDIES PRINTING | Х | | | | | |
| DATA PROCESSING (MAIN OFFICE) | | Х | | | | |
| REFERENCE MATERIALS | | | Х | | | |
| SHOP DRAWING PRINTING | | | | Х | | |
| ON-SITE FAX AND COPIER | | | Х | | | |
| DATA PROCESSING (ON-SITE) MAINTENANCE MANUALS (FROM SUBS) AND OPERATIONS MANUALS (FROM SUBS) | | | X | X | | |

Responsibility:

Designated Services and Method of Payment

| CONSTRUCTION MANAGEMENT SERVICES | REQUIR | ED OF CONS | TRUCTION M | IANAGER | REQUIRED OF ARCH | REQUIRED OF OWNER |
|--|---------------------------|-------------------|---------------|---------------------------|---------------------|----------------------|
| QUALITY CONTROL | PRE- CONST SVCS FEE | CONST SVCS FEE | GEN CONDS. | DIRECT COST OF WORK | | |
| FIELD INSPECTOR (AS REQUIRED) | | | Х | | | |
| INSPECTORS' OFFICE (AS REQUIRED) | | | Х | | | |
| INSPECTORS' TRANSPORTATION (AS REQUIRED) | | | Х | | | |
| INSPECTORS' EQUIPMENT (AS REQUIRED) | | | Х | | | |
| SPECIAL INSPECTION CONSULTANTS | | | | | | Х |
| SPECIAL TESTING CONSULTANTS | | | | | | х |
| CONCRETE SUBSTRUCTURE- OBSERVATIONS | | | | | | Х |
| CONCRETE TESTING | | | | | | Х |
| MASONRY TESTING | | | | | | Х |
| COMPACTION TESTING | | | | | | Х |
| WELDING TESTING | | | | | | Х |
| PIER INSPECTION/TESTING | | | | | | Х |
| SOILS INVESTIGATION | | | | | | Х |
| SPECIAL TESTING SERVICES (EXCEPT AS NOTED) | | | | | | Х |
| PROJECT PHOTOGRAPHS | | | Х | | | |
| WARRANTY INSPECTIONS / REWORK | | 1 | | | 2 | |
| AIR AND WATER BALANCING | | | | Х | | |
| OPERATOR ON-SITE TRAINING | | | X | | | |
| PREPARE OPERATION/MAINTENANCE MANUALS | | | 2 | 1 | | |

Responsibility:

Designated Services and Method of Payment

| CONSTRUCTION MANAGEMENT SERVICES | REQUIRED OF CONSTRUCTION MANAGER | | | | REQUIRED OF ARCH | REQUIRED OF OWNER |
|---|----------------------------------|----------------------|---------------|---------------------------|---------------------|----------------------|
| PERMITS AND SPECIAL FEES | PRE- CONST SVCS FEE | CONST SVCS FEE | GEN CONDS. | DIRECT COST OF WORK | | |
| STORAGE YARD RENTAL | | | | Х | | |
| PARKING LOT RENTALS AND SHUTTLE EXPENSES (AS REQUIRED) | | | | 2 | | 1 |
| FIELD OFFICE STAFF PARKING FEES | | | Х | | | |
| SIGN PERMITS | | | Х | | | |
| STREET/CURB PERMIT | | | | Х | | |
| BUILDING PERMITS | | | | | | X |
| PLAN CHECK FEES | | | | | | X |
| WATER SYSTEM DEV. FEE | | | | | | х |
| SEWER USE & DRAINAGE PERMIT/DEV. FEE | | | | | | Х |
| STORM CONNECTION FEE GAS AND POWER SERVICE CHARGE (PERMANENT) | | | | | | X x |
| GAS AND POWER SERVICE CHARGE (TEMPORARY) | | | Х | | | |
| STEAM SERVICE CHARGE | | | | | | Х |
| CHILLER WATER SERVICE CHARGE | | | | | | Х |
| SPECIAL TAP FEES | | | | | | X |
| CONTRACTORS LICENSES | | Х | | | | |
| CONSTRUCTION EQUIPMENT LICENSES | | Х | | | | |
| CONSTRUCTION EQUIPMENT PERMITS | | | | X | | |

Responsibility:

Designated Services and Method of Payment

| CONSTRUCTION MANAGEMENT SERVICES | REQUIRED OF CM/GC | | | | REQUIRED OF ARCH | REQUIRED OF OWNER |
|--|------------------------------|----------------------|---------------|------------------------------|---------------------|----------------------|
| INSURANCE AND BONDS | PRE- CONST SVCS FEE | CONST SVCS FEE | GEN CONDS. | DIRECT COST OF WORK | | |
| BUILDERS RISK INSURANCE | | | X | | | |
| GENERAL LIABILITY, INCLUDING AUTOMOBILE | | | Х | | | |
| PRODUCT LIABILITY | | | Х | | | |
| EXCESS LIABILITY COVERAGE | | | Х | | | |
| WORKERS COMPENSATION (FIELD OFFICE STAFF) | | | Х | | | |
| FICA INSURANCE (FIELD OFFICE STAFF) | | | Х | | | |
| FEDERAL UNEMPLOYMENT (FIELD OFFICE STAFF) | | | Х | | | |
| STATE UNEMPLOYMENT (FIELD OFFICE STAFF) | | | Х | | | |
| CONSTRUCTION MANAGER'S PAYMENT BOND | | | Х | | | |
| CONSTRUCTION MANAGER'S PERFORMANCE BOND | | | х | | | |
| STATE/LOCAL BONDS | | | | Х | | |
| * SUBCONTRACTOR BONDS | | | | X | | |

Responsibility:

^{*} Only as mutually agreed upon between the principal representative and the CONSTRUCTION MANAGER.

Designated Services and Method of Payment

| CONSTRUCTION MANAGEMENT SERVICES | REQUIRED OF CONSTRUCTION MANAGER | | | | REQUIRED OF ARCH | REQUIRED OF OWNER |
|---|----------------------------------|----------------------|---------------|---------------------------|---------------------|----------------------|
| OTHER COSTS | PRE- CONST SVCS FEE | CONST SVCS FEE | GEN CONDS. | DIRECT COST OF WORK | | |
| CONSTRUCTION EQUIPMENT | | | | Х | | |
| CONSTRUCTION SERVICES LABOR | | | | Х | | |
| CONSTRUCTION MATERIALS | | | | Х | | |
| COST OF DESIGN AND ENGINEERING | | | | | Х | |
| A/E FAST TRACK COST EXTRAS | | | | | Х | |
| PRELIMINARY SOILS INVESTIGATION | | | | | | Х |
| TITLE/DEVELOPMENT COST | | | | | | Х |
| BUILDING OPERATION AFTER MOVE-IN | | | | | | Х |
| BUILDING MAINTENANCE AFTER MOVE-IN | | | | | | Х |
| MOVING COORDINATION | | | | | | Х |
| MOVING COSTS | | | | | | Х |
| COSTS OF EMERGENCY WORK | | | | X | | |
| CONSTRUCTION MANAGER GENERAL OVERHEAD COST | | X | | | | |
| CONSTRUCTION MANAGER PROFIT MARGIN | | Х | | | | |
| GMP FINANCIAL RESPONSIBILITIES | | Х | | | | |
| STATE REQUIRED INSPECTIONS | | | | | | X |

Responsibility:

EXHIBIT A

Designated Services and Method of Payment

| CONSTRUCTION MANAGEMENT SERVICES | REQUIRE | D OF CONS | REQUIRED OF ARCH | REQUIRED OF OWNER | | |
|--|------------------------------|----------------------|---------------------|---------------------------|--|--|
| OFF-SITE SERVICES | PRE- CONST SVCS FEE | CONST SVCS FEE | GEN CONDS. | DIRECT COST OF WORK | | |
| CORPORATE EXECUTIVES (AS REQUIRED) | X | Х | | | | |
| PRINCIPAL IN CHARGE (AS REQUIRED) | X | X | | | | |
| PROJECT EXECUTIVE (AS REQUIRED) | Х | X | | | | |
| LEGAL - BASIC SERVICES (AS REQUIRED) | X | X | | | | |
| ACCOUNTING (AS REQUIRED) | | X | | | | |
| PURCHASING (AS REQUIRED) | Х | | | | | |
| SAFETY OFFICER (AS REQUIRED) | | X | | | | |
| EEO OFFICER (AS REQUIRED) | X | х | | | | |
| SECRETARIAL AND CLERK-TYPIST (AS REQUIRED) | Х | Х | | | | |
| BENEFITS AND VACATIONS FOR ABOVE | Х | X | | | | |
| STAFF BONUSES | | X | | | | |

Responsibility:

x = Total 1 = Primary 2 = Secondary

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT B CONSTRUCTION MANAGER'S CERTIFICATION

CONSTRUCTION MANAGER'S CERTIFICATION

| eby ce | rtify: |
|--------|--|
| a. | That I am the and duly authorized representative of the firm of:; |
| | and |
| b. | That the wage rates and other factual unit costs supporting the compensation to be paid by the State for these professional services and other services are accurate, complete, and current; and |
| c. | That I understand the original contract price and any additions shall be adjusted to exclude any significant sums by which the State determines the contract price had been increased due to inaccurate, incomplete, or non-current wage rates and other factual unit costs; and |
| d. | That all such contract adjustments shall be made within one year following the end of this contract. |
| со | NSTRUCTION MANAGER |
| Sig | nature - , Owner |

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT C REQUEST FOR PROPOSAL

REQUEST FOR PROPOSAL (Dated Aug 29, 2023) & RFP ADDENDUMS

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT D CONSTRUCTION MANAGER'S FEE PROPOSAL

- D.1 Construction Manager's Fee Proposal
- D.2 Construction Manager's Certificate of Insurance
- D.3 Construction Manager's Direct Labor Burden Calculation

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT D.1 CONSTRUCTION MANAGER'S FEE PROPOSAL

<u>Note:</u> Supervisory rates and equipment rental rates included in the proposal are for estimating purposes only. These rates are subject to review/audit prior to the first billing and will be adjusted to actual rates for specific individuals and agreed upon rates for equipment.

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT D.2 CONSTRUCTION MANAGER'S CERTIFICATE OF INSURANCE

After signing this document, you must supply your Certificates of Insurance for review before the University of Colorado Anschutz Medical Campus can continue processing this agreement.

Please email your insurance certificates to:

Elaine Rydberg, elaine.rydberg@cuanschutz.edu and

Robert Holzwarth, robert.holzwarth@cuanschutz.edu

Vendor Initial



CERTIFICATE OF LIABILITY INSURANCE

DATE (MM/DD/YYYY)
CURRENT DATE

THIS CERTIFICATE IS ISSUED AS A MATTER OF INFORMATION ONLY AND CONFERS NO RIGHTS UPON THE CERTIFICATE HOLDER. THIS CERTIFICATE DOES NOT AFFIRMATIVELY OR NEGATIVELY AMEND, EXTEND OR ALTER THE COVERAGE AFFORDED BY THE POLICIES BELOW. THIS CERTIFICATE OF INSURANCE DOES NOT CONSTITUTE A CONTRACT BETWEEN THE ISSUING INSURER(S), AUTHORIZED REPRESENTATIVE OR PRODUCER, AND THE CERTIFICATE HOLDER.

IMPORTANT: If the certificate holder is an ADDITIONAL INSURED, the policy(ies) must have ADDITIONAL INSURED provisions or be endorsed. If SUBROGATION IS WAIVED, subject to the terms and conditions of the policy, certain policies may require an endorsement. A statement on this certificate does not confer rights to the certificate holder in lieu of such endorsement(s).

| and to an action and the action and the action action and the action action and the action action action and the action a | | |
|--|--------------------------------------|-------|
| PRODUCER | CONTACT NAME: | |
| COMPANY | PHONE (A/C, No, Ext): FAX (A/C, No): | |
| ADDRESS | E-MAIL ADDRESS: | |
| CITY, STATE, ZIP CODE | INSURER(S) AFFORDING COVERAGE | NAIC# |
| | INSURER A: | |
| INSURED | INSURER B: | |
| INSURED NAME | INSURER C: | |
| INSURED ADDRESS | INSURER D: | |
| INSURED CITY, STATE, ZIP CODE | INSURER E : | |
| | INSURER F: | |

COVERAGES CERTIFICATE NUMBER: REVISION NUMBER:

THIS IS TO CERTIFY THAT THE POLICIES OF INSURANCE LISTED BELOW HAVE BEEN ISSUED TO THE INSURED NAMED ABOVE FOR THE POLICY PERIOD INDICATED. NOTWITHSTANDING ANY REQUIREMENT, TERM OR CONDITION OF ANY CONTRACT OR OTHER DOCUMENT WITH RESPECT TO WHICH THIS CERTIFICATE MAY BE ISSUED OR MAY PERTAIN, THE INSURANCE AFFORDED BY THE POLICIES DESCRIBED HEREIN IS SUBJECT TO ALL THE TERMS, EXCLUSIONS AND CONDITIONS OF SUCH POLICIES. LIMITS SHOWN MAY HAVE BEEN REDUCED BY PAID CLAIMS.

| | ACCUSIONS AND CONDITIONS OF SUCH POLICIES. LIMITS SHOWN MAY HAVE BEEN REDUCED BY PAID CLAIMS. ADDLISURE POLICY EFF POLICY EXP | | | | | | | |
|-------------|--|---|-----|----------------------------|--------------|------------------------------|---|------------------------|
| INSR LTR | TR TYPE OF INSURANCE | | WVD | POLICY NUMBER | (MM/DD/YYYY) | (MM/DD/YYYY) | LIMIT | S |
| | CLAIMS-MADE X OCCUR | | | | | | EACH OCCURRENCE DAMAGE TO RENTED PREMISES (Ea occurrence) | \$ 1,000,000 \$ |
| | X POLLUTION LIABILITY | | | | | | MED EXP (Any one person) | \$ |
| Α | | Υ | | POLICY NUMBER | 01/01/2019 | 01/01/2020 | PERSONAL & ADV INJURY | \$ 1,000,000 |
| | GEN'L AGGREGATE LIMIT APPLIES PER: | | | | | | GENERAL AGGREGATE | \$ 2,000,000 |
| | POLICY X PRO- JECT LOC | | | | | | PRODUCTS - COMP/OP AGG | \$ 2,000,000 |
| | OTHER: | | | | | | | \$ |
| | AUTOMOBILE LIABILITY | | | | | | COMBINED SINGLE LIMIT (Ea accident) | \$ 1,000,000 |
| | ANY AUTO | | | | | | BODILY INJURY (Per person) | \$ |
| В | X OWNED SCHEDULED AUTOS | | | POLICY NUMBER 01/01/2019 0 | 01/01/2020 | BODILY INJURY (Per accident) | \$ | |
| | X HIRED X NON-OWNED AUTOS ONLY | | | | | | PROPERTY DAMAGE (Per accident) | \$ |
| | | | | | | | | \$ |
| | UMBRELLA LIAB OCCUR | | | | | | EACH OCCURRENCE | \$ |
| | EXCESS LIAB CLAIMS-MADE | | | | | | AGGREGATE | \$ |
| | DED RETENTION \$ | | | | | | | \$ |
| | WORKERS COMPENSATION AND EMPLOYERS' LIABILITY | | | | | | X PER OTH- | |
| n | D ANYPROPRIETOR/PARTNER/EXECUTIVE OFFICER/MEMBER EXCLUDED? (Mandatory in NH) | | Υ | POLICY NUMBER | 01/01/2019 | 01/01/2020 | E.L. EACH ACCIDENT | \$ 100,000 |
| | | | ' | 1 OLIO I NOMBER | 01/01/2013 | 0 1/0 1/2020 | E.L. DISEASE - EA EMPLOYEE | \$ 100,000 |
| | If yes, describe under DESCRIPTION OF OPERATIONS below | | | | | | E.L. DISEASE - POLICY LIMIT | \$ 500,000 |
| Е | PROFESSIONAL LIABILITY | | Y | POLICY NUMBER | 01/01/2019 | 01/01/2020 | Each Occurrence Aggregate | 2,000,000 2,000,000 |

DESCRIPTION OF OPERATIONS / LOCATIONS / VEHICLES (ACORD 101, Additional Remarks Schedule, may be attached if more space is required)

The Regents of the University of Colorado, a Body Corporate are named as Additional Insured as respects General, Pollution and Automobile Liability policies.

The Automobile, Workers Compensation and Professional Liability policies are endorsed to include a Waiver of Subrogation in favor of The Regents of the University of Colorado, a Body Corporate.

| CERTIFICATE HOLDER | CANCELLATION |
|---|--|
| The Regents of the University of Colorado Attn: Project Management | SHOULD ANY OF THE ABOVE DESCRIBED POLICIES BE CANCELLED BEFORE THE EXPIRATION DATE THEREOF, NOTICE WILL BE DELIVERED IN ACCORDANCE WITH THE POLICY PROVISIONS. |
| 1945 North Wheeling Street, Campus Mail stop F-418 Aurora, CO 80045 | AUTHORIZED REPRESENTATIVE Authorized Representative Signature |

© 1988-2015 ACORD CORPORATION. All rights reserved.

ACORD 25 (2016/03)

The ACORD name and logo are registered marks of ACORD

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT D.3 CONSTRUCTION MANAGER'S DIRECT LABOR BURDEN CALCULATION

EXAMPLE BURDEN WORKSEET

COMPANY NAME DATE

| | | | | Fringe Bene | efits (\$/hr) | | | | | | Burden | (\$/hr) | | | |
|---------------|------------------------|----------|---------|-------------|---------------|----------|------------|---------|---------|----------|--------|----------|--------|------|-----------|
| | | | Base | | Health/ | | | | | Small | | | | | FULLY |
| | | Salary/H | Hourly | 401k/Pensi | Dental/ | Vacation | | | Union | Tools & | FUTA/ | Occupati | Wokers | | BURDEN |
| Employee Name | Job Title | ourly | Rate | on | Vision/Life | Leave | Sick Leave | Vehicle | Package | Supplies | SUTA | onal Tax | Comp | FICA | RATE |
| Employee #1 | Project Manager | Salary | \$35.50 | 0.55 | 4.69 | 1.76 | 0.25 | 1.99 | 0.00 | 0.00 | 0.01 | 0.01 | 0.01 | 0.0 | 1 \$44.74 |
| Employee #2 | General Superintendent | Salary | \$35.50 | 0.55 | 4.69 | 1.76 | 0.25 | 1.99 | 0.00 | 0.02 | 0.01 | 0.01 | 0.01 | 0.0 | 1 \$44.76 |
| Employee #3 | Project Engineer | Salary | \$30.00 | 0.55 | 4.69 | 1.76 | 0.25 | 0.00 | 0.00 | 0.00 | 0.01 | 0.01 | 0.01 | 0.0 | 1 \$37.25 |
| Employee #4 | Intern | Hourly | \$15.00 | 0.55 | 4.69 | 1.76 | 0.25 | 0.00 | 0.00 | 0.00 | 0.01 | 0.01 | 0.01 | 0.0 | 1 \$22.25 |
| Employee #5 | Carpenter | Hourly | \$25.00 | 0.00 | 0.00 | 0.00 | 0.00 | 0.00 | 5.99 | 0.75 | 0.01 | 0.01 | 0.01 | 0.0 | 1 \$31.78 |
| | | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | | |

THE (FICTIOUS) INFORMATION ABOVE IS AN EXAMPLE OF THE LEVEL OF DETAIL REQUIRED TO BE SUBMITTED AS STAFFING COST DETAIL FOR CM/GC AND SUBCONTRACTORS. ALL EMPLOYEES INCLUDED IN CM/GC PAY APPLICATIONS MUST BE INCLUDED. ADD COLUMNS AS NECESSARY FOR MISSING CATEGORIES.

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT E BUILDING CODE COMPLIANCE POLICY

BUILDING CODE COMPLIANCE POLICY: COORDINATION OF APPROVED BUILDING CODES, PLAN REVIEWS AND BUILDING INSPECTIONS

Refer to the Office of the State Architect State Buildings Building Codes Webpage for:

Building Code Compliance Policy (Rev. 7/2018); and

Exhibit A of the *Building Codes* dated (**Rev. 8/2023**), including the Amendment to Chapter 1 of the International Building Code

The State Architect Office's Building Codes Webpage is available at:

https://osa.colorado.gov/state-buildings/building-codes

The CU Denver | Anschutz Guidelines and Standards for Design and Construction Projects

https://www.cuanschutz.edu/offices/facilities-management/construction-projects/guidelines-and-standards

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT F LIST OF PRE-QUALIFIED SUBCONTRACTORS (BASED ON SCHEMATIC DESIGN DOCUMENTS)

Ex F

SC-6.5

Rev. 03/2023

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT G SCHEMATIC DESIGN ESTIMATE SUMMARY AND UPDATED SUMMARIES

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT H AMENDMENT (INCORPORATING GMP) EXHIBITS BASED ON DESIGN DEVELOPMENT DOCUMENTS

| Exhibit H.1 | GMP Documents, Drawings and Specifications including Addenda and Modifications (When approved by the Principal Representative) |
|-------------|--|
| Exhibit H.2 | Schedule of Bid Package Descriptions and Issuance Dates (as applicable) |
| Exhibit H.3 | Schedule of Values (prepared at the time of the GMP Amendment) |
| Exhibit H.4 | Allowance Schedule (prepared at the time of the GMP Amendment) |
| Exhibit H.5 | Detailed Critical Path Method Construction Schedule (Prepared at the time of the GMP Amendment) |

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT H.1 GMP Documents, Drawings and Specifications including Addenda and Modifications

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT H.2 Schedule of Bid Package Descriptions and Issuance Dates (as applicable)

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT H.3 Schedule of Values (prepared at the time of the GMP Amendment)

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT H.4 Allowance Schedule (prepared at the time of the GMP Amendment)

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT H.5 Detailed Critical Path Method Construction Schedule (Prepared at the time of the GMP Amendment)

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT I SUBSEQUENT AMENDMENTS (INCORPORATING BID PACKAGES) EXHIBITS

| Exhibit I.1 | Drawings, Specifications and Addenda |
|--------------|--|
| Exhibit I.2 | Schedule of Values (Consistent with GMP Schedule of Values) |
| Exhibit I.3 | Labor Overhead (SBP-6.18) for each Subcontractor to be applied to all change orders and amendments) |
| Exhibit I.4 | Allowance Schedule (consistent with GMP Allowance Schedule) |
| Exhibit I.5 | Performance Bond (Form SC-6.22) |
| Exhibit I.6 | Labor and Material Payment Bond (Form SC-6.221) |
| Exhibit I.7 | Insurance Certificates |
| Exhibit I.8 | Detailed Critical Path Method Construction Schedules (when approved by the Principal Representative) |
| Exhibit I.9 | Applicable Prevailing Wage Determinations (If applicable) |
| Exhibit I.10 | Apprenticeship Utilization Certifications (SBP-2.1) (If applicable) |

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT I.1 Drawings, Specifications and Addenda(When approved by the Principal Representative)

SC-6.5

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT I.2 Schedule of Values (Consistent with GMP Schedule of Values)

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT I.3 Labor Overhead (SBP-6.18) for each Subcontractor to be applied to all change orders and amendments)



DIRECT LABOR BURDEN CALCULATION

| Institution/Agency: | University of Colorado Anschutz Medical Campus |
|---------------------|--|
| Project No./Name: | 1 |
| | |

This form is required to be submitted for review prior to execution of a construction agreement.

List items below by the percentage of what makes up the total labor burden; Items include benefits that a contractor pays to employees on their payroll. Examples include taxes, pension cost, health and dental insurance etc. The Labor Burden amount must be agreed to by both the contractor and Principal Representative and will be included in the contract as part of Exhibit A and will be used in the calculation of any future Change Order Proposals (SC-6.312) Line 2.

Major sub-contractors defined as electricians, plumbers, mechanical contractors, excavators, millwork, concrete, masons etc. Please provide one (1) Labor Burden Calculation Sheet per contractor and for each sub-contractor. These labor burdens shall be used in the calculation of any future Change Order Proposals (SC-6.312) Line 10.

State reserves the right to require back-up confirmation of all information included in this calculation.

| Contractor/Subcontractor | Name: | |
|--------------------------|------------------------|--------------|
| | Percent of Salary Paid | |
| Payroll Taxes | | |
| Pension Costs | | |
| Health Insurance | | |
| Dental Insurance | | |
| Life Insurance | | |
| Other (Specify) | | Description: |
| Other (Specify) | | Description: |
| | | |
| Total Labor Burden: | 0% | |

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT I.4 Allowance Schedule (consistent with GMP Allowance Schedule)

SC-6.5

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT I.5 Performance Bond (Form SC-6.22)

COLORADO PERFORMANCE BOND

| | ·// (1102 50115 | | | |
|---|----------------------------|---------------------|-----------------------|--------------|
| Institution/Agency: University | of Colorado Anschutz Me | dical Campus | | |
| Project No./Name: / | | | | |
| | | | | |
| | | | | |
| BONDING COMPANY: | DO NOT MAKE ANY CHAI | NGES TO THE LAI | NGUAGE IN THIS BOI | ND. |
| | | | | |
| KNOW ALL PERSONS BY TH | ESE PRESENTS: | | | |
| That the Contractor | | | | |
| | | | | |
| as Principal and hereinafter call | ed "Principal," | | | |
| and | | | | |
| as Surety and hereinafter called | "Surety," a corporation or | ganized and exis | ting under the laws o | f |
| | are held and firmly bour | nd unto the STA | TE OF COLORADO | acting by |
| and through the Institution/Ager sum of: | cy identified above herein | after called the "F | Principal Representat | ive", in the |
| | | Dollars (\$ | |) |
| (Written Amount) | | <u>(N</u> | lumerical Amount) | <u> </u> |
| for the payment whereof the Pr successors and assigns, jointly | | | eirs, executors, adm | inistrators, |
| WHEREAS, the Principal and the have entered into a certain Con | | | h the Principal Repr | esentative |
| (Leave blank, to be completed by | / Institution/Agency) | for the construc | tion of a PROJECT | |
| | | | | |

State Form SC-6.22 (Rev. 7/2022)

identified above, which Contract is hereby by reference made a part hereof;

NOW, THEREFORE, THE CONDITION OF THIS OBLIGATION, is such that, if the Principal shall promptly, fully and faithfully perform all the undertakings, covenants, terms, conditions and agreements of said Contract during the original term of said Contract any extensions thereof that may be granted by the Principal Representative with or without notice to the Surety, and during the life of any guaranty required under the Contract, and shall also well and truly perform and fulfill all undertakings, covenants, terms, conditions and agreements of any and all duly authorized modifications of said Contract that may hereafter be made, notice of which modifications to the Surety being hereby waived, then this obligation shall be null and void; otherwise it shall remain in full force and effect.

AND THE SAID SURETY, for value received hereby stipulates and agrees that whenever the Principal shall be, and declared by the Principal Representative to be in default under said Contract, the State of Colorado having performed its obligations thereunder, the Surety may promptly remedy the default or shall promptly (1) Complete the Contract in accordance with its terms and conditions, or (2) Obtain a bid or bids for submittal to the Principal Representative for completing the Contract in accordance with its terms and conditions, and upon determination by the Principal Representative and Surety of the lowest responsible bidder, arrange for a contract between such bidder and the State of Colorado acting by and through the Principal Representative and make available as work progresses (even though there should be a default or a succession of defaults under the contract or contracts of completion arranged under this paragraph) sufficient funds to pay the cost of completion, less the balance of the contract price but not exceeding, including other costs and damages for which the Surety may be liable hereunder, the amount hereinbefore set forth. The term "balance of the contract price" as herein used shall mean the total amount payable to the Principal under the Contract and any amendments thereto, less the amount properly paid by the State of Colorado to the Contractor.

No right of action shall accrue on this bond to or for the use of any person or corporation other than the State of Colorado.

IN WITNESS WHEREOF said Principal and Surety have executed this Bond, on

| (If left blank, the Institution/Agency will date this bond to match the C | ontract date) |
|---|---------------------|
| (Corporate Seal) | THE PRINCIPAL |
| ATTEST: | By: |
| Secretary | |
| (Corporate Seal) | SURETY |
| | By:Attorney-in-fact |

THIS BOND MUST BE ACCOMPANIED BY POWER OF ATTORNEY, EFFECTIVELY DATED

Note: This bond is issued simultaneously with another bond conditioned for the full and faithful payment for all labor and material of the contract.

State Form SC-6.22 (Rev. 7/2022)

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT I.6 Labor and Material Payment Bond (Form SC-6.221)

COLORADO LABOR AND MATERIAL BOND

| Institution/Agency: University of Colorado Anschutz | z Medical Campus |
|---|---|
| Project No./Name: / | |
| | |
| BONDING COMPANY: DO NOT MAKE AN | Y CHANGES TO THE LANGUAGE IN THIS |
| | |
| KNOW ALL PERSONS BY THESE PRESENTS: | |
| That the Contractor | |
| | |
| as Principal and hereinafter called "Principal," | |
| | |
| and | |
| | |
| as Surety and hereinafter called "Surety," a corporation | on organized and existing under the laws of |
| are held and firmly and through the Institution/Agency identified above he subcontractors and any others who have supplied or machinery, tools, or equipment actually used in the pwho have performed or shall perform labor in the phereinafter called "Obligees" in the sum of: | furnished or shall supply or furnish materials, rental performance of the hereinafter identified Contract, or |
| | Dollars (\$) (Numerical Amount) |
| (Written Amount) | (Numerical Amount) |
| together with interest at the rate of eight per cent (accordance with said Contract, from the time such paymade, for the payment of which, well and truly made themselves, their heirs, executors, administrators, such these presents. | yments shall become due until such payment shall be e to the Obligees, the Principal and the Surety bind |
| WHEREAS, the Principal and the State of Colorado have entered into a certain Contract, hereinafter calle | |
| | for the construction of a PROJECT |
| (Leave blank, to be completed by Institution/Agency) | |
| identified above, which Contract is hereby by reference | ce made a part hereof; |

SC-6.221 (Rev. 7/2022 R1) NOW, THEREFORE, THE CONDITION OF THIS OBLIGATION is such that if the Principal and the Surety shall fully indemnify and save harmless the State of Colorado and the Principal Representative from and against any and all costs and damages, including patent infringements, which either may suffer by reason of any failure or failures of the Principal promptly and faithfully to perform all terms and conditions of said Contract and shall fully reimburse and repay the State of Colorado and the Principal Representative all outlay and expense which the State of Colorado and the Principal Representative may incur in making good any such failure or failures, and further, if the Principal and his subcontractors shall duly and promptly pay for any and all labor, materials, team hire, sustenance, provisions, provender, rental machinery, tools, or equipment and other supplies which have been or shall be used or consumed by said Principal or his subcontractors in the performance of the work of said Contract, and it said Principal shall duly and promptly pay all his subcontractors the sums due them for any and all materials, rental machinery, tools, or equipment and labor that have been or shall be furnished, supplied, performed or used in connection with performance of said Contract, and shall also fully indemnify and save harmless the State of Colorado and the Principal Representative to the extent of any and all expenditures which either or both of them may be required to make by reason of any failures or defaults by the Principal or any subcontractor in connection with such payments: then this obligation shall be null and void, otherwise it shall remain in full force and effect.

It is expressly understood and agreed that any alterations which may be made in the terms of said Contract or in the work to be done under said Contract, or any extension(s) of time for the performance of the Contract, or any forbearance on the part of either the State of Colorado or the Principal to any of the others, shall not in any way release the Principal and the Surety, or either of them, their heirs, executors, administrators, successors or assigns from their liability hereunder, notice to the Surety of any such alteration, extension or forbearance being hereby waived.

IN WITNESS WHEREOF, the Principal and the Surety have executed this Bond, , on

| (If left blank, the Institution/Agency will date this bond to m | natch the Contract date) | |
|---|--------------------------|--|
| (Corporate Seal) | THE PRINCIPAL | |
| | | |
| ATTEST: | | |
| | Ву: | |
| | Title: | |
| Secretary | | |
| (Corporate Seal) | SURETY | |
| | | |
| | | |
| | Ву: | |
| | Attorney-in-fact | |

THIS BOND MUST BE ACCOMPANIED BY POWER OF ATTORNEY, EFFECTIVELY DATED

Note: This bond is issued simultaneously with another bond conditioned for the full and faithful performance of the contract.

SC-6.221 Page 2 of 2

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT I.7 Insurance Certificates

After signing this document, you must supply your Certificates of Insurance for review before the University of Colorado Anschutz Medical Campus can continue processing this agreement.

Please email your insurance certificates to:

Elaine Rydberg, elaine.rydberg@cuanschutz.edu and

Robert Holzwarth, robert.holzwarth@cuanschutz.edu

Vendor Initial



CERTIFICATE OF LIABILITY INSURANCE

DATE (MM/DD/YYYY)
CURRENT DATE

THIS CERTIFICATE IS ISSUED AS A MATTER OF INFORMATION ONLY AND CONFERS NO RIGHTS UPON THE CERTIFICATE HOLDER. THIS CERTIFICATE DOES NOT AFFIRMATIVELY OR NEGATIVELY AMEND, EXTEND OR ALTER THE COVERAGE AFFORDED BY THE POLICIES BELOW. THIS CERTIFICATE OF INSURANCE DOES NOT CONSTITUTE A CONTRACT BETWEEN THE ISSUING INSURER(S), AUTHORIZED REPRESENTATIVE OR PRODUCER, AND THE CERTIFICATE HOLDER.

IMPORTANT: If the certificate holder is an ADDITIONAL INSURED, the policy(ies) must have ADDITIONAL INSURED provisions or be endorsed. If SUBROGATION IS WAIVED, subject to the terms and conditions of the policy, certain policies may require an endorsement. A statement on this certificate does not confer rights to the certificate holder in lieu of such endorsement(s).

| PRODUCER | CONTACT | | | | |
|-------------------------------|-------------------------------|-------|--|--|--|
| | NAME: PHONE FAX | | | | |
| COMPANY | (A/C, No, Ext): | | | | |
| ADDRESS | E-MAIL ADDRESS: | | | | |
| CITY, STATE, ZIP CODE | INSURER(S) AFFORDING COVERAGE | NAIC# | | | |
| | INSURER A: | | | | |
| INSURED | INSURER B: | | | | |
| INSURED NAME | INSURER C: | | | | |
| INSURED ADDRESS | INSURER D: | | | | |
| INSURED CITY, STATE, ZIP CODE | INSURER E: | | | | |
| | INSURER F: | | | | |
| | DEL/GLOUD MURED | | | | |

COVERAGES CERTIFICATE NUMBER: REVISION NUMBER:

THIS IS TO CERTIFY THAT THE POLICIES OF INSURANCE LISTED BELOW HAVE BEEN ISSUED TO THE INSURED NAMED ABOVE FOR THE POLICY PERIOD INDICATED. NOTWITHSTANDING ANY REQUIREMENT, TERM OR CONDITION OF ANY CONTRACT OR OTHER DOCUMENT WITH RESPECT TO WHICH THIS CERTIFICATE MAY BE ISSUED OR MAY PERTAIN, THE INSURANCE AFFORDED BY THE POLICIES DESCRIBED HEREIN IS SUBJECT TO ALL THE TERMS, EXCLUSIONS AND CONDITIONS OF SUCH POLICIES. LIMITS SHOWN MAY HAVE BEEN REDUCED BY PAID CLAIMS.

| NSR LTR | | TYPE OF INSURANCE | | SUBR WVD | POLICY NUMBER | POLICY EFF (MM/DD/YYYY) | POLICY EXP (MM/DD/YYYY) | LIMIT | S |
|------------|---|--|---|-------------|---------------|----------------------------|-----------------------------------|---|--------------------|
| A | X | CLAIMS-MADE X OCCUR | Y | .,,, | POLICY NUMBER | 01/01/2019 | 01/01/2020 | EACH OCCURRENCE DAMAGE TO RENTED PREMISES (Ea occurrence) | \$ 1,000,000 \$ |
| | \times | POLLUTION LIABILITY | | | | | | MED EXP (Any one person) | \$ |
| | | | | | | | | PERSONAL & ADV INJURY | \$ 1,000,000 |
| | GEN | I'L AGGREGATE LIMIT APPLIES PER: | | | | | | GENERAL AGGREGATE | \$ 2,000,000 |
| | | POLICY X PRO- | | | | | | PRODUCTS - COMP/OP AGG | \$ 2,000,000 |
| | | OTHER: | | | | | | | \$ |
| | AUT | AUTOMOBILE LIABILITY | | | | | | COMBINED SINGLE LIMIT (Ea accident) | \$ 1,000,000 |
| В | | ANY AUTO | | | POLICY NUMBER | 01/01/2019 | 01/01/2020 | BODILY INJURY (Per person) | \$ |
| | X | OWNED SCHEDULED AUTOS ONLY | Υ | Υ | | | | BODILY INJURY (Per accident) | \$ |
| | HIRED NON-OWNED AUTOS ONLY | | | | | | PROPERTY DAMAGE (Per accident) | \$ | |
| | | | | | | | | | \$ |
| | UMBRELLA LIAB OCCUR | | | | | | | EACH OCCURRENCE | \$ |
| | | EXCESS LIAB CLAIMS-MADE | | | | | | AGGREGATE | \$ |
| | | DED RETENTION\$ | | | | | | | \$ |
| | WORKERS COMPENSATION AND EMPLOYERS' LIABILITY | | | | | 01/01/2019 | 01/01/2020 | X PER OTH- STATUTE ER | |
| D | ANYF | ANYPROPRIETOR/PARTNER/EXECUTIVE OFFICER/MEMBER EXCLUDED? (Mandatory in NH) | | Υ | POLICY NUMBER | | | E.L. EACH ACCIDENT | \$ 100,000 |
| | (Man | | | ' | | | | E.L. DISEASE - EA EMPLOYEE | \$ 100,000 |
| | If yes | s, describe under CRIPTION OF OPERATIONS below | | | | | | E.L. DISEASE - POLICY LIMIT | \$ 500,000 |
| Е | PF | ROFESSIONAL LIABILITY | | Υ | POLICY NUMBER | 01/01/2019 | 01/01/2020 | Each Occurrence | 2,000,000 |
| | | | | | | | | Aggregate | 2,000,000 |

DESCRIPTION OF OPERATIONS / LOCATIONS / VEHICLES (ACORD 101, Additional Remarks Schedule, may be attached if more space is required)

The Regents of the University of Colorado, a Body Corporate are named as Additional Insured as respects General, Pollution and Automobile Liability policies.

The Automobile, Workers Compensation and Professional Liability policies are endorsed to include a Waiver of Subrogation in favor of The Regents of the University of Colorado, a Body Corporate.

CERTIFICATE HOLDER CANCELLATION

The Regents of the University of Colorado Attn: Project Management 1945 North Wheeling Street, Campus Mail s

Authorized Representative

Authorized Representative Signature

1945 North Wheeling Street, Campus Mail stop F-418
Aurora. CO 80045

© 1988-2015 ACORD CORPORATION. All rights reserved.

SHOULD ANY OF THE ABOVE DESCRIBED POLICIES BE CANCELLED BEFORE THE EXPIRATION DATE THEREOF, NOTICE WILL BE DELIVERED IN ACCORDANCE WITH THE POLICY PROVISIONS.

ACORD 25 (2016/03)

The ACORD name and logo are registered marks of ACORD

SC-6.5

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT I.8 Detailed Critical Path Method Construction Schedules (when approved by the Principal Representative)

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT I.9 Applicable Prevailing Wage Determinations (If applicable)

ADD WAGE DETERMINATIONS FROM RFQ FOR PREVAILING WAGE PROJECTS.

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT I.10 Apprenticeship Utilization Certifications (SBP-2.1) (If applicable)

SC-6.5



APPRENTICESHIP UTILIZATION CERTIFICATION (Public Projects of \$1 million or more)

| Institution/Agency: | University of Colorac | do Anschutz Medica | al Campus |
|---|---|--|---|
| Project No./Name: | | | |
| General Contractor: | | | |
| | low attach documentation Statement below. | on that all firms ide | entified participate in apprenticeship programs as |
| TRADE | SUBCONTRA | CTOR | UNION CERTIFICATION |
| Mechanical | | | |
| Sheet Metal | | | |
| Fire Suppression | | | |
| Sprinkler Fitting | | | |
| Plumbing | | | |
| Electrical | | | |
| Employment and Training and have a proven record past five years. The Gene office of apprenticeship vertice above documentation from when it is submitted. The General Contractor apprenticeship training proposed to terminate the subcontract the discretion of the incontractor pursuant to § 20 | Administration or state ap of graduating apprentices ral Contractor shall supply erifying the certification. In shall be made publicly available shall agree to provide a rograms relating to the requestation or willfully misrepactor contract immediately | prenticeship councils at a minimum of fifts supporting documental dilable by the contract dilable by the contract uirements above. If a presented their qualified and the subcontract of the of Personnel, the pursue any other results and the results of the pursue any other results and the subcontract of the pursue any other results and the subcontract of the pursue any other results and the subcontract of the s | gistered with the United States Department of Labor's recognized by the United States Department of Labor een percent of its apprentices for at least three of the nation from the United States Department of Labor's eting agency through its website within thirty (30) days eation to the contracting agency regarding affected a contracting agency determines that a subcontractor fications, the agency shall direct the General Contractor or will be immediately removed from the public project. State may initiate the process to debar the General emedy provided by law. |
| Full Legal Name | | | |
| BY: | | Owner | |

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT J Notice to Proceed to Commence Construction Phase (Form SC-7.26)



COLORADO NOTICE TO PROCEED TO COMMENCE CONSTRUCTION PHASE (CM/GC CONTRACT)

| , | |
|--|---|
| Date of Notice: | |
| Date to be inserted by the Principal Representative Amendment No./Date: | |
| Bid Package(s) No.: | |
| Institution/Agency: University of Colorado Anschutz Medical Campus | |
| Project No./Name: / | |
| · | |
| | |
| | |
| Attach Notice of Code Compliance from Code Review Agent/Building Official for Documents Listed Above | Э |
| | |
| | |
| To: , Owner | |
| | |
| | |
| | |
| | |
| This is to advise you that your Performance Bond, Labor and Material Payment Bond, Insurance Policy and | |
| Certificates of Insurance have been received. Our issuance of this Notice does not relieve you of | |
| responsibility to assure that the bond and insurance requirements of the Contract Documents are met for the duration of the Agreement. The Amendment # for the above described work has been fully | |
| executed. | |
| | |
| You are hereby authorized and directed to proceed within ten (10) days from date of this Authorization as | |
| required in the Agreement. Any liquidated damages for failure to achieve Substantial Completion by the | |
| date agreed that may be applicable to this contract will be calculated using the date of this Notice for the | |
| date of the commencement of the Work. | |
| | |
| | |
| By By | |
| State Buildings Program Date Principal Representative Date | |
| (or Authorized Delegate) (Institution or Agency) | |
| | |
| | |
| | |
| When completely executed, this form is to the Construction Manager by the Principal Representative. | |
| | |
| | |
| | |
| | |
| | |

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT M SUPPLEMENTARY GENERAL CONDITIONS: FEDERAL PROVISIONS

IF APPLICABLE TO THIS PROJECT, THE FOLLOWING EXHIBITS WILL BE INCLUDED.

Supplementary General Conditions Federal Provisions

SLFRF Federal Funds: Contractor Terms and Conditions Certification

SLFRF Federal Funds: Contractor Terms and Conditions

SC-6.5 Rev. 03/2023

Ex M

SUPPLEMENTARY GENERAL CONDITIONS: FEDERAL PROVISIONS

1. APPLICABILITY OF PROVISIONS.

- 1.1. The Contract to which these Federal Provisions are attached has been funded, in whole or in part, with an Award of Federal funds. In the event of a conflict between the provisions of these Federal Provisions, the Special Provisions, the body of the Contract, or any attachments or exhibits incorporated into and made a part of the Contract, the provisions of these Federal Provisions shall control.
- 1.1. The State of Colorado is accountable to Treasury for oversight of their subrecipients, including ensuring their subrecipients comply with the SLFRF statute, SLFRF Award Terms and Conditions, Treasury's Final Rule, and reporting requirements, as applicable.
- 1.2. Additionally, any subrecipient that issues a subaward to another entity (2nd tier subrecipient), must hold the 2nd tier subrecipient accountable to these provisions and adhere to reporting requirements.
- 1.3. These Federal Provisions are subject to the Award as defined in §2 of these Federal Provisions, as may be revised pursuant to ongoing guidance from the relevant Federal or State of Colorado agency or institutions of higher education.

2. DEFINITIONS.

- 2.1. For the purposes of these Federal Provisions, the following terms shall have the meanings ascribed to them below.
- 2.1.1. "Award" means an award of Federal financial assistance, and the Contract setting forth the terms and conditions of that financial assistance that a non-Federal Entity receives or administers.
- 2.1.1.1. Awards may be in the form of:
- 2.1.1.1.1. Grants;
- 2.1.1.1.2. Contracts;
- 2.1.1.1.3. Cooperative Contracts, which do not include cooperative research and development Contracts (CRDA) pursuant to the Federal Technology Transfer Act of 1986, as amended (15 U.S.C. 3710);
- 2.1.1.1.4. Loans;
- 2.1.1.1.5. Loan Guarantees;
- 2.1.1.1.6. Subsidies;
- 2.1.1.1.7. Insurance:
- 2.1.1.1.8. Food commodities;
- 2.1.1.1.9. Direct appropriations;
- 2.1.1.1.10. Assessed and voluntary contributions; and
- 2.1.1.11. Other financial assistance transactions that authorize the expenditure of Federal funds by non-Federal Entities.
- 2.1.1.12. Any other items specified by OMB in policy memoranda available at the OMB website or other source posted by the OMB.
- 2.1.1.2. Award does not include:
- 2.1.1.2.1. Technical assistance, which provides services in lieu of money;
- 2.1.1.2.2. A transfer of title to Federally-owned property provided in lieu of money; even if the award is called a grant;

- 2.1.1.2.3. Any award classified for security purposes; or
- 2.1.1.2.4. Any award funded in whole or in part with Recovery funds, as defined in section 1512 of the American Recovery and Reinvestment Act (ARRA) of 2009 (Public Law 111-5).
- 2.1.2. "Contract" means the Contract to which these Federal Provisions are attached and includes all Award types in §2.1.1.1 of this Exhibit.
- 2.1.3. "Contractor" means a non-Federal Entity (or a Federal agency under an Agreement to a non-Federal Entity) receiving Federal funds through a Prime Recipient to support the performance of the Federal project or program for which the Federal funds were awarded. A Contractor is subject to the terms and conditions of the Federal Award to the Prime Recipient, including program compliance requirements. The term "Contractor" includes and may be referred to as "Subcontractor". The term does not include an individual who is a beneficiary of a federal program.
- 2.1.4. "Data Universal Numbering System (DUNS) Number" means the nine-digit number established and assigned by Dun and Bradstreet, Inc. to uniquely identify a business entity. Dun and Bradstreet's website may be found at: http://fedgov.dnb.com/webform.
- 2.1.5. "Entity" means all of the following as defined at 2 CFR part 25, subpart C; 2.1.5.1. A governmental organization, which is a State, local government, or Indian Tribe; 2.1.5.2. A foreign public entity;
- 2.1.5.3. A domestic or foreign non-profit organization;
- 2.1.5.4. A domestic or foreign for-profit organization; and
- 2.1.5.5. A Federal agency, but only a Subrecipient under an Award or Subaward to a non-Federal entity.
- 2.1.6. "Executive" means an officer, managing partner or any other employee in a management position.
- 2.1.7. "Federal Award Identification Number (FAIN)" means an Award number assigned by a Federal agency to a Prime Recipient.
- 2.1.8. "Federal Awarding Agency" means a Federal agency providing a Federal Award to a Recipient as described in 2 CFR §200.37
- 2.1.9. "FFATA" means the Federal Funding Accountability and Transparency Act of 2006 (Public Law 109-282), as amended by §6202 of Public Law 110-252. FFATA, as amended, also is referred to as the "Transparency Act."
- 2.1.10. "Federal Provisions" means these Federal Provisions subject to the Transparency Act and Uniform Guidance, as may be revised pursuant to ongoing guidance from the relevant Federal or State of Colorado agency or institutions of higher education.
- 2.1.11. "OMB" means the Executive Office of the President, Office of Management and Budget.
- 2.1.12. "Prime Recipient" means a Colorado State agency or institution of higher education that receives an Award.
- 2.1.13. "Subaward" means an award by a Recipient to a Subrecipient funded in whole or in part by a Federal Award. The terms and conditions of the Federal Award flow down to the Award unless the terms and conditions of the Federal Award specifically indicate otherwise in accordance with 2 CFR §200.38. The term does not include payments to a contractor or payments to an individual that is a beneficiary of a Federal program.
- 2.1.14. Not used
- 2.1.15. "Contractor Parent UEI Number" means the subrecipient parent organization's 9-digit Universal Entity ID (UEI) number that appears in the subrecipient's System for Award Management (SAM) profile, if applicable.
- 2.1.16. "System for Award Management (SAM)" means the Federal repository into which an Entity must enter the information required under the Transparency Act, which may be found at http://www.sam.gov.
- 2.1.17. "Total Compensation" means the cash and noncash dollar value earned by an Executive during the Prime Recipient's or Subrecipient's preceding fiscal year and includes the following:
- 2.1.17.1. Salary and bonus;

- 2.1.17.2. Awards of stock, stock options, and stock appreciation rights, using the dollar amount recognized for financial statement reporting purposes with respect to the fiscal year in accordance with the Statement of Financial Accounting Standards No. 123 (Revised 2005) (FAS 123R), Shared Based Payments;
- 2.1.17.3. Earnings for services under non-equity incentive plans, not including group life, health, hospitalization or medical reimbursement plans that do not discriminate in favor of Executives and are available generally to all salaried employees;
- 2.1.17.4. Change in present value of defined benefit and actuarial pension plans; 2.1.17.5. Above-market earnings on deferred compensation which is not tax-qualified;
- 2.1.17.6. Other compensation, if the aggregate value of all such other compensation (e.g. severance, termination payments, value of life insurance paid on behalf of the employee, perquisites or property) for the Executive exceeds \$10,000.
- 2.1.18. "Transparency Act" means the Federal Funding Accountability and Transparency Act of 2006 (Public Law 109-282), as amended by §6202 of Public Law 110-252. The Transparency Act also is referred to as FFATA.
- 2.1.19. "Uniform Guidance" means the Office of Management and Budget Uniform Administrative Requirements, Cost Principles, and Audit Requirements for Federal Awards, which supersedes requirements from OMB Circulars A-21, A-87, A-110, and A-122, OMB Circulars A-89, A-102, and A-133, and the guidance in Circular A-50 on Single Audit Act follow-up. The terms and conditions of the Uniform Guidance flow down to Awards to Subrecipients unless the Uniform Guidance or the terms and conditions of the Federal Award specifically indicate otherwise.
- 2.1.20 "Unique Entity ID" means the Unique Entity ID established by the federal government for a Grantee at https://sam.gov/content/home.
- 2.1.20. "Vendor" means a dealer, distributor, merchant or other seller providing property or services required for a project or program funded by an Award. A Vendor is not a Prime Recipient or a Subrecipient and is not subject to the terms and conditions of the Federal award. Program compliance requirements do not pass through to a Vendor.

3. COMPLIANCE.

- 3.1. Contractor shall comply with all applicable provisions of the Transparency Act, all applicable provisions of the Uniform Guidance, and the regulations issued pursuant thereto, including but not limited to these Federal Provisions. Any revisions to such provisions or regulations shall automatically become a part of these Federal Provisions, without the necessity of either party executing any further instrument. The State of Colorado may provide written notification to Contractor of such revisions, but such notice shall not be a condition precedent to the effectiveness of such revisions.
- 3.2. Per US Treasury Final Award requirements, programs or services must not include terms or conditions that undermine efforts to stop COVID-19 or discourage compliance with recommendations and CDC guidelines.

4. SYSTEM FOR AWARD MANAGEMENT (SAM) AND DATA UNIVERSAL NUMBERING SYSTEM (DUNS) REQUIREMENTS.

- 4.1. SAM. Contractor shall maintain the currency of its information in SAM until the Contractor submits the final financial report required under the Award or receives final payment, whichever is later. Contractor shall review and update SAM information at least annually after the initial registration, and more frequently if required by changes in its information.
- 4.2. UEI. Grantee shall provide its Unique Entity ID to its Prime Recipient, and shall update Grantee's information in SAM.gov at least annually after the initial registration, and more frequently if required by changes in Grantee's information.

5. TOTAL COMPENSATION.

- 5.1. Contractor shall include Total Compensation in SAM for each of its five most highly compensated Executives for the preceding fiscal year if:
- 5.1.1. The total Federal funding authorized to date under the Award is \$30,000 or more; and
- 5.1.2. In the preceding fiscal year, Contractor received:
- 5.1.2.1. 80% or more of its annual gross revenues from Federal procurement contracts and subcontracts and/or Federal financial assistance Awards or Subawards subject to the Transparency Act; and
- 5.1.2.2. \$25,000,000 or more in annual gross revenues from Federal procurement contracts and subcontracts and/or Federal financial assistance Awards or Subawards subject to the Transparency Act; and
- 5.1.3. The public does not have access to information about the compensation of such Executives through periodic reports filed under section 13(a) or 15(d) of the Securities Exchange Act of 1934 (15 U.S.C. 78m(a), 78o(d) or § 6104 of the Internal Revenue Code of 1986.

6. REPORTING.

6.1. Contractor shall report data elements to SAM and to the Prime Recipient as required in this Exhibit if Contractor is a Subrecipient for the Award pursuant to the Transparency Act. No direct payment shall be made to Contractor for providing any reports required under these Federal Provisions and the cost of producing such reports shall be included in the Contract price. The reporting requirements in this Exhibit are based on guidance from the US Office of Management and Budget (OMB), and as such are subject to change at any time by OMB. Any such changes shall be automatically incorporated into this Contract and shall become part of Contractor's obligations under this Contract.

7. EFFECTIVE DATE AND DOLLAR THRESHOLD FOR REPORTING.

- 7.1. Reporting requirements in §8 below apply to new Awards as of October 1, 2010, if the initial award is \$30,000 or more. If the initial Award is below \$30,000 but subsequent Award modifications result in a total Award of \$30,000 or more, the Award is subject to the reporting requirements as of the date the Award exceeds \$30,000. If the initial Award is \$30,000 or more, but funding is subsequently de-obligated such that the total award amount falls below \$30,000, the Award shall continue to be subject to the reporting requirements. If the total award is below \$30,000 no reporting required; if more than \$30,000 and less than \$50,000 then FFATA reporting is required; and, \$50,000 and above SLFRF reporting is required.
- 7.2. The procurement standards in §9 below are applicable to new Awards made by Prime Recipient as of December 26, 2015. The standards set forth in §11 below are applicable to audits of fiscal years beginning on or after December 26, 2014.

8. SUBRECIPIENT REPORTING REQUIREMENTS.

- 8.1. If Contractor is a Subrecipient, Contractor shall report as set forth below.
- 8.1.1. To SAM. A Subrecipient shall register in SAM and report the following data elements in SAM for each Federal Award Identification Number no later than the end of the month following the month in which the Subaward was made:
- 8.1.1.1. Subrecipient UEI Number;
- 8.1.1.2. Subrecipient UEI Number + 4 if more than one electronic funds transfer (EFT) account;
- 8.1.1.3. Subrecipient Parent UEI Number;
- 8.1.1.4. Subrecipient's address, including: Street Address, City, State, Country, Zip + 4, and Congressional District;
- 8.1.1.5. Subrecipient's top 5 most highly compensated Executives if the criteria in §4 above are met; and
- 8.1.1.6. Subrecipient's Total Compensation of top 5 most highly compensated Executives if criteria in §4 above met.
- 8.1.2. To Prime Recipient. A Subrecipient shall report to its Prime Recipient, upon the effective date of the Contract, the following data elements:

- 8.1.2.1. Subrecipient's UEI Number as registered in SAM.
- 8.1.2.2. Primary Place of Performance Information, including: Street Address, City, State, Country, Zip code + 4, and Congressional District.

9. PROCUREMENT STANDARDS.

- 9.1. Procurement Procedures. A Subrecipient shall use its own documented procurement procedures which reflect applicable State, local, and Tribal laws and regulations, provided that the procurements conform to applicable Federal law and the standards identified in the Uniform Guidance, including without limitation, §§200.318 through 200.326 thereof.
- 9.2. Procurement of Recovered Materials (2 CFR 200.322). If a Subrecipient is a State Agency or an agency of a political subdivision of the State, its contractors must comply with section 6002 of the Solid Waste Disposal Act, as amended by the Resource Conservation and Recovery Act. The requirements of Section 6002 include procuring only items designated in guidelines of the Environmental Protection Agency (EPA) at 40 CFR part 247 that contain the highest percentage of recovered materials practicable, consistent with maintaining a satisfactory level of competition, where the purchase price of the item exceeds \$10,000 or the value of the quantity acquired during the preceding fiscal year exceeded \$10,000; procuring solid waste management services in a manner that maximizes energy and resource recovery; and establishing an affirmative procurement program for procurement of recovered materials identified in the EPA guidelines.
- 9.3 Domestic preference for procurements (2 CFR 200.322). As appropriate and to the extent consistent with law, the non-Federal entity should, to the greatest extent practicable under a Federal award, provide a preference for the purchase, acquisition, or use of goods, products, or materials produced in the United States (including but not limited to iron, aluminum, steel, cement, and other manufactured products). The requirements of this section must be included in all subawards including all Agreements and purchase orders for work or products under this award.

10. ACCESS TO RECORDS

10.1. A Subrecipient shall permit Recipient and auditors to have access to Subrecipient's records and financial statements as necessary for Recipient to meet the requirements of §200.331 (Requirements for pass-through entities), §§200.300 (Statutory and national policy requirements) through 200.309 (Period of performance), and Subpart F-Audit Requirements of the Uniform Guidance. 2 CFR §200.331(a)(5).

11. SINGLE AUDIT REQUIREMENTS

- 11.1. If a Subrecipient expends \$750,000 or more in Federal Awards during the Subrecipient's fiscal year, the Subrecipient shall procure or arrange for a single or program-specific audit conducted for that year in accordance with the provisions of Subpart F-Audit Requirements of the Uniform Guidance, issued pursuant to the Single Audit Act Amendments of 1996, (31 U.S.C. 7501-7507). 2 CFR §200.501.
- 11.1.1. Election. A Subrecipient shall have a single audit conducted in accordance with Uniform Guidance §200.514 (Scope of audit), except when it elects to have a program-specific audit conducted in accordance with §200.507 (Program-specific audits). The Subrecipient may elect to have a program-specific audit if Subrecipient expends Federal Awards under only one Federal program (excluding research and development) and the Federal program's statutes, regulations, or the terms and conditions of the Federal award do not require a financial statement audit of Prime Recipient. A program-specific audit may not be elected for research and development unless all of the Federal Awards expended were received from Recipient and Recipient approves in advance a program-specific audit.
- 11.1.2. Exemption. If a Subrecipient expends less than \$750,000 in Federal Awards during its fiscal year, the Subrecipient shall be exempt from Federal audit requirements for that year, except as noted in 2 CFR \$200.503 (Relation to other audit requirements), but records shall be available for review or audit by appropriate officials of the Federal agency, the State, and the Government Accountability Office.
- 11.1.3. Subrecipient Compliance Responsibility. A Subrecipient shall procure or otherwise arrange for the audit required by Part F of the Uniform Guidance and ensure it is properly performed and submitted when due in accordance with the Uniform Guidance. Subrecipient shall prepare appropriate financial statements, including the schedule of expenditures of Federal awards in accordance with Uniform Guidance §200.510 (Financial statements) and provide the auditor with access to personnel, accounts, books, records,

supporting documentation, and other information as needed for the auditor to perform the audit required by Uniform Guidance Part F-Audit Requirements.

12. CONTRACT PROVISIONS FOR CONTRACTORS AND SUBCONTRACTORS

- 12.1. Contractors shall comply with and shall include all of the following applicable provisions in all subcontracts entered into by it pursuant to this Contract and any provisions required by 2 CFR 200 Appendix II.
- 12.1.1. Equal Employment Opportunity. Except as otherwise provided under 41 CFR Part 60, all contracts that meet the definition of "federally assisted construction contract" in 41 CFR Part 60-1.3 shall include the equal opportunity clause provided under 41 CFR 60-1.4(b), in accordance with Executive Order 11246, "Equal Employment Opportunity" (30 FR 12319, 12935, 3 CFR Part, 1964-1965 Comp., p. 339), as amended by Executive Order 11375, "Amending Executive Order 11246 Relating to Equal Employment Opportunity," and implementing regulations at 41 CFR part 60, "Office of Federal Contract Compliance Programs, Equal Employment Opportunity, Department of Labor.
- 12.1.1.1. During the performance of this contract, the contractor agrees as follows:
- 12.1.1.1. Contractor will not discriminate against any employee or applicant for employment because of race, color, religion, sex, or national origin. The contractor will take affirmative action to ensure that applicants are employed, and that employees are treated during employment, without regard to their race, color, religion, sex, or national origin. Such action shall include, but not be limited to the following: Employment, upgrading, demotion, or transfer, recruitment or recruitment advertising; layoff or termination; rates of pay or other forms of compensation; and selection for training, including apprenticeship. The contractor agrees to post in conspicuous places, available to employees and applicants for employment, notices to be provided by the contracting officer setting forth the provisions of this nondiscrimination clause.
- 12.1.1.1.2. Contractor will, in all solicitations or advertisements for employees placed by or on behalf of the contractor, state that all qualified applicants will receive consideration for employment without regard to race, color, religion, sex, or national origin.
- 12.1.1.1.3. Contractor will send to each labor union or representative of workers with which he has a collective bargaining Contract or other contract or understanding, a notice to be provided by the agency contracting officer, advising the labor union or workers' representative of the contractor's commitments under section 202 of Executive Order 11246 of September 24, 1965, and shall post copies of the notice in conspicuous places available to employees and applicants for employment.
- 12.1.1.1.4. Contractor will comply with all provisions of Executive Order 11246 of September 24, 1965, and of the rules, regulations, and relevant orders of the Secretary of Labor.
- 12.1.1.1.5. Contractor will furnish all information and reports required by Executive Order 11246 of September 24, 1965, and by the rules, regulations, and orders of the Secretary of Labor, or pursuant thereto, and will permit access to his books, records, and accounts by the contracting agency and the Secretary of Labor for purposes of investigation to ascertain compliance with such rules, regulations, and orders.
- 12.1.1.1.6. In the event of Contractor's non-compliance with the nondiscrimination clauses of this contract or with any of such rules, regulations, or orders, this contract may be canceled, terminated or suspended in whole or in part and the contractor may be declared ineligible for further Government contracts in accordance with procedures authorized in Executive Order 11246 of September 24, 1965, and such other sanctions may be imposed and remedies invoked as provided in Executive Order 11246 of September 24, 1965, or by rule, regulation, or order of the Secretary of Labor, or as otherwise provided by law.
- 12.1.1.1.7. Contractor will include the provisions of paragraphs (1) through (7) in every subcontract or purchase order unless exempted by rules, regulations, or orders of the Secretary of Labor issued pursuant to section 204 of Executive Order 11246 of September 24, 1965, so that such provisions will be binding upon each subcontractor or vendor. The contractor will take such action with respect to any subcontract or purchase order as may be directed by the Secretary of Labor as a means of enforcing such provisions including sanctions for noncompliance: Provided, however, that in the event Contractor becomes involved in, or is threatened with, litigation with a subcontractor or vendor as a result of such direction, the contractor may request the United States to enter into such litigation to protect the interests of the United States."

- 12.1.2. Davis-Bacon Act. Davis-Bacon Act, as amended (40 U.S.C. 3141-3148). When required by Federal program legislation, all prime construction contracts in excess of \$2,000 awarded by non-Federal entities must include a provision for compliance with the Davis-Bacon Act (40 U.S.C. 3141-3144, and 3146-3148) as supplemented by Department of Labor regulations (29 CFR Part 5, "Labor Standards Provisions Applicable to Contracts Covering Federally Financed and Assisted Construction"). In accordance with the statute, contractors must be required to pay wages to laborers and mechanics at a rate not less than the prevailing wages specified in a wage determination made by the Secretary of Labor. In addition, contractors must be required to pay wages not less than once a week. The non-Federal entity must place a copy of the current prevailing wage determination issued by the Department of Labor in each solicitation. The decision to award a contract or subcontract must be conditioned upon the acceptance of the wage determination. The non-Federal entity must report all suspected or reported violations to the Federal awarding agency. The contracts must also include a provision for compliance with the Copeland "Anti-Kickback" Act (40 U.S.C. supplemented by Department of Labor regulations (29 CFR Part 3, "Contractors and Subcontractors on Public Building or Public Work Financed in Whole or in Part by Loans or Grants from the United States"). The Act provides that each contractor or Subrecipient must be prohibited from inducing, by any means, any person employed in the construction, completion, or repair of public work, to give up any part of the compensation to which he or she is otherwise entitled. The non-Federal entity must report all suspected or reported violations to the Federal awarding agency.
- 12.1.3. Rights to Inventions Made Under a Contract or Contract. If the Federal Award meets the definition of "funding Contract" under 37 CFR §401.2 (a) and Subrecipient wishes to enter into a contract with a small business firm or nonprofit organization regarding the substitution of parties, assignment or performance of experimental, developmental, or research work under that "funding Contract," Subrecipient must comply with the requirements of 37 CFR Part 401, "Rights to Inventions Made by Nonprofit Organizations and Small Business Firms Under Government Grants, Contracts and Cooperative Contracts," and any implementing regulations issued by the awarding agency.
- 12.1.4. Clean Air Act (42 U.S.C. 7401-7671q.) and the Federal Water Pollution Control Act (33 U.S.C. 1251-1387), as amended. Contracts and subgrants of amounts in excess of \$150,000 must contain a provision that requires the non-Federal award to agree to comply with all applicable standards, orders or regulations issued pursuant to the Clean Air Act (42 U.S.C. 7401-7671q) and the Federal Water Pollution Control Act as amended (33 U.S.C. 1251-1387). Violations must be reported to the Federal awarding agency and the Regional Office of the Environmental Protection Agency (EPA).
- 12.1.5. Debarment and Suspension (Executive Orders 12549 and 12689). A contract award (see 2 CFR 180.220) must not be made to parties listed on the government wide exclusions in the System for Award Management (SAM), in accordance with the OMB guidelines at 2 CFR 180 that implement Executive Orders 12549 (3 CFR part 1986 Comp., p. 189) and 12689 (3 CFR part 1989 Comp., p. 235), "Debarment and Suspension." SAM Exclusions contains the names of parties debarred, suspended, or otherwise excluded by agencies, as well as parties declared ineligible under statutory or regulatory authority other than Executive Order 12549.
- 12.1.6. Byrd Anti-Lobbying Amendment (31 U.S.C. 1352). Contractors that apply or bid for an award exceeding \$100,000 must file the required certification. Each tier certifies to the tier above that it will not and has not used Federal appropriated funds to pay any person or organization for influencing or attempting to influence an officer or employee of any agency, a member of Congress, officer or employee of Congress, or an employee of a member of Congress in connection with obtaining any Federal contract, grant or any other award covered by 31 U.S.C. 1352. Each tier must also disclose any lobbying with non-Federal funds that takes place in connection with obtaining any Federal award. Such disclosures are forwarded from tier to tier up to the non-Federal award.
- 12.1.7. Prohibition on certain telecommunications and video surveillance services or equipment (2 CFR 200.216). Grantee is prohibited from obligating or expending loan or grant funds on certain telecommunications and video surveillance services or equipment pursuant to 2 CFR 200.216.
- 12.1.8. Never Contract with the Enemy (2 CFR 200.215). Federal awarding agencies and recipients are subject to the regulations implementing "Never Contract with the Enemy" in 2 CFR part 183. The regulations in 2 CFR part 183 affect covered Agreements, grants and cooperative agreements that are expected to exceed \$50,000 within the period of performance, are performed outside the United States and its territories, and

- are in support of a contingency operation in which members of the Armed Forces are actively engaged in hostilities.
- 12.1.9. Title VI of the Civil Rights Act. The Subgrantee, Contractor, Subcontractor, transferee, and assignee shall comply with Title VI of the Civil Rights Act of 1964, which prohibits recipients of federal financial assistance from excluding from a program or activity, denying benefits of, or otherwise discriminating against a person on the basis of race, color, or national origin (42 U.S.C. § 2000d et seq.), as implemented by the Department of Treasury's Title VI regulations, 31 CFR Part 22, which are herein incorporated by reference and made a part of this Agreement (or agreement). Title VI also includes protection to persons with "Limited English Proficiency" in any program or activity receiving federal financial assistance, 42 U.S. C. § 2000d et seq., as implemented by the Department of the Treasury's Title VI regulations, 31 CRF Part 22, and herein incorporated by reference and made part of this Agreement or agreement.
- 12.1.10 Contract Work Hours and Safety Standards Act (40 U.S.C. 3701-3708). Where applicable, all contracts awarded by the non-Federal entity in excess of \$100,000 that involve the employment of mechanics or laborers must include a provision for compliance with 40 U.S.C. 3702 and 3704, as supplemented by Department of Labor regulations (29 CFR Part 5). Under 40 U.S.C. 3702 of the Act, each contractor must be required to compute the wages of every mechanic and laborer on the basis of a standard work week of 40 hours. Work in excess of the standard work week is permissible provided that the worker is compensated at a rate of not less than one and a half times the basic rate of pay for all hours worked in excess of 40 hours in the work week. The requirements of 40 U.S.C. 3704 are applicable to construction work and provide that no laborer or mechanic must be required to work in surroundings or under working conditions which are unsanitary, hazardous or dangerous. These requirements do not apply to the purchases of supplies or materials or articles ordinarily available on the open market, or contracts for transportation or transmission of intelligence.
- 12.1.11 Procurement of Recovered Materials (2 CFR 200.323) See Section 9.2
- 12.1.12 Domestic preference for procurements (2 CFR 200.322) See Section 9.3
- 12.1.13 Real Property Disposition (2 CFR 200.311) When real property is no longer needed for the originally authorized purpose, the non-Federal entity must obtain disposition instructions from the Federal awarding agency or pass-through entity.

13. CERTIFICATIONS.

13.1. Unless prohibited by Federal statutes or regulations, Recipient may require Contractor to submit certifications and representations required by Federal statutes or regulations on an annual basis. 2 CFR §200.208. Submission may be required more frequently if Contractor fails to meet a requirement of the Federal award. Contractor shall certify in writing to the State at the end of the Award that the project or activity was completed or the level of effort was expended. 2 CFR §200.201(3). If the required level of activity or effort was not carried out, the amount of the Award must be adjusted.

14. EXEMPTIONS.

- 14.1. These Federal Provisions do not apply to an individual who receives an Award as a natural person, unrelated to any business or non-profit organization he or she may own or operate in his or her name.
- 14.2. A Contractor with gross income from all sources of less than \$300,000 in the previous tax year is exempt from the requirements to report Subawards and the Total Compensation of its most highly compensated Executives.

15. EVENT OF DEFAULT AND TERMINATION.

- 15.1. Failure to comply with these Federal Provisions shall constitute an event of default under the Contract and the State of Colorado may terminate the Contract upon 30 days prior written notice if the default remains uncured five calendar days following the termination of the 30 day notice period. This remedy will be in addition to any other remedy available to the State of Colorado under the Contract, at law or in equity.
- 15.2. Termination (2 CFR 200.340). The Federal Award may be terminated in whole or in part as follows:
 - § By the Federal Awarding Agency or Prime Recipient, if a Contractor fails to comply with the terms and conditions of a Federal Award;

- § By the Federal awarding agency or Prime Recipient, to the greatest extent authorized by law, if an award no longer effectuates the program goals or agency priorities;
- § By the Federal awarding agency or Prime Recipient with the consent of the Contractor, in which case the two parties must agree upon the termination conditions, including the effective date and, in the case of partial termination, the portion to be terminated;
- § By the Prime Recipient upon sending to the Federal Awarding Agency or Prime Recipient written notification setting forth the reasons for such termination, the effective date, and, in the case of partial termination, the portion to be terminated. However, if the Federal Awarding Agency or Prime Recipient determines in the case of partial termination that the reduced or modified portion of the Federal Award or Contract will not accomplish the purposes for which the Federal Award was made, the Federal Awarding Agency or Prime Recipient may terminate the Federal Award in its entirety; or by the Federal Awarding Agency or Prime Recipient pursuant to termination provisions included in the Federal Award.

End of Supplementary General Conditions: Federal Provision

SLFRF FEDERAL FUNDS CONTRACTOR TERMS AND CONDITIONS

FEDERAL AWARD(S) APPLICABLE TO THIS GRANT AWARD

| Federal Awarding Office | US Department of the Treasury |
|---|---|
| Grant Program | Coronavirus State and Local Fiscal Recovery Funds |
| Assistance Listing Number | 21.027 |
| Federal Award Number | SLFRP0126 |
| Federal Award Date * | May 18, 2021 |
| Federal Award End Date | December 31, 2024 |
| Federal Statutory Authority | Title VI of the Social Security Act, Section 602 |
| Total Amount of Federal Award (this is <u>not</u> the | |
| amount of this grant agreement) | \$3,828,761,790 |
| Approved Expenditure Category | 6.1 |

^{*} Funds may not be available through the Federal Award End Date subject to the provisions in §2 and §5 below.

Section 602(b) of the Social Security Act (the Act), as added by section 9901 of the American Rescue Plan Act (ARPA), Pub. L. No. 117-2 (March 11, 2021), authorizes the Department of the Treasury (Treasury) to make payments to certain recipients from the Coronavirus State Fiscal Recovery Fund. As a condition of your organization receiving federal recovery funds from the State, the authorized representative below hereby (i) certifies that your organization will carry out the activities listed in section 602(c) of the Act and (ii) agrees to the terms attached hereto. Your organization also agrees to use the federal recovery funds as specified in bills passed by the General Assembly and signed by the Governor.

Under penalty of perjury, the undersigned official certifies that the authorized representative has read and understood the organization's obligations in the Assurances of Compliance and Civil Rights Requirements, that any information submitted in conjunction with this assurances document is accurate and complete, and that the organization is in compliance with the nondiscrimination requirements.

| Contractor Name |
|----------------------------|
| |
| Authorized Representative: |
| Authorized Representative. |
| |
| Title: |
| |
| Signature: |
| ~ 10 |

SLFRF FEDERAL FUNDS CONTRACTOR TERMS AND CONDITIONS

1. Use of Funds.

- a. Contractor understands and agrees that the funds disbursed under this award may only be used in compliance with section 602(c) of the Social Security Act (the Act) and Treasury's regulations implementing that section and guidance.
- b. Contractor will determine prior to engaging in any project using this assistance that it has the institutional, managerial, and financial capability to ensure proper planning, management, and completion of such project.
- 2. <u>Period of Performance</u>. The period of performance for this agreement is shown on page one of this Agreement. Contractor may use funds to cover eligible costs incurred, as set forth in Treasury's implementing regulations, during this period of performance.
- 3. <u>Reporting</u>. Contractor agrees to comply with any reporting obligations established by Treasury as they relate to this agreement. Contractor also agrees to comply with any reporting requirements established by the Governor's Office and Office of the State Controller. The State will provide notice of such additional reporting requirements in writing.

4. Maintenance of and Access to Records

- a. Contractor shall maintain records and financial documents sufficient to evidence compliance with section 602(c), Treasury's regulations implementing that section, and guidance issued by Treasury regarding the foregoing.
- b. The Treasury Office of Inspector General and the Government Accountability Office, or their authorized representatives, shall have the right of access to records (electronic and otherwise) of Contractor in order to conduct audits or other investigations.
- c. Records shall be maintained by Contractor for a period of five (5) years after all funds have been expended or returned to Treasury, whichever is later.
- 5. <u>Pre-award Costs.</u> Pre-award costs, as defined in 2 C.F.R. § 200.458, may not be paid with funding from this award.
- 6. <u>Administrative Costs.</u> Contractor may use funds provided under this agreement to cover both direct and indirect costs.
- 7. <u>Conflicts of Interest</u>. Contractors must disclose in writing to the contracting entity, as appropriate, any potential conflict of interest affecting the awarded funds in accordance with 2 C.F.R. § 200.112.

- 8. Compliance with Applicable Law and Regulations.
 - a. Contractor agrees to comply with the requirements of section 602 of the Act, regulations adopted by Treasury pursuant to section 602(f) of the Act, and guidance issued by Treasury regarding the foregoing. Contractor also agrees to comply with all other applicable federal statutes, regulations, and executive orders, and Contractor shall provide for such compliance by other parties in any agreements it enters into with other parties relating to this award.
 - i. Per US Treasury Final Award requirements, programs and services must not include terms or conditions that undermine efforts to stop COVID-19 or discourage compliance with recommendations and CDC guidelines.
 - b. Federal regulations applicable to this award include, without limitation, the following:
 - i. Uniform Administrative Requirements, Cost Principles, and Audit Requirements for Federal Awards, 2 C.F.R. Part 200, other than such provisions as Treasury may determine are inapplicable to this Agreement and subject to such exceptions as may be otherwise provided by Treasury. Subpart F Audit Requirements of the Uniform Guidance, implementing the Single Audit Act, shall apply to this award.
 - ii. Universal Identifier and System for Award Management (SAM), 2 C.F.R. Part 25, pursuant to which the award term set forth in Appendix A to 2 C.F.R. Part 25 is hereby incorporated by reference.
 - iii. Reporting Subaward and Executive Compensation Information, 2 C.F.R. Part 170, pursuant to which the award term set forth in Appendix A to 2 C.F.R. Part 170 is hereby incorporated by reference.
 - iv. OMB Guidelines to Agencies on Government wide Debarment and Suspension (Nonprocurement), 2 C.F.R. Part 180, including the requirement to include a term or condition in all lower tier covered transactions (Agreements and Subcontractors described in 2 C.F.R. Part 180, subpart B) that the award is subject to 2 C.F.R. Part 180 and Treasury's implementing regulation at 31 C.F.R. Part 19.
 - v. Subrecipient Integrity and Performance Matters, pursuant to which the award term set forth in 2 C.F.R. Part 200, Appendix XII to Part 200 is hereby incorporated by reference.
 - vi. Government wide Requirements for Drug-Free Workplace, 31 C.F.R. Part 20.
 - vii. New Restrictions on Lobbying, 31 C.F.R. Part 21.
 - viii. Uniform Relocation Assistance and Real Property Acquisitions Act of 1970

(42 U.S.C. §§ 4601-4655) and implementing regulations.

- ix. Generally applicable federal environmental laws and regulations.
- c. Statutes and regulations prohibiting discrimination applicable to this agreement include, without limitation, the following:
 - i. Title VI of the Civil Rights Act of 1964 (42 U.S.C. §§ 2000d et seq.) and Treasury's implementing regulations at 31 C.F.R. Part 22, which prohibit discrimination on the basis of race, color, or national origin under programs or activities receiving federal financial assistance;
 - ii. The Fair Housing Act, Title VIII of the Civil Rights Act of 1968 (42 U.S.C. §§ 3601 et seq.), which prohibits discrimination in housing on the basis of race, color, religion, national origin, sex, familial status, or disability;
 - iii. Section 504 of the Rehabilitation Act of 1973, as amended (29 U.S.C. § 794), which prohibits discrimination on the basis of disability under any program or activity receiving federal financial assistance;
 - iv. The Age Discrimination Act of 1975, as amended (42 U.S.C. §§ 6101 et seq.), and Treasury's implementing regulations at 31 C.F.R. Part 23, which prohibit discrimination on the basis of age in programs or activities receiving federal financial assistance; and
 - v. Title II of the Americans with Disabilities Act of 1990, as amended (42 U.S.C. §§ 12101 et seq.), which prohibits discrimination on the basis of disability under programs, activities, and services provided or made available by state and local governments or instrumentalities or agencies thereto.
- 9. Remedial Actions. In the event of any noncompliance with section 602 of the Act, other applicable laws, Treasury's implementing regulations, guidance, or any reporting or other program requirements, Treasury may impose additional conditions on the receipt of a subsequent tranche of future award funds, if any, or take other available remedies as set forth in 2 C.F.R. § 200.339. In the case of a violation of section 602(c) of the Act regarding the use of funds, previous payments shall be subject to recoupment as provided in section 602(e) of the Act and any additional payments may be subject to withholding as provided in sections 602(b)(6)(A)(ii)(III) of the Act, as applicable.
- 10. <u>Hatch Act.</u> Recipient agrees to comply, as applicable, with requirements of the Hatch Act (5 U.S.C.§§ 1501-1508 and 7324-7328), which limit certain political activities of State or local government employees whose principal employment is in connection with an activity financed in whole or in part by this federal assistance.
- 11. <u>False Statements</u>. Recipient understands that making false statements or claims in connection with this award is a violation of federal law and may result in criminal, civil, or administrative

- sanctions, including fines, imprisonment, civil damages and penalties, debarment from participating in federal awards or Agreements, and/or any other remedy available by law.
- 12. <u>Publications</u>. Any publications produced with funds from this award must display the following language: "This project [is being] [was] supported, in whole or in part, by federal award number SLFRF0126 awarded to the State of Colorado by the U.S. Department of the Treasury."

13. Debts Owed the Federal Government.

- a. Any funds paid to the Contractor (1) in excess of the amount to which the Contractor is finally determined to be authorized to retain under the terms of this award; (2) that are determined by the Treasury Office of Inspector General to have been misused; or (3) that are determined by Treasury to be subject to a repayment obligation pursuant to sections 602(e) and 603(b)(2)(D) of the Act and have not been repaid by the Contractor shall constitute a debt to the federal government.
- b. Any debts determined to be owed to the federal government must be paid promptly by the Contractor. A debt is delinquent if it has not been paid by the date specified in Treasury's initial written demand for payment, unless other satisfactory arrangements have been made or if the Contractor knowingly or improperly retains funds that are a debt as defined in paragraph 14(a). Treasury will take any actions available to it to collect such a debt.

14. Disclaimer.

- a. The United States expressly disclaims any and all responsibility or liability to Contractor or third persons for the actions of Contractor or third persons resulting in death, bodily injury, property damages, or any other losses resulting in any way from the performance of this agreement or any other losses resulting in any way from the performance of this Agreement or any Agreement, or Subcontractor under this award.
- b. The acceptance of this agreement by Contractor does not in any way establish an agency relationship between the United States and Contractor.

15. Protections for Whistleblowers.

- a. In accordance with 41 U.S.C. § 4712, Contractor may not discharge, demote, or otherwise discriminate against an employee in reprisal for disclosing to any of the list of persons or entities provided below, information that the employee reasonably believes is evidence of gross mismanagement of a federal Agreement or grant, a gross waste of federal funds, an abuse of authority relating to a federal Agreement or grant, a substantial and specific danger to public health or safety, or a violation of law, rule, or regulation related to a federal Agreement (including the competition for or negotiation of an Agreement) or grant.
- b. The list of persons and entities referenced in the paragraph above includes the following:
 - i. A member of Congress or a representative of a committee of Congress;
 - ii. An Inspector General;
 - iii. The Government Accountability Office;

- iv. A Treasury employee responsible for Agreement or grant oversight or management;
- v. An authorized official of the Department of Justice or other law enforcement agency;
- vi. A court or grand jury; or
- vii. A management official or other employee of Subrecipient, Contractor, or Subcontractor who has the responsibility to investigate, discover, or address misconduct.
- c. Contractor shall inform its employees in writing of the rights and remedies provided under this section, in the predominant native language of the workforce.
- 16. <u>Increasing Seat Belt Use in the United States</u>. Pursuant to Executive Order 13043, 62 FR 19217 (Apr. 18, 1997), Prime Recipient should encourage its Contractors to adopt and enforce onthe-job seat belt policies and programs for their employees when operating company-owned, rented or personally owned vehicles.
- 17. Reducing Text Messaging While Driving. Pursuant to Executive Order 13513, 74 FR 51225 (Oct. 6, 2009), Prime Recipient should encourage its employees, Subrecipients, and Contractors to adopt and enforce policies that ban text messaging while driving, and Contractors should establish workplace safety policies to decrease accidents caused by distracted drivers.

ASSURANCES OF COMPLIANCE WITH CIVIL RIGHTS REQUIREMENTS

ASSURANCES OF COMPLIANCE WITH TITLE VI OF THE CIVIL RIGHTS ACT OF 1964

As a condition of receipt of federal financial assistance from the Department of the Treasury, through the Prime Recipient, the Contractor provides the assurances stated herein. The federal financial assistance may include federal grants, loans and Agreements to provide assistance to the Prime Recipient's residents, the use or rent of Federal land or property at below market value, Federal training, a loan of Federal personnel, subsidies, and other arrangements with the intention of providing assistance. Federal financial assistance does not encompass Agreements of guarantee or insurance, regulated programs, licenses, procurement Agreements by the Federal government at market value, or programs that provide direct benefits.

The assurances apply to all federal financial assistance from or funds made available through the Department of the Treasury, including any assistance that the Prime Recipient may request in the future.

The Civil Rights Restoration Act of 1987 provides that the provisions of the assurances apply to all of the operations of the Prime Recipient's program(s) and activity(ies), so long as any portion of the Prime Recipient's program(s) or activity(ies) is federally assisted in the manner prescribed above.

- 1. Contractor ensures its current and future compliance with Title VI of the Civil Rights Act of 1964, as amended, which prohibits exclusion from participation, denial of the benefits of, or subjection to discrimination under programs and activities receiving federal financial assistance, of any person in the United States on the ground of race, color, or national origin (42 U.S.C. § 2000d *et seq.*), as implemented by the Department of the Treasury Title VI regulations at 31 CFR Part 22 and other pertinent executive orders such as Executive Order 13166, directives, circulars, policies, memoranda, and/or guidance documents.
- 2. Contractor acknowledges that Executive Order 13166, "Improving Access to Services for Persons with Limited English Proficiency," seeks to improve access to federally assisted programs and activities for individuals who, because of national origin, have Limited English proficiency (LEP). Contractor understands that denying a person access to its programs, services, and activities because of LEP is a form of national origin discrimination prohibited under Title VI of the Civil Rights Act of 1964 and the Department of the Treasury's implementing regulations. Accordingly, Contractor shall initiate reasonable steps, or comply with the Department of the Treasury's directives, to ensure that LEP persons have meaningful access to its programs, services, and activities. Contractor understands and agrees that meaningful access may entail providing language assistance services, including oral interpretation and written translation where necessary, to ensure effective communication in the Contractor's programs, services, and activities.

- 3. Contractor agrees to consider the need for language services for LEP persons when Contractor develops applicable budgets and conducts programs, services, and activities. As a resource, the Department of the Treasury has published its LEP guidance at 70 FR 6067. For more information on taking reasonable steps to provide meaningful access for LEP persons, please visit http://www.lep.gov.
- 4. Contractor acknowledges and agrees that compliance with the assurances constitutes a condition of continued receipt of federal financial assistance through the Prime Recipient and is binding upon Contractor and Contractor's successors, transferees, and assignees for the period in which such assistance is provided.
- 5. Contractor acknowledges and agrees that it must require any sub-grantees, contractors, subcontractors, successors, transferees, and assignees to comply with assurances 1-4 above, and agrees to incorporate the following language in every Agreement or agreement subject to Title VI and its regulations between the Contractor and the Contractor's sub-grantees, Contractors, Subcontractors, successors, transferees, and assignees:

The sub-grantee, Contractor, Subcontractor, successor, transferee, and assignee shall comply with Title VI of the Civil Rights Act of 1964, which prohibits Prime Recipients and Contractors of Prime Recipients of federal financial assistance from excluding from a program or activity, denying benefits of, or otherwise discriminating against a person on the basis of race, color, or national origin (42 U.S.C. § 2000d et seq.), as implemented by the Department of the Treasury's Title VI regulations, 31 CFR Part 22, which are herein incorporated by reference and made a part of this Agreement (or agreement). Title VI also includes protection to persons with "Limited English Proficiency" in any program or activity receiving federal financial assistance, 42 U.S.C. § 2000d et seq., as implemented by the Department of the Treasury's Title VI regulations, 31 CFR Part 22, and herein incorporated by reference and made a part of this Agreement or agreement.

- 6. Contractor understands and agrees that if any real property or structure is provided or improved with the aid of federal financial assistance by the Department of the Treasury, this assurance obligates the Prime Recipient and Contractor, or in the case of a subsequent transfer, the transferee, for the period during which the real property or structure is used for a purpose for which the federal financial assistance is extended or for another purpose involving the provision of similar services or benefits. If any personal property is provided, this assurance obligates the Contractor for the period during which it retains ownership or possession of the property.
- 7. Contractor shall cooperate in any enforcement or compliance review activities by the Department of the Treasury of the aforementioned obligations. Enforcement may include investigation, arbitration, mediation, litigation, and monitoring of any settlement agreements that may result from these actions. The Contractor shall

- comply with information requests, on-site compliance reviews and reporting requirements.
- 8. Contractor shall maintain a complaint log and inform the Department of the Treasury of any complaints of discrimination on the grounds of race, color, or national origin, and limited English proficiency covered by Title VI of the Civil Rights Act of 1964 and implementing regulations and provide, upon request, a list of all such reviews or proceedings based on the complaint, pending or completed, including outcome. Contractor also must inform the Department of the Treasury if Contractor has received no complaints under Title VI.
- 9. Contractor must provide documentation of an administrative agency's or court's findings of non-compliance of Title VI and efforts to address the non-compliance, including any voluntary compliance or other agreements between the Contractor and the administrative agency that made the finding. If the Contractor settles a case or matter alleging such discrimination, the Contractor must provide documentation of the settlement. If Contractor has not been the subject of any court or administrative agency finding of discrimination, please so state.
- 10. If the Contractor makes sub-contracts to other agencies or other entities, the Contractor is responsible for ensuring that sub-contractors also comply with Title VI and other applicable authorities covered in this document Contractors that make sub-contracts must have in place standard contract assurances and review procedures to demonstrate that that they are effectively monitoring the civil rights compliance of sub- contractors.

The United States of America has the right to seek judicial enforcement of the terms of this assurances document and nothing in this document alters or limits the federal enforcement measures that the United States may take in order to address violations of this document or applicable federal law.

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT N NOTICE OF (PARTIAL) SUBSTANTIAL COMPLETION

(when issued)

SC-6.5 Rev. 03/2023



NOTICE OF PARTIAL SUBSTANTIAL COMPLETION

| Date of Partial Subs | stantial Completion: | | | | |
|---|---|---|---|--|--|
| | | Date to be insert | ted by the Principal Representative | | |
| Institution/Agency: | University of Colora | ado Anschutz | Medical Campus | | |
| Project No./Name: | 1 | | | | |
| TO: | | | | | |
| Principal Representativ | e | | | | |
| and | | | | | |
| (\$VendorAddress) | | | | | |
| Contractor | | | | | |
| Architect/Engineer, to b General Conditions of the suitable for occupancy determined to be fully at A punch list of work to hereto, along with the Subcontractor or trade period indicated in the Except as stated on the | e substantially complet ne Contract in SC-6.23, b) inspected for code and comfortably usable be completed, work no Contractor's schedule responsible for the work Agreement for punch line reverse side of this | e as of the date and SC-8.1 or a compliance we had also and d) fully club, and d) fully club, and the complete, and the date st completion provided in the complete of Particle of Particle and the date of Particle of Particle and the date of Particle of Particle and SC-8.1 or and the date of Particle of Particle and SC-8.1 or and the date of Particle and SC-9.1 or | e noted above in accordance with Article 17.3 in SC-6.4 and the Specith Building Inspection Records seaned and appropriate for presen with the Drawings or Specification of each and every item identification to Final Acceptance. tial Substantial Completion, all m | knowledge, information and belief of the the criteria outlined in Article 41 of The iffications, including without limitation a igned by code officials for the State, of tation to the public. Instantial and unsatisfactory work is attached in the punch list specifying the becommenced and finished within an anufacturers' warranties, other species on the Date of Substantial Completice. | ne a) c) ed ne ny al |
| executed on the reversubstantially complete | se by the Contractor a as of the Date of Subs | and the Princip tantial Complet | oal Representative. The Principa | Substantial Completion only when ful I Representative accepts the Work a agrees to complete or correct the Work letion schedule | as |
| Architect/Engineer | | Date | Contractor | Date | |
| State Buildings Pro (or Authorized Dele | | Date | Principal Representative (Institution or Agency) | Date | |

| The responsibilities of the Principal Representative and the Contractor for security, maintenance, heat, utilities, and insurance shabe as specified in the Contract Documents or as otherwise hereafter noted: | all |
|---|-----|
| Exceptions, if any, to the commencement of warranties shall be: | |
| The attached final punch list consists of pages, and the attached Contractor's schedule showing the dates of commencement and completion of each punch list item consists of pages. | |
| When completely executed, this form shall be sent to the Contractor and the Principal Representative with a copy to State Buildings Program. | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |



NOTICE OF SUBSTANTIAL COMPLETION

| Date of Substantial Comp | pletion: | | |
|--|--------------------------------|--|-------------------------|
| · | Date to be inserted by t | he Principal Representative | |
| Institution/Agency: Univ | ersity of Colorado Anschut | z Medical Campus | |
| Project No./Name: / | | | |
| | | | |
| TO. | | | |
| ТО: | | | |
| | | | |
| | | | |
| Principal Representative | | | |
| | | | |
| and | | | |
| | | | |
| | | | |
| | | | |
| Contractor | | | |
| | | | |
| | | ected and determined, to the best know | |
| | | e as of the date noted above in accord tract in SC-6.23 and SC-8.1 or Article 1 | |
| | | occupancy, b) inspected for code com | |
| | | c) determined to be fully and comfortab | ly usable, and d) fully |
| cleaned and appropriate for p | resentation to the public. | | |
| A punch list of work to be co | ompleted, work not in complia | nce with the Drawings or Specification | ns, and unsatisfactory |
| work is attached hereto, along | g with the Contractor's schedu | lle for the completion of each and every | item identified on the |
| | | e for the work, and the dates the comple in the Agreement for punch list com | |
| Acceptance. | i within any penod indicated | in the Agreement for punch list con | ipietion phor to Final |
| · | | | |
| | | antial Completion, all manufacturers' wa | |
| warranties and the Contractor Completion noted above. | s one-year obligation to peno | rm remedial work, shall commence on t | ne Date of Substantial |
| о от при от | | | |
| | | l establish the Date of Substantial Com | |
| | | ative. The Principal Representative abletion herein noted. The Contractor a | |
| | | to do so in accordance with attached | |
| schedule. | , | | |
| | | | |
| Architect/Engineer | Date | Contractor | Date |
| , a strike of Engineer | Date | Contractor | Date |
| | | | |
| | | | |
| State Buildings Program | Date | Principal Representative | Date |
| (or Authorized Delegate) | | (Institution or Agency) | |

| The responsibilities of the Principal Representative and the Contractor for security, maintenance, heat, utilities, and insurance be as specified in the Contract Documents or as otherwise hereafter noted: |
|--|
| Exceptions, if any, to the commencement of warranties shall be: |
| The attached final punch list consists of pages, and the attached Contractor's schedule showing the dates of commencement and completion of each punch list item consists of pages. |
| When completely executed, this form shall be sent to the Contractor and the Principal Representative with a copy to State Buildings Program |
| |
| |
| |
| |
| |
| |
| |
| |
| |
| |

shall

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT O NOTICE OF APPROVAL OF OCCUPANCY/USE

(when issued)

SC-6.5 Rev. 03/2023



NOTICE OF APPROVAL OF OCCUPANCY/USE

| Date of Occup | pancy: | |
|-----------------|-------------|---|
| | | Date to be inserted by the Architect/Engineer after consulation with Principal Representative |
| Institution/Age | ency: | University of Colorado Anschutz Medical Campus |
| Project No./Na | ame: | |
| | | |
| Portion(s) of p | roject for | which occupancy is approved: |
| | | |
| | | |
| | | |
| Type of Occup | pancy: | ☐ Total or ☐ Partial |
| .) | , - | |
| | | |
| The items ide | entified be | low if applicable must be completed before Occupancy is approved. |
| | | |
| Date | A/E | |
| Completed | Signoff | |
| | | 1a. The Notice of Substantial Completion has been issued. |
| | | |
| | | 1b. The Building Inspection Record is completely signed-off and attached. |
| | | |
| | | 2a. Notification has been made to the local Fire Department concerning which |
| | | portion(s) of the building will be occupied and the date(s). |
| | | |
| | | 2b. Fire alarms, smoke detection systems and building fire sprinkler systems |
| | | have been fully checked and are operable. |
| | | 2c. The building's fire connections must be installed and operable, if applicable. |
| | | |
| | | 3. Coordination for final utility and service connections and meters (water, gas, |
| | | sewer, electricity and telecommunication) has been made and systems are in |
| | | full operating order. |
| | | 4. Sterilization of plumbing systems has been performed. |
| | | |
| | | |
| | | 5. Operational test of systems and equipment has been performed as required. |
| | | |
| | | 6. Systems adjustments such as balancing, equipment operations, etc., have |
| | | been performed. Reports have been submitted to the Architect/Engineer for |
| | | approval. |
| | | 7. Principal Representative furnished equipment and furnishings are coordinated |
| | | and placed. |
| | | 8. All elements left unfinished must be in such condition that there would be no |
| | | hazard to the health or safety of the occupants. |

| | | 9. All restroom facilities must be fully functional and operable. |
|--------------|--------------|--|
| | | 10. All light fixtures must be installed and operable. |
| | | 11. All exit lights and emergency lighting systems have been checked and are operable. |
| | | 12. All windows have been glazed and hardware is available for ventilation purposes. |
| | | 13. All routes of egress must be clear of construction materials and debris at all times. |
| | | 14. There must be a means of pedestrian access to each building. Contractor must have sidewalks installed before occupancy and pedestrian barricades and other means of public protection as required. |
| | | |
| Occupancy do | es not const | titute acceptance of the project as being complete. It simply provides the Principal |

Occupancy does not constitute acceptance of the project as being complete. It simply provides the Principal Representative the opportunity to occupy/use the project or the applicable portion thereof prior to final completion and acceptance. Occupants can expect to be impacted by the Contractor's efforts to complete the project. The Contractor would not repair any damage caused by the occupants.

| Architect/Engineer | Date | Principal Representative (Institution or Agency) | Date |
|--|------|--|------|
| State Buildings Program (or Authorized Delegate) | Date | Contractor | Date |

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT P NOTICE OF (PARTIAL) FINAL ACCEPANCE

(when issued)

SC-6.5 Rev. 03/2023



NOTICE OF PARTIAL FINAL ACCEPTANCE

| Date of Notice of Par | tial Acceptance: | | | |
|---|----------------------|--------------------------|--|----------------------------|
| | | | by A/E after consultation with the Princip | pal Representative |
| Institution/Agency: | University of Co | olorado Anschutz | Medical Campus | |
| Project No./Name: | | | | |
| Portion(s) of Project | for which final acc | ceptance is appro | oved: | |
| , , | | | | |
| | | | | |
| | | | | |
| | | | | |
| TO: | | | | |
| | | | | |
| | | | | |
| Notice is hereby given | that the State of | Colorado actino | by and through the Board of Reg | nents of the University of |
| Colorado, a body corp | | | versity of Colorado Denver, acce | |
| numbered project. | | | | |
| | | | | |
| | | | | |
| State Buildings Progr | ram | Date | Principal Representative | Date |
| (or Authorized Deleg | | Date | (Institution or Agency) | Date |
| | | | | |
| | | | | |
| | | | | |
| | | | | |
| | | | | |
| *When completely exe Representative. | ecuted, this form is | s to be sent by <u>e</u> | ertified mail email to the Contra | ctor by the Principal |
| | | | | |
| | | | | |
| | | | | |
| | | | | |
| | | | | |
| | | | | |



NOTICE OF FINAL ACCEPTANCE

| Date of Notice of Accepta | ance: | | |
|--|---------------------------|--|--------|
| | Date to be inserted by | A/E after consultation with the Principal Represen | tative |
| Institution/Agency: Univ | ersity of Colorado Ansch | utz Medical Campus | |
| Project No./Name: / | | | |
| | | | _ |
| | | | |
| TO: | | | |
| | | | |
| | | | |
| | | | |
| | ody corporate, for and or | cting by and through the <u>Board of Reger</u> n behalf of the University of Colorado Der | |
| | | | |
| | | | |
| | | | |
| State Buildings Program (or Authorized Delegate) | Date | Principal Representative (Institution or Agency) | Date |

^{*}When completely executed, this form is to be sent by <u>certified mail</u> to the Contractor by the Principal Representative or delivered by any other means to which the parties agree.

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT Q NOTICE OF CONTRACTOR'S SETTLEMENT

(when issued)

SC-6.5 Rev. 03/2023



NOTICE OF CONTRACTOR'S SETTLEMENT

| Ins | titution/Agency | University of Colorado Anschutz Medical Campus |
|------------|---|--|
| No | tice Number: | |
| Pro | ject No./Title: | 1 |
| | | |
| settle | ement will be ma | that on at <u>1945 Wheeling Street, Mail Stop F418, Aurora, CO 80045,</u> final e by the STATE OF COLORADO with, hereinafter called the "CONTRACTOR", for contract for the construction of a PROJECT as referenced above. |
| 1. | on account of tools, or equipment about the perfo | partnership, association or corporation who has an unpaid claim against the said project, for one furnishing of labor, materials, team hire, sustenance, provisions, provender, rental machinery ent and other supplies used or consumed by such Contractor or any of his subcontractors in one nance of said work, may at any time up to and including said time of such final settlement, file at the amount due and unpaid on account of such claim |
| 2. | All such claims | hall be filed with the Authority for College, Institution, Department or Agency. |
| 3. | | rt of a creditor to file such statement prior to such final settlement will relieve the State of ny and all liability for such claim |
| Auth | norized Facility | lanager or Authorized Individual |
| Nai | me: | |
| | oroval Date: | |
| Age | | niversity of Colorado Anschutz Medical ampus |
| Pho | one: | априз |
| Fax | · — | |
| Em | ail: | |
| MED | OIA OF PUBLICA | TION: |
| <u>PUB</u> | LICATION DATI First: | <u>5:</u> |
| | Second: | (At least ten (10) days prior to above settlement date) |
| <u>NOT</u> | ES TO EDITOR | |
| | | |

State Form SBP-7.3 Rev. 7/2010

Transmit two (2) copies of the Affidavit of Publication, and invoice, to:

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT S University of Colorado Denver | Anschutz Medical Campus Supplementary General Conditions

UNIVERSITY OF COLORADO DENVER | ANSCHUTZ MEDICAL CAMPUS CONSTRUCTION MANAGER/GENERAL CONTRACTOR SUPPLEMENTARY GENERAL CONDITIONS Updated 12/21/2021

The Construction Manager/General Contractor Agreement shall be amended as follows:

1. The terms University, University of Colorado, University of Colorado Denver, University of Colorado Anschutz Medical Campus, CU Denver, CU Anschutz, CU, and Principal Representative, are interchangeable for this replacement of Article 25.

Replace Article 25 as follows:

ARTICLE 25. INSURANCE

For purposes of this supplement "Contractor" as used herein shall mean, as appropriate to the State Contract form being used, Contractor, Standing Order Contractor, Construction Manager/General Contractor, or Design/Build Entity.

COVERAGES AND LIMITS OF INSURANCE

Contractor shall provide coverage with limits of liability not less than those stated below.

A General

The Contractor shall obtain and maintain, at its own expense and for the duration of the contract including any warranty periods under the Contract are satisfied, the insurance coverages set forth below.

By requiring such insurance, the Principal Representative shall not be deemed or construed to have assessed the risk that may be applicable to the Contractor its agents, representatives, employees or subcontractors under this contract. The insurance requirements herein for this Contract in no way limit the indemnity covenants contained in the Contract. The Principal Representative in no way warrants that the limits contained herein are sufficient to protect the Contractor from liabilities that might arise out of the performance of the work under this Contract by the Contractor, its agents, representatives, employees, or subcontractors. The Contractor shall assess its own risks and if it deems appropriate and/or prudent, maintain higher limits and/or broader coverages. The Contractor is not relieved of any liability or other obligations assumed or pursuant to the Contract by reason of its failure to obtain or maintain insurance in sufficient amounts, duration, or types.

B Commercial General Liability – ISO CG 0001 or equivalent. Coverage to include:

- 1. Premises and Operations
- 2. Explosions, Collapse and Underground Hazards
- 3. Personal / Advertising Injury
- 4. Products / Completed Operations
- 5. Liability assumed under an Insured Contract (including defense costs assumed under contract)
- 6. Independent Contractors
- 7. Designated Construction Projects(s) General Aggregate Limit, ISO CG 2503 (1997 Edition)
- 8. Additional Insured—Owners, Lessees or Contractors Endorsement, ISO Form 2010 (2004 Edition or equivalent)
- 9. Additional Insured—Owners, Lessees or Contractors Endorsement (Completed Operations), ISO CG 2037 (7/2004 Edition or equivalent)
- 10. The policy shall be endorsed to include the following additional insured language on the Additional Insured Endorsements specified above: "The Regents of the University of Colorado, a Body Corporate, named as an additional insured with

Rev 12/21/2021 Page 1 of 7.

respect to liability and defense of suits arising out of the activities performed by, or on behalf of the Contractor, including completed operations".

- 11. Commercial General Liability Completed Operations policies must be kept in effect for up to three (3) years after completion of the project. For buildings with a construction cost greater than \$99 million, the Commercial General Liability Completed Operations policies must be kept in effect for up to eight (8) years after the completion of the project.
- 12. An umbrella and/or excess liability policy may be used to meet the minimum liability requirements provided that the coverage is written on a "following form" basis.

| Liability Limits | General Aggregate | Products/Completed Operation Aggregate | Each Occurrence | Personal/ Advertising Injury |
|-------------------------------------|----------------------|---|--------------------|------------------------------------|
| Primary General Liability | \$2,000,000 | \$2,000,000 | \$1,000,0000 | \$1,000,000 |
| Umbrella or Excess Liability* | \$5,000,000 | \$5,000,000 | \$5,000,000 | \$5,000,000 |

^{*}Umbrella or Excess Liability does not apply to projects totaling \$500,000 or under.

13. The following exclusionary endorsements are prohibited in the CGL policy:

- i. Damage to work performed by subcontract/vendor (CG 22-94 or similar);
- ii. Contractual liability coverage exclusion modifying or deleting the definition of an "insured contract":
- iii. If applicable to the work to be performed: Residential or multi-family;
- iv. If applicable to the work to be performed: Exterior insulation finish systems;
- v. If applicable to the work to be performed: Subsidence or earth movement.

C Automobile Liability

Bodily Injury and Property Damage for any owned, hired, and non-owned vehicles used in the performance of this contract

Minimum Limits:

Bodily Injury/Property Damage (Each Accident) \$ 1,000,000

D Workers Compensation

- 1. Statutory Benefits (Coverage A)
- 2. Employers Liability (Coverage B)
- Policy shall contain a waiver of subrogation in favor of the Principal Representative.
- 4. This requirement shall not apply when a contractor or subcontractor is exempt under Colorado Workers' Compensation Act., AND when such contractor or subcontractor executes the appropriate sole proprietor waiver form.

Minimum Limits:

Coverage A (Workers' Compensation) Statutory

Coverage B (Employers Liability)

Each accident \$ 100,000

Disease each employee \$ 100,000

Disease policy limit \$ 500,000

E NOT USED

F Builder's Risk / Installation Floater

Unless otherwise provided or instructed by the Principal Representative, the Contractor shall purchase and maintain, in a company or companies lawfully authorized to do business in the

Rev 12/21/2021 Page 2 of 7.

jurisdiction in which the project is located, Builder's Risk Insurance in the amount of the initial contract amount as well as subsequent modifications for the entire project at the site on a replacement cost basis without optional deductibles. This coverage is required for new buildings or additions to existing buildings and for materials and equipment to be installed in existing structures.

- 1. Covered Cause of Loss: Special Form
- 2. Include Theft and Vandalism
- 3. Labor costs to repair damaged work
- 4. Shall be written for 100% of the completed value (replacement cost basis)
- 5. Deductible maximum is \$50,000.00
- 6. Waiver of Subrogation is to apply
- 7. The Regents of the University of Colorado, a body corporate, shall be added as Additional Named Insured on Builders Risk.
- 8. Policy must provide coverage from the time any covered property becomes the responsibility of the Contractor, and continue without interruption during construction, renovation, or installation, including any time during which the covered property is being transported to the construction installation site, or awaiting installation, whether on or off site.
- 9. The Policy shall be maintained, unless otherwise provided in the contract documents or otherwise agreed in writing by all persons and entities who are beneficiaries of such insurance, until final payment has been made or until no person or entity other than the Principal Representative has insurable interest in the property to be covered, whichever is later.
- 10. The Builder's Risk insurance shall include interests of the Principal Representative, and if applicable, affiliated or associated entities, the General Contractor, subcontractors and sub-tier contractors in the project.
- 11. Builders' Risk Coverage shall be on a **Special** Covered Cause of Loss Form and shall include theft, vandalism, malicious mischief, collapse, false-work, temporary buildings and debris removal including demolition, increased cost of construction, architect's fees and expenses, flood (including water damage), earthquake, and if applicable, all below and above ground structures, piping, foundations including underground water and sewer mains, piling including the ground on which the structure rests and excavation, backfilling, filling, and grading. Equipment Breakdown Coverage (a.k.a. Boiler & Machinery) shall be included as required by the Contract Documents or by law, which shall specifically cover insured equipment during installation and testing (including hot testing, where applicable). Other coverages may be required if provided in contract documents.
- 12. The Builders' Risk shall be written for 100% of the completed value (replacement cost basis) of the work being performed. The Builders' Risk shall include the following provisions:
 - i. Replacement Cost Basis including modification of the valuation clause to cover all costs needed to repair the structure or work (including overhead and profits) and will pay based on the values figured at the time of rebuilding or repairing, not at the time of loss
 - ii. Modify or delete exclusion pertaining to damage to interior of building caused by an perils insured against are covered; also provide coverage for water damage.

Note, if the addition, or renovation is to an existing building, The Principal Representative requires that the Contractor provide as an option to include the existing building into the Builders' Risk Policy. The Principal Representative shall provide the replacement cost value of the existing building.

- 13. At the option of the Principal Representative, the Principal Representative may include Soft Costs (including Loss of Use)/Delay in Opening Endorsement under the builder's risk policy. The Principal Representative agrees to provide the necessary exposure base information for quotation by the Builder's Risk carrier. The Principal Representative agrees to pay the premium associated with the Soft Costs coverage, the Principal Representative decides to purchase this coverage.
- 14. The Builders' Risk Policy shall specifically permit occupancy of the building during construction. Partial occupancy or use of the work shall not commence until the insurance company or companies providing insurance have consented to such partial occupancy or use. The Principal Representative and Contractor shall take reasonable steps to obtain consent of the insurance company or companies and delete any

Rev 12/21/2021 Page 3 of 7.

- provisions with regard to restrictions within any Occupancy Clauses within the Builders' Risk Policy. The Builders' Risk Policy shall remain in force until acceptance of the project by the Principal Representative.
- 15. The deductible shall not exceed \$50,000 and shall be the responsibility of the Contractor except for losses such as flood (not water damage), earthquake, windstorm, tsunami, volcano, etc. Losses in excess of \$50,000 insured shall be adjusted in conjunction with the Principal Representative. Any insurance payments/proceeds shall be made payable to the Principal Representative subject to requirements of any applicable mortgagee clause.

The Contractor shall pay subcontractors their just shares of insurance proceeds received by the Contractor, and by appropriate agreements, written where legally required for validity, shall require subcontractors to make payments to their sub-subcontractors in similar manner.

The Principal Representative shall have the authority to adjust and settle any losses in excess of \$50,000 with insurers unless one of the parties in interest shall object in writing within five days after occurrence of loss to the Principal Representative exercise of this power. It is expressly agreed that nothing in this section shall be subject to arbitration and any references to arbitration are expressly deleted.

16. The Contractor is responsible for providing 45 days' notice of cancellation to the Principal Representative. The policy shall contain all generally applicable conditions, definitions, exclusions and endorsements related to the Project.

If the Contractor does not intend to purchase such Builder's Risk Insurance required by the Contract and with all of the coverages in the amount described above, the Contractor shall so inform the Principal Representative as stated in writing prior to commencement of the work. The Principal Representative may then affect insurance that will protect the interests of the Principal Representative, the General Contractor, Subcontractors and sub-tier contractors in the project. Coverages applying shall be the same as stated above including other coverages that may be required by the Principal Representative. The cost shall be charged to the Contractor. Coverage shall be written for 100% of the completed value of the work being performed, with a deductible not to exceed \$50,000 per occurrence for most projects.

All deductibles will be assumed by the Contractor. Waiver of Subrogation is to apply against all parties named as insureds, but only to the extent the loss is covered, and Beneficial Occupancy Endorsements are to apply.

If the Principal Representative is damaged by the failure or neglect of the Contractor to purchase or maintain insurance as described above, without so notifying the Principal Representative, then the Contractor shall bear all reasonable costs properly attributable thereto.

G Contractors Pollution Liability

- 1. Coverage shall apply to sudden and gradual pollution conditions resulting from the escape of release of smoke, vapors, fumes, acids, alkalis, toxic chemicals, liquids, or gases, natural gas, waste materials, or other irritants, contaminants, or pollutants (including asbestos). Policy shall cover the Contractor's completed operations.
- If the coverage is written on a claims-made basis, the Contractor warrants that any
 retroactive date applicable to coverage under the policy precedes the effective date of
 this Contract; and that continuous coverage will be maintained or an extended discovery
 period will be exercised for a period of three (3) years beginning from the time that work
 under this contract is completed.
- 3. The policy shall be endorsed to include the following as Additional Insureds: The Regents of the University of Colorado, a Body Corporate, named as an additional insured with respect to liability and defense of suits arising out of the activities

Rev 12/21/2021 Page 4 of 7.

performed by, or on behalf of the Construction Manager, including completed operations.

- 4. Endorsements CA9948 and MCS-90 are required on the Automobile Liability Coverage if the Contractor is transporting any type of hazardous materials.
- 5. Contractors Pollution Liability policies must be kept in effect for up to three (3) years after completion of the project.

Minimum Limits (Projects at or under \$500,000):

Per Loss \$ 1,000,000 Aggregate \$ 1,000,000

Minimum Limits (Projects over \$500,000):

Per Loss \$ 2,000,000 Aggregate \$ 2,000,000

H Professional Liability (Errors and Omissions)

(This Professional Liability requirement applies only to Design/Build Entity SC-8.0 and 9.0.)

1. The Contractor shall maintain Errors and Omissions Liability covering negligent acts, errors and/or omissions, including design errors of the Contractor for damage sustained by reason of or in the course of operations under this Contract. The policy/coverages shall be amended to include the following:

Amendment of any Contractual Liability Exclusion to state: "This exclusion does not apply to any liability of others which you assume under a written contract provided such liability is caused by your negligent acts."

- 2. In the event that any professional liability insurance required by this Contract is written on a claims-made basis, Contractor warrants that any retroactive date under the policy shall precede the effective date of this Contract; and that either continuous coverage will be maintained or an extended discovery period will be exercised for a period of three (3) years beginning at the time work under this Contract is completed.
- 3. Policy shall contain a waiver of subrogation against The Regents of the University of Colorado, a Body Corporate.

 Wrongful Act
 \$ 2,000,000

 General Aggregate
 \$ 2,000,000

I ADDITIONAL INSURANCE REQUIREMENTS

- 1. All insurers must be licensed or approved to do business within the State of Colorado, and unless otherwise specified, all policies must be written on a per occurrence basis.
- 2. Contractor's insurance carrier should possess a minimum A.M. Best's Insurance Guide rating of A- VI.
- On insurance policies where the Principal Representative are named as additional insureds, the Principal Representative shall be additional insureds to the full limits of liability purchased by the Contractor even if those limits of liability are in excess of those required by this Contract.
- 4. Contractor shall furnish the Principal Representative with certificates of insurance (ACORD form or equivalent approved by the Principal Representative) as required by this Contract. The certificates for each insurance policy are to be signed by a person authorized by that insurer to bind coverage on its behalf.

All certificates and any required endorsements are to be received and approved by the Principal Representative before work commences.

Each insurance policy required by this Contract must be in effect at or prior to commencement of work under this Contract and remain in effect for the duration of the project. Failure to maintain the insurance policies as required by this Contract or to provide evidence of renewal is a material breach of contract.

Rev 12/21/2021 Page 5 of 7.

- 5. Upon request by the Principal Representative, Contractor must provide a copy of the actual insurance policy effecting coverage(s) required by the contract.
- 6. The Contractor's insurance coverage shall be primary insurance and non-contributory with respect to all other available resources.
- 7. The Contractor shall advise the Principal Representative in the event any general aggregate or other aggregate limits are reduced below the required per occurrence limit. At their own expense, the Contractor will reinstate the aggregate limits to comply with the minimum requirements and shall furnish to the Principal Representative a new certificate of insurance showing such coverage is in force.
- 8. Provide a minimum of thirty (30) days advance written notice to the Principal Representative for cancellation, non-renewal, or material changes to policies required under the Contract (45 days for builders' risk coverage.
- 9. Certificate Holder: The Regents of the University of Colorado, Project Management, 1945. North Wheeling Street, Campus Mail stop F-418, Aurora, CO 80045.

Failure of the Contractor to fully comply with these requirements during the term of the Contract may be considered a material breach of contract and may be cause for immediate termination of the Contract at the option of the Principal Representative. The Principal Representative reserves the right to negotiate additional specific insurance requirements at the time of the contract award.

Subcontractors

Contractor's certificate(s) shall include all subcontractors as additional insureds under its policies **or** subcontractors shall maintain separate insurance as determined by the Contractor, however, subcontractor's limits of liability shall not be less than \$1,000,000 per occurrence / \$2,000,000 aggregate.

Non-Waiver

The parties hereto understand and agree that The Principal Representative is relying on, and does not waive or intend to waive by any provision of this Contract, the monetary limitations or any other rights, immunities, and protections provided by the Colorado Governmental Immunity Act, et seq., as from time to time amended, or otherwise available to the Principal Representative or its officers, employees, agents, and volunteers.

Mutual Cooperation

The Principal Representative and Contractor shall cooperate with each other in the collection of any insurance proceeds which may be payable in the event of any loss, including the execution and delivery of any proof of loss or other actions required to effect recovery.

(Major Revision 01-21-11)

5. ARTICLE 53. MISCELLANEOUS. PROVISIONS

- a. Delete the following section except for Projects that are ARRA funded:
 - I. STATEWIDE CONTRACT MANAGEMENT SYSTEM
- b. Add the Following:
 - K. UNIVERSITY OF COLORADO DENVER | ANSCHUTZ MEDICAL CAMPUS POLICY ON SEXUAL HARASSMENT
 - The Contractor shall vigorously pursue to the greatest extent possible, adherence to the University of Colorado Denver Policy on Sexual Harassment and also require all employees, and employees of all subcontractors of any kind, working on this project to adhere to this Policy.
 - 2. Statement of Policy: It is the policy of the University of Colorado Denver to maintain the community as a place of work, study, and residence free of sexual harassment or exploitation of students, faculty, staff, and administrators. Sexual

Rev 12/21/2021 Page 6 of 7.

harassment is prohibited on campus and in university programs. The university is committed to taking appropriate action against any of its officials, employees or students who violate the policy prohibiting sexual harassment.

- 3. Definition of Sexual Harassment: For purposes of this Policy, sexual harassment is defined as conduct which is unwelcome and consists of:
 - a. sexual advances;
 - b. requests for sexual favors; or
 - c. other verbal or physical conduct of a sexual nature when submission to such conduct is made either explicitly or implicitly a term or condition of an individual's employment or academic decisions affecting the individual; or when such conduct has the purpose or effect, of unreasonably interfering with an individual's work or academic performance by creating an intimidating, hostile, or offensive working or educational environment.

Conduct prohibited under this policy may occur between persons of the same sex or of different sexes and may manifest itself in different ways. For example, sexual harassment may be as undisguised as a direct solicitation of sexual favors, or arise from behavior which has the effect of creating an intimidating, hostile, or offensive educational or working environment. In this regard, the following types of acts, if pervasive and continuous, are more likely than not to be considered sexual harassment: unwelcome physical contact, sexual remarks about a person's clothing, body, or sexual relations, conversation of a sexual nature or similar jokes and stories, and the display of sexually explicit materials in the workplace or their use in the classroom without defensible educational purpose.

- 4. Consequence of Sexual Offenses: The university may require the Contractor to remove from university property any individual or individuals who violate the policy prohibiting sexual harassment.
- 5. The full Sexual Misconduct policy of the University of Colorado hall be followed and can be found at: https://www.cu.edu/ope/aps/5014
- L. UNIVERSITY OF COLORADO DENVER | ANSCHUTZ MEDICAL CAMPUS POLICY ON SECURITY BADGING
 - 1. All costs and time associated with obtaining a University security badges for Contractor employees and subcontractors working on campus shall be borne by the Contractor. Badging shall be covered by Preconstruction Services Fee or Construction Services Fee and not directly reimbursable.

Rev 12/21/2021 Page 7 of 7.

STATE OF COLORADO OFFICE OF THE STATE ARCHITECT STATE BUILDINGS PROGRAM

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT T SALES AND USE TAX FORM

SALES AND USE TAX FORM

(Attach when executed)

- 1. Colorado Department of Revenue Certificate of Exemption for State Sales/Use Tax, dated August 25, 2017.
- 2. Colorado Department of Revenue Sales Tax Exemption Certificate Multi-Jurisdiction
- 3. State of Colorado letter confirming Adams County, RTD, Stadium, and Cultural Tax Exemptions dated April 7, 2006
- 4. City of Aurora Sales and use Tax exemption dated March 12, 2001
- 5. City and County of Denver confirmation of tax exemption status dated February 19, 2014
- 6. Colorado Department of Revenue Contractor Application for Exemption certification

DR 0160 (02/16/11) COLORADO DEPARTMENT OF REVENUE DENVER CO 80261-0013

CERTIFICATE OF EXEMPTION FOR STATE SALES/USE TAX ONLY

| THIS LICENSE IS |
|------------------|
| NOT TRANSFERABLE |

| USE ACCOUNT NUMBER for all references | LIABILITY INFORMATION | | ISSUE DATE | |
|---------------------------------------|-----------------------|--------|-------------|--|
| 09802565 | G | 010180 | Aug 25 2017 | |

յրուկիրընդրորդումիորիիդիրերերի

STATE OF COLORADO/ OFFICE OF STATE CONTROLLER ATTN: OFFICE OF UNIVERSITY CONTROLLER 1800 N GRANT ST STE 600 DENVER CO 80203-1148

Executive Director Department of Revenue



Sales Tax Exemption Certificate Multi - Jurisdiction

| See page 2 for instruct | ions | | | | | |
|--|---|--|-------------------------------------|--------------------------|-----------------------|--------------------|
| Last Name or Business Nam | е | First Name | | | | Middle Initial |
| Address | | | | | | |
| City | | | State | ZIP | | |
| | | | | | | |
| No. of Fig. (D. or) | | Certify That | | | | |
| Name of Firm (Buyer) The Regents of University | sity of Colorado | | | | | |
| Address | sity of Colorado | | | | | |
| 1800 Grant Street, Suit | te 600 | | | | | |
| City | | | State | ZIP | | |
| Denver | | | co | 80203 | | |
| | Qualifies As (Ch | heck each applic | able ite | m) | | |
| ☐Wholesaler | ☐ Retailer | ☐ Manufa | acturer | | ☐ Charitable or | Religious |
| ⊠ Political Subdivision | or Governmental Agency | ☐ Other (| Specify) | | | |
| If Other, specify here | | | | | | St. 1911 1911 1911 |
| 1) and is registered with | the below listed states and citie | s within which you | ır firm wo | ould delive | r purchases to us | |
| | | | | estitution o | of Higher Education | in |
| which are for resale or le | ease by us in the normal course of | of our business wh | ich is 🗀 | TOUTOUT C | 7 Tilgitor Eddodilo | or or |
| 2) that such purchases a | are exempt from payment of sale | es or use tax in su | ch states | and cities | s because our buy | er is: |
| ⊠ Political Subdivision | or Governmental Agency C | haritable or Religi | ous 🗆 | Otherwise | e Exempt By Statu | te (Specify) |
| If Otherwise Exempt By Statu | e, specify here | | | | | |
| City or State | State Registration or ID Number | er City or State | | | State Registration or | ID Number |
| City of Aurora | 98-00799-0000 | Colorado (I | Boulder | campus) | 98-02915-0000 | |
| City or State | State Registration or ID Number | er City or State | | 100 | State Registration or | ID Number |
| Colorado | 98-02565-0000 | Texas | | | 32002730391 | |
| City or State | State Registration or ID Number | er City or State | | | State Registration or | ID Number |
| I further certify that if any Use Tax we will pay the tax billing. This certificat shall be called until cand | cities is more than six(6), atta y property so purchased tax free tax due direct to proper taxing a e shall be part of each order who celed by us in writing or revoked | e is used or consul authority when sta nich we may herea | med by to te law s after give | he firm as o provides | s or inform the sell | er for added |
| General Description of produc | ets to be purchased from seller | | | | | |
| | ry, I swear or affirm that the infor | mation on this forr | n is true | and corre | ct as to every mate | erial matter. |
| Authorized Signature (owner, | | Title Associate V | ice Pres | sident/Univ | versity Controller | Date (MM/DD/YY) |

STATE OF COLORADO

TAXPAYER SERVICE DIVISION

Department of Revenue

1375 Sherman Street Denver, Colorado 80261



Bill Owens Governor

M. Michael Cooke Executive Director

Neil Tillquist

Michael J. Barden University of Colorado at Denver and Health Sciences Center(UCDHSC) Building 500, Mail Stop F418 P.O. Box 6508 Aurora CO 80045

April 7, 2006

Dear Mr. Barden:

This is in response to your letter of March 1, 2006, to Bruce Nelson of the Department of Revenue regarding sales tax exemption from county and special district sales taxes for UCDHSC construction projects at the Fitzsimons campus. Mr. Nelson has left the Department, so I am responding to your inquiry.

In regards to Adams County sales and use tax, the sales tax is collected by the Department of Revenue, not the city of Aurora. Use tax on building materials is collected by the county when issuing building permits. Under 29-2-105(d), 39-26-708(1)(a) and 39-26-708(2)(a), C.R.S., UCDHSC and its contractors and sub-contractors are exempt from county sales and use tax on construction and building materials for State/UCDHSC owned real property.

In regards to special district sales and use taxes, UCDHSC and its contractors and sub-contractors are exempt from sales and use tax pursuant to the exemptions granted in 39-26-708(1)(a) and 39-26-708(2)(a), C.R.S., for the Regional Transportation District under 32-9-119(2)(c)(II), C.R.S, for the Scientific and Cultural District under 32-13-107(2), C.R.S, and for the Metropolitan Football Stadium District under 32-15-110(2)(a), C.R.S.

Additionally, for construction projects in the City and County of Denver, UCDHSC and its contractors and sub-contractors are exempt from the aforementioned special district sales and use taxes, as well as state sales and use tax.

Should you have additional questions regarding these matters, feel free to contact me.

Respectfully,

Steve Asbell

tene

Taxpayer Service Policy Group Colorado Dept of Revenue

Ph:303.866.3889 email: sasbell@spike.dor.state.co.us



Administration 1470 South Havana Street Aurora, Colorado 80012 303-739-7055 FAX: 303-739-7068

March 12, 2001

Wayne F. Henderson Vice Chancellor for Administration and Finance University of Colorado Health Sciences Center Fitzsimons, Building 500, Room C1003 P.O. Box 6508 Aurora, Colorado 80045-0508

RE: Letter of Commitment

Dear Mr. Henderson:

I am in receipt of your letter dated February 27, 2001, requesting that I issue a letter of commitment to the University of Colorado Health Sciences Center ("UCHSC") pursuant to *City Code Section 130-63(c)*. It is my understanding that UCHSC is part and parcel of the University of Colorado, a public institution of higher education of the State of Colorado. § 23-20-101, et seq., C.R.S. You have asked for some assurance that UCHSC is exempt from the payment of City sales and use tax, as well as the employer portion of the City occupational privilege tax.

City Code Section 130-157(1) exempts all sales of tangible personal property and taxable services to the various political subdivisions of this state from imposition of City sales tax. Identical exemptions exist in both the City Use Tax ordinance (City Code § 130-198(5)) and the City Employer Occupational Privilege Tax ordinance (City Code § 130-405(1)). Accordingly, UCHSC falls squarely within each of these three exemptions.

It should be noted, however, that these exemptions do not extend to the collection of City tax. For instance, UCHSC must collect, report, and remit City sales tax on any retail sale of tangible personal property or taxable services it makes to a non-exempt third party. City Code § 130-160. Likewise, UCHSC

Wayne F. Henderson March 12, 2001 Page Two

must also collect, report, and remit the employee portion of the City occupational privilege tax for each person it employs within the City for any period of time within a calendar month sufficient to receive no less than \$250.00 as compensation for such employment. City Code § 130-464.

With respect to the deposit and ultimate payment of City use tax on construction materials, it is the longstanding policy of the City that the party who contracts for and directs and controls the construction of building improvements is liable for such tax. See <u>Fifteenth Street Investment Co. v. People</u>, 102 Colo. 571, 81 P.2d 764 (1938). Under the circumstances described in your request, it is UCHSC, and not its contractors, upon whom sole liability for the payment of City use tax would rest. Because UCHSC is an exempt entity, no use tax is due and owing on the purchase and subsequent use of construction materials for the development of UCHSC's property at the Fitzsimons site.

With regard to your additional requests, the City has no objection if UCHSC's contractors wish to use this letter to present to City building officials and third-party retailers as evidence of UCHSC's tax exemption. As for any future revocation of this letter, unless the status of UCHSC as a political subdivision changes, the various City tax exemptions which UCHSC is entitled to claim cannot be lawfully repealed without the prior approval of the City's voters. <u>See Colo. Const. Art. X, § 20(4)(a)</u>. Therefore, the City believes UCHSC will be adequately informed in the event that the City decides to seek approval for any change in its tax laws that would impact UCHSC's tax-exempt status.

Very truly yours,

John Gross Director of Finance



201 W Colfax Ave #1009 Denver, CO 80202 fax: 720- 913-9455 www.denvergov.org/treasury

February 19, 2014

University of Colorado Procurement Service Center 1800 Grant Street, Suite 500 Denver, CO 80203

Ladies/Gentlemen:

The above named entity is exempt from the Denver sales tax per Sec. 53-26(1) of the City Retail Sales Tax Article:

Sec. 53-26 (1) Exemptions

There shall be exempt from taxation under the provisions of this Article the following: (1) All sales to the United States Government, to the State, its departments and institutions and the political subdivisions thereof, only when purchased in their governmental capacities.

To qualify for the exemption, purchases must be billed direct to the organization, and payment made from funds of the organization.

The exemption does not extend to construction contractors who may perform contracts for you; they are the consumer of all property purchased and used in the performance or contracts for others. Nor does the exemption apply to purchases by employees or members for their own personal use.

You may reproduce this letter to furnish to suppliers as needed.

Sincerely,

Donald Korte, Audit Manager Tax Compliance/Audit Section

720-913-9339



DR 0172 (05/01/18)
COLORADO DEPARTMENT OF REVENUE
Denver CO 80261 - 0009
(303) 238-SERV (7378)

Special Notice

Purpose of this application

The exemption certificate for which you are applying must be used only for the purpose of purchasing construction and building materials for the exempt project described below. This exemption does not include or apply to the purchase or rental of equipment, supplies, and materials which are purchased, rented, or consumed by the contractor and which do not become a part of the structure, highway, road, street, or other public works **owned** and **used** by the exempt organization.

Any unauthorized use of the exemption certificate will result in revocation of your exemption certificate and other penalties provided by law.

A separate certificate is required for each project.

Colorado Withholding Account Number

A Colorado Account Number (CAN) should be provided in this field. Applications that are left blank or list N/A will not be processed and will be returned.

Subsidiary:

This box is marked when a subsidiary is using the parents withholding account number (only when it does not have its own.) Provide the parents CAN.

Subcontractor:

This box is marked when a contractor does not have employees of their own and outsources their employees through a subcontractor. List the subcontractor or subcontractors name and CAN(s).

Staffing Agency:

This box is marked when a contractor does not have employees of their own and outsources their employees through a staffing agency. Provide the Staffing Agency's name and CAN.

No employees/no subcontractors:

For contractors with no employees, no subcontractors/ staffing agencies:

Write no employees in the (CAN) box and provide explanation. For example, I have no employees or subcontractors and perform all of the work myself.

Subcontractors:

Subcontractors will not be issued Certificates of Exemption by the Department of Revenue. Upon receipt of the Certificate, the prime contractor should make a copy for each subcontractor involved in the project and complete it by filling in the subcontractor's name and address and signing it. The original Certificate should always be retained by the prime contractor. Copies of all Certificates that the prime contractor issued to subcontractors should be kept at the prime contractor's place of business for a minimum of three years and be available for inspection in the event of an audit.

See FYI Sales 95 for information about qualifying affordable housing projects.

To avoid a returned application ensure you have done the following:

| fol | lowing: |
|-----|---|
| | Accurately completed all applicable boxes of the form. |
| | Provided a copy of the Contract or agreement page. The Contract or Agreement page lists the type and scope of work. |
| | Bid amount on Contract or Agreement page matches the amount listed on the application (to the penny). |
| | Contract or Agreement page contains the signatures of the contracting parties. |
| | The form DR0172 (application) is signed. |
| | The exempt organizations number was provided and is correct. |



DR 0172 (05/01/18)
COLORADO DEPARTMENT OF REVENUE
Denver CO 80261 - 0009
(303) 238-SERV (7378)

Contractor Application for Exemption Certificate

This exemption does not include or apply to the purchase or rental of equipment, supplies, and materials which are purchased, rented, or consumed by the contractor and which do not become a part of the structure, highway, road, street, or other public works **owned** and **used** by the exempt organization.

Any unauthorized use of the exemption certificate will result in revocation of your exemption certificate and other penalties provided by law. A separate certificate is required for each contract.

Send completed forms to: Colorado Department of Revenue, Denver, CO 80261-0009

Failure to accurately complete all boxes of the form or provide all supporting documentation will cause the application to be denied.

| supporting docur | | | | | | · | | |
|--|--------------|--------------|------------------------|-------------------------|-----------------|--------------|---------------|--------|
| For Depart | tment Us | e Only. I | Do not write | in this secti | on. | | | |
| Contractor/Account No. | | | | Period (MM/Y | Y-MM/YY) | | | |
| 89- | | | | | | | | |
| | Must be | complete | ed by appli | cant | | | | |
| Contractor Information | | | - a , a.p.p | | | | | |
| Trade name/DBA | | | | | | | | |
| | | | | | | | | |
| Owner partner or corporate last name | | First Nam | | | | | Middle Ini | tial |
| Owner, partner or corporate last name | | First Ivaiii | е | | | | Ivildale IIII | ııaı |
| | | | | | | | | |
| Mailing Address | City | | | | | State | Zip | |
| | | | | | | | | |
| E-Mail Address | | FEIN | | Bid amour | t for your cont | tract (Must | match to the | penny) |
| | | | | \$ | | | | |
| Fax number | | | Business Phor | ne number | | | | |
| | | | | | | | | |
| Colorado withholding tax account number | 10 | | | | | 0. 55 4 | | |
| (See instructions) | Subsidian | | Subcor | | | Staffing Ag | jency | |
| No Employees/Subcontractors. (Provide explanation o | | | ntractors (see | below) | | | | |
| 140 Employees/Subcontractors, (Frovide explanation of | attacii a ie | tter or expi | anauonj. | | | | | |
| Exemption Information type of work | ntract or a | agreeme | nt page, ide | entifying the | contracting | parties, | bid amount | i, |
| type of work, | and sign | atures of | contracting | | | | | |
| Name of exempt organization (as show on contract) | | | | | ganization's n | umber | | |
| | | | | 98 | | | | |
| Address of exempt organization | City | | | | | State | Zip | |
| | | | | | | | | |
| Principal contact at exempt organization-Last Name | | First Nam | е | | | | Middle Ini | tial |
| | | | | | | | | |
| Housing Authority (if applicable) | | N | ame of Projec | t (if applicable) | | | | |
| | | | | | | | | |
| Owner of the Project (if applicable) | | | | | | | | |
| owner of the Project (if applicable) | | | | | | | | |
| Dharian I analisa of antinatally (also natural address and | | | | | | | | |
| Physical location of project site (give actual address when the street of the street o | | bla and Ciki | | | | -4-4\ | | |
| | hen applical | ble and Citi | es and/or Cou | nty (ies) where | project is loca | ated) | | |
| | hen applical | ble and Citi | | | | | | |
| City | hen applical | ble and Citi | es and/or Cou | nty (ies) where | | | lephone numb | oer |
| City | hen applical | ble and Citi | | | | | lephone numb | oer |
| City Scheduled construction start date (MM/DD/YY) | hen applical | ble and Citi | State | | Principal o | | lephone numb | oer |
| | hen applical | ble and Citi | State | Zip | Principal o | | elephone numb | per |
| Scheduled construction start date (MM/DD/YY) | | | State Estimated con | Zip mpletion date (N | Principal o | contact's te | elephone numb | oer |
| Scheduled construction start date (MM/DD/YY) I declare under penalty of perjury in the second of | | | State Estimated con | Zip mpletion date (N | Principal o | contact's te | elephone numb | per |
| Scheduled construction start date (MM/DD/YY) | degree tha | | State Estimated con | Zip mpletion date (n | Principal o | e and | elephone numb | per |

STATE OF COLORADO OFFICE OF THE STATE ARCHITECT STATE BUILDINGS PROGRAM

CONSTRUCTION MANAGER/GENERAL CONTRACTOR (CM/GC) AGREEMENT (STATE FORM SC-6.5)

EXHIBIT Y Service-Disabled Veteran-Owned Small Business and Minority/Women Business Enterprise Participation Report



SERVICE-DISABLED VETERAN-OWNED SMALL BUSINESS AND MINORITY/WOMEN BUSINESS ENTERPRISE PARTICIPATION REPORT

| Institution/Agency: | University of Colorado Anschutz Medical Campus / GFE |
|---|---|
| Project No./Name: | 1 |
| • | |
| | |
| | GIBLE FOR AWARD OF THIS CONTRACT, EACH CONTRACTOR (INCLUDING GINEER/CONSULTANT/CONTRACTOR) IS REQUESTED TO COMPLY WITH THESE REQUIREMENTS. |
| venture) [](is) [undersigned Archit venture) [](is) [Architect/Engineer/ [](is not)* a wor *Persons signing hereby behalf and acknowledge title and will not be ac | rchitect/Engineer/Consultant/Contractor hereby certifies that the [](company) [] (joint](is not)* a service-disabled veteran-owned enterprise as defined in this report. The ect/Engineer/Consultant/Contractor hereby certifies that the [](company) [](joint](is not)* a minority enterprise as defined in this report. The undersigned Consultant/Contractor hereby certifies the [](company) [](joint venture) [](is) nan-owned business enterprise as defined. (*Strike out where inapplicable.) y swear and affirm that they are authorized to act on Architect/Engineer/Consultant/Contractor's e that the State is relying on their representations to that effect. Principal is not a recognized cepted ER/CONSULTANT/CONTRACTOR |
| ARCHITECT/ENGINE | ER/CONSULTANT/CONTRACTOR |
| Legal Name of Contra | cting Entity |
| *Signature | |
| By: , Owner Name (print) | Title |
| Date: | |

II. It is the general policy of the State of Colorado to be as inclusive as possible to all member communities when spending taxpayer dollars. It is also the intent of the State to address the goals of the HB14-1224

CRS 24-103-211 of at least 3% of all contracts by dollar value to be awarded to SDVOSBs.

III. REQUIREMENTS

- A. Service-Disabled Veteran-Enterprise (SDVE) means for the purpose of this report, a business who must be incorporated or organized in Colorado or they must maintain a place of business or have an office in Colorado and who are officially registered and verified as a SDVOSB by the Center for Veteran Enterprise within the U.S. Department of Veterans Affairs (www.vip.vetbiz.gov) per CRS 24-103-211
- B. Minority Business Enterprise (MBE) means, for the purpose of this report, a business enterprise at least 51 percent that is owned and controlled by minority group members, or, in the case of a publicly owned business, at least 51 percent of the stock of which is owned and controlled by minority group members. Eligible persons are expected to be engaged full time in the day-to-day operation and management of the business. Minority group members are ethnic minorities including African American, Hispanic American, Native American or Asian/Pacific American.
- C. Women Business Enterprise (WBE) means, for the purpose of this report, a business enterprise of at least 51 percent of which is owned and controlled by a woman or women, or, in the case of a publicly-owned business, at least 51 percent of the stock of which is owned and controlled by women. Women are expected to be engaged full time in the day-to-day operation and management of the business.

- D. The University of Colorado Denver | Anschutz Medical Campus does not have a certification process for nor does it require MBE's and WBE's to be certified.
- E. The percentages of service-disabled veteran, minority and women-owned business participation will be determined by dollar value of the work subcontracted to or joint ventured with service-disabled veteran, minority, and women-owned firms, as compared to the total dollar value of the bid amount for all work bid under this contract.
- F. Prior to the award of this contract, the contractor will be required to provide to the Principal Representative a list of SDV/M/WBE enterprises, stipulating the dollar amount of each subcontract or supplier of materials on page 2 of this Service-Disabled Veteran, Minority and Women Business Enterprises Participation Report.

| G. The contractor will retain completion of this contract reasonable times and plactor its designated represer | t. These records ar | nd documents, or o y an authorized re | copies thereof, will presentative of the | be made available at Principal Representative, |
|---|-----------------------------|--|---|---|
| ARCHITECT/ENGINEER/CONSUI SDVOE: Yes [] MBE: No [] | | | | |
| Total Contract Amount: \$, | | | | |
| Name and Address of SDV/M/WBE Subcontractors and/or Suppliers and/or Self- Performed Work by SDV/M/WBE Primes* | SDVE Contract Amounts | MBE Contract Amounts | WBE Contract Amounts | Type of Work |
| | | | | |
| | | | | |
| | | | | |
| | | | | |
| | | | | |
| | | | | |
| | | | | |
| | | | | |
| | | | | |
| Indicate ethnicity based on Pa | ragraph III. A. ab | ove. | | |
| Total SDVE Contracts \$ | | | | |
| Total MBE Contracts: \$ | | | <u> </u> | |
| Total WBE Contracts: \$ | | | <u> </u> | |
| Total SDVE % | | | <u></u> | |
| Total MBE %: | | | <u></u> | |
| Total WBE %: | | | <u></u> | |

STATE OF COLORADO OFFICE OF THE STATE ARCHITECT STATE BUILDINGS PROGRAM

CONTRACTOR'S DESIGN/BID/BUILD AGREEMENT

(STATE FORM SC-6.21)

EXHIBIT Z: MISCELLANEOUS DOCUMENTS

Appendix D

DIRECT LABOR BURDEN (SBP-6.18)



STATE OF COLORADO OFFICE OF THE STATE ARCHITECT STATE BUILDINGS PROGRAM

DIRECT LABOR BURDEN CALCULATION

| Institution/Agency: | | |
|---|--|--|
| Project No./Name: | | |
| | | |
| | | |
| This form is required to be sul | bmitted for review prior to ex | ecution of a construction agreement. |
| contractor pays to employees insurance etc. The Labor B | s on their payroll. Examples Burden amount must be ag Bluded in the contract as part | total labor burden; Items include benefits that a include taxes, pension cost, health and dental greed to by both the contractor and Principal of Exhibit A and will be used in the calculation of |
| concrete, block layers etc. Ple each sub-contractor. These Proposals (SC-6.312) Line 10 | ease provide one (1) Labor labor burdens shall be used | , mechanical contractors, excavators, millwork, Burden Calculation Sheet per contractor and for d in the calculation of any future Change Order all information included in this calculation. |
| | | |
| Contractor/Subcontractor | Name: | |
| | Percent of Salary Paid | |
| Payroll Taxes | | |
| Pension Costs | | |
| Health Insurance | | |
| Dental Insurance | | |
| Life Insurance | | |
| Other (Specify) | | Description: |
| Other (Specify) | | Description: |

0%

Total Labor Burden:

Appendix G

PROGRAM PLAN ETC

University of Colorado Anschutz Medical Campus

R1 Vivarium HVAC Study Program Plan

13001 E 17th PL Aurora, CO 80045



CU Anschutz Medical Campus

June 22, 2022 CRA# 2022-193



Cator, Ruma & Associates, Co. Colorado | Idaho | Wyoming

896 Tabor Street, Lakewood, CO 80401 P (303) 232-6200 www.catorruma.com

www.catorruma.com

1. EXECUTIVE SUMMARY

1.1 CRA was hired to evaluate the cost to upgrade the ventilation system for the vivarium on Research 1. Two projects are needed, one to replace the 2 exhaust fans on the roof and the other to replace all of the supply and exhaust air valves and associated controls in the interstitial space above the vivarium.

The Primary components required to be upgraded are as follows:

- A. Exhaust Fans
- B. Supply and Exhaust Air Valves
- C. Reheat Coils and Piped Accessories
- D. Associated Controls

Following this summary, the report will identify the observations from plan review and site visits as well as two detailed cost opinions.

The proposed construction cost for these two projects are as follows:

- 1) Exhaust Fans = \$400,000.
- 2) Air Valves and Controls = \$2,600,000

Design, other soft costs and escalation should be added for the total project funding request.

2. NOTED OBSERVATIONS FROM PLAN REVIEW:

2.1 The ventilation system for the vivarium resides in the interstitial space above the vivarium. There are 145 supply air valves with reheat coils, 114 general exhaust air valves, 39 cage rack exhaust valves, 16 Biosafety Cabinets exhaust valves and 5 other miscellaneous exhaust valves.

The existing air valves are pressure independent venturi style with internal plunger and spring.

These air valves do not have airflow measurement, therefore separate cross-ring airflow stations were provided to report airflow to the building automaton system (BAS).

There are two 34,300 CFM exhaust fans on the roof, in a parallel configuration. The connected load from all exhaust air valves for this system is 58,235 cfm. This puts the existing exhaust fans each at 60% of the connected load. If one fan is disabled, there may still be enough air to satisfy the spaces for a short period.

www.catorruma.com

When the energy recovery coils were added to these exhaust branches going to the fans, they were sized at 40,000 cfm each. Since the coils are sized larger, there is the possibility of increasing the size of the fans to get closer to N+1 redundancy.

NOTED OBSERVATIONS FROM THE FIELD REVIEW:

3.1 The interstitial space above the vivarium is comprised of catwalks to service the ventilation system components and other equipment. Most of the air valves are located adjacent to the catwalk. Therefore, most air valves are readily accessible for replacement.

However, there are approximately 35 air valves that are too far away from the edge of the catwalk and will require scaffolding or other platforms to be installed to access air valves for replacement. Depending on how the contractor arranges scaffolding, there will likely need to be 20-30 areas of scaffolding or platforms constructed in order to replace these hard-to-reach air valves.

The reheat coils do not have true isolation valves since "valve packs" were used where balancing valves are also intended for isolation.

Therefore, drain-down of the heating water system and additional "quadrant" isolation valves are recommended to be installed for future isolation without having to affect other areas of the vivarium. Control valves for the coils, while still operational are beyond their useful life.

4. ISSUES AND NEEDS:

- 4.1 There are two primary issues with the existing ventilation system.
 - A. Age. The system is over 20 years old and reaching the end of its useful life. As this is a critical system, for the welfare of animals, it must be reliable.
 - B. Clogging of the exhaust air valves and associated airflow measuring devices. The animal air and dander travel from the cage racks into the exhaust ducts. This then collects on the internal plunger and airflow measuring sensors causing them not to measure correctly. Maintenance to clean these components is difficult, time consuming and requires lengthy outages to the animal holding rooms.
- 4.2 The proposed replacement air valves use a type of airflow measuring called vortices shedding. This uses a "Bluff Body" that created vortexes that are frequencies measured by a microphone. These types of sensors are self-cleaning and do not clog like a pitot tube measuring device.

www.catorruma.com

- 4.3 The proposed air valves by Accutrol also have access doors to easily clean the Bluff Body and blade dampers inside the valve, compared to having to remove the air valve to clean the plunger.
- 4.4 The proposed replacement of the exhaust fans is at a minimum to replace like for like, but increase the size of the two exhaust fans to 40,000 cfm to match the capacity of the energy recovery coil.
- 4.5 Another option to provide redundancy for the exhaust system is to add another equal sized exhaust fan at 40,000 cfm to create a true N+1 redundancy for the system. This would require a new stack, branch duct between one of the coil outlets to the fan, power connection, VFD, programming updates and roof repair.

5. SCOPE OF WORK FOR THE PROJECT:

- 5.1 The scope of work required is as follows:
 - A. Exhaust fans: remove the existing fans and install the replacement fans in the same location. A large crane will be needed for this lift as the roof is 10 stories high (approximately 150 ft high).
 - 1) Alternate add a 3rd fan of equal size to create the N+1 redundancy for the exhaust system. This is not currently budgeted.
 - B. Air Valves: Remove and replace all of the room supply and general exhaust valves. Since the new air valves take up less length, additional sheet metal will be required to make up this difference. A Laboratory Air Control System (LACS) will accompany these valves.
 - C. Reheat Coils: The goal is to replace these in the same location to minimize the cost of accessories. Additional isolation valves will be required for system isolation for future maintenance.
 - D. Controls: All of the reheat coil water control valves will be replaced with new Siemens valves. Siemens also provides the wiring and labor to install the wiring and sensors provided with the LACS. Upgrading of the controls for the new exhaust will also be required.

STC/bqg

P:\CU\Anschutz\2022-193 R1 Vivarium HVAC Study\Sup-Docs\Reports\22193.R1 Vivarium HVAC Study Program Plan.20220608.docx

P18 VIVARIUM GENERAL EXHAUST INVESTIGATION



SUMMARY

Recent issues with doors slamming and incorrect airflow in several rooms in vivarium suite 404 led to a wider investigation into the P18 general exhaust system to address similar issues that continue to recur over time throughout the vivarium.

Since exhaust air valves are typically implicated in these issues, an investigation was conducted to identify factors contributing to performance deficiencies in the P18 vivarium general exhaust system and associated venturi control valves.

Thermiq Services, LLC worked with CU's Building Performance Group who performed field testing and provided all data and trending for the investigation.

Multiple factors were identified that are likely contributing to performance issues in the problem spaces. Several exhaust air valves, or EAVs, are installed in the incorrect orientation, causing erratic or poor control of exhaust airflow in these spaces. Some EAV controllers are inaccurately sensing exhaust airflow due to sensors that have drifted out of calibration, and in some cases due to defective controllers. Many EAVs are out of range of the rated pressure differential required across the venturi valve, preventing proper valve performance and airflow control. These EAVs are likely affected by low fan-side static pressure, since all deficient EAV's identified are located on the same general exhaust trunk. There are indications that this duct trunk may be near capacity. A past project transferred a portion of the ABSL-3 exhaust load to the vivarium general exhaust system, further increasing load on the duct trunk in question.

Based on findings, several corrective measures have been identified that may mitigate or resolve vivarium exhaust issues. These are listed under the "recommendations" section in this report. Due to the age of vivarium systems and equipment, several recommendations involve a larger scope of recommissioning for the vivarium exhaust equipment.

Further investigation or sample testing may be warranted, depending on the implementation approach taken for the listed recommendations. See the "Recommendations" section for additional detail.

APPROACH

For cost and time-efficiency reasons, a targeted approach was taken to isolate problem EAVs. Sample testing was then performed, typically on a subset of the problem EAVs.

Problem EAVs were isolated by using BAS trending to identify EAVs that failed to meet their airflow setpoints for a significant amount of time during the trending period of several months. 22 "problem EAVs" in the vivarium were identified on the P18 general exhaust (GEX) system through BAS trending. Trending showed negligible issues for SAV boxes.

Potential causes for the vivarium exhaust problems were investigated at both the fan system-level, and at the EAV component level with the objective of identifying contributing factors and ruling out others.

The following potential issues were investigated: installation deficiencies, failed components or maintenance-related issues, excessive pressure drop across duct or system components, air valve pressure drop outside of manufacturer's required range, installation deficiencies, sensor drift.

TESTING RESULTS

| | EAV Box | Orientation | Valve PD (in.wc) | Silencer PD (in.wc) | Calibration | Setpoints |
|----|---------|-------------|---------------------|------------------------|------------------|-----------|
| 1 | EAV 032 | HORIZ | 1.23 | | | VERIFY |
| 2 | EAV 033 | HORIZ | 0.31 | 0.34 | req. replacement | VERIFY |
| 3 | EAV 035 | HORIZ | 0.13 | 0.44 | req. replacement | VERIFY |
| 4 | EAV 036 | HORIZ | 0.52 | | | ОК |
| 5 | EAV 144 | VERT | 0.27 | | | VERIFY |
| 6 | EAV 001 | HORIZ | 0.13 | | | VERIFY |
| 7 | EAV 146 | HORIZ | 0.11 | | | VERIFY |
| 8 | EAV 037 | HORIZ | 0.90 | | | VERIFY |
| 9 | EAV 055 | HORIZ | 0.41 | | OUT OF RANGE | ОК |
| 10 | EAV 046 | HORIZ | 0.39 | | req. replacement | ОК |
| 11 | EAV 044 | HORIZ | 0.56 | 0.17 | OUT OF RANGE | ОК |
| 12 | EAV 028 | HORIZ | 0.19 | | OUT OF RANGE | OK |
| 13 | EAV 030 | HORIZ | 0.39 | | | ОК |
| 14 | EAV 051 | HORIZ | 0.54 | | OUT OF RANGE | OK |
| 15 | EAV 058 | HORIZ | 0.63 | | OUT OF RANGE | ОК |
| 16 | EAV 004 | HORIZ | 0.11 | | | ОК |
| 17 | EAV 142 | HORIZ | 0.18 | | | ОК |
| 18 | EAV 002 | HORIZ | 0.17 | | | OK |
| 19 | EAV 043 | HORIZ | 0.55 | | OUT OF RANGE | OK |
| 20 | EAV 041 | HORIZ | 0.57 | 0.30 | | OK |
| 21 | EAV 024 | HORIZ | 0.30 | 0.47 | | OK |
| 22 | EAV 022 | HORIZ | 0.11 | | OUT OF RANGE | OK |

^{* ---} indicates EAV was not included for that test type

FAN SYSTEM

Fan speed control

BAS trending of fan speed confirmed stable and correct fan control.

GEX duct static pressure sensor accuracy

Comparison of pressure sensor readings to calibrated manometer readings confirmed accurate pressure sensor readings.

Duct static pressure drop or duct constraints

All 22 EAVs were found to be connected to the same major duct "trunk" that serves the vivarium's north side. This trunk is one of three and represents approximately half of the vivarium's general exhaust capacity. A basic static pressure profile was developed by measuring static pressure at specified duct locations to quantify pressure drop through this trunk.

Changes in airflow capacity requirements

Drawings and field conditions show that a past project transferred 5 exhaust air valves from the ABSL-3 exhaust system to the GEX exhaust system, adding 730 cfm of demand on the strained capacity of this trunk.

COMPONENT LEVEL

Controller airflow setpoints

TEC controller subpoint reports were used to compare current CFM setpoint values with design CFM setpoint values. 7 of the 22 boxes had setpoints that deviated from design values and that may adversely affect space conditions or decrease available exhaust at nearby EAVs.

Actuator failure

One non-responsive actuator was determined to have failed and was replaced at EAV-IN 041.

Air valve installation orientation

8 of the 22 "problem boxes" are installed in the wrong orientation, causing erratic or poor control of airflow rates.

Pressure differential at air valve

The venturi air valves in the vivarium require a differential pressure across the valve in the range of 0.6 - 3.0 in.w.g. in order to control and perform correctly. 18 of the 22 "problem boxes" tested had differential pressures below the minimum required. Low

differential pressure at the valve can be caused by both excessive pressure drop on the room-side of the valve, as well as by inadequate exhaust static pressure on the fan-side of the valve.

Pressure drop at sound attenuator

To find a reason for high static pressures on the room-side of many venturi valves, pressure drop was measured at 5 of the 22 sound attenuators for the problem EAVs and compared to rated pressure drop from the manufacturer at the current air velocities. All of the boxes that were tested had pressure drops that exceeded the expected pressure drop by 2-4x the rated pressure drop.

Unknown restriction (i.e. exhaust intake filters)

No ceiling exhaust intake filters were found at any of the 22 boxes, and no other unexpected components or restrictions were found.

Airflow sensor accuracy

Controller airflow sensing accuracy was tested by taking pitot traverse airflow measurements at 10 of the 22 EAVs and comparing measured CFM values to CFM values sensed by the BAS system. 8 of the 10 EAVs tested were out range for calibration. 3 of these are suspected to be due to TEC controller failures. 6 of the 8 EAVs had higher airflow than what was measured by the BAS system, indicating that the system would likely benefit from a net gain in capacity if valves were calibrated.

RECOMMENDATIONS

Due to the complexity and age of vivarium HVAC systems, a broader scope of recommissioning performed by a TAB firm is recommended to improve performance of the existing exhaust equipment. Several of the measures listed below that would best fit within a phased recommissioning of the vivarium exhaust system include calibration of box airflows, an audit of space pressure differential setpoints, and rebalance of exhaust inlet dampers where required.

Steps that can be taken to alleviate general exhaust system deficiencies for the vivarium are described below in order of importance.

For areas affected by incorrect venturi valve orientation, the priority should be to install a valve with the correct orientation. This is likely a major performance issue for these boxes since venturi valves are manufactured to work for a specified orientation (vertical or horizontal). Venturi valves use an internal valve spring calibrated for the intended orientation. Incorrect orientation will result in erratic or poor control of exhaust airflow cfm. A practical approach would likely require ordering and installing replacement venturi valves made for the current installation orientation.

Calibration of vivarium general exhaust EAV airflows is recommended to reclaim excess airflow from non-problem EAVs. 75% of EAV boxes out of calibration range were high in airflow, indicating that calibration of air valves will result in a net increase in available capacity as excess airflow is reclaimed by accurate calibration. Calibration could be conducted in phases, starting with the EAV boxes served by the north vivarium ducting. The 3 EAV controllers requiring unusually large calibration coefficients should be replaced due to the likelihood of a failed controller being at fault.

Some EAV setpoints may need to be reviewed and corrected to achieve desired space performance and pressure for the areas served by these EAVs.

There is capacity at the fan level for a small increase in fan speed. This is recommended as a last step to address EAV performance issues, as the small increase available is unlikely to resolve or mask the existing issues on its own. Implementing the other recommendations before modifying the exhaust fan setpoints will achieve the most effective and long-term improvement. When testing began general exhaust fans 6 and 7 were operating at a setpoint of 2.5" and a speed of approximately 83.5%. If required, an incremental increase in duct static pressure could be made until pressure requirements are met at the problem EAVs. It is not recommended to exceed 95% fan speed or to exceed 3" duct static pressure without verifying duct pressure class rating.

POTENTIAL FOLLOW-UP

Duct leakage testing was not included in the investigation due to time and equipment constraints. Additionally, there were no indications of abnormal duct leakage. However, on a large ducting system leakage rate can be a relevant factor, especially if the system is near capacity. Qualitative testing using a smoke generator would be recommended initially, since a quantitative duct leakage rate test would require system shutdowns and may be impractical to implement.

The cause of high pressure drops through many of the sound attenuators is still unknown. Any follow-up investigation should include evaluating the installed configuration for system effects, comparing pressure readings with other non-problem EAV attenuators, and inspecting the interior condition of one or two sound attenuators to determine if debris buildup has contributed to higher pressure drop. Potential corrective action could be determined based on the findings.

The position of individual exhaust intake balance dampers could be surveyed in spaces served by EAVs with high room-side static pressure drop. A correctly balanced system should have at least one damper in the wide-open position. Excessively closed dampers could prevent an EAV box from operating within its design differential pressure range.

Correct vivarium space pressurization depends on the airflow differential between exhaust and supply air, as well as the space pressurization of adjacent spaces. An audit of the net pressure

gradient balance for larger vivarium suite sections may help identify additional pressurization issues, especially where suites adjoin corridors or adjacent suites.

The scope of testing already performed could be further expanded to other EAVs where useful. A broader recommissioning could implement the above recommendations in a systematic way to achieve the best performance improvement for the existing system.

Jordan Eads Thermiq Services, LLC

6/1/23